### SALISBURY TOWNSHIP

Lehigh County, Pennsylvania Board of Commissioners Meeting Regular Meeting Agenda—May 23, 2019 7:00 PM

- 1. Call to Order
- 2. Pledge of Allegiance
- 3. Notifications:
  - All public sessions of the Salisbury Township Board of Commissioners regular meeting are electronically monitored (tape recorded) for the purpose of taking the Minutes. Public comment on agenda items will be entertained at the time the item is up for vote, non-agenda related public comments will be entertained after the agenda.
  - Public Comment Period Policy—Three (3) Minutes Rule
- 4. Financial Report (Apr 2019) and Bills Payable Period 5/4/19 5/17/19
- 5. Approval of the Minutes –May 9, 2019
- 6. New Business
  - A. Conditional Use Hearing South Mall Food Truck Events
  - **B. ORDINANCES** 
    - 1. Ordinance to Approve Amendments to the Current Non-Uniformed Defined Benefit Pension Plan and Establish a New Defined Contribution Pension Plan for New Full-Time Non-Uniformed Employees

### C. RESOLUTIONS

1. Resolution to Approve Myron Haydt's request for SALDO Waiver for 2217 North Melrose Ln re: maximum driveway slope

### **D. MOTIONS**

1. Motion - Discussion regarding Lafayette College Meyner Center's Proposal for Police Chief recruitment services

### **E. PRESENTATIONS**

- 1. Presentation by Ms. Mary Erdman regarding Fair Districts PA Legislation
- 7. Courtesy of the Floor: (opportunity for public comment on non-agenda items)
- 8. Adjournment

Salisbury Township			l Report April 20 iod: 04/19	)19			Page:
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
General Fund							
Property Taxes							
01-301.100	Real Estate Tax - Current	2,477,000	425,262	17.17%	2,470,000	963	0.04%
01-301.400	Real Estate Tax - Claims	35,000	15,981	45.66%	30,000	18,516	61.72%
01-301.600	Real Estate Tax - Exceptions	3,000	739	24.63%	3,000	.00	.00
Total Prope	rty Taxes:	2,515,000	441,982	17.57%	2,503,000	19,480	0.78%
₋ocal Enabling T	axes						
01-310.100	Realty Transfer Tax	600,000	124,128	20.69%	480,000	249,312	51.94%
01-310.200	Earned Income Tax	2,350,000	727,140	30.94%	2,300,000	658,825	28.64%
01-310.400	Local Services Tax	520,000	123,765	23.80%	520,000	128,952	24.80%
Total Local	Enabling Taxes:	3,470,000	975,033	28.10%	3,300,000	1,037,089	31.43%
Business Licens	es & Permits						
01-321.600	Sign Permits	3,000	1,266	42.20%	1,500	3,760	250.63%
01-321.800	Cable Franchise Fees	255,000	59,820	23.46%	255,000	119,885	47.01%
Total Busine	ess Licenses & Permits:	258,000	61,086	23.68%	256,500	123,644	48.20%
Non-Business Lic	censes & Permit						
01-322.810	Pole Permits	.00	.00	.00	.00	.00	.00
01-322.820	Street-Opening Permits	4,000	1,250	31.25%	2,000	575	28.75%
)1-322.830	Curbing Permits	100	.00	.00	100	.00	.00
01-322.840	Moving Permits	2,000	540	27.00%	2,000	580	29.00%
)1-322.850	Solicitation Permit	200	.00	.00	100	200	200.00%
Total Non-B	Business Licenses & Permit:	6,300	1,790	28.41%	4,200	1,355	32.26%
Fines							
01-331.110	Vehicle Code Violations	70,000	12,617	18.02%	78,000	13,407	17.19%
01-331.120	Violation of Ordinances	20,000	4,046	20.23%	25,000	6,246	24.98%
01-331.130	State Police Fines	8,600	.00	.00	7,500	.00	.00
Total Fines:		98,600	16,664	16.90%	110,500	19,653	17.79%
nterest							
01-341.000	Interest Income	29,000	14,462	49.87%	9,000	3,439	38.21%
01-341.100	Lien Interest Income	12,000	5,434	45.28%	9,000	6,967	77.41%
Total Interes	st:	41,000	19,896	48.53%	18,000	10,406	57.81%
Rents & Royalties	S						
01-342.100	Verizon Lease Payments	29,686	11,847	39.91%	29,479	11,776	39.95%
01-342.200	Magistrate Office Rent	31,762	13,104	41.26%	31,137	12,844	41.25%
01-342.300	Franko Lease Payments	.00	.00	.00	8,400	3,500	41.67%
Total Rents	& Royalties:	61,448	24,951	40.60%	69,016	28,120	40.74%
Federal Grants							
-euerai Granis							
1-351.025	Bulletproof Vest Grant	1,800	.00	.00	1,800	.00	.00

Period: 04/19 2019-19 04/19 2018-18 04/18 **Current Year** Current YTD % of Prior Year Prior YTD % of Account Number Account Title Budget Actual Budget Budget Actual Budget .00 Total Federal Grants: 1,800 .00 .00 1,800 .00 **State Grants** 01-354.010 Police Dept Grants 10,000 .00 .00 .00 2,783 .00 .00 01-354.020 Safe Streets Grant .00 .00 .00 .00 .00 2,000 2,000 352 17.60% 01-354.025 Seatbelt Program Grant .00 .00 01-354.026 Child Passenger Safety Grant 1.200 .00 .00 1,200 .00 .00 01-354.030 SO/Aggressive Driver Grant 4,000 375 9.37% 4,000 1,124 28.09% 01-354.040 35,000 33,093 94.55% 35,000 .00 Recycling Grant .00 01-354.050 Keystone Grant .00 .00 .00 .00 .00 .00 01-354.100 Main Street Initiatives Grant .00 .00 .00 .00 .00 .00 01-354.110 .00 .00 .00 .00 .00 .00 CDBG: Storm Sewer Improv Proj .00 .00 .00 01-354.120 Gaming Grant: Casino Corr Traf 6,800 .00 118,343 .00 .00 .00 01-354.130 Regional Init Grant: SALDO Upd .00 .00 .00 Park Development Grant(s) .00 .00 .00 637,000 60,000 9.42% 01-354.140 01-354.150 PA DEP Grant(s) .00 .00 .00 .00 .00 .00 **Total State Grants:** 59,000 33,468 56.72% 797,543 64,259 8.06% State-Shared Revenue 01-355.010 Public Utility Tax 6,100 .00 .00 6,100 .00 .00 01-355.040 Beverage License & Tavern Tax 4,600 1,520 33.04% 5,000 2.804 56.08% 01-355.050 Foreign Casualty Insurance Tax 346,645 .00 00 344,119 .00 .00 01-355.070 Foreign Fire Insurance Tax 83,000 .00 .00 92,000 .00 .00 Total State-Shared Revenue: 440,345 1,520 447,219 2,804 0.63% 0.35% In Lieu Of Taxes 01-359.100 Payments in Lieu of Taxes 77,000 .00 .00 67,000 67,000 100.00% Total In Lieu Of Taxes: .00 67,000 100.00% 77.000 .00 67,000 Service Fees .00 Fees for Svcs--DPW .00 01-360.100 .00 .00 .00 .00 01-360.200 Fees for Svcs--Police Security 15,000 1,259 15,000 32,243 214.95% 8.39% 18,000 34.06% 01-360.250 Fees for Svcs--Police SRO 64,000 21,800 54,000 33.33% Fees for Svcs--Finance 15,000 3.40% 01-360.300 2,000 1,124 56.18% 509 Total Service Fees: 81,000 24,182 29.85% 84,000 50,752 60.42% **Review Fees** 250 3,000 26.33% 01-361.310 Planning & SALDO Fees 8,000 3.13% 790 01-361.330 10.000 7.520 75.20% 12,000 2,720 22.67% Zoning Appeals & Fees 01-361.340 **Building Code Appeal** 1,500 00 1,500 00 .00 00 10,000 7,941 10,000 3,825 38.25% 01-361.350 Legal Review Fees 79.41% 20,051 30,000 **Engineering Review Fees** 30,000 66.84% 11,960 39.87% 01-361.360 Sale - Maps/Copies/Publication 254 50.80% 01-361.500 500 140 27.95% 500 Total Review Fees: 60,000 35,902 59.84% 57,000 19,549 34.30% **Public Safety Fees** 95.00% 01-362.130 Security Alarm Monitoring 500 150 30.00% 500 475 01-362.410 **Building Permits** 60,000 19.795 32.99% 60,000 22,440 37.40% 15.76% 01-362,415 Mechanical Permits 15,000 2,695 17.97% 15,000 2,365 **Electrical Permits** 30,000 10,389 34.63% 30,000 5,732 19.11% 01-362.420

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
01-362.430	Plumbing Permits	20,000	4,713	23.57%	20,000	3,933	19.66%
01-362.440	On-Site Sewage Permits	10,000	2,110	21.10%	5,000	5,460	109.20%
01-362.450	Re-Inspection Fee	4,000	1,780	44.50%	5,000	500	10.00%
01-362.460	PA State Fee - Permits	.00	.00	.00	.00	.00	.00
01-362.470	Fire Inspection Fee	45,000	15,330	34.07%	35,000	18,600	53.14%
01-362.480	Rental/Resale Inspection Fee	35,000	210	0.60%	.00	.00	.00
Total Public	c Safety Fees:	219,500	57,172	26.05%	170,500	59,504	34.90%
Snow Removal							
01-363.510	PennDOT/Snow Removal	2,050	.00	.00	1,700	.00	.00
Total Snow	Removal:	2,050	.00	.00	1,700	.00	.00
Sanitation Fees							
01-364.300	Solid Waste Collection	.00	.00	.00	.00	.00	.00
01-364.400	Freon Decal Sales	.00	.00	.00	.00	.00	.00
01-364.500	Recycling Container Sales	.00	.00	.00	.00	.00	.00
01-364.600	Recycling Proceeds	.00.	.00	.00	.00	.00	.00
01-364.700	PA Refuse Surcharge	.00.	.00.	.00.	.00	.00	.00.
01-364.900	Scrap Metal Recycling	3,000	213	7.09%	1,800	1,443	80.14%
Total Sanita	ation Fees:	3,000	213	7.09%	1,800	1,443	80.14%
Membership Fee							
01-365.600	Ambulance Subscriptions	24,000	10,521	43.84%	24,000	6,915	28.81%
Total Memb	pership Fees:	24,000	10,521	43.84%	24,000	6,915	28.81%
Recreational Use							
01-367.200	Fundraising Proceeds	.00	.00	.00	.00.	.00	.00.
01-367.500	Park/Pavilion Usage Fee	2,400		13.54%	2,400		22.92%
Total Recre	ational User Fees:	2,400	325	13.54%	2,400	550	22.92%
Miscellaneous							
01-380.000	Miscellaneous Revenue	600	195	32.58%	400	233	58.22%
01-380.001	Misc Utility Revenue	12,000	160	1.33%	12,000	12,895	107.46%
01-380.002	Utility Lien Fees	5,000	1,401-	-28.02%	.00	.00	.00
01-380.003	Utility Legal Fees	2,000	2,210	110.50%	.00	.00	.00
01-380.004	Utility Certification Fees	5,000	1,660	33.20%	.00.	.00	.00
01-380.100	Forfeited/Returned Deposits			50.00%		25	.00
Total Misce	llaneous:	24,700	2,874	11.64%	12,400	13,153	106.07%
Special Assessm							
01-383.100	Curbing Assessments	.00	.00	.00	.00	.00	.00
Total Speci	al Assessments:	.00	.00	.00	.00	.00	.00
Contributions							
01-387.000	Contributions	65,000	.00	.00	89,800	65,000	72.38%
01-387.215	ContributionsPolice/K-9 Prog	1,000	.00	.00	1,800	.00	.00
	ibutions:	66,000	.00	.00	91,600	65,000	70.96%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
	7,000411,7110						
Asset Disposal							=
01-391.100	Sale of Fixed Assets	.00.	307	.00	4,000	29,677	741.93%
01-391.200	Loss of Fixed Assets	.00	.00	.00	.00	.00	.00
Total Asset	Disposal:	.00.	307	.00	4,000	29,677	741.93%
Interfund Transfe	ers						
01-392.030	Transfer From Fire Fund	.00	.00	.00	.00	.00	.00
01-392.060	Transfer From Water Fund	107,000	.00	.00	135,300	.00.	.0
01-392.080	Transfer From Sewer Fund	107,000	.00	.00	135,300	.00	.0
01-392.100	Tr fr Refuse & Recycling Fund	119,000	.00	.00	184,200	.00	.0
01-392.910	Transfer From Developers Fund	300	513	171.16%	300	229	76.37%
Total Interfu	und Transfers:	333,300	513	0.15%	455,100	229	0.05%
Prior Year Proce	eds						
01-393.120	Note Proceeds	.00	.00	.00	.00	.00	.00
01-393.121	Bond Proceeds	.00	.00	.00	.00	.00	.00
01-393.200	Premium on Bonds Sold	.00	.00	.00	.00	.00	.00
Total Prior `	Year Proceeds:	.00	.00	.00	.00	.00	.00
Prior Year Exp							
01-395.000	Refund of Prior Year Expenses	.00	325	.00	.00	.00	.00
01-395.100	Refund of Unused Premiums	25,000	47,917	191.67%	60,000	.00	.00
Total Prior \	Year Exp.	25,000	48,242	192.97%	60,000	.00	.00
	·						
Prior Year Reserv 01-396.000	ves Prior Year Reserves	151,995	.00	.00	404,134	.00	.00
Total Prior `	Year Reserves:	151,995	.00	.00	404,134	.00	.00
Legislative						***************************************	
01-400.113	Commissioners	13,000	4,500	34.62%	13,000	4,500	34.62%
01-400.200	Volunteer & Public Events	1,500	1,026	68.39%	1,500	548	36.54%
01-400.240	Supplies	600	20	3.33%	300	85	28.37%
01-400.420	Dues/Subscriptions/Memberships	3,300	3,270	99.09%	3,300	3,270	99.09%
01-400.460	Conferences & Training	100	.00	.00	100	.00	.00
Total Legisl	ative:	18,500	8,816	47.65%	18,200	8,403	46.17%
Executive		***************************************					
01-401.121	Manager	45,960	13,829	30.09%	44,200	13,729	31.06%
01-401.122	Assistant Manager	.00	.00	.00	.00	.00	00.
01-401.139	Caretaker	.00.	.00	.00	.00	.00	.00
01-401.141	ClericalFull Time	112,800	34,147	30.27%	109,500	33,699	30.78%
01-401.149	ClericalPart Time	21,200	6,910	32.60%	30,000	5,776	19.25%
01-401.212	Newsletter	5,600	5,650	100.90%	5,000	5,617	112.34%
01-401.231	Vehicle Fuel	1,800	212	11.79%	1,200	550	45.87%
01-401.240	Office Supplies	4,500	917	20.39%	4,000	826	20.64%
01-401.251	Vehicle Maintenance	500	.00	.00	500	85	17.00%
01-401.260	Minor Equipment & Small Tools	1,000	.00	.00	1,000	.00	.0
01-401.312	Consulting Services	2,880	1,800	62.50%	20,000	1,660	8.30%
01-401.315	General Services	10,700	1,694	15.83%	8,000	1,804	22.55%
01-401.320	Telephone	16,140	4,953	30.69%	16,800	5,444	32.40%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
01-401.325	Postage	8,000	2,892	36.15%	8,000	2,236	27.95%
01-401,341	Advertising	4,000	1,856	46.39%	4,000	1,262	31.55%
01-401.342	Printing	1,000	.00	.00	1,000	196	19.57%
01-401.343	Right-to-Know Request Fees	.00	57	.00	.00	.00	.00
01-401.375	Equip Maint & Lease Agreements	11,000	1,801	16.37%	10,000	3,146	31.46%
01-401.420	Dues/Subscriptions/Memberships	2,900	540	18.62%	3,800	867	22.82%
01-401.421	Training	1,000	319	31.90%	1,000	582	58.16%
01-401.460	Conferences	1,800	.00	.00	.00	.00	.00
01-401.700	Capital Equipment	.00	.00	.00	.00	.00	.00
Total Execu	ıtive:	252,780	77,577	30.69%	268,000	77,479	28.91%
Finance							
01-402.122	Finance Director	89,200	26,990	30.26%	85,000	25,648	30.17%
01-402.123	Accounting Supervisor	74,250	21,880	29.47%	70,600	20,787	29.44%
01-402.240	Supplies	2,000	293	14.63%	1,000	750	75.00%
01-402.260	Minor Equipment & Small Tools	1,000	.00	.00	500	.00	.00
01-402.311	Auditing & Accounting Services	12,000	4,000	33.33%	12,000	.00	.00
01-402.320	Telephone	660	165	25.03%	660	165	24.99%
01-402.420	Dues/Subscriptions/Memberships	900	627	69.67%	750	569	75.89%
01-402.421	Training	1,000	200	19.96%	1,000	150	15.00%
01-402.460	Conferences	1,800	425	23.61%	.00	.00	.00.
Total Finan	ce:	182,810	54,578	29.86%	171,510	48,069	28.03%
Tax Collection							
01-403.114	Tax Collector	10,000	3,462	34.62%	10,000	3,462	34.62%
01-403.240	Supplies	500	198	39.62%	500	82	16.41%
01-403.260	Minor Equipment & Small Tools	.00	.00	.00	.00	.00	.00.
01-403.325	Postage	4,200	756	18.01%	4,200	2,708	64.47%
01-403.353	Tax Collector's Bond	580	428	73.79%	550	428	77.82%
01-403.420	Dues/Subscriptions/Memberships	100	.00.	.00.	100	.00	.00.
01-403.452	EIT Collection Fee	30,550	7,987	26.14%	29,900	8,254	27.61%
01-403.453	LST Collection Fee	11,700	2,631	22.49%	11,700	2,632	22.49%
01-403.454	Real Estate Tax Collections	3,100		1.35%	3,100	3,074	99.15%
Total Tax C	ollection:	60,730	15,504	25.53%	60,050	20,639	34.37%
Legal	Township Solicitor	E0 000	04.500	49 4 40/	E0 000	7 445	14.000/
01-404.310	•	50,000	21,568	43.14%	50,000	7,115	14.23%
01-404.314	Special Legal & Consulting Svc	3,500	217	6.19%	3,000	.00	.00
01-404.318	Reimbursable Legal Services	10,000	2,647	26.47%	10,000	3,657	36.57%
01-404.410	Judgements & Settlements	.00	.00	.00	.00	.00	.00.
Total Legal:		63,500	24,431	38.47%	63,000	10,773	17.10%
Personnel Admir			_				
01-406.150	Compensated Absences Expense	.00.	.00	.00	.00.	.00	.00.
01-406.171	HRA Employee Reimbs	9,000	959	10.66%	9,000	1,928	21.43%
01-406.172	Retirement Incentive Pmts	30,720	12,800	41.67%	30,720	7,680	25.00%
01-406.240	Supplies & Minor Equipment	500	153	30.70%	500	.00	.00.
01-406.314	Special Legal & Consult'g Svcs	1,500	2,767	184.43%	5,000	496	9.92%
01-406.315 01-406.341	General Services	7,000	.00	.00	5,000	3,297	65.93%
D3_4D6_341	Advertising	3,000	508	16.94%	2,000	1,329	66.46%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total Perso	onnel Admin:	51,720	17,187	33.23%	52,220	14,730	28.21%
Data Processing							
01-407.261	Computer Equip & Software	9,500	14,256	150.06%	10,500	7,065	67.28%
01-407.319	Computer Maint & Support	33,000	16,483	49.95%	19,000	13,481	70.95%
Total Data	Processing:	42,500	30,739	72.33%	29,500	20,546	69.65%
Engineering							
01-408.313	Township Engineer	10,000	10,445	104.45%	20,000	5,252	26.26%
01-408.314	Special Engineering Services	.00	.00	.00	500	.00	.00
01-408.318	Reimbursable Engineering Svcs	30,000	4,287	14.29%	30,000	5,691	18.97%
Total Engin	eering:	40,000	14,732	36.83%	50,500	10,943	21.67%
Buildings & Plan	t						
01-409.144	Maintenance Custodian	53,000	16,984	32.05%	47,200	15,821	33.52%
01-409.192	Work Boot & Clothing Allowance	250	.00	.00	250	.00	.00
01-409.230	Heating Fuel	34,000	23,440	68.94%	33,000	30,258	91.69%
01-409.232	Generator Fuel	600	1,217	202.81%	600	285	47.48%
01-409.240	Supplies	4,200	786	18.71%	3,500	1,558	44.53%
01-409.260	Minor Equipment & Small Tools	1,000	.00	.00	2,000	.00	.00
01-409.320	TELEPHONE	660	165	25.03%	.00	.00	.00
01-409.361	Electric	30,000	8,259	27.53%	30,000	9,255	30.85%
01-409.366	Water	2,220	538	24.24%	2,100	660	31.42%
01-409.373	Facilities Maintenance	25,000	12,229	48.92%	30,000	8,879	29.60%
01-409.600	Capital Construction	.00	.00	.00	36,000	.00	.00
01-409.601 01-409.699	Stormwater Basin Retrofit Capital Reserve	.00 .00	.00 .00	.00 .00	.00 .00	.00 .00	00. 00.
Total Buildii	ngs & Plant:	150,930	63,618	42.15%	184,650	66,716	36.13%
Police			-		***************************************		
01-410.122	Police Chief	106,780	32,129	30.09%	103,700	31,898	30.76%
01-410.130	Sergeants	275,250	81,571	29.64%	267,000	80,655	30.21%
01-410.131	Detective & Corporals	346,120	103,995	30.05%	419,000	75,745	18.08%
01-410.132	Police Officers	899,770	267,585	29.74%	780,000	249,166	31.94%
01-410.139	Police Officers - Part Time	45,000	3,618	8.04%	45,000	12,460	27.69%
01-410.140	Animal Control Officer	10,000	3,042	30.42%	10,000	3,080	30.80%
01-410.141	ClericalFull Time	101,060	30,410	30.09%	107,400	25,195	23.46%
01-410.148	Crossing Guards	23,000	8,620	37.48%	24,000	8,412	35.05%
01-410.181	Double Time	6,000	1,849	30.82%	2,000	1,624	81.22%
01-410.182	Longevity	17,900	.00	.00	17,500	.00	.00.
01-410.183	Overtime	70,000	32,568	46.53%	70,000	48,455	69.22%
01-410.184	Reimbursable Overtime	30,000	.00	.00 22.72%	30,000	.00	.00 29.54%
01-410.185 01-410.186	Holiday Pay Shift Differential	40,000 13,500	9,087 478	3.54%	35,000 10,000	10,341 394	3.94%
01-410.187	College Credit Compensation	3,000	218	7.25%	2,500	216	8.64%
01-410.188	Court Time	15,000	3,169	21.12%	10,000	4,130	41.30%
01-410.189	K-9 Care Compensation	3,000	.00	.00	3,000	.00	.00
01-410.191	Uniform Maintenance Allowance	4,800	5,325	110.94%	4,600	4,320	93.91%
01-410.200	Community Programs & Events	3,500	.00	.00	3,500	.00	.00.
01-410.205	Bike Patrol	.00	.00	.00	.00	.00	.00
01-410.210	QRS Supplies	15,700	3,634	23.15%	15,700	2,079	13.24%
	• •	, -	1,837	24.17%	7,600	2,099	27.62%

Period: 04/19

		2019-19 Current Year	04/19 Current YTD	% of	2018-18 Prior Year	04/18 Prior YTD	% of
Account Number	Account Title	Budget	Actual	Budget	Budget	Actual	Budget
01-410.220	MERT Program	.00	.00	.00	2,000	.00	.00
01-410.231	Vehicle Fuel	45,000	15,491	34.42%	38,000	16,007	42.12%
01-410.238	Uniforms	11,550	2,378	20.58%	10,500	2,457	23.40%
01-410.240	Office Supplies	3,500	1,733	49.52%	3,500	1,703	48.66%
01-410.241	Patrol Supplies	3,100	97	3.11%	3,100	740	23.88%
01-410.242	Firearms Supplies	10,850	2,057	18.96%	10,850	2,015	18.57%
01-410.243	Animal Control Supplies	1,000	.00	.00	1,000	175	17.50%
01-410.250	Police Vehicles	.00	.00	.00	40,000	58,984	147.46%
01-410.251	Vehicle Maintenance	27,000	10,533	39.01%	27,000	10,256	37.99%
01-410.260	Minor Equipment & Small Tools	25,000	793-	-3.17%	25,000	15,549	62.20%
01-410.261	Comp Equip & SWPD	10,000	4,294	42.94%	16,000	13,036	81.47%
01-410.315	General Services	5,600	1,596	28.49%	5,600	1,680	29.99%
01-410.317	Contracted SvcsAnimal Contrl	10,000	2,500	25.00%	10,000	2,500	25.00%
01-410.319	Comp Maint & SupportPD	45,000	18,434	40.97%	48,000	17,843	37.17%
01-410.320	Telephone	10,800	2,626	24.31%	9,000	2,566	28.51%
01-410.342	Printing	2,250	1,038	46.14%	2,250	330	14.67%
01-410.375	Equipment Maintenance	5,000	2,999	59.98%	5,000	1,996	39.92%
01-410.420	Dues/Subscriptions/Memberships	2,500	730	29.20%	2,500	1,273	50.94%
01-410.421	Training	15,000	7,010	46.74%	15,000	1,715	11.43%
01-410.460	Conferences	1,800	.00	.00	1,800	.00	.00
01-410.700	Capital Equipment	.00	.00	.00	269,155	.00	.00
Total Police	<b>9</b> :	2,271,930	661,855	29.13%	2,512,755	711,094	28.30%
Fire							
01-411.139	Fire InspectorPart Time	52,600	19,315	36.72%	59,000	22,938	38.88%
01-411.238	Fire Inspector Uniforms	6,000	114	1.90%	6,000	362	6.03%
01-411.240	Supplies & Minor Equipment	7,000	.00	.00	7,000	862	12.31%
01-411.320	Telephone	750	103	13.72%	750	103	13.72%
01-411.421	Fire Inspector Training	1,250	600	48.00%	1,250	895	71.61%
01-411.540	Firemen's Relief Contribution	83,000	.00	.00	92,000	.00	.00
Total Fire:		150,600	20,132	13.37%	166,000	25,159	15.16%
Code Enforceme							
01-413.131	Code Enforcement Officer/BCO	68,250	21,099	30.91%	66,300	20,387	30.75%
01-413.139	Code Enf InspPart Time	31,200	546	1.75%	10,000	.00	.00
01-413.141	ClericalFull Time	.00	.00	.00	.00	.00	.00.
01-413.306	Commercial Inspection Service	5,000	1,908	38.15%	5,000	770	15.41%
01-413.307	Residential Inspection Service	10,000	2,610	26.10%	10,000	4,635	46.35%
01-413.308	Plan & Drawing Review	50,000	9,919	19.84%	40,000	14,930	37.33%
01-413.317	C/SWeed & Code Violations	.00	.00	.00	.00	175-	.00.
01-413.320	Telephone	660	195	29.58%	600	165	27.49%
01-413.420	Dues/Subscriptions/Memberships	210	.00	.00	200	210	105.00%
01-413.421 01-413.460	Training Conferences	1,000 .00	.00 .00	.00 .00	1,000 .00	.00	1.26% .00
	Enforcement:	166,320	36,276	21.81%	133,100	40,935	30.76%
						-10,000	
Planning & Zonii 01-414.113	ng Zoning Hearing Board	1,000	130	13.00%	2,000	230	11.50%
01-414.122	Planning & Zoning Officer	89,500	26,938	30.10%	87,000	26,744	30.74%
	ClericalFull Time	52,100	15,676	30.09%	51,000	15,563	30.52%
01-414.141		1,080	310	28.70%	1,500	490	32.67%
	Planning Commission						
01-414.141 01-414.220 01-414.240	Planning Commission Supplies	3,000	2,959	98.63%	3,000	267	8.88%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
01-414.310	Planning Solicitor	5,000	481	9.62%	5,000	221	4.42%
01-414.314	Zoning Solicitor	5,000	488	9.75%	5,000	2,013	40.25%
01-414.315	General Services	400	.00	.00	200	.00	.00
01-414.316	Stenograhper	2,500	150	6.00%	2,500	628	25.12%
01-414.317	Contracted Services	.00	.00	.00	.00	.00	.00
01-414.318	DCED/UCC Fees	1,500	.00	.00	1,800	239	13.25%
01-414.320	Telephone	660	165	25.03%	660	165	24.99%
01-414.341	Advertising	1,500	700	46.67%	1,500	380	25.31%
01-414.342	Printing	500	.00	.00	1,500	475	31.67%
01-414.420	Dues/Subscriptions/Memberships	102	477	467.65%	200	102	51.00%
01-414.421	Training	300	.00	.00	300	30	10.00%
01-414.460	Conferences	.00	.00	.00	.00	.00	.00
Total Plann	ning & Zoning:	164,642	48,473	29.44%	164,660	47,545	28.87%
Emergency Mana	•						/
01-415.139	Emergency Mgmt Coordinator	3,000	1,333	44.45%	3,000	1,000	33.33%
01-415.149	Emergency Mgmt Deputy	1,000	333	33.33%	1,000	.00	.00
01-415.240	Supplies	.00	.00	.00	500	.00	.00.
01-415.260	Minor Equipment & Small Tools	5,000	.00	.00	1,000	987	98.69%
01-415.320	Telephone	5,100	652	12.79%	2,100	639	30.42%
01-415.421	Training	1,000	.00	.00	1,000	.00.	.00
01-415.700	Cap EquipEmergency Mgmt	.00.	.00	.00	.00.	.00	.00
Total Emer	gency Management:	15,100	2,319	15.36%	8,600	2,626	30.53%
Sanitation						0.0	0.0
01-427.147	Recycling Center - Part Time	.00	.00	.00	.00	.00	.00
01-427.300	Refuse Collection Services	.00	.00	.00	.00	.00	.00
01-427.301	PA Refuse Collection Surcharge	.00.	.00 .00	.00 .00	.00 .00	.00. 00.	.00. 00.
01-427.302 01-427.303	Recycling Costs Grass Collection Fees	.00	.00	.00	.00	.00.	.00.
01-427.303	Collection Agency Services	.00	.00	.00	.00	.00	.00
Total Sanita	ation:	.00	.00	.00	.00	.00	.00
DPW - Wages		-					
01-430.122	Public Works Director	40,870	12,296	30.09%	39,680	12,208	30.77%
01-430.130	DPWHighway Supervisor	76,000	23,893	31.44%	77,100	22,700	29.44%
01-430.141	ClericalFull Time	55,840	16,801	30.09%	54,200	16,680	30.78%
01-430.143	DPW - Full Time	964,670	252,518	26.18%	942,000	276,345	29.34%
01-430.149	DPW - Part Time	.00	.00	.00	.00	.00	.00
01-430.181	Double Time	6,000	4,953	82.55%	6,000	3,117	51.95%
01-430.183	Overtime	32,000	27,698	86.56%	27,000	22,946	84.99%
01-430.189	On-Call	11,000	8,094	73.58%	12,400	7,672	61.87%
01-430.192	Work Boot & Clothing Allowance	5,500	359	6.52%	5,500	1,414	25.70%
01-430.231	Vehicle Fuel	47,000	16,114	34.28%	41,000	16,557	40.38%
01-430.240	Supplies	10,000	4,885	48.85%	8,000	4,247	53.09%
<del></del>	Minor Equipment & Small Tools	4,000	4,087	102.17%	6,800	5,621	82.67%
01-430.260			.00	.00	2,500	1,384	55.36%
	General Services	2,500	.00	.00			00.0070
01-430.260 01-430.315 01-430.320	General Services Telephone	2,500 1,920				482	
01-430.315		1,920 500	482	25.13%	1,860 500		25.92%
01-430.315 01-430.320	Telephone	1,920			1,860	482	25.92%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total DPW	- Wages:	1,258,300	372,578	29.61%	1,225,040	391,433	31.95%
Composting							
01-431.303	Composting Costs	.00	.00.	.00	.00.	.00	.00
Total Comp	posting:	.00	.00	.00	.00	.00	.00
Traffic Control							
01-433.246	Signs & Street Markings	5,000	1,160	23.21%	5,000	656	13.11%
01-433.317	Contracted Services	.00	.00	.00	.00	.00	.00
01-433.362	Traffic Signals	7,090	6,284	88.64%	7,090	3,270	46.12%
01-433.450	Street Line Painting	7,500	.00	.00	5,000	.00	.00
Total Traffic	c Control:	19,590	7,445	38.00%	17,090	3,925	22.97%
Storm Sewers							
01-436.131	MS4 Coordinator	15,000	4,263	28.42%	15,000	4,616	30.77%
01-436.139	MS4 InspectorPart Time	.00	.00	.00	.00	.00	.00
01-436.240	MS4 Supplies	3,000	7,490	249.66%	3,000	.00	.00
01-436.312	Consulting Svcs	.00	.00	.00	.00	.00	.00
01-436.313	Engineering Svcs	50,000	2,205	4.41%	70,000	14,284	20.41%
01-436.315	MS4 General Services	5,000	.00	.00	5,000	539	10.78%
01-436.370	Repairs & Maint - Storm Sewers	14,000	249	1.78%	14,000	987	7.05%
01-436.600	Capital ConstructionStorm Sw	.00	.00	.00	.00	.00	.00
01-436.601	Public/Kline Drainage Proj	.00	.00	.00	.00	.00	.00
01-436.700	Capital Equipment	.00.	.00	.00	.00	.00	.00
Total Storm	Sewers:	87,000	14,207	16.33%	107,000	20,426	19.09%
Tools & Machine	ery						
01-437.251	Vehicle Maintenance	25,000	6,471	25.89%	22,500	16,799	74.66%
01-437.375	Equipment Maintenance	22,500	23,573	104.77%	15,000	8,098	53.99%
Total Tools	& Machinery:	47,500	30,044	63.25%	37,500	24,898	66.39%
Streets & Bridge	s						
01-438.245	Road Materials	210,000	3,777	1.80%	160,000	6,727	4.20%
01-438.450	Road Program	.00.	.00	.00	.00	.00	.00
Total Street	ts & Bridges:	210,000	3,777	1.80%	160,000	6,727	4.20%
Capital							
01-439.600	Capital Construction	.00	.00	.00	.00	.00	.00
01-439.601	Dauphin Street Traffic Light	.00	.00	.00	.00	.00	.00
01-439.700	Capital Equipment	.00	.00	.00	13,000	.00	.00
Total Capita	al:	.00	.00	.00	13,000	.00	.00.
Storm Water Mg	mt			<u></u>			
01-446.370	Storm Water Management	.00	.00	.00	.00	.00	.00
01-446.603	Gilmore Street Drainage	.00	.00	.00	.00	.00	.00
	Filbert St Stormwater Project	.00	.00	.00	.00	.00	.00
01-446.604	i libert of otorniwater i rolest						
01-446.604 01-446.605	Maumee Ave Stormwater Project	.00	.00	.00	.00	.00	.00

## BOC Financial Report-- April 2019

Period: 04/19

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total Storn	n Water Mgmt:	.00	.00	.00	.00	.00	.00
Recreation Adm	in						
01-451.240	Supplies	3,000	620	20.66%	3,000	.00	.00
01-451.260	Minor Equipment & Small Tools	4,000	.00	.00	9,600	.00.	.00
01-451.312	Consulting SvcsLindberg MSP	.00	7,532	.00	48,000	27,820	57.96%
01-451.313	Consulting SvcsLaubach MSP	.00	.00	.00	.00	6,973	.00
01-451.315	General Services	3,000	857	28.57%	3,500	806	23.04%
01-451.361	Electric	2,800	423	15.12%	3,100	502	16.21%
01-451.371	Property Maint (Grounds)	7,000	2,141	30.58%	7,000	449	6.42%
01-451.373	Facilities Maint (Structures)	2,000	.00	.00	2,000	.00	.00
01-451.375	Equipment Maintenance	3,000	3,569	118.98%	3,000	.00	.00
01-451.600	Capital Construction-Parks	.00	.00	.00	.00	.00	.00
01-451.601	Lindberg Pk MSP: Park Improvmt	.00	.00	.00	869,000	1,164	0.13%
01-451.602	Lindberg Pk MSP: Bike Ln Proj	.00	.00	.00	.00	.00	.00
01-451.603	Laubach Pk MSP: Park Improvmt	.00	.00	.00	.00	.00	.00
01-451.700	Capital Equipment	.00	.00	.00	.00.	.00	.00.
Total Recre	eation Admin:	24,800	15,142	61.06%	948,200	37,715	3.98%
Participant Recr	eation						
01-452.129	Recreation Director	15,600	5,400	34.62%	15,600	5,400	34.62%
01-452.149	Seasonal Employees	20,000	.00	.00	21,500	.00	.00
01-452.200	Community Events	1,200	.00	.00	1,000	341	34.10%
01-452.240	Supplies & Minor Equipment	5,000	.00	.00	5,000	.00	.00.
01-452.315	Fundraising Services	.00	.00	.00	.00.	.00	.00
01-452.320	Telephone	660	165	25.03%	600	165	27.49%
Total Partic	cipant Recreation:	42,460	5,565	13.11%	43,700	5,906	13.51%
Environmental A	Advisory						
01-461.240	Supplies	500	.00	.00	500	.00	.00
01-461.317	Contracted Svc	.00	.00	.00	.00	.00	.00
01-461.341	Advertising	400	.00	.00	400	98	24.38%
01-461.342	Printing	200	.00	.00	200	.00	.00
01-461.420	Dues/Subscriptions/Memberships	450	.00	.00	200	440	220.00%
01-461.421	Training	.00	.00	.00		.00	.00.
Total Enviro	onmental Advisory:	1,550	.00	.00	1,500	538	35.83%
Contributions	Overtedial Oc.	40.000		47.0401	10.00-	40.000	400 ====
01-465.309	Custodial Services	10,000	1,704	17.04%	10,000	10,250	102.50%
01-465.500	Volunteer Fire Co Incentives	40,000	.00	.00.	36,000	.00	.00.
)1-465.540	Contributions	11,350		2.20%	11,300		2.21%
Total Contr	ibutions:	61,350	1,954	3.19%	57,300	10,500	18.33%
Debt Service - Pi	-						
)1-471.350	Principal - Lease Pmt	.00.	.00	.00	.00	.00	.00
Total Debt	Service - Principal:	.00	.00	.00	.00	.00	.00
Debt Service - In							
)1-472.350	Interest - Lease Pmt	.00	.00	.00	.00	.00	.00

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total Debt	Service - Interest:	.00	.00	.00	.00	.00	.00
Other Expenditu	Irac						
01-474.430	Real Estate Taxes	.00.	406	.00	.00	406	.00
Total Othe	r Expenditures:	.00	406	.00	.00	406	.00
Fiscal Agent Fee	as a						
01-475.000	Fiscal Agent Fees	.00	.00	.00	.00	.00	.00
Total Fisca	al Agent Fees:	.00	.00	.00	.00	.00	.00
Fees & Miscella	nooue	,					•
01-480.000	Miscellaneous Expense	300	110	36.67%	300	102	34.00%
01-480.001	Disability Pay Expense	.00	114	.00	.00	.00	.00
01-480.005	Financial Service Fees	.00	40	.00	90	.00	.00
01-480.010	Credit Card Service Fees	.00.	.00	.00	.00	.00	.00
Total Fees	& Miscellaneous:	300	264	87.85%	390	102	26.16%
Ingurance							
Insurance 01-486.351	Insurance - Commercial	48,000	.00	.00	48,000	47,307	98.56%
01-486.351	Insur - Business Auto	65,000	.00 67,444	.00 103.76%	59,000	62,781	106.41%
01-486.353	Insur - Professional	55,000	56,233	103.70%	51,000	54,560	106.98%
01-486.354	Insur - Workers Comp	110,000	.00	.00	106,000	129,504	122.17%
01-486.355	Insur - Twp Official's Bond	4,000	3,525	88.13%	4,500	3,953	87.84%
Total Insur	ance:	282,000	127,202	45.11%	268,500	298,106	111.03%
Employee Benef	iits						
01-487.156	Insurance - Health	1,012,000	426,219	42.12%	930,000	377,569	40.60%
01-487.158	Insurance - Life & Disability	22,680	8,674	38.25%	26,000	9,535	36.67%
01-487.160	Pension	869,342	.00	.00	795,197	.00	.00
01-487.161	Social Security Tax	186,754	55,973	29.97%	180,000	55,898	31.05%
01-487.162	Unemployment Compensation	.00	.00	.00	.00	.00	.00
Total Empl	oyee Benefits:	2,090,776	490,865	23.48%	1,931,197	443,002	22.94%
Reserves							
01-490.740	Transfer to Capital Reserves	.00.	.00	.00	8,400	.00	.00
Total Rese	rves:	.00	.00	.00	8,400	.00	.00
Prior Year	Defineds of Drive V and Drive V		00	22	22	20	
01-491.000	Refunds of Prior Year Revenue	.00		.00	.00	.00	.00.
Total Prior	Year:	.00		.00	.00	.00	.00.
Interfund Transf							
01-492.010	Transfer to Capital Fund	200,000	.00	.00	.00	.00	.00
01-492.030	Transfer to Fire Fund	63,750	.00	.00	63,200	.00	.00
01-492.060	Transfer to Water Fund	.00	.00	.00	.00	.00	.00
01-492.080	Transfer to Sewer Fund	.00	.00	.00	.00	.00	.00
01-492.200	Transfer to Debt Service Fund	.00	.00	.00	178,650	647	0.36%
01-492.910	Transfer to SubDiv Escrow Fund	.00	.00	.00	.00.	.00	.00

Salisbury Township			l Report April 20 riod: 04/19	19			Page: 12
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total Interfund T	ransfers:	263,750	.00	.00	241,850	647	0.27%
<b>Department: 999</b> 01-999.999 Rou	nding Account	.00	.00	.00	.00	.00	.00
Total Departmen	t: 999:	.00	.00	.00	.00	.00	.00
General Fund Re	venue Total:	8,021,438	1,756,641	21.90%	8,943,412	1,620,581	18.12%
General Fund Ex	penditure Total:	8,021,438	2,145,688	26.75%	8,943,412	2,349,989	26.28%
Net Total Genera	l Fund:	.00	389,047-	.00	.00	729,408-	.00

Salisbury Township			l Report April 20 iod: 04/19	)19			Page: 1
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Fire Fund							
Property Taxes 03-301.100 03-301.400 03-301.600	Real Estate Tax - Current Real Estate Tax - Claims Real Estate Tax - Exceptions	268,000 5,000 300	46,052 1,760 80	17.18% 35.19% 26.67%	271,000 4,000 360	104 2,288 .00	0.04% 57.20% .00
Total Prope	erty Taxes:	273,300	47,891	17.52%	275,360	2,392	0.87%
nterest 03-341.000	Interest Income	3,000	2,125	70.85%	480	193	40.16%
Total Intere	st:	3,000	2,125	70.85%	480	193	40.16%
Miscellaneous 03-380.000	Miscellaneous Revenue	.00	.00	.00	.00	.00	.00
Total Misce	llaneous:	.00	.00	.00	.00	.00	.00
Contributions 03-387.000	Contributions	.00	.00	.00	.00	.00	.00
Total Contri	ibutions:	.00.	.00	.00	.00	.00	.00
Asset Disposal 03-391.100	Sale of Fixed Assets	.00	.00	.00	.00	.00	.00
Total Asset	Disposal:	.00	.00	.00	.00	.00	.00
nterfund Transfe 3-392.010	ers Transfer from General Fund	13,000	.00	.00	63,200	.00	.00
Total Interfu	und Transfers:	13,000	.00	.00	63,200	.00	.00
Prior Year 3-393.122	Loan Proceeds	.00	.00	.00	.00	.00	.00
Total Prior `	Year:	.00	.00	.00	.00	.00	.00
Prior Year Resv 3-396.000	Prior Year Reserves	.00	.00	.00	.00	.00	.00
Total Prior `	Year Resv:	.00	.00	.00	.00	.00	.00
astern Salisbur 3-411.213	y Fire P&I- 2005 Fire Truck	.00	.00	.00	.00	.00	.00
3-411.214	P & I - 2007 Fire Truck	.00	.00	.00	.00	.00	.00
3-411.214 3-411.215	P & I - 2007 Fire Truck	.00.	.00	.00	.00	.00	.00
3-411.215 3-411.216	P & I - 2009 Fire Truck	.00.	.00	.00	.00	.00	.00.
3-411.216 3-411.217	P & I - 2013 Fire Truck	.00.	.00	.00	.00	.00	.00.
3-411.217	Operating Supplies	21,000	.00 8,530	.00 40.62%	.00 18,350	.00 5,408	.00 29.47%
3-411.240 3-411.260	Minor Equip & Small Tools	21,000 8,900	834	40.62% 9.37%	7,400	5,408 616	29.47% 8.32%
J-+11.200	Utilities	13,000	3,500	26.93%	15,000	2,515	16.77%
3-411 320	O LINUCO	15,000	5,500	20.0070	10,000	۷,010	10.11/0
		13 500	19 571	93 12%	12 000	12 567	104 73%
03-411.320 03-411.350 03-411.374	Insurances RepairsMachinery/Equip	13,500 24,315	12,571 2,502	93.12% 10.29%	12,000 27,400	12,567 2,447	104.73% 8.93%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
03-411.600	Capital Construction	.00	.00	.00	.00	.00	.00
03-411.700	Capital Equipment	.00	.00.	.00	.00	.00.	.00
Total Easte	rn Salisbury Fire:	84,615	29,502	34.87%	84,050	25,429	30.25%
Western Salisbu	ry Fire						
03-412.240	Operating Supplies	7,050	1,583	22.45%	7,300	4,973	68.13%
03-412.260	Minor Equip & Small Tools	12,635	1,695	13.42%	14,800	14,792	99.95%
03-412.320	Telephone/Tablet Chgs	4,139	5,478	132.34%	3,000	515	17.15%
03-412.350	Insurances	22,500	17,337	77.05%	24,000	17,486	72.86%
03-412.374	RepairsMachinery/Equip	31,837	9,950	31.25%	28,500	3,920	13.75%
03-412.421	Training	5,250	2,811	53.55%	4,640	1,550	33.41%
03-412.600	Capital Construction	.00	.00	.00	.00	.00	.00
03-412.700	Capital Equipment	.00.	.00	.00	.00.	.00	.00.
Total West	ern Salisbury Fire:	83,411	38,854	46.58%	82,240	43,236	52.57%
Fuel 03-413.231	Vehicle Fuel	10,000	2,534	25.34%	9,000	3,388	37.64%
Total Fuel:		10,000	2,534	25.34%	9,000	3,388	37.64%
Debt Service - Pr	•						
03-471.212	Principal - 1999 Fire Truck	.00	.00	.00	.00	.00	.00
03-471.213	Principal - 2005 Fire Truck	.00	.00	.00	11,065	11,065	100.00%
03-471.214	Principal - 2007 Fire Truck	.00	.00	.00	6,982	6,982	100.00%
03-471.215	Principal - 2009 Fire Truck	.00	.00	.00	10,215	10,215	100.00%
03-471.216	Principal - 2013 Fire Truck	.00	.00	.00	6,329	6,329	100.00%
03-471.217	Principal - 2017 Fire Truck	.00	.00	.00	8,244	8,244	100.00%
Total Debt	Service - Principal:	.00	.00	.00	42,835	42,834	100.00%
Debt Service - In	terest						
03-472.212	Interest - 1999 Fire Truck	.00	.00	.00	.00	.00	.00
03-472.213	Interest - 2005 Fire Truck	.00	.00	.00	518	518	100.06%
03-472.214	Interest - 2007 Fire Truck	,00	.00	.00	740	740	100.00%
03-472.215	Interest - 2009 Fire Truck	.00	.00	.00	1,368	1,368	100.02%
03-472.216	Interest - 2013 Fire Truck	.00	.00	.00	1,393	1,393	100.03%
03-472.217	Interest - 2017 Fire Truck	.00	.00	.00	1,358	3,898	287.01%
Total Debt S	Service - Interest:	.00	.00	.00	5,377	7,918	147.25%
"							
Fees & Miscellan			22	22		25	
03-480.005	Financial Service Fees	.00	.00	.00	.00	.00	.00.
03-480.454	Real Estate Tax Collections	360		1.34%	450	333	73.97%
Total Fees	& Miscellaneous:	360	5	1.34%	450	333	73.97%
Reserves 03-490.740	Transfer to Capital Reserves	110,914	.00	00	115,088	00	00
	·			.00		.00	.00.
Total Reser	ves:	110,914	.00	.00	115,088	.00	.00
Prior Year 03-491.000	Refunds of Prior Year Revenue	.00	.00	.00	.00	1,290	.00

Salisbury Township		BOC Financial Report April 2019 Period: 04/19					Page: 15	
Account Number	er Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget	
Total Prio	or Year:	.00	.00	.00	.00	1,290	.00	
Interfund Trans 03-492.010 03-492.430	i <b>sfers</b> Transfer to General Fund Transfer to Fire Capital Fund	.00.	.00 50,752	.00 .00	.00 .00	.00 .00	.00 .00	
Total Inte	erfund Transfers:	.00	50,752	.00	.00.	.00	.00	
Fire Fund	d Revenue Total:	289,300	50,017	17.29%	339,040	2,585	0.76%	
Fire Fund	d Expenditure Total:	289,300	121,646	42.05%	339,040	124,427	36.70%	
Net Total	l Fire Fund:	.00	71,630-	.00	.00	121,842-	.00	

Salisbury Townsh	nip		l Report April 20 riod: 04/19	019			Page: 16
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Library Fund							
Property Taxes 04-301.100 04-301.400 04-301.600	Real Estate Tax - Current Real Estate Tax - Claims Real Estate Tax - Exceptions	78,000 1,400 50	13,441 514 23	17.23% 36.68% 46.70%	78,000 1,400 100	30 668 .00	0.04% 47.69% .00
Total Prope	erty Taxes:	79,450	13,978	17.59%	79,500	698	0.88%
Interest 04-341.000	Interest Income	2,100	645	30.69%	300	125	41.69%
Total Intere	est:	2,100	645	30.69%	300	125	41.69%
Miscellaneous 04-380.000	Miscellaneous Revenue	.00	.00	.00	.00	.00	.00
Total Misce	ellaneous:	.00	.00	.00	.00	.00	.00
Prior Year Resv 04-396.000	Prior Year Reserves	110	.00	.00	1,980	.00	.00
Total Prior	Year Resv:	110	.00	.00	1,980	.00	.00
Library Services 04-456.305	Allentown Library Services	81,560	.00	.00	81,600	.00	.00
Total Librar	y Services:	81,560	.00	.00	81,600	.00	.00
Fees & Miscellar 04-480.005 04-480.454	neous Financial Service Fees Real Estate Tax Collections	.00 100	.00	.00 1.45%	.00 180	.00 97	.00 53.97%
Total Fees	& Miscellaneous:	100	1	1.45%	180	97	53.97%
<b>Prior Year</b> 04-491.000	Refunds of Prior Year Revenue	.00	.00	.00	.00	.00	.00
Total Prior	Year:	.00	.00	.00	.00	.00	.00
Library Fun	d Revenue Total:	81,660	14,622	17.91%	81,780	823	1.01%
Library Fun	d Expenditure Total:	81,660	1	.00	81,780	97	0.12%
Net Total L	ibrary Fund:	.00	14,621	.00	.00	726	.00

Salisbury Townsh	ip	BOC Financial Report April 2019 Period: 04/19					
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Water Fund							
Interest							
06-341.000	Interest Income	15,600	3,900	25.00%	16,800	4,873	29.01%
Total Intere	st:	15,600	3,900	25.00%	16,800	4,873	29.01%
System Revenue							
06-378.100 06-378.910	Metered Sales Tapping Fees	1,789,000 1,500	504,493 .00	28.20% .00	1,708,000 1,000	440,375 654	25.78% 65.40%
Total Syste	m Revenue:	1,790,500	504,493	28.18%	1,709,000	441,029	25.81%
Miscellaneous							
06-380.000	Miscellaneous Revenue	360	40	11.11%	.00	89	.00
Total Misce	llaneous:	360	40	11.11%	.00	89	.00
Asset Disposal 06-391.100	Sale of Fixed Assets	.00	.00	.00	.00	.00	.00
Total Asset	Disposal:	.00	.00	.00	.00	.00	.00
Prior Year				<del></del>			
06-393.120	Note Proceeds	.00	.00	.00	.00	.00	.00
06-393.121	Bond Proceeds	.00	.00	.00	.00	.00	.00
06-393.200	Premium on Bonds Sold	.00	.00	.00	.00	.00	.00
Total Prior `	Year:	.00	.00	.00	.00.	.00	.00
Prior Year Exp							
06-395.000	Refund of Prior Year Expenses	.00.	.00	.00	.00	.00	.00
Total Prior `	Year Exp:	.00	.00	.00	.00	.00	.00
Prior Year Resv 06-396.000	Prior Year Reserves	.00	.00	.00	702,068	.00	.00
Total Prior `	Year Resv:	.00	.00	.00	702,068	.00	.00
Admin Salaries -	Executive						
06-401.121 06-401.122	Manager Assistant Manager	22,980 .00	6,914 .00	30.09% .00	22,300 .00	6,865 .00	30.78% .00
Total Admir	n Salaries - Executive:	22,980	6,914	30.09%	22,300	6,865	30.78%
Other General Go					•-		
06-406.000	Other General Gov't Admin	.00.	.00	.00	.00	.00	.00
Total Other	General Gov't Admin:	.00	.00	.00	.00	.00	.00
Admin Salaries - 06-430.122	<b>DPW</b> Public Works Director	30,650	9,222	30.09%	29,760	9,156	30.77%
	n Salaries - DPW:	30,650	9,222	30.09%	29,760	9,156	30.77%

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Water System Op	perations						
06-448.130	DPWUtility Supervisor	43,360	13,046	30.09%	42,100	12,952	30.77%
06-448.141	ClericalFull Time	19,570	5,753	29.39%	18,050	5,505	30.50%
06-448.142	Aide to Public Works Director	32,650	9,826	30.09%	31,710	9,755	30.76%
06-448.143	DPW - Full Time	148,770	44,765	30.09%	142,500	44,930	31.53%
06-448.149	DPWPart Time	.00	.00	.00	.00	.00	.00
06-448.181	Double Time	1,500	773	51.53%	1,500	547	36.46%
06-448.183	Overtime	6,000	4,769	79.49%	5,000	3,559	71.19%
06-448.189	On - Call	15,500	4,670	30.13%	15,000	4,702	31.34%
06-448.231	Vehicle Fuel	8,000	2,740	34.24%	8,000	2,922	36.52%
06-448.232	Generator Fuel	.00	.00	.00	.00	.00	.00
06-448.240	Supplies	10,000	1,858	18.58%	19,000	1,668	8.78%
06-448.251	Vehicle Maintenance	5,000	443	8.86%	5,000	506	10.12%
06-448.260	Minor Equipment & Small Tools	7,200	319	4.44%	7,200	484	6.72%
06-448.261	Computer Equip & Software	4,200	4,185	99.65%	4,000	2,166	54.16%
06-448.310	Legal Services	.00	.00	.00	.00	.00	.00
06-448.311	Auditing & Accounting Services	1,500	625	41.67%	1,200	.00	.00
06-448.312	Consulting Services	.00	.00.	.00	.00	.00	.00
06-448.313	Engineering Services	10,000	597	5.97%	10,000	1,530	15.30%
06-448.315	General Services	1,000	60	6.00%	600	570	94.98%
06-448.316	Testing & Calibration Services	5,000	880	17.60%	5,000	840	16.80%
06-448.317	Contracted Services	3,000	.00	.00	3,000	893	29.75%
06-448.318	DEP Annual Fee	.00	10,000	.00	.00	.00	.00
06-448.319	Computer Maint & Support	11,000	6,483	58.93%	9,500	6,741	70.95%
06-448.320	Telephone	1,560	449	28.78%	1,500	404	26.93%
06-448.325	Postage	3,760	654	17.38%	3,500	1,480	42.27%
06-448.342	Printing	2,080	314	15.08%	2,300	.00	.00
06-448.361	Electric	7,000	2,556	36.51%	7,200	1,900	26.39%
06-448.363	Hydrant Rental	14,500	4,498	31.02%	14,000	2,640	18.86%
06-448.367	Water Purchases - LCA/Altn	877,000	163,062	18.59%	854,000	133,322	15.61%
06-448.368	Water Purchases - Bethlehem	4,200	1,310	31.19%	4,500	979	21.77%
06-448.369	Water Purchase-South Whitehall	7,000	.00	.00	7,500	3,375	45.00%
06-448.373	Facilities Maintenance	3,000	.00	.00	4,000	490	12.24%
06-448.375	Equipment Maintenance	2,500	106	4.22%	3,500	418	11.94%
06-448.421	Training	1,000	530	53.00%	1,000	.00	.00
06-448.600	Capital Construction	.00	.00	.00	12,000	2,196	18.30%
06-448.605	Flexer Avenue Waterline	.00	.00	.00	.00	.00	.00
06-448.606	Clearwood Dr Waterline	.00	.00	.00	.00	.00	.00
06-448.607	Ellsworth Water Main	.00	.00	.00	.00	.00	.00
06-448.608	Edgemont Drive Water Main	.00	.00	.00	.00	629	.00
06-448.609	Flexer Pump Stn Generator Repl	.00	.00	.00	.00	.00	.00
06-448.610	Potomac Waterline Replcmt	.00	.00	.00	.00	.00	.00
06-448.611	Water Meter Replacement Proj	.00	.00	.00	.00	61,894	.00
06-448.612	Edgemont, Meadowbrook, Bellair	.00	.00	.00	.00	708	.00
06-448.613	Meadowbrook SOUTH Waterline	.00	.00	.00	.00	464	.00
06-448.614	Bellair Dr Loop Waterline Proj	.00	.00	.00	.00	755	.00
06-448.615	Paxford Rd Waterline Project	.00	.00	.00	225,000	4,536	2.02%
06-448.616	Maumee Ave Waterline Project	.00	.00	.00	250,000	3,901	1.56%
06-448.617	Montgomery St Waterline Projec	.00	.00	.00	100,000	.00	.00
06-448.699	Capital Reserve	.00	.00	.00	.00	.00	.00
06-448.700	Capital Equipment	.00	.00	.00	27,500	.00	.00
06-448.740	Maintenance Reserve	.00	.00	.00	.00	.00	.00
06-448.800	Depreciation	.00.	.00	.00	.00	.00	.00
Total Water	System Operations:	1,256,850	285,269	22.70%	1,845,860	320,360	17.36%

2019-19 04/19 2018-18 04/18 **Current Year Current YTD** % of Prior Year Prior YTD % of Account Number Account Title Budget Actual Budget Budget Actual Budget Debt Service - Principal .00 06-471.202 Principal - 2010 Bond .00 .00 .00 .00 .00 06-471.203 Principal - 2016 Bonds .00 .00 .00 137,750 .00 .00 .00 06-471.350 Principal - Lease Pmt .00 .00 .00 .00 .00 Total Debt Service - Principal: .00 .00 .00 137,750 .00 .00 Debt Service - Interest 06-472.202 Interest - 2010 Bonds .00 .00 .00 .00 .00 .00 .00 73,375 06-472.203 Interest - 2016 Bonds .00 .00 .00 .00 06-472.350 .00 Interest - Lease Pmt .00 .00 .00 .00 .00 Total Debt Service - Interest: .00 .00 .00 73,375 .00 .00 **Fiscal Agent Fees** 06-475.000 Fiscal Agent Fees .00 .00 .00 .00 .00 .00 Total Fiscal Agent Fees: .00 .00 .00 .00 .00 .00 Fees & Miscellaneous 06-480.000 Miscellaneous Expense .00 .00 .00 .00 .00 .00 06-480.005 Financial Service Fees .00 .00 .00 300 .00 .00 06-480.010 6,000 2,277 37.94% Credit Card Service Fees 5,800 3,233 55.74% Total Fees & Miscellaneous: 5,800 3,233 6,300 2,277 55.74% 36.14% Insurance 06-486.351 Insurance - Commercial 2,700 2,700 2,579 95.51% .00 .00 06-486.352 Insurance - Business Auto 4.400 3,747 85.16% 3,300 4,163 126.15% 06-486.354 Insurance - Workers Comp 12,000 .00 .00 11,700 14,214 121.49% 17,700 20,956 Total Insurance: 19,100 3,747 19.62% 118.39% **Employee Benefits** 37,094 06-487.156 41,617 37.50% 92,500 40.10% Insurance - Health 110,975 36.95% 06-487.158 Insurance - Life & Disability 3,100 1,097 35.39% 2,100 776 06-487.160 Pension 95,087 .00 .00 45,667 .00 .00 06-487.161 Social Security Tax 24,555 7,767 31.63% 23,556 7,542 32.02% 163,823 45,412 27.72% Total Employee Benefits: 233,717 50,480 21.60% Department: 490 .00 .00 .00 06-490.740 Transfer to Capital Fund 130,363 .00 .00 .00 .00 .00 Total Department: 490: 130,363 .00 .00 **Prior Year** 06-491.000 Refunds of Prior Year Revenue .00 .00 .00 .00 .00 .00 .00 .00 Total Prior Year: .00 .00 .00 .00 Interfund Transfers .00 .00 06-492.010 Transfer to General Fund 107,000 .00 131,000 .00 Total Interfund Transfers: .00 131,000 .00 .00 107,000 .00

Salisbury Townsh	nip		l Report April 20 iod: 04/19	)19			Page: 20
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Department: 999 06-999.999	Rounding Account	.00	.00	.00	.00	.00	.00
Total Depa	rtment: 999:	.00.	.00	.00	.00.	.00	.00
Water Fund	d Revenue Total:	1,806,460	508,433	28.15%	2,427,868	445,991	18.37%
Water Fund	d Expenditure Total:	1,806,460	358,865	19.87%	2,427,868	405,024	16.68%
Net Total V	Vater Fund:	.00	149,568	.00	.00	40,967	.00

Salisbury Township			l Report April 20 iod: 04/19	019			Page: 21
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Sewer Fund							
Interest 08-341.000	Interest Income	12,000	10.260	86.40%	7,200	3,062	42.52%
			10,369				
Total Intere	st:	12,000	10,369	86.40%	7,200	3,062	42.52%
<b>State Grants</b> 08-354.110	CDBG:Sanitary Swr System Rehab	.00	.00	.00	.00	.00	.00
Total State	Grants:	.00	.00	.00	.00	.00	.00
Sanitation Fees 08-364.110	Tapping Fees	3,000	1,600	53.33%	3,000	.00	.00
08-364.120	Sewer Rent	1,953,000	550,292	28.18%	1,800,000	469,242	26.07%
08-364.301	Sewer Assessments	.00	.00	.00	.00.	.00	.00
08-364.310	Sale of Capacity	.00	.00	.00	.00	.00	.00
Total Sanita	ation Fees:	1,956,000	551,892	28.22%	1,803,000	469,242	26.03%
Miscellaneous 08-380.000	Miscellaneous Revenue	.00	.00	.00	.00	.00	.00
Total Misce	llaneous:	.00	.00	.00	.00	.00	.00
Asset Disposal				7.007			
08-391.100	Sale of Fixed Assets	.00	.00	.00	.00.	.00	.00
Total Asset	Disposal:	.00	.00	.00	.00.	.00	.00
Prior Year							
08-393.120	Note Proceeds	.00	.00	.00	.00	.00	.00
08-393.121	Bond Proceeds	.00	.00	.00	.00.	.00	.00
08-393.200	Premium on Bonds Sold	.00	.00	.00	.00	.00	.00
Total Prior `	Year:	.00	.00	.00	.00	.00	.00.
Prior Year Exp	Defund of Dries Very Evenness	00	00	00	00	00	00
08-395.000	Refund of Prior Year Expenses	.00	.00	.00	.00	.00	.00
Total Prior `	Year Exp:	.00	.00	.00	.00	.00	.00
Prior Year Resv	<b>-</b>				404 ===0		
08-396.000	Prior Year Reserves	.00	.00	.00	404,773	.00	.00
Total Prior `	Year Resv:	.00	.00	.00	404,773	.00	.00
Admin Salaries -	Executive						
08-401.121	Manager	22,980	6,914	30.09%	22,300	6,865	30.78%
08-401.122	Assistant Manager	.00	.00	.00	.00.	.00	.00
Total Admir	n Salaries - Executive:	22,980	6,914	30.09%	22,300	6,865	30.78%
Other General Go	ovt Admin Other General Govt Admin	.00	.00	.00	.00	.00	.00

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Total Othe	r General Govt Admin:	.00	.00	.00	.00	.00	.00
Sewer System C	Onerations (						
08-429.130	DPWUtility Supervisor	43,360	13,046	30.09%	42,100	12,952	30.77%
08-429.141	ClericalFull Time	19,570	5,753	29.39%	18,050	5,505	30.50%
08-429.142	Aide to Public Works Director	32,650	9,825	30.09%	31,710	9,755	30.76%
08-429.143	DPW - Full Time	148,770	44,765	30.09%	142,500	44,930	31.53%
08-429.149	DPWPart Time	.00	.00	.00	.00	.00	.00
08-429.181	Double Time	1,800	773	42.94%	1,500	547	36.45%
08-429.183	Overtime	6,000	4,769	79.49%	5,200	3,559	68.45%
08-429.189	On - Call	15,000	4,670	31.13%	15,000	4,702	31.34%
08-429.231	Vehicle Fuel	9,000	2,740	30.44%	8,400	2,922	34.78%
08-429.232	Generator Fuel	.00	.00	.00	.00	392	.00
08-429.240	Supplies	10,000	4,852	48.52%	10,000	1,579	15.79%
08-429.251	Vehicle Maintenance	5,000	443	8.86%	5,000	506	10.12%
08-429.260	Minor Equipment & Small Tools	5,000	547	10.93%	5,000	484	9.67%
08-429.261	Computer Equip & Software	4,200	4,396	104.67%	4,000	2,166	54.16%
08-429.310	Legal Services	10,000	13,468	134.68%	10,000	3,488	34.88%
08-429.311	Auditing & Accounting Services	1,500	625	41.67%	1,200	.00	.00
08-429.313	Engineering Services	20,000	17,606	88.03%	20,000	3,407	17.03%
08-429.315	General Services	1,000	.00	.00	600	670	111.65%
08-429.317	Contracted Services	20,000	2,067	10.33%	20,000	893	4.46%
08-429.319	Computer Maint & Support	11,000	6,483	58.93%	9,500	6,741	70.95%
08-429.320	Telephone	438	111	25.31%	400	149	37.32%
08-429.325	Postage	3,760	915	24.33%	2,500	1,480	59.18%
08-429.342	Printing	2,080	439	21.11%	2,000	.00	.00
08-429.361	Electric	10,500	4,739	45.13%	10,000	3,729	37.29%
08-429.367	Disposal Cost - LCA/Altn	550,000	.00	.00	680,000	.00	.00
08-429.368	Disposal Cost - Bethlehem	200,000	.00	.00	180,000	.00	.00
08-429.372	I&IRepairs & Maintenance	.00	9,703	.00	150,000	9,799	6.53%
08-429.373	Facilities Maintenance	6,000	1,106	18.43%	6,000	1,035	17.25%
08-429.375	Equipment Maintenance	2,500	106	4.22%	2,500	418	16.72%
08-429.421	Training	300	.00	.00	300	.00	.00
08-429.530	Transmission - LCA/Altn	6,000	.00	.00	5,400	.00	.00
08-429.531	Transmission - Fountain Hill	1,200	658	54.83%	1,000	313	31.27%
08-429.532	Transmission - Emmaus	12,000	3,672	30.60%	11,000	.00	.00
08-429.533	Transmission - LCA	22,000	.00	.00	15,000	.00	.00
08-429.534	Debt Service - LCA/Altn	136,000	.00	.00	70,000	.00	.00
08-429.535	Debt Service - Bethlehem	22,801	11,401	50.00%	21,855	10,928	50.00%
08-429.536	Debt Service - Fountain Hill	2,400	600	25.00%	2,400	600	25.00%
08-429.600	Capital Construction	.00	.00	.00	12,000	.00	.00
08-429.601	Patricia Drive Pump Station	.00	.00	.00	.00	.00	.00
08-429.603	Cardinal Dr PumpStn Force Main	.00	.00	.00	.00	569	.00
08-429.604	Riverside Dr Pump Stn Rehab	.00	1,915	.00	100,000	.00	.00
08-429.605	Montgomery St Sewer Replcmnt	.00	.00	.00	.00	.00	.00
08-429.606	Trout Creek Int'ceptor Replcmt	.00	.00.	.00	.00	.00	.00.
08-429.607	CuredInPlace Pipe Lining Projs	.00	7,812	.00	125,000	16,502	13.20%
08-429.699	Capital Reserve	.00	.00	.00	.00.	.00	.00
08-429.700	Capital Equipment	.00	.00	.00	27,500	.00	.00
08-429.740	Maintenance Reserve	.00	.00	.00	.00	.00	.00
08-429.800	Depreciation	.00	.00	.00	.00	.00	.00
Total Sawa	er System Operations:	1,341,829	180,000	13.41%	1,774,615	150,717	8.49%

2019-19 04/19 2018-18 04/18 Current Year % of Prior Year % of Current YTD Prior YTD Account Number Account Title Budget Actual Budget Budget Actual Budget Admin Salaries - DPW 08-430.122 **Public Works Director** 30,650 9,222 30.09% 29,760 9,156 30.77% Total Admin Salaries - DPW: 30,650 9,222 30.09% 29,760 9,156 30.77% **Debt Service - Principal** 08-471.202 Principal - 2010 Bonds .00 .00 .00 .00 .00 .00 08-471.203 Principal - 2016 Bonds .00 .00 .00 44,400 .00 .00 Principal - Lease Pmt 08-471.350 .00 .00 .00 .00 .00 .00 Total Debt Service - Principal: .00 .00 .00 44,400 .00 .00 Debt Service - Interest 08-472.202 Interest - 2010 Bonds .00 .00 .00 .00 .00 .00 08-472.203 Interest - 2016 Bonds .00 .00 .00 25,160 .00 .00 08-472.350 Interest - Lease Pmt .00 .00 .00 .00 .00 .00 Total Debt Service - Interest: .00 .00 .00 25,160 .00 .00 **Fiscal Agent Fees** 08-475.000 Fiscal Agent Fees .00 .00 .00 .00 .00 .00 Total Fiscal Agent Fees: .00 .00 .00 .00 .00 .00 Fees & Miscellaneous 08-480.000 Miscellaneous Expense .00 .00 .00 .00 .00 .00 08-480.005 Financial Service Fees .00 .00 .00 200 .00 .00 08-480.010 Credit Card Service Fees 53.88% 6,000 3,233 6,000 2,277 37.94% Total Fees & Miscellaneous: 53.88% 6,000 6,200 3,233 2,277 36.72% Insurance 08-486.351 2,700 2,700 Insurance - Commercial .00 .00 2,579 95.51% 08-486.352 Insurance - Business Auto 4,400 3,747 85.16% 3,300 4,163 126.15% 08-486.354 Insurance - Workers Comp 12,000 .00 .00 11,700 14,214 121.49% Total Insurance: 19,100 3,747 19.62% 17,700 20,956 118.39% **Employee Benefits** 08-487.156 Insurance - Health 104,000 41,617 40.02% 92,500 37,094 40.10% 08-487.158 Insurance - Life & Disability 35.39% 36.95% 3,100 1,097 2,100 776 08-487.160 Pension 95,087 45,667 .00 .00 .00 .00 08-487.161 Social Security Tax 24,540 7,766 31.65% 23,571 7,542 32.00% Total Employee Benefits: 226,727 50,480 22.26% 163,838 45,411 27.72% Department: 490 08-490.740 Transfer to Capital Fund 213,714 .00 .00 .00 .00 .00 .00 .00 Total Department: 490: 213,714 .00 .00 .00 **Prior Year** 08-491.000 Refunds of Prior Year Revenue .00 .00 .00 .00 .00 .00 Total Prior Year: .00 .00 .00 .00 .00 .00

Salisbury Townsh	ip		l Report April 20 riod: 04/19	019		Page: 24	
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Interfund Transfer 08-492.010	rs Transfer to General Fund	107,000	.00	.00	131,000	.00	.00
Total Interf	und Transfers:	107,000	.00	.00	131,000	.00	.00
Sewer Fund	d Revenue Total:	1,968,000	562,261	28.57%	2,214,973	472,304	21.32%
Sewer Fund	d Expenditure Total:	1,968,000	253,596	12.89%	2,214,973	235,380	10.63%
Net Total S	ewer Fund:	.00	308,665	.00	.00	236,923	.00

Salisbury Townsh	ip		l Report April 20 iod: 04/19	019			Page: 2
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Refuse & Recycl	ing Fund						
Interest							
10-341.000	Interest Income	5,400	3,457	64.03%	1,800	510	28.36%
Total Intere	st:	5,400	3,457	64.03%	1,800	510	28.36%
Sanitation Fees							
0-364.300	Solid Waste Collection	1,420,000	690,544	48.63%	1,398,000	689,423	49,31%
0-364.400	Freon Decal Sales	1,200	360	30.00%	1,500	220	14.67%
0-364.500	Recycling Container Sales	1,500	617	41.13%	1,500	380	25.33%
0-364.600	Recycling Proceeds	.00	.00	.00	.00	.00	.00
10-364.700	PA Refuse Surcharge	.00.	64-	.00	.00	8	.00
Total Sanita	ation Fees:	1,422,700	691,457	48.60%	1,401,000	690,031	49.25%
Prior Year Reser	ves						
10-396.000	Prior Year Reserves	.00	.00	.00	.00.	.00	.00
Total Prior	Year Reserves:	.00	.00	.00	.00	.00	.00
Administration							
0-401.325	Postage	1,880	1,046	55.62%	2,500	1,480	59.18%
0-401.342	Printing	1,040	502	48.26%	2,000	.00	.00.
Total Admir	nistration:	2,920	1,547	53.00%	4,500	1,480	32.88%
Other General G							
10-406.000	Other General Govt Admin	.00	.00	.00	.00		.00
Total Other	General Govt Admin:	.00	.00	.00	.00	.00	.00
Data Processing							
0-407.261	Computer Equip & Software	3,600	3,739	103.87%	4,000	2,166	54.16%
0-407.319	Computer Maint & Support	9,000	5,559	61.76%	9,500	6,741	70.95%
Total Data I	Processing:	12,600	9,298	73.79%	13,500	8,907	65.98%
Buildings & Plan				••	400		
0-409.240	Supplies	.00	.00	.00	100	.00	.00.
Total Buildi	ngs & Plant:	.00	.00	.00	100	.00	.00
Vages	Manager	00.000	0.044	20.000/	00.000	0.005	00 700/
0-426.121	Manager	22,980	6,914	30.09%	22,300	6,865	30.78%
0-426.122	Assistant Manager	.00	.00	.00	.00	.00	.00
0-426.141	ClericalFull Time	19,570	5,900	30.15%	18,100	5,671	31.33%
0-426.147	Recycling Center - Part Time	10,600	1,235	11.65%	9,500	1,210	12.74%
Total Wage	s:	53,150	14,049	26.43%	49,900	13,746	27.55%
Sanitation							
0-427.300	Refuse Collection Services	1,100,000	364,141	33.10%	1,075,000	353,646	32.90%
0-427.301	PA Refuse Collection Surcharge	.00	.00	.00	.00	.00	.00
10-427.302	Recycling Costs	60,000	21,082	35.14%	12,000	9,694	80.78%
10-427.303	Grass Collection Services	8,400	.00	.00	9,000	.00	.00

Salisbury Townsh	nip		l Report April 20 iod: 04/19	)19			Page: 2	
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget	
Total Sanit	ation:	1,168,400	385,222	32.97%	1,096,000	363,340	33.15%	
Department: 431 10-431.303	Composting Costs	12,000	.00	.00	12,000	662	5.52%	
Total Depa	rtment: 431:	12,000	.00	.00	12,000	662	5.52%	
Debt Service - P 10-471.350	rincipal Principal - Least Pmt	.00	.00	.00	.00	.00	.00	
Total Debt	Service - Principal:	.00	.00	.00	.00	.00	.00	
Debt Service - In 10-472.350	i <b>terest</b> Interest - Lease Pmt	.00	.00	.00	.00	.00	.00	
Total Debt	Service - Interest:	.00	.00	.00	.00	.00	.00	
Fees & Miscellar 10-480.005 10-480.010	neous Financial Services Fee Credit Card Service Fees	.00 5,600	.00 3,233	.00 57.73%	100 5,600	.00 2,277	.00 40.65%	
Total Fees	& Miscellaneous:	5,600	3,233	57.73%	5,700	2,277	39.94%	
Employee Benef 10-487.156 10-487.158 10-487.160 10-487.161	its Insurance - Health Insurance - Life & Disability Pension Social Security Tax	25,893 624 19,393 4,066	9,372 215 .00 1,087	36.20% 34.45% .00 26.73%	.00 .00 .00 3,817	.00 .00 .00 1,052	.00 .00 .00 27.55%	
Total Empl	oyee Benefits:	49,976	10,674	21.36%	3,817	1,052	27.55%	
Collections 10-489.317	Collection Agency Services	.00	.00	.00	.00	.00	.00	
Total Collec	ctions:	.00	.00	.00	.00	.00	.00	
Reserves 10-490.740	Tr to Reserves	4,454	.00	.00	36,283	.00	.00	
Total Rese	rves:	4,454	.00	.00	36,283	.00	.00	
Prior Year 10-491.000	Refunds of Prior Year Revenue	.00	.00	.00	.00	.00	.00	
Total Prior	Year:	.00.	.00	.00	.00.	.00	.00	
nterfund Transfe 10-492.010	ers Transfer to General Fund	119,000	.00	.00	181,000	.00	.00	
Total Interf	und Transfers:	119,000	.00	.00	181,000	.00	.00	
Refuse & R	Recycling Fund Revenue Total:	1,428,100	694,914	48.66%	1,402,800	690,541	49.23%	
Refuse & R	Recycling Fund Expenditure Total:	1,428,100	424,023	29.69%	1,402,800	391,462	27.91%	

Salisbury Township		BOC Financial Report April 2019 Period: 04/19					Page:	27
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget	_
Net Total Refuse	& Recycling Fund:	.00.	270,891	.00	.00	299,079	.00	ı

Salisbury Township			ll Report April 20 riod: 04/19	)19			Page: 28
Account Numb	er Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Debt Service I	Fund						
Interfund Tran							
20-392.010	Transfer from General Fund	183,032	.00	.00	178,650	.00	.00
Total Inte	Total Interfund Transfers:		.00	.00	178,650	.00	.00
Debt Service -	- Principal						
20-471.202	Principal - 2010 Bonds	.00	.00	.00	.00	.00	.00
20-471.203	Principal - 2016 Bonds	137,000	.00	.00	127,850	.00	.00
Total Debt Service - Principal:		137,000	.00	.00	127,850	.00	.00
Debt Service -	- Interest						
20-472.202	Interest - 2010 Bonds	.00	.00	.00	.00	.00	.00
20-472.203	Interest - 2016 Bonds	46,032	.00.	.00	50,200	.00	.00
Total De	bt Service - Interest:	46,032	.00	.00	50,200	.00	.00
Fees & Miscel	laneous						
20-480.005	Financial Service Fees	.00.	.00	.00	600	.00	.00
Total Fee	es & Miscellaneous:	.00	.00	.00	600	.00	.00
Debt Service Fund Revenue Total:		183,032	.00	.00	178,650	.00	.00
Debt Ser	rvice Fund Expenditure Total:	183,032	.00	.00	178,650	.00	.00
Net Total Debt Service Fund:		.00	.00	.00	.00	.00	.00

Salisbury Township		BOC Financial Report April 2019 Period: 04/19					Page: 29
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Highway Aid Fur	nd						
Interest							
35-341.000	Interest Income	3,000	1,919	63.98%	1,800	442	24.56%
Total Intere	est:	3,000	1,919	63.98%	1,800	442	24.56%
State-Shared Re	venue						
35-355.020 35-355.030	Motor Vehicle Fuels Tax Road Turnback	475,458 1,080	488,362 1,080	102.71% 100.00%	469,000 1,100	477,094 1,080	101.73% 98.18%
Total State	-Shared Revenue:	476,538	489,442	102.71%	470,100	478,174	101.72%
Prior Year Exp 35-395.000	Refund of Prior Year Expenses	.00	.00	.00	.00	.00	.00
Total Prior	Year Exp:	.00	.00	.00	.00	.00	.00
Prior Year Resv							
35-396.000	Prior Year Reserves	.00	.00	.00	.00	.00	.00
Total Prior	Year Resv:	.00.	.00	.00	.00	.00.	.00
Snow Removal							
35-432.245	Snow Removal Salt	65,000	38,892	59.83%	65,000	35,708	54.94%
Total Snow	Removal:	65,000	38,892	59.83%	65,000	35,708	54,94%
Traffic Control							
35-433.246 35-433.362	Signs & Street Markings Traffic Signals	.00 5,000	.00 1,731	.00 34.63%	.00 4,400	.00 1,521	.00 34.56%
Total Traffic	c Control:	5,000	1,731	34.63%	4,400	1,521	34.56%
Street Lighting							
35-434.361	Electric	160,000	44,867	28.04%	168,000	57,213	34.06%
Total Stree	t Lighting:	160,000	44,867	28.04%	168,000	57,213	34.06%
Streets & Bridge							
35-438.450 35-438.700	Road Program Capital Equipment	.00. 80,000	.00 .00	.00 .00	.00 70,000	.00 .00	.00 .00
35-438.740	Equipment Reserve	.00	.00	.00	.00	.00	.00
Total Stree	ts & Bridges:	80,000	.00	.00	70,000	.00	.00
<b>Debt Service - Pt</b> 35-471.350	rincipal Principal - Lease Pmt	.00	.00	.00	.00	.00	.00
Total Debt	Service - Principal:	.00	.00	.00	.00	.00	.00
Debt Service - In	terest						
35-472.350	Interest - Lease Pmt	.00	.00	.00	.00	.00	.00
Total Debt	Service - Interest:	.00	.00	.00	.00	.00	.00

Salisbury Townsh	nip		BOC Financial Report April 2019 Period: 04/19				Page: 3
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Reserves	T ( ) 0 % IB	400 500			404.500	00	22
35-490.740	Transfer to Capital Reserves	169,538	.00	.00	164,500	.00	.00
Total Rese	rves:	169,538	.00	.00	164,500	.00	.00
Highway Aid Fund Revenue Total:		479,538	491,361	102.47%	471,900	478,616	101.42%
Highway A	id Fund Expenditure Total:	479,538	85,491	17.83%	471,900	94,442	20.01%
Net Total H	lighway Aid Fund:	.00.	405,870	.00	.00	384,174	.00

Salisbury Townsh	ip		BOC Financial Report April 2019 Period: 04/19				
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Fund: 41							
Interest Income							
41-341.000	Interest Income	12,500	.00	.00	.00	.00	.00
Total Intere	st Income:	12,500	.00	.00	.00	.00	.00.
<b>Source: 342</b> 41-342.300	Franko Lease Payments	8,400	3,500	41.67%	.00	.00	.00
Total Source	e: 342:	8,400	3,500	41.67%	.00	.00	.00
Source: 354							
41-354.040	Recycling Grant Revenue	221,499	.00	.00	.00	.00	.00
41-354.120	Gaming Grant: Casino Corr Tra	107,000	.00	.00	.00	.00	.00
41-354.140	Park Development Grant(s)	.00	.00	.00	.00	.00	.00
Total Source	e: 354:	328,499	.00	.00	.00.	.00	.00
Source: 391							
41-391.100	Sale of Fixed Assets	5,000	.00	.00	.00	.00	.00
Total Source	e: 391;	5,000	.00	.00	.00	.00	.00
Source: 392	T ( ( ) ( ) ( )						•
41-392.010	Transfer from General Fund	200,000	.00	.00	.00	.00	.00
Total Sourc	e: 392:	200,000	.00	.00	.00	.00	.00
Source: 393	D 10						
41-393.121	Bond Proceeds	.00	.00	.00	.00	.00	.00
41-393.200	Premium on Bonds Sold	.00	.00	.00	.00.	.00	.00
Total Sourc	e: 393:	.00	.00	.00	.00	.00	.00
PY Reserves for 41-396.000		224 222	00	00	00	00	00
	PY Reserves for Bond Projects	321,233	.00	.00	.00.	.00	.00.
Total PY Re	eserves for Bond Projects:	321,233	.00	.00	.00	.00	.00
Department: 401	Canital Favinanant Admin	00	00	00	00	00	00
41-401.700	Capital Equipment- Admin	.00	.00	.00	.00	.00	.00
Total Depar	tment: 401:	.00	.00	.00	.00	.00	.00
Department: 407 41-407.261	Computer Equip & Software-Adm	.00	.00	.00	.00.	.00	.00
Total Depar	tment: 407:	.00	.00	.00	.00	.00	.00
Department: 409							
41-409.600	Capital Construction- Facilit	10,000	.00	.00	.00	.00	.00
41-409.700	Capital Equip- Facilities	21,000	.00	.00	.00	.00	.00

Page:	32
raye.	52

Salisbury Township

Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Department: 410							
41-410.250	Police Vehicles	110,000	.00	.00	.00	.00.	.00
41-410.261	Computer Eq & Software- Police	.00.	.00	.00	.00	.00	.00
41-410.700	Capital Equipment- Police	27,000	.00	.00	.00.	.00	.00
Total Depa	rtment: 410:	137,000	.00	.00	.00.	.00	.00
Department: 411							
41-411.700	Capital Equip- Fire Inspection	.00.	.00	.00	.00.	.00	.00
Total Depa	rtment: 411;	.00	.00	.00	.00	.00	.00
Department: 412							
41-412.600	Earmarked for WSFD Request	215,000	.00	.00	.00	.00	.00
Total Depar	rtment: 412:	215,000	.00	.00	.00	.00	.00
Department: 415 41-415.700		0.000	00	00	00	00	00
41-415.700	Capital Equipment- Emerg Mgmt	9,000	.00	.00	.00	.00	.00
Total Depar	rtment: 415:	9,000	.00	.00	.00	.00	.00
Department: 436 41-436.700	Capital Equipment- Storm Sewer	.00	.00	.00	.00	.00	.00
Total Depar	rtment: 436:	.00	.00	.00	.00	.00	.00
Department: 439							
41-439.700	Capital Equipment- Highway	255,000	.00	.00	.00	.00	.00
Total Depar	rtment: 439:	255,000	.00	.00	.00	.00	.00
Department: 451							
41-451.600	Capital Construction- Parks	45,000	.00	.00	.00.	.00	.00
41-451.700	Capital Equipment- Parks	.00	.00	.00	.00	.00	.00
Total Depar	rtment: 451:	45,000	.00.	.00	.00	.00	.00
Department: 492							
41-492.200	Transfer to Debt Service Fund	183,632	23,663	12.89%	.00.	.00	.00
Total Depar	rtment: 492:	183,632	23,663	12.89%	.00	.00	.00
Fund: 41 R	evenue Total:	875,632	3,500	0.40%	.00	.00	.00
Fund: 41 E	xpenditure Total:	875,632	23,663	2.70%	.00	.00	.00
Net Total F	und: 41:	.00	20,163-	.00	.00	.00	.00
HOL FOLDIT	wild, it.		20,100-		.00		

Salisbury Township			l Report April 20 riod: 04/19	)19			Page: 33
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Fund: 43							
Interest Income		4 000		00		00	20
43-341.000	Interest Income	1,600	.00	.00	.00	.00	.00
Total Intere	st Income:	1,600	.00	.00	.00	.00	.00
Source: 392							
43-392.010	Transfer from General Fund	50,750	.00	.00	.00	.00	.00
43-392.030	Transfer from Fire Fund	101,000	50,752	50.25%	.00	.00	.00
Total Source	e: 392:	151,750	50,752	33.44%	.00	.00	.00
Department: 411							
43-411.600	Capital Construction- ESFD	15,000	.00	.00	.00	.00	.00
43-411.700	Capital Equipment- ESFD	.00.	.00	.00	.00	.00	.00
Total Depar	tment: 411:	15,000	.00	.00	.00	.00	.00
Department: 412							
43-412.600	Capital Construction- WSFD	.00	.00	.00	.00	.00	.00
43-412.700	Capital Equipment- WSFD	.00.	.00	.00	.00	.00	.00
Total Depar	tment: 412:	.00	.00	.00	.00	.00	.00
Department: 471							
43-471.213	Principal- 2005 Fire Truck	11,288	11,288	100.00%	.00	.00	.00
43-471.214	Principal- 2008 Fire Truck	7,123	7,123	100.00%	.00	.00	.00
43-471.215	Principal- 2009 Fire Truck	10,421	10,421	100.00%	.00	.00	.00
43-471.216	Principal- 2013 Fire Truck	6,456	6,457	100.01%	.00	.00	.00
43-471.217	Principal- 2017 Fire Truck	8,410	8,410	100.00%	.00	.00	.00
Total Depar	tment: 471:	43,698	43,699	100.00%	.00	.00	.00
Department: 472							
43-472.213	Interest- 2005 Fire Truck	295	295	99.99%	.00	.00	.00
43-472.214	Interest- 2008 Fire Truck	599	599	100.01%	.00	.00	.00
43-472.215	Interest- 2009 Fire Truck	1,162	1,162	100.01%	.00	.00	.00
43-472.216	Interest- 2013 Fire Truck	1,266	1,266	99.97%	.00	.00	.00
43-472.217	Interest- 2017 Fire Truck	3,731	3,731	100.00%	.00.	.00	.00
Total Depar	tment: 472:	7,053	7,053	100.00%	.00	.00	.00
Department: 490							
43-490.740	Transfer to Capital Reserves	87,599	.00	.00	.00	.00	.00
Total Depar	tment: 490:	87,599	.00	.00	.00.	.00	.00
Fund: 43 R	evenue Total:	153,350	50,752	33.10%	.00	.00	.00
Fund: 43 Ex	kpenditure Total:	153,350	50,752	33.10%	.00	.00	.00
Not Takai m	und: 42:	00	00	00	00	00	00
Net Total F	unu. 43.	.00	.00	.00	.00	.00	.00

Salisbury Township		BOC Financial Report April 2019 Period: 04/19					Page: 34
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Fund: 46							
Interest Income	Interest Income	0.000	00	00	00	00	00
46-341.000	Interest Income	6,300	.00	.00	.00	.00	.00
Total Intere	est Income:	6,300	.00	.00	.00	.00	.00
Source: 392							
46-392.060	Transfer from Water Fund	130,000	.00	.00	.00	.00	.00
Total Source	ce: 392:	130,000	.00	.00	.00	.00	.00
Prior Year Reser	avos						
46-396.000	Prior Year Reserves	86,039	.00	.00	.00	.00	.00
Total Prior	Year Reserves:	86,039	.00	.00	.00	.00	.00
Department: 448							
46-448.600	Capital Construction-Fac Maint	12,000	.00	.00	.00	.00	.00
46-448.615	Paxford Rd Waterline Project	.00	.00	.00	.00	.00	.00
46-448.616	Maumee Ave Waterline Project	.00	.00	.00	.00	.00	.00
46-448.617	Montgomery St Waterline Proj	.00.	.00	.00	.00	.00	.00
46-448.700	Capital Equipment	.00.	.00	.00	.00.	.00	.00
Total Depa	rtment: 448:	12,000	.00	.00	.00	.00	.00
Department: 471							
46-471.203	Principal - 2016 Bonds	142,400	.00	.00	.00	.00	.00
Total Depa	rtment: 471:	142,400	.00	.00	.00	.00	.00
Department: 472				•			
46-472.203	Interest- 2016 Bonds	67,939	33,970	50.00%	.00	.00	.00
Total Depa	rtment: 472:	67,939	33,970	50.00%	.00	.00	.00
Fund: 46 R	evenue Total:	222,339	.00	.00	.00	.00	.00
Fund: 46 E	xpenditure Total:	222,339	33,970	15.28%	.00.	.00	.00
		·					
Net Total F	und: 46:	.00	33,970-	.00	.00	.00	.00

Salisbury Township		BOC Financia Per	Page: 35				
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget
Fund: 48							
Interest Income		7.400	20	20	00	00	00
48-341.000	Interest Income	7,100	.00	.00	.00	.00	.00
Total Intere	est Income:	7,100	.00	.00	.00	.00	.00
Source: 392							
48-392.080	Transfer from Sewer Fund	200,000	.00	.00	.00	.00	.00
Total Source	ce: 392:	200,000	.00	.00	.00	.00	.00
Prior Year Reser	ves						
48-396.000	Prior Year Reserves	61,881	.00	.00	.00	.00	.00
Total Prior Year Reserves:		61,881	.00	.00	.00	.00	.00
Department: 429							
48-429.372	I&I - Repairs & Maintenance	200,000	.00	.00	.00	.00	.00
48-429.600	Capital Const- Facilities Main	.00	.00	.00	.00	.00	.00
48-429.604	Riverside/Cardinal Pump Stn	.00	.00	.00	.00	.00	.00
48-429.607	Cured in Place Pipe Lining	.00	.00	.00	.00	.00	.00
48-429.700	Capital Equipment	.00.	.00	.00	.00.	.00	.00
Total Depa	rtment: 429:	200,000	.00	.00	.00	.00	.00
Department: 471							
48-471.203	Principal- 2016 Bonds	45,600	.00	.00	.00	.00	.00
Total Depa	rtment: 471:	45,600	.00	.00	.00	.00	.00
Department: 472							
48-472.203	Interest- 2016 Bonds	23,381	11,690	50.00%	.00	.00	.00
Total Depa	rtment: 472:	23,381	11,690	50.00%	.00	.00	.00
Fund: 48 R	evenue Total:	268,981	.00	.00	.00	.00	.00
Fund: 48 E	xpenditure Total:	268,981	11,690	4.35%	.00	.00	.00
Net Total F	und: 48:	.00	11,690-	.00	.00	.00	.00

Salisbury Township			BOC Financial Report April 2019 Period: 04/19					
Account Number Account Title		2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget	
Payroll Fund								
Interest Income 90-341.000 Inte	erest Income	.00.	155	.00	.00	57	.00	
Total Interest Inc	come:	.00	155	.00	.00	57	.00	
Payroll Fund Re	venue Total:	.00	155	.00	.00	57	.00	
Payroll Fund Exp	penditure Total:	.00	.00	.00	.00	.00	.00	
Net Total Payrol	l Fund:	.00	155	.00	.00	57	.00	

Salisbury Township			BOC Financial Report April 2019 Period: 04/19						
Account Number	Account Title	2019-19 Current Year Budget	04/19 Current YTD Actual	% of Budget	2018-18 Prior Year Budget	04/18 Prior YTD Actual	% of Budget		
Subdivision Escrow	Fund								
Interfund Transfers 91-492.060 Tra	nsfer to Water Fund	.00.	.00.	.00	.00	.00	.00		
Total Interfund	Fransfers:	.00	.00	.00	.00	.00	.00		
Subdivision Esc	row Fund Revenue Total:	.00	.00	.00	.00	.00	.00		
Subdivision Esc	row Fund Expenditure Total:	.00	.00	.00	.00	.00	.00		
Net Total Subdiv	vision Escrow Fund:	.00	.00	.00	.00	.00	.00		
Total Asset:		.00.	.00	.00	.00	.00	.00		
Total Liability:		.00	.00	.00	.00	.00	.00		
Total Equity:		.00	.00	.00	.00	.00	.00		
Total Revenue:		15,777,830	4,132,655	26.19%	16,060,423	3,711,499	23.11%		
Total Expenditure	:	15,777,830	3,509,386	22.24%	16,060,423	3,600,822	22.42%		
Net Grand Totals	:	.00	623,269	.00	.00	110,677	.00		

#### Report Criteria:

Detail report.

Invoice detail records above \$0 included.

Paid and unpaid invoices included.

Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date P
A-B-E Laboratory							
A-B-E Laboratory	4302019	06-448.316	2019-Apr water analysis	04/30/19	300.00	.00	
Total A-B-E Laborato	ory:				300.00	.00	
Belgray Oil							
Belgray Oil	3161	01-430.240	Grease	05/01/19	188.00	.00	
Belgray Oil	3161	06-448.240	Grease	05/01/19	188.00	.00	
Belgray Oil	3161	08-429.240	Grease	05/01/19	188.00	.00	
Total Belgray Oil:					564.00	.00	
Berkheimer Associates							
erkheimer Associates	04302019	01-403.453	2019-Apr-LST Comm	04/30/19	153.64	.00	
Total Berkheimer As	sociates:				153.64	.00	
Bethlehem, City of							
Bethlehem, City of	5112019	06-448.363	2019-Mar-Hydrant rental	05/11/19	880.00	.00	
Bethlehem, City of	5112019	06-448.368	2019-Mar-4 resale customers	05/11/19	357.86	.00	
Total Bethlehem, Cit	ty of:				1,237.86	.00	
rown, James P.							
Brown, James P.	5142019	01-414.220	Planning Commission Mtg-5/14/2019	05/14/19	25.00	.00	
Total Brown, James	P.:				25.00	.00	
Budget Instant Print							
sudget Instant Print	15387	01-410.342	Rental Inspection Checklist	04/25/19	198.50	.00	
Total Budget Instant	Print:				198.50	.00	
CodeMaster							
CodeMaster	CDM-014-166	01-413.307	2019-Apr-Residential inspections	04/30/19	95.00	.00	
CodeMaster	CDM-014-166	01-413.308	2019-Apr-Plan Review	04/30/19	2,200.00	.00	
CodeMaster	CDM-014-166	01-413.306	2019-Apr-Commercial inspections	04/30/19	60.00	.00	
odeMaster	CDM-014-166	01-413.306	2019-Apr-Commercial inspections	04/30/19	18,425.52	.00	
Total CodeMaster:					20,780.52	.00	
coopersburg Materials							
Coopersburg Materials	18169	01-438.245	19MM/9.5MM	04/30/19	7,365.37	.00	
Total Coopersburg N	Naterials:				7,365.37	.00	
Davison & McCarthy							
Davison & McCarthy	22479	01-408.318	2018-Aug-Plot 886 LLC	01/02/19	26.00	.00	
Davison & McCarthy	22589	01-408.318	2018-Sep-Plot 886 LLC	01/02/19	312.00	.00	
Davison & McCarthy	22760	01-408.318	2018-Oct-Plot 886 LLC	01/02/19	52.00	.00	
Davison & McCarthy	23646	01-404.318	2019-Mar-Resubdivision of 2242 &	04/05/19	39.00	.00	
Davison & McCarthy	23752	01-404.318	2310 Church Rd 2019-Apr-Samuel M Gentile Minor Subdivision	04/29/19	557.00	.00	

Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
Davison & McCarthy	23818	01-404.310	2019-Apr-General Matters	05/06/19	1,077.70	.00	
Davison & McCarthy	23819	01-404.310	2019-Apr-Amendment to SALDO	05/06/19	39.00	.00	
Davison & McCarthy	23820	01-404.310	2019-Apr-Bankruptcy of Residents	04/30/19	1,234.00	.00	
Davison & McCarthy	23821	01-414.310	2019-Apr-Planning Commission	05/06/19	312.00	.00	
Davison & McCarthy	23822	01-404.310	2019-Apr-Cornerstone Consultant Co- Conditional Use Application	05/06/19	39.00	.00	
Davison & McCarthy	23823	01-404.310	2019-Apr-Board of Commissioners	05/06/19	598.00	.00	
Davison & McCarthy	23824	08-429.310	2019-Apr-Sewer Overpayments to Allentown	05/06/19	39.00	.00	
Davison & McCarthy	23825	01-404.310	2019-Apr-Vanessa Holding Company (Second Appeal)	05/06/19	130.00	.00	
Davison & McCarthy	23826	01-404.318	2019-Apr-621 E Lynnwood St Major Subdivision	05/06/19	39.00	.00	
Davison & McCarthy	23827	01-404.310	2019-Apr-3115 W Emmaus Ave	05/06/19	351.00	.00	
Davison & McCarthy	23828	01-404.310	2019-Apr-321 E Emmaus Ave (rear)	05/06/19	39.00	.00	
Davison & McCarthy	23829	01-404.310	2019-Apr-Contract with LANTA-Bus Shelter License Agreement	05/06/19	234.00	.00	
Davison & McCarthy	23830	01-404.318	2019-Apr-1239 Vortman Ave	05/06/19	143.00	.00	
Davison & McCarthy	23831	01-404.310	2019-Apr-Liens-2019	05/06/19	390.00	.00	
Davison & McCarthy	23832	01-404.310	2019-Apr-Satisfactions for 2019	05/06/19	182.00	.00	
Davison & McCarthy	23834	01-404.318	2019-Apr-415 Mountain Blvd	05/06/19	39.00	.00	
Davison & McCarthy	23835	01-404.318	2019-Apr-Resubdivision of 2242 & 2310 Church Rd	05/06/19	299.00	.00	
Davison & McCarthy	23836	01-404.310	2019-Apr-Suit by DiCesare	05/06/19	928.98	.00	
Davison & McCarthy	23837	01-404.310	2019-Apr-South Mall-Metro Commercial Mgmt Conditional Use- Food Truck Event	05/06/19	793.00	.00	
Davison & McCarthy	23838	01-404.318	2019-Apr-1651 Bobalew Trail Grading Plan	05/06/19	117.00	.00	
Total Davison & McCa	rthy:				8,009.68	.00	
Document Systems Ltd.	0.5000014	04.404.0==		05/04/40		••	
Document Systems Ltd.	G52669M	01-401.375	Neopost envelope opener	05/01/19	300.00	.00	
Total Document System	ms Ltd.:				300.00	.00	
Ecco Communications LLC							
Ecco Communications LLC	74591	06-448.375	#35-Replacement lamp	05/01/19	80.00	.00	
Ecco Communications LLC	74591	08-429.375	#35-Replacement lamp	05/01/19	80.00	.00	
Ecco Communications LLC	74598	01-410.251	#216-Antenna	05/02/19	14.00	.00	
Total Ecco Communica	ations LLC:				174.00	.00	
Faust Hauling Inc.							
Faust Hauling Inc.	19-263	01-451.240	18 yds topsoil	04/30/19	504.00	.00	
Total Faust Hauling Ind	o.:				504.00	.00	
Ferguson Waterworks							
Ferguson Waterworks	0470983	01-436.240	Pipe for storm sewer-Allen St	02/27/19	35.98	.00	
Ferguson Waterworks	0473836	01-410.700	Pipe for pole building	03/29/19	98.58	.00	
Ferguson Waterworks	0476641	06-448.240	4" x 1" Saddle/1" Corporation stop	05/17/19	111.20	.00	
Ferguson Waterworks	0476685	06-448.240	O-rings/meter chambers/strainers/gaskets/meter liners/freeze bottoms/meter bolts	05/07/19	1,433.00	.00	
Ferguson Waterworks	472185	01-436.240	Pipe for storm sewer-Gaskill Ave	03/13/19	169.00	.00	
Total Ferguson Waterv	works:				1,847.76	.00	
	41:4						
First Regional Compost Au First Regional Compost Au	=	10-427.303	Grass collection-30 yds-4/25/19	05/04/19	100.00	.00	

		Kep	ort dates. 5/4/2019-5/17/2019		ivia	y 17, 2019 10.2	.ZAWI
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
Total First Regional Co	ompost Authority:				100.00	.00	
Forthman, Letitia Forthman, Letitia	5062019	01-406.171	2019 Health Reimbursement	05/06/19	170.00	.00	
Total Forthman, Letitia	ı:				170.00	.00	
Gero, Deidre							
Gero, Deidre	5162019	01-410.315	Reimb-Repair CP screen	05/16/19	190.79	.00	
Total Gero, Deidre:					190.79	.00	
<b>Grainger</b> Grainger	9153512059	01-430.240	Leather gloves/sting relief wipes/first	04/23/19	92.78	.00	
Grainger	9153512059	06-448.240	aid kit Leather gloves/sting relief wipes/first	04/23/19	92.78	.00	
Grainger	9153512059	08-429.240	aid kit Leather gloves/sting relief wipes/first aid kit	04/23/19	92.78	.00	
Total Grainger:					278.34	.00	
Granturk Equipment Comp	any Inc						
Granturk Equipment Comp Granturk Equipment Comp	1132105-01 1132105-01	06-448.375 08-429.375	#35-Auto wind shoe #35-Auto wind shoe	05/02/19 05/02/19	46.79 46.79	.00	
Total Granturk Equipm	nent Company Inc:				93.58	.00	
Gross McGinley LLP Gross McGinley LLP	5012019	01-414.314	2019-Apr-Hearings/decisions/misc	05/01/19	1,237.50	.00	
Total Gross McGinley			2010 / p. 100go 00000	00.0 10	1,237.50	.00	
Hassick, Richard Hassick, Richard	5142019	01-414.220	Planning Commission Mtg-5/14/19	05/14/19	25.00	.00	
Total Hassick, Richard	l:				25.00	.00	
Herbein's Garden Center In Herbein's Garden Center In		01-451.240	Rye grass	04/29/19	110.00	.00	
Total Herbein's Garder	n Center Inc.:				110.00	.00	
Home Depot Credit Service	es						
Home Depot Credit Service	21507	01-409.240	Gutters & Downspouts for Annex Bldg	05/15/19	81.79	.00	
Home Depot Credit Service Home Depot Credit Service		01-430.260 01-410.700	Tape measure Expansion joint for pole bldg	04/25/19 04/25/19	13.94 35.60	.00	
Total Home Depot Cre	edit Services:				131.33	.00	
Hunter Keystone Peterbilt L Hunter Keystone Peterbilt	<b>-P</b> R202006390	03-411.374	#20-21-Inspection	04/22/19	253.00	.00	
Total Hunter Keystone			·		253.00	.00	
ICMA Membership Pensus	ı						
ICMA Membership Renewa ICMA Membership Renew	5062019	01-401.420	2019 Membership-#386449	05/06/19	910.24	.00	

		Kep					ZAIVI
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Pai
Total ICMA Membersh	ip Renewal:				910.24	.00	
Jahanan Orastaala Eina Bust	andiam I D						
Johnson Controls Fire Prot Johnson Controls Fire Prot		03-411.240	2019-ESFC bldg monitoring contract for alarm	05/02/19	402.85	.00	
Total Johnson Controls	s Fire Protection LP:				402.85	.00	
Keystone Consulting Engin	eers						
Keystone Consulting Engin		01-408.318	2018-Aug-2882 E Texas Blvd-Minor	01/02/19	501.50	.00	
Keystone Consulting Engin	161195	01-408.318	Subdivision 2018-Sep-2882 E Texas Blvd-Minor	01/02/19	211.00	.00	
Keystone Consulting Engin	161197	01-408.318	Subdivision 2018-Sept-1239 Vortman Ave Grading	01/02/19	370.00	.00	
Keystone Consulting Engin	161578	01-408.318	Plan 2018-Oct-2882 E Texas Blvd-Minor	01/02/19	396.00	.00	
Keystone Consulting Engin	161579	01-408.318	Subdivision 2018-Oct-1239 Vortman Ave Grading	01/02/19	633.00	.00	
Keystone Consulting Engin	162118	01-408.318	Plan 2018-Nov-Cedar Crest Professional	12/13/18	685.75	.00	
			Park Grading Plan				
Keystone Consulting Engin	162415	01-408.318	2018-Dec-Cedar Crest Professional Park Grading Plan	01/09/19	633.00	.00	
Ceystone Consulting Engin	163732	01-408.318	2019-Feb-Weil Street Subdivision	04/11/19	217.00	.00	
Keystone Consulting Engin	163734	01-408.318	2019-Feb-1651 Bobalew Trail Grading Plan	04/11/19	217.00	.00	
eystone Consulting Engin	163744	01-408.318	2019-Mar-2242 Church Rd Minor Subdivision	04/11/19	1,139.25	.00	
eystone Consulting Engin	164145	08-429.604	2019-Riverside Dr-Pump Station Generator Replacement	05/03/19	271.25	.00	
eystone Consulting Engin	164146	01-408.318	2019-Apr-Lot 2A Southbury Park	05/03/19	434.00	.00	
eystone Consulting Engin	164147	08-429.607	Grading Plan 2019-Apr-Sanitary Sewer CIPP Lining	05/03/19	949.88	.00	
Ceystone Consulting Engin	164148	01-408.318	Project 2018 2019-Apr-Emmaus Reservoir Grading	05/03/19	537.75	.00	
Keystone Consulting Engin	164149	01-408.318	Plan 2019-Apr-LVHN Ed Expansion	05/03/19	2,760.13	.00	
Keystone Consulting Engin	164150	01-408.318	2019-Apr-2882 E Texas Blvd-Minor	05/03/19	325.50	.00	
Keystone Consulting Engin	164151	01-408.318	Subdivision 2019-Apr-1651 Bobalew Trail Grading	05/03/19	815.01	.00	
Keystone Consulting Engin	164152	08-429.313	Plan 2019-Apr-General Engineering-Sewer	05/03/19	2,194.64	.00	
(eystone Consulting Engin	164153	01-408.313	Matters 2019-Apr-General Engineering-Traffic	05/03/19	217.00	.00	
Keystone Consulting Engin	164154	08-429.372	2019-Apr-I&I Removal Program	05/03/19	1,078.00	.00	
Keystone Consulting Engin	164155	08-429.607	2019-Apr-Sanitary Sewer CIPP Lining	05/03/19	1,858.51	.00	
ovetone Consulting Engin	164156	01-408.318	Project 2018	05/03/19	4,522.00	.00	
Ceystone Consulting Engin Ceystone Consulting Engin	164157	01-408.318	2019-Apr-Plot 886 LLC 2019-Apr-2242 Church Rd Minor	05/03/19	4,322.00	.00	
,			Subdivision				
Keystone Consulting Engin Keystone Consulting Engin	164172 164173	01-408.313 01-408.313	2019-Apr-Planning Commission 2019-Apr-Lindberg Park	05/03/19 05/03/19	1,519.00 4,680.38	.00	
teystone Consulting Engin	104173	01-400.313	Perimeter/Trail	03/03/19	4,060.36	.00	
Total Keystone Consul	ting Engineers:				27,628.30	.00	
Cijak, Mark							
Kijak, Mark	5142019	01-414.220	Planning Commission Mtg-5/14/19	05/14/19	25.00	.00	
Total Kijak, Mark:					25.00	.00	
Klocek, Jessica							
Klocek, Jessica	5142019	01-414.220	Planning Commission Mtg-5/14/19	05/14/19	25.00	.00	
Total Klocek, Jessica:					25.00	.00	

		<u> </u>					
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
Knerr, David G. Knerr, David G.	2429	01-406.314	DPW Contract review	05/08/19	6,900.00	.00	
Total Knerr, David G.:					6,900.00	.00	
Kobalt Construction Kobalt Construction	4302019	01-200202	Lindberg Ph3 Proj-Final Rehabilitation -Thru 4/30/19-Pmt #9	04/30/19	20,160.00	20,160.00	05/10/19
Total Kobalt Construct	ion:				20,160.00	20,160.00	
Kress, Kevin J.							
ress, Kevin J.	5082019	01-410.421	Reimb-Lodging & meals for Evidence Room Seminar	05/08/19	189.66	.00	
Total Kress, Kevin J.:					189.66	.00	
_aky, Jason							
∟aky, Jason	5082019	01-410.215	Reimb -Vet bills-K-9 Rocket	05/08/19	1,556.00	.00	
Total Laky, Jason:					1,556.00	.00	
Lehigh County Law Departi Lehigh County Law Depart		01-400.240	File Ord: #05-2019-622 Non-Uniform Pension Plan Changes	05/16/19	20.00	20.00	05/16/19
Total Lehigh County L	aw Department:				20.00	20.00	
_YNX Computer Technolog	jies						
YNX Computer Technolo	585121	01-407.319	LYNXcare Platinum-May-2019	05/08/19	1,678.00	.00	
YNX Computer Technolo	585121	01-410.319	LYNXcare Platinum-May-2019	05/08/19	2,310.00	.00	
YNX Computer Technolo	585121	06-448.319	LYNXcare Platinum-May-2019	05/08/19	361.00	.00	
YNX Computer Technolo	585121	08-429.319	LYNXcare Platinum-May-2019	05/08/19	361.00	.00	
YNX Computer Technolo	585121	10-407.319	LYNXcare Platinum-May-2019	05/08/19	130.00	.00	
Total LYNX Computer	Technologies:				4,840.00	.00	
Miller, Glenn							
∕liller, Glenn	5142019	01-414.220	Planning Commission Mtg-5/14/19	05/14/19	25.00	.00	
Total Miller, Glenn:					25.00	.00	
Morning Call, The							
Morning Call, The	4202019	01-401.420	Subscription renewal-acct#20063974 thru 11/02/2019	04/20/19	342.00	.00	
Total Morning Call, Th	e:				342.00	.00	
MP Outfitters							
MP Outfitters	32907-5	01-410.238	Alterations/badge holder-Nothstein	04/30/19	19.98	.00	
MP Outfitters	33138-5	01-410.238	Jackets-Stiles	05/09/19	87.99	.00	
Total MP Outfitters:					107.97	.00	
NAPA Auto Parts-Emmaus							
NAPA Auto Parts-Emmaus	2100-503268	01-437.251	#12-Tool box	05/14/19	47.98	.00	
Total NAPA Auto Parts	s-Emmaus <sup>.</sup>				47.98	.00	

		Nep	Joil dates. 3/4/2015-3/17/2015		IVIA	y 17, 2019 10.2	
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
PA Municipal Health Ins Coop	)						
PA Municipal Health Ins Co	5062019	01-487.156	2019 June coverage	05/06/19	87,778.41	.00	
PA Municipal Health Ins Co	5062019	06-487.156	2019 June coverage	05/06/19	8,376.52	.00	
PA Municipal Health Ins Co	5062019	08-487.156	2019 June coverage	05/06/19	8,376.52	.00	
PA Municipal Health Ins Co	5062019	10-487.156	2019 June coverage	05/06/19	1,849.87	.00	
Total PA Municipal Hea	alth Ins Coop:				106,381.32	.00	
PA One Call System Inc.							
PA One Call System Inc.	813288	01-401.320	2019-May-Voice/Fax messages	04/30/19	651.86	.00	
Total PA One Call Syst	tem Inc.:				651.86	.00	
Patten, Ronald C.							
Patten, Ronald C.	5102019	01-410.242	reimb-Shipping costs to return M-14	05/10/19	107.36	.00	
Total Patten, Ronald C	.:				107.36	.00	
PPL Electric Utilities	5000040	05 400 000	00540 00000	05/00//-	<b>^=</b> ==		
PPL Electric Utilities	5032019	35-433.362	96549-22003	05/03/19	37.67	.00	
PPL Electric Utilities	5032019	35-433.362	97159-57000	05/03/19	42.48	.00	
PPL Electric Utilities	5032019	08-429.361	98810-20005	05/03/19	161.99	.00	
PPL Electric Utilities	5032019	35-433.362	32810-04003	05/03/19	37.36	.00	
PPL Electric Utilities	5032019	35-433.362	58010-04003 70300 43006	05/03/19	39.13	.00	
PPL Electric Utilities	5032019	35-434.361	79200-43006	05/03/19 05/03/19	10,531.25 508.69	.00	
PPL Electric Utilities PPL Electric Utilities	5032019 5032019	35-434.361 06-448.361	79000-43000 59410-04007	05/03/19	30.85	.00	
PPL Electric Utilities	5032019	35-433.362	59610-04003	05/03/19	41.34	.00	
PPL Electric Utilities	5032019	08-429.361	95425-66011	05/03/19	30.61	.00	
PPL Electric Utilities	5032019	08-429.361	49200-35001	05/03/19	35.95	.00	
PPL Electric Utilities	5062019	08-429.361	47020-14000	05/06/19	30.35	.00	
PPL Electric Utilities	5092019	08-429.361	08450-07007	05/09/19	29.28	.00	
PPL Electric Utilities	5092019	08-429.361	15450-07002	05/09/19	28.77	.00	
PPL Electric Utilities	5092019	08-429.361	32650-08001	05/09/19	29.34	.00	
PPL Electric Utilities	5092019	08-429.361	42850-08008	05/09/19	29.34	.00	
PPL Electric Utilities	5092019	06-448.361	30460-06001	05/09/19	532.55	.00	
PPL Electric Utilities	5092019	01-451.361	54060-05002	05/09/19	28.15	.00	
PPL Electric Utilities	5092019	08-429.361	56060-06003	05/09/19	28.91	.00	
Total PPL Electric Utilit	ties:				12,234.01	.00	
Print-O-Stat Inc. Print-O-Stat Inc.	A0106820	01-414.240	Saddlebrook/Devonshire/Flash drive	04/26/19	16.60	.00	
		01-414.240	Saddleblook/Devolishiller lash drive	04/20/19			
Total Print-O-Stat Inc.:					16.60	.00	
ReadyRefresh by Nestle ReadyRefresh by Nestle	09E0441704392	01-409.366	2019-Apr-Acct#0441704392-Franko	05/08/19	4.23	.00	
ReadyRefresh by Nestle	19E0436481527	01-409.366	2019-Apr-Acct#0436481527-Admin	05/08/19	59.86	.00	
ReadyRefresh by Nestle	19E0436481547	01-409.366	2019-Apr-Acct#0436481543-Police	05/08/19	59.86 67.05	.00	
ReadyRefresh by Nestle	19E0436481576	01-409.366	2019-Apr-Acct#0436481576-DPW	05/08/19	22.35	.00	
Total ReadyRefresh by	Nestle:				153.49	.00	
Russo, Jevin							
Russo, Jevin	2019-013	03-412.421	CPR/AED First Aid Training	05/08/19	40.00	.00	

Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Pai
Total Russo, Jevin:					40.00	.00	
Sabo, Donald							
Sabo, Donald	5072019	01-410.187	Tuition/book reimbPersonnel Management course	05/07/19	217.50	.00	
Total Sabo, Donald:					217.50	.00	
Salisbury Township School	Dist						
Salisbury Township School	5062019	01-465.309	Custodian charges- Basketball/Baseball	05/06/19	7,790.41	.00	
Total Salisbury Townsh	nip School Dist:				7,790.41	.00	
Schreiter P.L.S., Richard L.							
Schreiter P.L.S., Richard L.	5142019	01-414.220	Planning Commission Mtg-5/14/19	05/14/19	25.00	.00	
Total Schreiter P.L.S.,	Richard L.:				25.00	.00	
Selex ES	00000	04 440 700	Discourse	0./20//-	01.010	<b>a</b> -	
Selex ES	33668	01-410.700	Pole cameras	04/30/19	21,240.00	.00	
Total Selex ES:					21,240.00	.00	
ervice Electric Cable TV							
ervice Electric Cable TV	5012019	01-410.315	Prolog Express -May 2019	05/01/19	80.56	.00	
Total Service Electric C	Cable TV:				80.56	.00	
Shammy Shine Car Washes							
Shammy Shine Car Washe	043019	01-410.315	Police Car Washes-4/1-4/30/19	04/30/19	120.00	.00	
Total Shammy Shine C	ar Washes:				120.00	.00	
HRM HRM	4302019	01-401.420	2019-Membership to human resource	04/30/19	209.00	.00	
or it vivi	4302013	01-401.420	management-Wilson	04/30/13			
Total SHRM:					209.00	.00	
South Whitehall Township							
South Whitehall Township	14877	06-448.369	1Q2019 water/Acct#289	04/30/19	1,365.00	.00	
Total South Whitehall	Township:				1,365.00	.00	
Staples Advantage							
taples Advantage	3412778866	01-401.240	2GB USB Drives/Stylus pen	05/03/19	119.16	.00	
Total Staples Advantag	ge:				119.16	.00	
tauffer Insurance, C. M.							
Stauffer Insurance, C. M.	112603	01-486.351	2019-Comm Liab Policy-vacant land	01/29/19	894.00	.00	
tauffer Insurance, C. M.	112640	01-486.351	2019-Commercial Umbrella policy	01/17/19	10,138.50	.00	
tauffer Insurance, C. M.	112640	06-486.351	2019-Commercial Umbrella policy	01/17/19	563.25	.00	
tauffer Insurance, C. M.	112640	08-486.351	2019-Commercial Umbrella policy	01/17/19	563.25	.00	
tauffer Insurance, C. M.	113101	01-486.354	2019-Workers Compensation	03/13/19	193,724.18	.00	
Stauffer Insurance, C. M.	113101	06-486.354	2019-Workers Compensation	03/13/19	21,262.41	.00	
			•				

Vendor Name							,, 20.0 .0.2	
Studies   Stud	Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
Studies   Stud	Stauffer Insurance, C. M. 11	13113	01-486.354	2019-Commercial Package Policy	03/08/19	36,953.10	.00	
Statuffer insurance, C. M. 131313   08 488.351   2019-Commercial Package Policy   C304919   2,052.95   0.0		13113	06-486.351		03/08/19		.00	
State   Stat		13113	08-486.351	• •	03/08/19	2,052.95	.00	
Stautifer Insurance, C. M.   113366   06-486.354   2018-WC adjusted premium-audit   04/29/19   2,212.92   .00   assessment   2018-WC adjusted premium-audit   04/29/19   2,212.92   .00	Stauffer Insurance, C. M. 11	13366	01-486.354	, ,	04/29/19	20,162.16	.00	
Stauffer Insurance, C. M.   113866   094-86.354   2018-WC adjusted premium-audit   assessment   2019   2,212.02   0.00	Stauffer Insurance, C. M. 11	13366	06-486.354	2018-WC adjusted premium-audit	04/29/19	2,212.92	.00	
Statis   Statemanne   C. M.   13816   01-403.353   2019-Money & Securities Policy   05-0119   137.50   0.0	Stauffer Insurance, C. M. 11	13366	08-486.354	2018-WC adjusted premium-audit	04/29/19	2,212.92	.00	
Stites, Allen   5062019   01-410.460   Reimt-Tolls & Meals for FBULeeda   05/06/19   128.83   .00	Stauffer Insurance, C. M. 11	13616	01-403.353	2019-Money & Securities Policy-	05/01/19	137.50	.00	
Silbes, Allen   5062019	Total Stauffer Insurance, 0	C. M.:				314,192.50	.00	
Total Silles, Allen:   128.83   0.00	Stiles, Allen							
Stotz/Fatzinger Office Supply   Stotz/Fatzinger Office Sup   10128CM	Stiles, Allen 50	062019	01-410.460		05/06/19	128.83	.00	
Stotz/Fatzinger Office Sup   10128CM   01-410_240   Return-Envelopes   0425/19   21.26   0.0     Total Stotz/Fatzinger Office Sup   232513   01-401_240   Copy paper   05/15/19   164.95   .0.0     Total Stotz/Fatzinger Office Supply:	Total Stiles, Allen:					128.83	.00	
Stratix Systems Inc   399234	Stotz/Fatzinger Office Supply							
Total Stotz/Fatzinger Office Supply: 143.6900  Stratix Systems Inc 399199 01-401.375 Maint fee-police 1st fir copier-5/12/19- 05/06/19 448.3500  Stratix Systems Inc 399233 01-401.375 Maint fee-police 2nd fir copier-5/12/19 05/06/19 198.9100  Stratix Systems Inc 399234 01-401.375 Maint fee-police 2nd fir copier-5/12/19 05/06/19 198.9100  Stratix Systems Inc 399234 01-401.375 Maint fee-police 2nd fir copier-5/12/19 05/06/19 642.8000  Stratix Systems Inc 399234 01-401.375 Maint fee Admin-5/12/2019-8/11/2019 05/06/19 642.8000  Total Stratix Systems Inc 01156 01-414.240 Magenta toner for P/Z 05/15/19 167.0000  Suburban Propane-2022  Suburban Propane-2022 805888 01-410.231 unleaded 05/03/19 15.7400  Suburban Propane-2022 805888 01-410.231 unleaded 05/03/19 1,148.3800  Suburban Propane-2022 805888 01-430.221 unleaded 05/03/19 1,148.0000  Suburban Propane-2022 805888 08-432.231 unleaded 05/03/19 203.1000  Suburban Propane-2022 805888 08-432.231 unleaded 05/03/19 203.1000  Suburban Propane-2022 805888 08-432.231 unleaded 05/03/19 203.1000  Suburban Propane-2022 805888 08-432.231 unleaded 05/03/19 187.8200  Suburban Propane-2022 80588 08-432.231 unleaded 05/03/19 863.300  Suburban Propane-2022 80588 08-432.231 diesel 05/03/19 629.9800  Suburban Propane-2022 95082 01-410.231 diesel 05/03/19 629.9800  Suburban Propane-2022 95082 01-430.231 diesel 05/03/19 111.4100  Suburban Propane-2022 95082 08-432.231 diesel 05/03/19 111.4100  Suburban Propane-2022 95082 08-432.231 diesel 05/03/19 111.4100  Total Suburban Propane-2022 95082 08-432.231 diesel 05/03/19 103.0300  Total Tim	Stotz/Fatzinger Office Sup 10	0128CM	01-410.240	Return-Envelopes	04/25/19	21.26-	.00	
Stratix Systems Inc         399199         01-401.375         Maint fee-police 1st fir copier-5/12/19         05/06/19         448.35         .00           Stratix Systems Inc         399233         01-401.375         Maint fee-police 2nd fir copier-5/12/19         05/06/19         198.91         .00           Stratix Systems Inc         399234         01-401.375         Maint fee-police 2nd fir copier-5/12/19         05/06/19         642.80         .00           Stratix Systems Inc         401156         01-414.240         Magenta toner for P/Z         05/15/19         167.00         .00           Total Stratix Systems Inc:         1,457.06         .00           Suburban Propane-2022           Suburban Propane-2022         805888         01-401.231         unleaded         05/03/19         15.74         .00           Suburban Propane-2022         805888         01-401.231         unleaded         05/03/19         1,194.60         .00           Suburban Propane-2022         805888         01-402.31         unleaded         05/03/19         1,194.60         .00           Suburban Propane-2022         805888         04-422.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022	Stotz/Fatzinger Office Sup 23	32513	01-401.240	Copy paper	05/15/19	164.95	.00	
Stratix Systems Inc   399199   01-401.375   Maint fee-police 1st fir copier-5/12/19   05/06/19   448.35   .00   8/11/2019   05/06/19   198.91   .00	Total Stotz/Fatzinger Offic	e Supply:				143.69	.00	
Stratix Systems Inc   399233   01-401.375   Maint fee-police 2nd fir copier-5/12/19   05/06/19   198.91   .00	Stratix Systems Inc							
Stratix Systems Inc   399234   01-401.375   Maint fee Admin-5/12/2019-8/11/2019   05/06/19   642.80   .00     Stratix Systems Inc   401156   01-414.240   Magenta toner for P/Z   05/15/19   167.00   .00     Total Stratix Systems Inc:   1,457.06   .00     Suburban Propane-2022   805888   01-401.231   unleaded   05/03/19   15.74   .00     Suburban Propane-2022   805888   01-410.231   unleaded   05/03/19   1,148.38   .00     Suburban Propane-2022   805888   01-430.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   805888   06-448.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   805888   06-448.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   805888   08-429.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   805888   03-413.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   805888   03-413.231   unleaded   05/03/19   203.10   .00     Suburban Propane-2022   95082   01-401.231   diesel   05/03/19   8.63   .00     Suburban Propane-2022   95082   01-410.231   diesel   05/03/19   629.98   .00     Suburban Propane-2022   95082   01-410.231   diesel   05/03/19   655.31   .00     Suburban Propane-2022   95082   01-430.231   diesel   05/03/19   655.31   .00     Suburban Propane-2022   95082   08-429.231   diesel   05/03/19   111.41   .00     Suburban Propane-2022   95082   03-413.231   diesel   05/03/19   111.41   .00     Total Suburban Propane-2022   95082   03-413.231   diesel   05/03/19   111.41   .00     Total Suburban Propane-2022   95082   03-413.231   diesel   05/03/19   103.03   .00     Total Suburban Propane-2022   950	Stratix Systems Inc 39	99199	01-401.375		05/06/19	448.35	.00	
Stratix Systems Inc	Stratix Systems Inc 39	99233	01-401.375		05/06/19	198.91	.00	
Total Stratix Systems Inc:   1,457.06   .00	Stratix Systems Inc 39	99234	01-401.375	Maint fee Admin-5/12/2019-8/11/2019			.00	
Suburban Propane-2022	Stratix Systems Inc 40	01156	01-414.240	Magenta toner for P/Z	05/15/19	167.00	.00	
Suburban Propane-2022         805888         01-401.231         unleaded         05/03/19         15.74         .00           Suburban Propane-2022         805888         01-410.231         unleaded         05/03/19         1,148.38         .00           Suburban Propane-2022         805888         01-430.231         unleaded         05/03/19         1,194.60         .00           Suburban Propane-2022         805888         06-448.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         08-429.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         08-429.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         03-413.231         unleaded         05/03/19         187.82         .00           Suburban Propane-2022         95082         01-401.231         diesel         05/03/19         8.63         .00           Suburban Propane-2022         95082         01-430.231         diesel         05/03/19         655.31         .00           Suburban Propane-2022         95082         08-448.231         diesel         05/03/19         111.41         .	Total Stratix Systems Inc:					1,457.06	.00	
Suburban Propane-2022         805888         01-410.231         unleaded         05/03/19         1,148.38         .00           Suburban Propane-2022         805888         01-430.231         unleaded         05/03/19         1,194.60         .00           Suburban Propane-2022         805888         06-448.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         08-429.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         03-413.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         95082         01-401.231         diesel         05/03/19         8.63         .00           Suburban Propane-2022         95082         01-410.231         diesel         05/03/19         629.98         .00           Suburban Propane-2022         95082         01-402.31         diesel         05/03/19         655.31         .00           Suburban Propane-2022         95082         06-448.231         diesel         05/03/19         111.41         .00           Suburban Propane-2022         95082         08-429.231         diesel         05/03/19         113.41         .00 <td>Suburban Propane-2022</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Suburban Propane-2022							
Suburban Propane-2022         805888         01-430.231         unleaded         05/03/19         1,194.60         .00           Suburban Propane-2022         805888         06-448.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         08-429.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         03-413.231         unleaded         05/03/19         187.82         .00           Suburban Propane-2022         95082         01-401.231         diesel         05/03/19         8.63         .00           Suburban Propane-2022         95082         01-410.231         diesel         05/03/19         629.98         .00           Suburban Propane-2022         95082         01-430.231         diesel         05/03/19         655.31         .00           Suburban Propane-2022         95082         06-448.231         diesel         05/03/19         111.41         .00           Suburban Propane-2022         95082         08-429.231         diesel         05/03/19         111.41         .00           Suburban Propane-2022         95082         08-429.231         diesel         05/03/19         103.03         .00	Suburban Propane-2022 80	05888	01-401.231	unleaded	05/03/19	15.74	.00	
Suburban Propane-2022         805888         06-448.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         08-429.231         unleaded         05/03/19         203.10         .00           Suburban Propane-2022         805888         03-413.231         unleaded         05/03/19         187.82         .00           Suburban Propane-2022         95082         01-401.231         diesel         05/03/19         8.63         .00           Suburban Propane-2022         95082         01-410.231         diesel         05/03/19         629.98         .00           Suburban Propane-2022         95082         01-430.231         diesel         05/03/19         655.31         .00           Suburban Propane-2022         95082         06-448.231         diesel         05/03/19         655.31         .00           Suburban Propane-2022         95082         08-429.231         diesel         05/03/19         111.41         .00           Suburban Propane-2022         95082         08-429.231         diesel         05/03/19         111.41         .00           Suburban Propane-2022         95082         03-413.231         diesel         05/03/19         103.03         .00 <td>Suburban Propane-2022 80</td> <td>05888</td> <td>01-410.231</td> <td>unleaded</td> <td>05/03/19</td> <td>1,148.38</td> <td>.00</td> <td></td>	Suburban Propane-2022 80	05888	01-410.231	unleaded	05/03/19	1,148.38	.00	
Suburban Propane-2022       805888       08-429.231       unleaded       05/03/19       203.10       .00         Suburban Propane-2022       805888       03-413.231       unleaded       05/03/19       187.82       .00         Suburban Propane-2022       95082       01-401.231       diesel       05/03/19       8.63       .00         Suburban Propane-2022       95082       01-410.231       diesel       05/03/19       629.98       .00         Suburban Propane-2022       95082       01-430.231       diesel       05/03/19       655.31       .00         Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       113.41       .00         Times News         Times News       100459223-050120       01-401.341       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 80	05888	01-430.231	unleaded	05/03/19	1,194.60	.00	
Suburban Propane-2022       805888       03-413.231       unleaded       05/03/19       187.82       .00         Suburban Propane-2022       95082       01-401.231       diesel       05/03/19       8.63       .00         Suburban Propane-2022       95082       01-410.231       diesel       05/03/19       629.98       .00         Suburban Propane-2022       95082       01-430.231       diesel       05/03/19       655.31       .00         Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       111.41       .00         Total Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Times News         Times News       100459223-050120       01-401.341       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 80	05888	06-448.231	unleaded	05/03/19	203.10	.00	
Suburban Propane-2022       95082       01-401.231       diesel       05/03/19       8.63       .00         Suburban Propane-2022       95082       01-410.231       diesel       05/03/19       629.98       .00         Suburban Propane-2022       95082       01-430.231       diesel       05/03/19       655.31       .00         Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Total Suburban Propane-2022:       4,572.51       .00         Times News         Times News:       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 80	05888	08-429.231	unleaded	05/03/19	203.10	.00	
Suburban Propane-2022       95082       01-410.231       diesel       05/03/19       629.98       .00         Suburban Propane-2022       95082       01-430.231       diesel       05/03/19       655.31       .00         Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Total Suburban Propane-2022:       4,572.51       .00         Times News         Times News:       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 80	05888	03-413.231	unleaded	05/03/19	187.82	.00	
Suburban Propane-2022       95082       01-430.231       diesel       05/03/19       655.31       .00         Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Total Suburban Propane-2022:       4,572.51       .00         Times News         Times News:       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 95	5082	01-401.231	diesel	05/03/19	8.63	.00	
Suburban Propane-2022       95082       06-448.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Total Suburban Propane-2022:       4,572.51       .00         Times News         Times News       100459223-050120       01-401.341       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 95	5082	01-410.231	diesel	05/03/19	629.98	.00	
Suburban Propane-2022       95082       08-429.231       diesel       05/03/19       111.41       .00         Suburban Propane-2022       95082       03-413.231       diesel       05/03/19       103.03       .00         Total Suburban Propane-2022:       4,572.51       .00         Times News         Times News       100459223-050120       01-401.341       Ad:ZHB mtg 5/7/19       05/09/19       205.10       .00         Tractor Supply Credit Plan	Suburban Propane-2022 95	5082	01-430.231	diesel	05/03/19	655.31	.00	
Suburban Propane-2022         95082         03-413.231         diesel         05/03/19         103.03         .00           Total Suburban Propane-2022:         4,572.51         .00           Times News           Times News         100459223-050120         01-401.341         Ad:ZHB mtg 5/7/19         05/09/19         205.10         .00           Total Times News:         205.10         .00	Suburban Propane-2022 95	5082	06-448.231	diesel	05/03/19	111.41	.00	
Times News Times News   100459223-050120   01-401.341   Ad:ZHB mtg 5/7/19   05/09/19   205.10   .00  Total Times News:   205.10   .00  Tractor Supply Credit Plan	Suburban Propane-2022 95	5082	08-429.231	diesel	05/03/19	111.41	.00	
Times News Times News 100459223-050120 01-401.341 Ad:ZHB mtg 5/7/19 05/09/19 205.10 .00  Total Times News: 205.10 .00  Tractor Supply Credit Plan	Suburban Propane-2022 95	5082	03-413.231	diesel	05/03/19	103.03	.00	
Times News 100459223-050120 01-401.341 Ad:ZHB mtg 5/7/19 05/09/19 205.10 .00  Total Times News: 205.10 .00  Tractor Supply Credit Plan	Total Suburban Propane-2	2022:				4,572.51	.00	
Total Times News: 205.10 .00  Tractor Supply Credit Plan	Times News							
Tractor Supply Credit Plan	Times News 10	0459223-050120	01-401.341	Ad:ZHB mtg 5/7/19	05/09/19	205.10	.00	
	Total Times News:					205.10	.00	
Tractor Supply Credit Plan 892197 01-430.240 LED Light/Coupling/Level 05/08/19 43.96 .00	Tractor Supply Credit Plan							
	Tractor Supply Credit Plan 89	92197	01-430.240	LED Light/Coupling/Level	05/08/19	43.96	.00	

		кер	ort dates: 5/4/2019-5/17/2019		IVIa	y 17, 2019 10:2	ZZAWI
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid
Total Tractor Supply (	Credit Plan:				43.96	.00	
Upper Saucon Township							
Upper Saucon Township	190501	10-431.303	Compost site expenses	05/01/19	78.41	.00	
Total Upper Saucon 1	Fownship:				78.41	.00	
Verizon							
Verizon	4242019	08-429.320	2019-Apr-868-8310-Cardinal Dr Pump Stn	04/24/19	36.68	.00	
Verizon	4252019	01-401.320	2019-Apr-Mgr	04/25/19	54.96	.00	
Verizon	4252019	01-402.320	2019-Apr-Finance Director	04/25/19	54.96	.00	
Verizon	4252019	01-409.320	2019-Apr-Facility Maintenance	04/25/19	54.96	.00	
Verizon	4252019	01-410.320	2019-Apr-Police	04/25/19	481.84	.00	
Verizon	4252019	01-413.320	2019-Apr-Code Enf Ofcr	04/25/19	54.96	.00	
Verizon	4252019	01-414.320	2019-Apr-P/Z Officer	04/25/19	54.96	.00	
Verizon	4252019	01-415.320	2019-Apr-Tapler	04/25/19	78.03	.00	
Verizon	4252019	01-430.320	2019-Apr-DPW	04/25/19	140.56	.00	
Verizon	4252019	01-452.320	2019-Apr-Police tablet/air card	04/25/19	331.81	.00	
Verizon	4252019	06-448.320	charges 2019-Apr-Water Dept	04/25/19	66.08	.00	
Verizon	4252019	01-410.320	2019-Apr-Mgr/Tablet	04/25/19	10.02	.00	
Verizon	4252019	01-452.320	2019-Apr-Rec Dir Tablet	04/25/19	271.00	.00	
Verizon	4252019	01-411.320	2019-Apr-Fire Inspectors/Tablet	04/25/19	34.31	.00	
Verizon	4252019	01-413.320	2019-Apr-Code Enf Ofcr Tablet	04/25/19	10.02	.00	
Verizon	4252019	03-411.320	2019-Apr-ESFC/Tablet	04/25/19	108.69	.00	
Verizon	4252019	03-412.320	2019-Apr-WSFC/Tablet	04/25/19	181.55	.00	
Verizon	4252019	01-452.320	2019-Apr-Rec Dir	04/25/19	54.96	.00	
Verizon	5142019	06-448.320	610-435-4231-Flxr Ave wtr pump stn-	05/14/19	42.54	.00	
Verizon	5142019	01-410.320	5/2019 610-433-0563-monitor traffic signals CCB-5/2019	05/14/19	41.93	.00	
Total Verizon:					2,164.82	.00	
Whitehall Turf Equipment							
Whitehall Turf Equipment	403348	01-451.260	Weed wacker/backpack blower	01/26/19	1,242.45	.00	
Whitehall Turf Equipment	403791	01-451.240	1 gallon mix	05/01/19	74.75	.00	
Whitehall Turf Equipment	404244	01-451.375	open backpack rack/Trimmer rack	05/06/19	385.37	.00	
Whitehall Turf Equipment	404582	01-451.375	open trailer Throttle kit & air filter for Echo blower	05/08/19	43.99	.00	
Whitehall Turf Equipment	404583	01-451.375	Backpack blower repair	05/08/19	21.00	.00	
writteriali Turi Equipment	404303	01-401.070	Dackpack blower repair	03/00/19	21.00		
Total Whitehall Turf E	Equipment:				1,767.56	.00	
Ziegenfus, Paul							
Ziegenfus, Paul	5022019	01-402.420	Reimb-lodging/meals/mileage/tolls- GFOA conference	05/02/19	657.89	657.89	05/15/19
Ziegenfus, Paul	5162019	01-402.421	Reimb-Mileage for training	05/16/19	46.40	.00	
Total Ziegenfus, Paul	:				704.29	657.89	
Grand Totals:					583,439.87	20,837.89	

Salisbury Township		Payment Approval Report - BOC Report dates: 5/4/2019-5/17/2019			Page: 10 May 17, 2019 10:22AM			
Vendor Name	Invoice Number	GL Account Number	Description	Invoice Date	Net Invoice Amount	Amount Paid	Date Paid	
Report Criteria: Detail report. Invoice detail records Paid and unpaid invoi	•							

Salisbury Township

#### Payment Approval Report - Summary by Fund Report dates: 5/4/2019-5/17/2019

Page: 1 May 17, 2019 10:24AM

Vendor Name	Invoice Number	Description	GL Acct Number	Invoice Date	Net Invoice Amt	Amount Paid	Date Pai
General Fund							<u>-</u>
Total General Fund:					485,712.89	20,837.89	
ire Fund							<u>-</u>
Total Fire Fund:					1,276.94	.00	
Vater Fund							-
Total Water Fund:					40,670.21	.00	<u>-</u>
Sewer Fund							
Total Sewer Fund:					42,383.63	.00	-
Refuse & Recycling Fund	i						
Total Refuse & Recycling Fund:				2,158.28	.00	-	
lighway Aid Fund							
Total Highway Aid F	und:				11,237.92	.00	-
Grand Totals:					583,439.87	20,837.89	

Date: _	
President:	
Board of Commissioners	

# TOWNSHIP OF SALISBURY LEHIGH COUNTY, PENNSYLVANIA MINUTES FROM THE BOARD OF COMMISSIONERS May 9, 2019 REGULAR MEETING – 7:00 PM

The Public Meeting of the Salisbury Township Board of Commissioners was held on the above date in the Township Municipal Building located at 2900 South Pike Avenue, Allentown, Lehigh County, Pennsylvania.

#### PLEDGE OF ALLEGIANCE

Commissioner Martucci asked everyone to rise and recite the Pledge of Allegiance.

#### **CALL TO ORDER**

Commissioner Martucci called the meeting to order at 7:00 p.m.

#### **ROLL CALL**

#### **Board Members Present:**

Robert Martucci, Jr., President Debra Brinton, Vice-President James Seagreaves, President Pro-Tempore Joanne Ackerman Rodney Conn

#### **Staff Present:**

Cathy Bonaskiewich, Township Manager – EXCUSED
Sandy Nicolo, Assistant Township Manager/Code Enforcement Director, MS4 Coordinator
Paul Ziegenfus, Finance Director
John Andreas, Director of Public Works
Allen Stiles, Chief of Police
Cynthia Sopka, Director of Planning & Zoning
John Ashley, Esquire, rep of Davison & McCarthy, Township Solicitor
David Tettemer, rep of Keystone Consulting Engineers, Township Engineer

Commissioner Martucci announced that the Board will convene into an Executive Session regarding a personnel matter.

The Board reconvened and made the following Motion:

Motion by Commissioner Ackerman, seconded by Commissioner Conn, to accept the letter of resignation from Mr. Kyle Rehatchek from the rank of corporal and return to the rank of patrol officer effective 4/24/19.

#### **Roll Call:**

COMMISSIONER ACKERMAN – YES COMMISSIONER SEAGREAVES – YES COMMISSIONER BRINTON – YES COMMISSIONER CONN – YES COMMISSIONER MARTUCCI – YES

#### The Motion passed by 5-0.

Commissioner Martucci turned the proceedings over to Mr. Sandy Nicolo, Assistant Township Manager.

#### **NOTIFICATION**

Mr. Nicolo informed the attendees that all sessions of the Salisbury Township Board of Commissioners regular meetings are recorded electronically for the purpose of taking the Minutes. All public comments on agenda items will be taken prior to the vote. All public commented related to non-agenda items will be taken after the agenda has been satisfied.

Mr. Nicolo reminded everyone of the three-minute rule and asked that anyone who wishes to speak, come to the podium, announce oneself, and sign-in with name and address so as to assist Township administration with follow-ups.

#### APPROVAL OF THE LIST OF BILLS PAYABLE

Motion by Commissioner Ackerman, seconded by Commissioner Seagreaves, to approve the the list of Bills Payable for the period 4/20/2019-5/3/2019, broken down as follows:

\$34,481.45 = GENERAL \$1,451.32 = FIRE \$0 = LIBRARY \$2,088.56 = WATER \$2,616.94 = SEWER \$99,524.32 = REFUSE & RECYCLING \$238.55 = HIGHWAY AID \$0 = SUBDIVISION & ESCROW \$209,077.39 = GRAND TOTAL ALL FUNDS

#### **Roll Call:**

COMMISSIONER ACKERMAN – YES COMMISSIONER SEAGREAVES – YES COMMISSIONER BRINTON – YES COMMISSIONER CONN – YES COMMISSIONER MARTUCCI – YES

The Motion passed by 5-0.

#### **MINUTES**

#### **April 25, 2019**

Commissioner Martucci declared the Minutes of April 25, 2019 accepted as presented.

#### **NEW BUSINESS**

**ORDINANCES** 

None.

**RESOLUTIONS** 

None.

#### **MOTIONS**

#### Motion to Authorize Renewal of Franko Lease Agreement.

Mr. Nicolo noted that the lease agreement will be renewed annually.

Motion by Commissioner Conn, seconded by Commissioner Brinton, to authorize the renewal of the Franko Lease Agreement.

#### **Roll Call:**

COMMISSIONER ACKERMAN – YES COMMISSIONER SEAGREAVES – YES COMMISSIONER BRINTON – YES COMMISSIONER CONN – YES COMMISSIONER MARTUCCI – YES

The Motion passed by 5-0.

Motion to Approve Payment #9 to Kobalt Construction for Lindberg Park Phase 3 Contracted Work in the amount of \$20,160.00.

Motion by Commissioner Seagreaves, seconded by Commissioner Conn, to approve Payment #9 to Kobalt Construction for Lindberg Park Phase 3 Contracted Work in the amount of \$20,160.00.

#### **Roll Call:**

COMMISSIONER ACKERMAN – YES COMMISSIONER SEAGREAVES – YES COMMISSIONER BRINTON – YES

#### COMMISSIONER CONN – YES COMMISSIONER MARTUCCI – YES

The Motion passed by 5-0.

#### **PRIVLEGE OF THE FLOOR**

Ms. Patricia Heckman of 1623 Lehigh Avenue expressed concerns about water run-off from East Susquehanna Street and inquired if the Township had jurisdiction over that road. Commissioner Seagreaves responded that it is a state road. Mr. Nicolo noted that some of the water issues in the area will be looked at when the Township starts the Laubach Park Master Site Plan.

Ms. Heckman commented that there are a large number of cars parked on Lehigh Avenue and alleged that a mechanic who lives in the area is selling cars. Mr. Nicolo stated that he is familiar with the activity in the area and will follow-up with the resident.

Commissioner Brinton brought forth concerns about a resident holding concerts on East Rock Road. Ms. Sopka noted that she will follow-up with the property owner.

Commissioner Martucci commented that he received a call from Mr. Dennis Grejda regarding Salisbury Youth Association's use of Lindberg Park. A discussion ensued about the use of the fields by both SYA and Hamilton Park. Commissioner Martucci suggested that Mr. Grejda contact Ms. Genny Baillie and Mr. Nicolo noted that he would follow-up with Ms. Bonaskiewich.

Commissioner Martucci stated that the Drop-off Center Operators have expressed concerns about the placement of the newly-installed shelter. Mr. Andreas commented that he believes the shelter is strategically placed for the most effective use of the structure because of the need to check residency and materials. Commissioner Martucci noted that the operators cannot see the cars coming in so he suggested installing a window. Mr. Andreas stated that he can look into relocating it, but it would take more sitework. He also commented that there will be new signage at both locations.

Chief Stiles provided statistics to the Board relating to Drug-Take-Back Day.

Chief Stiles announced that the Police Department has received grant funding for the continuation of their accreditation program.

Chief Stiles thanked Mr. Andreas and Fountain Hill Borough for helping to install license plate reader system on Broadway.

#### **ADJOURNMENT**

Commissioner Brinton made a motion to adjourn the meeting. Seconded by Commissioner Ackerman. The time was 7:50 p.m.

Respectfully submitted,
Cathy Bonaskiewich Township Secretary
These constitute the official minutes of the Regular Meeting of the Board of Commissioners held on May 9, 2019.
Approved and certified on this date:
Cathy Bonaskiewich
Date:
CEAL
SEAL

## TOWNSHIP OF SALISBURY LEHIGH COUNTY, PENNSYLVANIA

Ordinance No. 05-2019-622

(Duly Adopted May 23, 2019)

AN ORDINANCE ELECTING TO AMEND THE SALISBURY TOWNSHIP NON-UNIFORMED EMPLOYEES DEFINED BENEFIT PENSION PLAN ADMINISTERED BY THE PENNSYLVANIA MUNICIPAL RETIREMENT SYSTEM PURSUANT TO ARTICLE IV OF THE PENNSYLVANIA MUNICIPAL RETIREMENT LAW TO CONFORM TO UPDATED SYSTEM PLAN DOCUMENTS, TO DESIGNATE REQUIRED EMPLOYEE CONTRIBUTIONS AS PRE-TAX CONTRIBUTIONS, AND TO CLOSE MEMBERSHIP IN THE PLAN SO THAT ONLY PERSONS EMPLOYED BEFORE JUNE 1, 2019 MAY BE COVERED BY THE PLAN; AGREEING TO BE ALL PROVISIONS OF THE PENNSYLVANIA BOUND BY MUNICIPAL RETIREMENT LAW AS AMENDED AND AS APPLICABLE TO **MEMBER** MUNICIPALITIES; **FURTHER** AMENDING THE NON-UNIFORMED EMPLOYEES DEFINED BENEFIT PENSION PLAN ADMINISTERED BY THE PENNSYL-VANIA MUNICIPAL RETIREMENT SYSTEM SO THAT EMPLOYEE CONTRIBUTIONS FOR CALENDAR YEAR 2019 ARE REDUCED TO ONE PERCENT (1.0%) OF COMPENSATION; AND ESTAB-LISHING A NEW DEFINED CONTRIBUTION PENSION PLAN FOR FULL-TIME NON-UNIFORMED EMPLOYEES FIRST EMPLOYED BY THE TOWNSHIP ON OR AFTER JUNE 1, 2019, PROVIDING FOR EMPLOYER CONTRIBUTIONS OF EIGHT PERCENT (8.0%) OF COMPENSATION, NO EMPLOYEE CONTRIBUTIONS. VESTING AFTER FIVE YEARS OF FULL-TIME SERVICE (WITH NO VESTING BEFORE), DISTRIBUTIONS AFTER SEPARATION FROM SERVICE, AND A PARTICIPANT-DIRECTED INVESTMENT PROGRAM.

WHEREAS, the Township of Salisbury, Lehigh County ("Township") has established a defined benefit pension plan for its non-uniformed employees administered by the Pennsylvania Municipal Retirement System ("PMRS"); and

WHEREAS, this defined benefit pension plan is currently governed by ordinance 05-2006-526, codified at Code of Ordinances § 1-621 through § 1-627, and an Agreement dated May 11, 2006 between the Township and the governing board of PMRS; and

WHEREAS, PMRS has substantially revised its pension plan documentation and obtained an advisory letter from the Internal Revenue Service, and has requested the Township to update its Agreement with PMRS accordingly; and

WHEREAS, the Township and the Salisbury Township Public Works Employees Association ("Association"), the collective bargaining representatives of the Township's non-managerial public works employees ("PW Employees"), have entered into a collective bargaining agreement ("CBA") effective from January 1, 2019 through December 31, 2022; and

WHEREAS, the CBA makes changes to the pension program applicable to PW Employees, and the Board of Commissioners ("Board") of the Township desires to comply with the CBA; and

WHEREAS, the Board desires to extend the pension program changes applicable to PW Employees to all non-uniformed employees of the Township; and

WHEREAS, the changes to the pension program include (1) designating employee contributions to the defined benefit plan as pre-tax contributions, (2) closing membership in the defined benefit plan so that only persons employed before June 1, 2019 may be covered by that plan, (3) reducing employee contributions for calendar year 2019 to one percent (1.0%) of compensation, and (4) establishing a new defined contribution plan for employees who first work an hour of service for the Township on or after June 1, 2019; and

WHEREAS, the new Salisbury Township Non-Uniformed Employees Defined Contribution Pension Plan provides contributions and benefits only for full-time non-uniformed employees of the Township (regularly scheduled for at least 35 hours per week) who first work on or after June 1, 2019, and provides for employer contributions of eight percent (8.0%) of compensation, no employee contributions, full vesting after

five years of full-time service (with no vesting before), distributions after separation from service, and a participant-directed investment program; and

**WHEREAS**, prior to the adoption of this Ordinance the Board reviewed the following cost estimate of the effect of the changes made herein:

The amendment to the defined benefit plan reducing the employee contribution rate for calendar year 2019 from three and one-half percent (3.5%) of compensation to one percent (1.0%) of compensation does not change the cost of the plan at all, but only the allocation of payment between the Township and the employees. So, for every dollar that the employee is not required to pay in to the plan, the Township will have to contribute an additional dollar. Accordingly, the reduction for 2019 means that the Township must pay an extra two and one-half percent (2.5%) of all members' compensation for 2019 more than it would have if there had been no reduction. However, for very many years before 2019, the Township eliminated all employee contributions each year, and so in 2019 the Township will actually be taking on a lesser portion of the "regular" employee contributions stated in the plan agreement than in previous years.

The amendment to the defined benefit plan designating employee contributions as pre-tax rather than after-tax contributions also has no effect on the cost of the plan. Employee contributions to the plan are in the same amount whether or not the employee pays tax on that income, and the total amount paid out by the plan with respect to any participant is the same whether or not the distributee pays or the plan withholds taxes.

The amendment of the defined benefit plan to conform to the language of the new PMRS base plan document and adoption agreement is not intended to make any actual change to the terms, conditions, and benefits of the plan, and so it has no effect on the cost of the plan.

The amendment of the defined benefit plan to close membership as of June 1, 2019 has no effect on the cost of the plan for existing plan members who continue in employment. However, if an existing member terminates, the normal cost that would have been associated with his/her compensation will no longer accrue and there will be no normal cost in the defined benefit plan associated with the compensation of any replacement employee. Thus, over time, it can be expected that the total compensation of members in the defined benefit plan will drop as members terminate employment, resulting in lower plan costs.

The establishment of the defined contribution plan will generally result in a cost to the Township of eight percent (8%) of the compensation of new non-uniformed full-time employees hired on or after June 1, 2019, although the long-term cost may actually be somewhat lower when employees leave with less than five years of full-time service and forfeit the amounts that were contributed by the Township to their accounts.

**NOW, THEREFORE**, be it **ORDAINED** and **ENACTED** by the Board of Commissioners of the Township of Salisbury, Lehigh County, Pennsylvania, as follows:

**SECTION 1.** Salisbury Township Code of Ordinances Chapter 1, Part 6(B), § 1-621 through § 1-627, and the Agreement dated May 11, 2006 between the Township and the Pennsylvania Municipal Retirement Board are hereby superseded by the following provisions of this Ordinance, effective as of January 1, 2019.

**SECTION 2.** Salisbury Township Code of Ordinances Chapter 1, Part 6 shall be amended by adding the following new subparts C and D:

# C Non-Uniformed Employees' Defined Benefit Pension Plan (2019-)

#### § 1-631 Amendment of Existing Plan.

Salisbury Township (the Township), having established a defined benefit pension plan for its non-uniformed employees administered by the Pennsylvania Municipal Retirement System (the System), hereby elects to amend its Non-Uniform Employees' Defined Benefit Pension Plan administered by the System in accordance with Article IV of the Pennsylvania Municipal Retirement Law, 53 P.S. §881.101 *et seq.* (Retirement Law), and does hereby agree to be bound by all the requirements and provisions of the Retirement Law and the Municipal Pension Plan Funding Standard and Recovery Act, 53 P.S. §895.101 *et seq.*, and to assume all obligations, financial and otherwise, placed upon member municipalities.

#### § 1-632 Authorization of Adoption Agreement.

The Board of Commissioners of the Township hereby authorizes and directs the President or Vice President of the Board and the Township Manager to execute on behalf of the Township an Adoption Agreement in the form attached to Ordinance 05-2019-622 as Exhibit 2. When fully executed, this Adoption Agreement and the underlying System Base Plan Document referenced therein (attached to Ordinance 05-2019-622 as Exhibit 1), shall constitute the Contract between the Township and the System with respect to the Salisbury Township Non-Uninform Employees' Defined Benefit Pension Plan, effective as of January 1, 2019.

#### § 1-633 Administration.

As part of this Ordinance, the Township agrees that the System shall administer and provide the benefits set forth in the Contract entered into between the Pennsylvania Municipal Retirement Board and the Township effective as of January 1, 2019.

#### § 1-634 Acceptance of Contract and Financial Obligations.

The Township acknowledges that by passage and adoption of this Ordinance, the Township officially accepts the Contract and the financial obligations resulting from the administration of the Contract.

#### § 1-635 Payments and Liability.

Payment for any obligation established by the adoption of this Ordinance and the Contract shall be made by the Township in accordance with the Retirement Law and the Municipal Pension Plan Funding Standard and Recovery Act. The Township hereby assumes all liability for any unfundedness created due to the benefit structure set forth in the Contract.

#### § 1-636 Intent.

The Township intends this Ordinance to be the complete authorization of the Contract, as amended and it shall become effective as of the date specified in the adoption agreement, which is the effective date of the Contract, as amended.

#### § 1-637 Filing with PMRS.

The Township shall file a duly certified copy of this Ordinance and an executed Contract with the System.

#### § 1-638 Amendments Regarding Employee Contributions.

(a) 2019. The Board of Commissioners of the Township hereby authorizes and directs the President or Vice President of the Board and the Township Manager to execute on behalf of the Township an Adoption Agreement Amendment in the form attached to Ordinance 05-2019-622 as Exhibit 3, in order to reduce employee contributions for calendar year 2019 only, subject to approval by the System. The Township shall file a duly certified copy of Ordinance 05-2019-622 and the executed Adoption Agreement Amendment with the System.

## D Non-Uniformed Employees Defined Contribution Pension Plan

#### **§ 1-641** Adoption.

The Board of Commissioners of Salisbury Township hereby establishes the Salisbury Township Nonuniformed Employees Defined Contribution Pension Plan in the form attached to Ordinance 05-2019-622 as Exhibit 4, effective June 1, 2019.

SECTION 3. The following exhibits are hereby incorporated into this Ordinance by reference:

Exhibit 1: Pennsylvania Municipal Retirement System Base Plan Document,

together with—

Addendum A: Applicable State Laws Addendum B: PMRS Policy Statements

Exhibit 2: Adoption Agreement 001 for PMRS Defined Benefit Base Plan Document.

Exhibit 3: Adoption Agreement Amendment to Reduce Employee Contributions to 1.0% for 2019 only.

Exhibit 4: Salisbury Township Nonuniformed Employees Defined Contribution Pension Plan.

**DULY ORDAINED** and **ENACTED** by the Board of Commissioners of the Township of Salisbury, Lehigh County, this 23<sup>rd</sup> day of May, 2019, in lawful session duly assembled.

	TOWNSHIP OF SALISBURY
	Robert Martucci, Jr., President
Attest:	
Cathy Bonaskiewich Secretary	

## **EXHIBIT 1**

# PENNSYLVANIA MUNICIPAL RETIREMENT SYSTEM BASE PLAN DOCUMENT

BPD201707-F Base Plan Document

## **TABLE OF CONTENTS**

	Page
ARTICLE I: DEFINITIONS	1
ARTICLE II: MEMBERSHIP	15
ARTICLE III: MUNICIPALITY CONTRIBUTIONS	17
ARTICLE IV: MEMBER CONTRIBUTIONS	18
ARTICLE V: SUPERANNUATION RETIREMENT PENSION	21
ARTICLE VI: EARLY RETIREMENT PENSION	24
ARTICLE VII: DISABILITY PENSION	26
ARTICLE VIII: DEFERRED VESTED PENSION	28
ARTICLE IX: DEATH BENEFITS	30
ARTICLE X: PAYMENT OF ACCRUED BENEFIT - OPTIONAL FORMS OF PAYMENT	32
ARTICLE XI: MISCELLANEOUS PROVISIONS AFFECTING THE CREDITING OF SERVICE	41
ARTICLE XII: MISCELLANEOUS PROVISIONS AFFECTING THE PAYMENT OF BENEFITS	49
ARTICLE XIII: ADMINISTRATIVE PROVISIONS	53
ARTICLE XIV: MEMBER ADMINISTRATIVE PROVISIONS	57
ARTICLE XV: PENNSYLVANIA MUNICIPAL RETIREMENT FUND	59
ARTICLE XVI: AMENDMENT	61
ARTICLE XVII: MISCELLANEOUS PROVISIONS	64
ADDENDUM A - APPLICABLE STATE LAWS	
ADDENDUM B - POLICY STATEMENTS	

#### Pennsylvania Municipal Retirement System Base Plan Document

Pursuant to Article IV of the Pennsylvania Municipal Retirement Law 53 P.S. § 881.101 <u>et. seq.</u>, ("PMRL"), (Sections of the PMRL referenced in this Base Plan Document are attached hereto in Addendum A) and Sections 104 and 401 thereof, the Pennsylvania Municipal Retirement Board has adopted this "Base Plan Document" which, together with the applicable executed "Adoption Agreement" shall constitute the "Plan" of a Municipality. Each such Plan is intended to be a tax qualified governmental plan under Sections 401, 414(d) and 501 of the Internal Revenue Code of 1986, as amended.

This Base Plan Document and its related Adoption Agreements are important legal documents under the Internal Revenue Code and Pennsylvania law. The Municipality is urged to consult with its legal counsel regarding the adoption of these instruments.

#### **ARTICLE I: DEFINITIONS**

All capitalized terms set forth shall have the meaning set forth below.

- **Accrued Benefit** means, subject to Plan termination provisions in Article XVI of the Plan, a Member's Superannuation Retirement Pension under Section 5.02 of the Plan. The Accrued Benefit shall include the value of the Member Contribution Account, if applicable, based on the subcategory of the type of contribution. For Cash Balance Plans the minimum annual Accrued Benefit is one half of one percent (0.5%) of Compensation.
- **Accumulated Deductions** means the total amount deducted from the salary or compensation of the Member and paid over by the Municipality or paid by the Member or from any existing pension or retirement system directly into the Fund and credited to the appropriate subaccount within the Member's Contribution Account, together with Regular Interest thereon.
- **1.03** Active Member means a Member who is accruing Credited Service under the Plan.
- **1.04** Actuarial Equivalence or Actuarial Equivalent means a benefit of equal value to the normal form of benefit (as defined in Article X of the Plan) whether in the form of an annuity, a lump sum or otherwise, based on the following:
  - (a) For Mortality:
    - (i) For Members under age 50:
      - (A) The RP-2000 Male Non-Annuitant and the RP-2000 Female Non-Annuitant Mortality Tables (as published by the Society of Actuaries) are used with a blend of seventy percent (70%) male and thirty percent (30%) female rates
        - 1. For male Members, the mortality table is projected 15 years with Scale AA mortality improvements
        - 2. For female Members, the mortality table is projected 15 years with Scale AA mortality improvements and also setback 5 years
    - (ii) For Members 50 years of age and older:
      - (A) The RP-2000 Male Annuitant and the RP-2000 Female Annuitant Mortality Tables (as published by the Society of Actuaries) are used with a blend of seventy percent (70%) male and thirty percent (30%) female rates
        - 1. For male Members, the mortality table is projected 5 years with Scale AA mortality improvements
        - 2. For female Members, the mortality table is projected 10 years with Scale AA mortality improvements
  - (b) For an Interest Rate, the rate of Regular Interest as defined in Section 1.77 of the Plan.
  - (c) With respect to benefits determined under a cash balance formula, (1) the Actual Equivalence shall be applied to the Member's Hypothetical Account Balance; and (2) lump sum payment and other benefits payable in a form that would be subject to the minimum present value requirements of Code Section 417(e)(3) (if applicable) will not be determined using the applicable mortality table and applicable latest rules under Code Section 417(e), this produces a benefit greater than the benefit determined using the Actual Equivalence assumption specified in this Section.
- **1.05** Actuarially Sound means a plan which is being operated under supervision of an Actuary and which is being funded annually at a level not lower than the normal cost of the plan plus a

contribution towards the unfunded accrued liability sufficient to complete the funding thereof within thirty years of the effective date of the system. If the unfunded accrued liability is increased subsequent to the effective date of the system, such additional liability shall be funded within a period not to exceed thirty years from the effective date of the increase. If deemed advisable by the actuary, the initial liability and any increase thereof, may be combined and amortized over a period of years, not to exceed thirty.

#### **1.06** Actuary means:

- (a) Member of the American Academy of Actuaries, and
- (b) an individual who has demonstrated to the satisfaction of the Insurance Commissioner of Pennsylvania that he had the educational background necessary for the practice of actuarial science and has had at least seven years of actuarial experience, or
- (c) a firm, partnership, or corporation of which one or more Members meets the requirements of subclauses (a) and (b) above.
- Adoption Agreement means the document executed by each Municipality, and approved by the Board, electing to join the System and setting forth its applicable terms. The terms of this Plan and Trust as modified by the terms of an adopting Municipality's Adoption Agreement shall constitute a separate Plan and Trust to be construed as a single Plan. Each elective provision of the Adoption Agreement shall correspond by section reference to the section of the Base Plan Document that grants the election.
- **1.08** <u>Alternate Payee</u> means a spouse, former spouse, child or dependent of a Member, who is recognized by an Approved Domestic Relations Order as having a right to receive all or a portion of the money payable to the Member under the Plan.
- **1.09 Annuitant** means a Member during the time period:
  - (a) beginning with the effective date of the Member's Retirement; and
  - (b) ending on the date of termination of the Member's annuity.
- **1.10** Approved Domestic Relations Order means a Domestic Relations Order which has been reviewed and approved by the Board.
- 1.11 <u>Average Annual Compensation</u> means the arithmetic annual average of Compensation paid to a Member by the Municipality for a specified number of consecutive years of Credited Service immediately preceding the Member's most recent Termination of Employment. The Municipality, in its Adoption Agreement, shall determine the years of Credited Service for use in the calculation and the maximum number of years of Credited Service to consider.

If a Member does not have a sufficient number of consecutive years of Credited Service as specified by the Municipality in its Adoption Agreement, the Average Annual Compensation shall be based on such actual lesser number of consecutive years of Credited Service.

If a Member (a) Terminates Employment and is later reemployed; or (b) has an unpaid Leave of Absence, the consecutive years of Credited Service prior to such Termination or Leave of Absence combined with the consecutive years of Credited Service after such Termination or Leave of Absence shall be considered consecutive for the purposes of this Section.

1.12 <u>Average Monthly Compensation and Final Monthly Compensation</u> means the arithmetic monthly average of Compensation paid to a Member by the Municipality for a specified number of consecutive months for purposes of determining the Member's Final Average Salary. The Municipality, in its Adoption Agreement, shall determine the maximum number of months of

Credited Service for use in the calculation and the specific months of Credited Service to consider (either the months producing the highest average or the final months immediately preceding Termination of Employment). If any partial months of Credited Service are within the maximum number of months of Credited Service considered, the partial months shall be included without adjustment.

If a Member does not have a sufficient number of consecutive months of Credited Service as specified by the Municipality in its Adoption Agreement, the Average Monthly Compensation or Final Monthly Compensation shall be based on such actual lesser number of consecutive months of Credited Service.

If a Member (a) Terminates Employment and is later reemployed; or (b) has an unpaid Leave of Absence, the consecutive months of Credited Service prior to such Termination or Leave of Absence combined with the consecutive months of Credited Service after such Termination or Leave of Absence shall be considered consecutive for the purposes of this Section.

- **1.13 Base Plan Document** means this document which sets forth the mandatory provisions of a Municipality's Plan.
- **1.14 Basic Benefit** means the pension benefit determined pursuant to the formula set forth in Section 5.02(a)(i) of the Plan.
- **1.15 Beneficiary** means a person last designated by a Member to receive benefits after the death of such Member
- **Benefit Commencement Date** means, with respect to a Member, Survivor Annuitant, or Beneficiary the first day of the first period for which payment of the benefit under the Plan is scheduled to commence, either as a result of the Member's written election or by operation of the Plan, whichever is later.
- 1.17 <u>Board</u> means the Pennsylvania Municipal Retirement Board as established pursuant to Section 103 of the PMRL. See Addendum A. The Board may act in the name of the System and shall designate, as appropriate, its Secretary to act on its behalf.
- 1.18 Break in Service means,
  - (a) with respect to an Employee:
    - (i) under the Hours of Service Method, any consecutive twelve (12) month period during which such Employee fails to complete at least fifty percent (50%) of the required number of hours designated in Section 1.40 of the Adoption Agreement for purpose of the Hours of Service Method. The consecutive twelve (12) month period shall be measured from the Employment Commencement Date or Reemployment Commencement Date and each anniversary thereof; and
    - (ii) under the Elapsed Time Method, a Period of Severance of twelve (12) consecutive months.
  - (b) For a Leave of Absence, including Military Leave under USERRA and FMLA Leave under the Family and Medical Leave Act of 1993, a Break in Service shall not be deemed to have occurred if the Employee returns to the Service of the Municipality following the Leave of Absence within the time required by the earlier of federal or state law.

- (c) For a Maternity or Paternity Leave, for purposes of determining when the measuring period for a Break in Service begins, the Employee shall receive additional Eligibility and Vesting Service as follows:
  - (i) Under the Hours of Service Method, an Employee shall be credited with the greater of:
    - (A) the number of Hours of Service actually worked, or
    - (B) the number of Hours of Service he normally would have been credited but for the Maternity or Paternity Leave.

Provided, however, the maximum number of Hours of Service credited for purposes of this Section shall not exceed fifty percent (50%) of the required number of hours under designated in Section 1.40 of the Adoption Agreement for purpose of the Hours of Service Method. Hours of Service credited shall be applied only to the Plan Year in which the Maternity or Paternity Leave begins unless such Hours of Service are not required to prevent the Employee from incurring a Break in Service, in which event such Hours of Service shall be credited to the Employee in the immediately following year.

- (ii) Under the Elapsed Time Method, the Termination of Employment Date of an Employee who is absent from employment beyond the first anniversary of his first date of absence is the second anniversary of the first date of absence. The period between the first and second anniversaries is neither a Period of Service nor a Period of Severance.
- (iii) No Service shall be credited due to Maternity or Paternity Leave as described in this Section unless the Employee furnishes proof satisfactory to the Municipality:
  - (A) that his absence from work was due to a Maternity or Paternity Leave, and
  - (B) of the number of days he was absent due to the Maternity or Paternity Leave.
- (iv) No Service shall be credited due to Maternity and Paternity Leave in excess of the amount of time the Employee was actually absent due to such Maternity and Paternity Leave.
- (v) An Employee shall not earn Credited Service while on Maternity or Paternity Leave.

The Municipality shall prescribe uniform and nondiscriminatory procedures by which to make the determinations required in this Section.

- 1.19 <u>Cash Balance Contribution Credit</u> means the sum of (i) required Municipal contributions under Article III; (ii) required Member contributions under Article IV; and (iii) Optional After-Tax contributions (if permitted) under Section 4.02(b).
- **1.20** <u>Cash Balance Interest Credit</u> means the amount credited to the Member's Hypothetical Account Balance as provided in Section 5.06.
- **1.21** Cash Balance Interest Crediting Rate means the Regular Interest rate and in no event shall not exceed six percent (6%) annually.
- 1.22 Code means the Internal Revenue Code of 1986 (Public Law 99-514, 26 U.S.C. § 1 et seq.).

- **1.23** <u>Compensation</u> means remuneration actually received for services rendered as a Municipal Employee, Municipal Firefighter or Municipal Police Officer:
  - (a) The following items are excluded from compensation:
    - (i) Reimbursement for expenses incidental to employment;
    - (ii) Income derived directly or by formula from the collection of fees or taxes;
    - (iii) Severance payments; and
    - (iv) Payments related to services rendered during a Period of Service in which an individual is an Ineligible Employee or an Ineligible Member.
  - (b) The following additional rules apply:
    - (i) Compensation shall be adjusted as appropriate pursuant to the terms of the Adoption Agreement.
    - (ii) For Members who are enrolled in a Plan that has adopted the provisions of Code Section 414(h), the term includes a contribution designated as a pickup contribution.
    - (iii) Compensation shall include payments pursuant to the Enforcement Officers Disability Benefits Law, 53 P.S. § 637. See Addendum A.
    - (iv) Notwithstanding any provision of this Plan to the contrary, a Member's Compensation shall not exceed the limitations under Code Section 401(a)(17), as adjusted in accordance with Code Section 401(a)(17)(B).
      - (A) The adjustment in effect for a calendar year applies to a period:
        - 1. which begins in the calendar year;
        - 2. which does not exceed twelve months; and
        - 3. over which Compensation is determined.
      - (B) If a determination period consists of fewer than twelve months, the Compensation limit shall be multiplied by a fraction:
        - 1. the numerator of which is the number of months in the determination period; and
        - 2. the denominator of which is twelve.
- 1.24 <u>Credited Service</u> means the measurement of a Member's Service that is used to determine the Member's Accrued Benefit. Credited Service shall not include Service while the individual is an Ineligible Employee or an Inactive Member unless otherwise specifically provided for by the Municipality in its Adoption Agreement.
  - (a) Credited Service shall be determined, as specified in each Municipality's Adoption Agreement, by either the:
    - (i) Elapsed Time Method or
    - (ii) Hours of Service Method.
  - (b) Each Municipality shall elect in its Adoption Agreement to include or exclude:
    - (i) A maximum number of years of Credited Service that will be used in Pension benefit calculations,
    - (ii) Service prior to the Original Effective Date of the Plan,

- (iii) Transferred and/or additional Credited Service as provided in Article XI,
- (iv) Service rendered under a Prior Plan of the Municipality; or
- (v) Unused sick leave and/or annual leave as specifically set forth in the Adoption Agreement.
- **1.25** <u>Deferred Vested Pension</u> means, with respect to a Member, the benefit described in Article VIII of the Plan.
- **1.26** <u>Disability or Disabled</u> means, upon certification by the Board, a Member who is unable to engage in any gainful employment by reason of any medically determinable physical or mental impairment.
- **1.27** <u>**Disability Date**</u> means the date on which the Disability applicant became Disabled, as determined by the Board, provided however, that the Disability Date shall not be later than the Member's Termination of Employment.
- **1.28** <u>Disability Pension</u> means, with respect to a Member, the benefit described in Article VII of the Plan.
- 1.29 <u>Domestic Relations Order</u> means any judgment, decree or order, including approval of a property settlement agreement, entered into on or after the effective date of this definition by a court of competent jurisdiction pursuant to a domestic relations law which relates to the marital property rights of the spouse or former spouse of a Member, including the right to receive all or a portion of the money payable to that Member under the Plan, in furtherance of the equitable distribution of marital assets that satisfies the definition of a "qualified domestic relations order" under Code Section 414(p). The term includes an "order of support" under 23 Pa.C.S. § 4302 (relating to definitions) and an order for the enforcement of arrearages under 23 Pa.C.S. § 3703 (relating to enforcement of arrearages).
- **1.30** Early Retirement Date means the date the Member becomes eligible for an Early Retirement Pension, as specified the Municipality's Adoption Agreement.
- **1.31 Early Retirement Pension** means, with respect to a Member, the benefit described in Article VI of the Plan.
- **1.32** Effective Date of Disability Retirement means the later of (i) the date following the last day for which the Member is paid for services as an Employee; or (ii) the date on which the Member files an application for Disability Pension
- **Elapsed Time Method** means a method of computing Service by reference to the total time (years, months and days) that elapses between the Employee's Employment Commencement Date or Reemployment Commencement Date and the Employee's Termination of Employment. The total time need not be consecutive.

For the purpose of calculating Eligibility Service and Vesting Service, a Member shall accrue one day of Service for each day in which he is credited with one Hour of Service as an Employee of the Municipality and shall accrue one Year of Service for each three hundred and sixty-five (365) days. The calculations shall be subject to the Break in Service provisions of the Plan.

For the purpose of calculating Credited Service, a Member shall accrue one day of Service for each day in which he is credited with one Hour of Service as an Employee of the Municipality and shall

- accrue one Year of Service for each three hundred and sixty-five (365) days. The calculations shall not be subject to the Break in Service provisions of the Plan.
- **Eligibility Service** means the measurement of an Eligible Employee's Service for purposes of determining when the Employee is eligible for membership in the Plan and is measured from the Employee's Employment Commencement Date or Reemployment Commencement Date, as applicable, to the date the Employee satisfies the requirements in Article II of the Plan and the Municipality's Adoption Agreement. Eligibility Service shall not include the Period of Service for which the Employee was considered an Ineligible Employee.

Eligibility Service shall be determined, as specified in each Municipality's Adoption Agreement, by either the:

- (a) Elapsed Time Method or
- (b) Hours of Service Method.
- 1.35 <u>Employee</u> means an individual employed by the Municipality. Employees designated in the Municipality's Adoption Agreement as eligible to participate in the Plan shall be considered an "Eligible Employee." An Employee designated as ineligible to participate in the Plan or not otherwise identified in the Adoption Agreement as eligible to participate in the Plan shall be considered an "Ineligible Employee." Individuals paid by the Municipality wholly on a fee basis shall be considered "Ineligible Employees."
- **Employment Commencement Date** means the date on which an Eligible Employee first performs an Hour of Service for the Municipality.
- **Excess Interest** means additional assets which may result when investment earnings from the Fund exceed those necessary to cover the Regular Interest or all reserves for the covered Plans and the current market asset value exceeds the Reserve Accounts plus expenses. Excess Interest is determined and declared pursuant to the Board policies attached in Addendum B.
- **1.38 <u>Final Average Salary</u>** means the Average Monthly Compensation or the Average Annual Compensation or Final Monthly Compensation as elected by the Municipality in its Adoption Agreement.
- **1.39 Fund** means the Pennsylvania Municipal Retirement Fund created by the PMRL and consisting of all property of every kind invested and administered by the Board pursuant thereto.
- **1.40 Hour of Service** means the increments of time, described in sections (a), (b), (c), and (d) hereof (as applicable), subject to any limitations set forth herein:
  - (a) Each hour for which the Municipality, either directly or indirectly, pays an Employee, or for which the Employee is entitled to payment, for the performance of duties during the Plan Year. The Municipality shall credit Hours of Service under this paragraph (a) to the Employees for the Plan Year(s) in which the Employee performs the duties, irrespective of when paid;
  - (b) Each hour for back pay, irrespective of mitigation of damages, to which the Municipality has agreed or for which the Employee has received an award. The Municipality shall credit Hours of Service under this paragraph (b) to the Employee for the Plan Year(s) to which the award or the agreement pertains rather than for the Plan Year(s) in which the award, agreement or payment is made; and
  - (c) Each hour for which the Municipality, either directly or indirectly, pays an Employee, or for which the Employee is entitled to payment (irrespective of whether the employment

relationship is terminated), for reasons other than for the performance of duties during a Plan Year, such as leave of absence, vacation, holiday, sick leave, illness, incapacity (including disability), layoff, jury duty, or military duty, provided:

- (i) A Municipality shall not credit more than fifty percent (50%) of the required number of hours under the Hours of Service Method to an Employee on account of any single continuous period during which the Employee does not perform any duties as an Employee (whether or not such period occurs during a single Plan Year);
- (ii) An hour for which an Employee is directly or indirectly paid, or entitled to payment, on account of a period during which he performs no duties as an Employee shall not be credited as an Hour of Service if such payment is made or due under a plan maintained solely to comply with applicable workers' compensation, unemployment compensation, or disability insurance laws; and
- (iii) Hours of Service shall not be credited to an Employee for a payment that solely reimburses such Employee for medical or medically related expenses incurred by him.
- (d) Each hour for which the Employee is required to be granted leave under USERRA.
- (e) A Municipality shall not credit an Hour of Service under more than one (1) of the above paragraphs (a), (b), (c) or (d). If the Service counted under this Section can be counted under more than one of these paragraphs, the rule crediting the greatest number of Hours of Service shall apply. The Municipality shall resolve any ambiguity with respect to the crediting of an Hour of Service in favor of the Employee.
- (f) The Municipality shall credit Hours of Service under this Section in accordance with Department of Labor Regulation Section 2530.200b-2(b) and (c), 29 CFR Part 2530, as amended, which the Plan, by this reference, specifically incorporates in full, or such other federal regulations as may from time to time be applicable.
- Hours of Service Method means a method for computing Service by reference to the number of Hours of Service performed by the Employee in a consecutive twelve (12) month period or any permitted equivalency. Any computations or calculations using the Hours of Service Method shall: (a) award a year of Service for any consecutive twelve (12) month period in which a Member is credited with the number of Hours of Service as set forth in the Adoption Agreement; and (b) be subject to the Break in Service provisions.
- **1.42 Hypothetical Account Balance** means the balance of a hypothetical account established for the benefit of a Member and credited with Cash Balance Contribution Credits and Cash Balance Interest Credits.
- **1.43 Inactive Member** means a Member who has experienced a Severance from Credited Service but has not received full payment of his Vested benefit under the Plan. A Member on Leave of Absence shall not be deemed an Inactive Member.
- **1.44** <u>In-service Distribution</u> means a distribution of a Member's Accrued Benefit commencing after the Member's Normal Retirement Age and before the Member's Severance from Credited Service.
- **1.45 Joint Coverage Member** means a Member who shall have become a Member of the System subsequent to the last date permitted by the Municipality employing him for statement of preference concerning social security coverage, or who, having become a Member on or before such date, shall have filed with the Municipality a written statement that he elects social security coverage

under an agreement with the Federal Secretary of Health, Education and Welfare entered into by the Commonwealth of Pennsylvania.

**1.46** <u>Leave of Absence</u> means a paid or unpaid excused leave of absence granted to an Employee in accordance with applicable federal or state law or the Municipality's personnel policies. Leave of Absence shall include the following:

#### (a) Military Leave

Employees who leave the service of a Municipality, voluntarily or involuntarily, to enter the Armed Forces of the United States and the Employee is legally entitled to certain rights under USERRA.

## (b) FMLA Leave

Employees who leave the service of the Municipality under the provisions of the Family and Medical Leave Act of 1993 ("FMLA") provided that the Employee returns to active employment within the time required under the FMLA.

### (c) Other Leave

Employees who leave the service of the Municipality under such other circumstances as approved by the Municipality and applied uniformly among Employees under similar circumstances

- **1.47** Limitation Year means the calendar year.
- **1.48 Look Back Month** means the December 1 immediately preceding a Stability Period.
- 1.49 <u>Maternity or Paternity Leave</u> means any period during which an Employee is absent from work with a Municipality by reason of (a) pregnancy of such Employee, (b) the birth of a child of such Employee, (c) the placement of a child with such Employee in connection with the adoption of a child by such Employee, or (d) for purposes of such Employee caring for such child immediately after such birth or placement.
- **1.50 Member** means an individual that is:
  - (a) a Municipal Officer, Employee, Firefighter or Police Officer; or
  - (b) an Employee of a municipal government association, who satisfies the conditions for membership as set forth in Article II.
- 1.51 <u>Member Contribution Account</u> means the notional account and subaccounts thereunder established by the Board to reflect Accumulated Deductions and other contributions made by the Member to the Trust, if any, plus Regular Interest credited thereon as required under the Plan. In addition to any other accounts the Board shall establish, the Board shall establish the following subaccounts under the Member Contribution Account for each Member to be designated as follows:
  - (a) "Pick-Up Contribution Account" shall reflect a Member's interest in the Municipality pick-up contributions made under Section 4.01 of the Plan and shall include any payments made by the Member under Section 4.08 of the Plan to repay such account.
  - (b) "Required After Tax Contribution Account" shall reflect a Member's interest in required after-tax contributions made by the Member under Section 4.02(a) of the Plan and shall include any payments made by the Member under Section 4.08 of the Plan to repay such account.
  - (c) "Optional After-Tax Contribution Account" shall reflect a Member's interest in elective after-tax contributions made by the Member under Section 4.02(b) of the Plan and shall

- include any payments made by the Member under Section 4.08 of the Plan to repay such account.
- (d) "Service Purchase Account" shall reflect a Member's interest in payments made by the Member under Section 4.03(a) of the Plan for the purpose of purchasing eligible Credited Service under the Plan.
- (e) "Transferred Contribution Account" shall reflect a Member's interest in amounts transferred by the Municipality under Section 4.03(b) of the Plan, from one or more of the Municipality's tax-qualified plans, that have been merged or combined with the Plan, directly to the Fund.
- (f) "Terminated Contribution Account" shall reflect a Member's interest in amounts transferred, at the election of the Member under Section 4.03(c) of the Plan, from one or more of the Municipality's previously terminated Prior Plans directly to the Fund.
- 1.52 <u>Member's Excess Investment Account</u> means the account maintained for each Member, to which shall be credited such Excess Interest deemed to be earned on the Member Contribution Account. For purposes of any distribution of the Member Contribution Account, said distribution shall include the Member's Excess Investment Account.
- **1.53** <u>Membership Commencement Date</u> means the date a Member first commences participation under the Plan.
- **Municipal Contribution Account** means, in the aggregate, the account and subaccounts thereunder established by the Board to reflect contributions made by the Municipality to the Trust, plus Regular Interest deemed to be earned by the Board thereon as required under the Plan. In addition to any other accounts the Board shall establish, the Board shall establish separate accounts for each Municipality to be designated as follows:
  - (a) "Defined Benefit Contribution Account" shall reflect the Municipality's required contributions and any additional contributions made by the Municipality under Sections 3.01 and 3.02 of the Plan.
  - (b) "Cash Balance Contribution Account" shall reflect the Municipality's contributions required Section 3.02 of the Plan.
- 1.55 <u>Municipal Employee</u> means a person holding an office or position, other than that of a Municipal Firefighter or Municipal Police Officer, under a Municipality or a municipal government association and paid on a regular salary or per diem basis. The term shall not include officers and employees paid wholly on a fee basis.
- **1.56** <u>Municipal Firefighter</u> means a person holding a full-time position in the fire department of a Municipality and who works for a stated salary or Compensation.
- **1.57** <u>Municipal Police Officer</u> means a person holding a full-time position in the police department of a Municipality and who works for a stated salary or Compensation.
- 1.58 <u>Municipality</u> means a city, borough, town, township, county, institution district, or any newly created governmental unit, or an authority created by a city, borough, town, township, county or county institution district, or jointly by any such political subdivisions, or an institution supported and maintained by a municipality or a municipal government association, or an industrial development agency which has adopted bylaws and the governing body of which is organized and holds regular public meetings.

- **New Member** means a municipal officer, employee, firefighter or police officer, or an employee of a municipal government association who first becomes a Member after the date the Municipality by which he is employed establishes a Plan.
- 1.60 Nonqualified Service Credit means Permissive Service Credit other than service rendered as an employee of the Government of the United States, any State or political subdivision thereof, any agency of the foregoing, service as an employee of an educational organization described in Code Section 170(b)(1)(A)(ii), or military service (other than qualified military service under Code Section 414(u).
- 1.61 Normal Retirement Age means such age as may be stipulated by the Municipality in the Adoption Agreement at which a Member is eligible to commence receipt of his Superannuation Retirement Pension under Section 5.01 without actuarial or other reduction for early commencement. The above notwithstanding, Normal Retirement Age shall be interpreted consistent with the requirements of "normal retirement age" as set forth in Treasury Regulation § 1.401(a)-1(b).
- **Original Member** means a Municipal Officer, Employee, Firefighter or Police Officer, or an employee of a municipal government association who was employed by the Municipality and becomes a Member in the Plan the date the Municipality establishes a Plan.
- 1.63 Original Plan means the tax-qualified plan of a Municipality that is (i) merged into this Plan; or (ii) administered by the Municipality or a third party other than Board, and is restated and superseded by this Plan document (provided that the corresponding assets of such plan are transferred into this Plan). If there is no such tax-qualified plan, then this Plan shall be the Original Plan of the Municipality.
- 1.64 Original Plan Effective Date shall mean the effective date of the Original Plan of the Municipality.
- **1.65** Original Plan Service means all service an Eligible Employee completed at the time the Municipality by which he is or was employed elected to join the System or the same Municipality under a prior name or classification elected to join the System.
- **Period of Service** means for purposes of the Elapsed Time Method of counting Service, the Employee's period of employment with the Municipality commencing with the Employment Commencement Date or the Reemployment Commencement Date, whichever is applicable, and ending on the Employee's Termination of Employment date.
- **1.67 Period of Severance** means, for purposes of the Elapsed Time Method of counting Service, a continuous period of time during which the Employee is not employed by the Municipality, commencing on the Employee's Termination of Employment date and ending on the Employee's Reemployment Commencement Date.
- 1.68 Permissive Service Credit means certain Credited Service that the Member may receive only by making a voluntary contribution which does not exceed the amount necessary to fund the benefit attributable to such service credit. Such service shall include service rendered as an employee of the Government of the United States, any State or political subdivision thereof, any agency of the foregoing, service as an employee of an educational organization described in Code Section 170(b)(1)(A)(ii), or military service (other than qualified military service under Code Section 414(u).
- **Plan** means the plan as set forth in this document, including the executed Adoption Agreement through which a Municipality establishes a Plan. A Municipality shall designate the name of the Plan in the Adoption Agreement. The Board shall administer and maintain the Plan of each

- adopting Municipality as a separate Plan and independent from the plan of any other Municipality, and any other plan offered by the adopting Municipality.
- **1.70 Plan Entry Date** means the date designated in the Municipality's Adoption Agreement as the Employee's Membership Commencement Date after satisfying the Plan eligibility requirements as specified in Article II of the Plan and the Municipality's Adoption Agreement.
- **1.71 Plan Year** means the twelve (12) month period specified in the Adoption Agreement.
- **1.72 PMRS Plan Effective Date** shall mean the date the Municipality establishes a plan with the Board pursuant to Sections 107 or 402 of the PMRL. See Addendum A.
- 1.73 Policy Statement means a statement of rule issued by the Board pursuant to its general powers under Section 104(10) of the PMRL. See Addendum A. In the event of a conflict between operative provisions of the Base Plan and a Policy Statement, the terms of the Base Plan shall control. All Policy Statements are incorporated by reference and attached at Addendum B.
- **1.74 Prior Plan** means a tax-qualified plan established by a Municipality that has been terminated or frozen for future benefit accruals.
- **1.75 Prior Plan Service** means all service recognized under a Prior Plan of the Municipality prior to the Original Plan.
- **1.76** Reemployment Commencement Date means the first date on which the Eligible Employee performs an Hour of Service that is required to be taken into account for Eligibility, Vesting or Credited Service, following a Break in Service or Period of Severance.
- **Regular Interest** means the rate fixed by the Board (from time to time, and in effect on the Look Back Month), to be applied to the Member's notional accounts, to the Municipal Accounts and to the Retired Member's Reserve Account. Effective January 1, 2017, the Regular Interest rate is five and one-quarter percent (5.25%).
- **1.78** Retire or Retirement means Termination of Employment with the Municipality or Severance from Credited Service on or after (i) the Member's Early Retirement Date or Superannuation Retirement Date or Disability Date and (ii) the Member has filed an application to retire.
- **1.79** Reserve Account means separate accounts and subaccounts established for the aggregation of monies held in the Trust and to be used for specified purposes. The Board shall establish a minimum number of such accounts to be designated as follows:

# (a) Disability Reserve Account

This account is the aggregation of monies collected from Municipalities that offer a Disability Pension and is used to supplement monies transferred from one or more Municipality Accounts and one or more Member Contribution Accounts if such transferred monies are insufficient to equal the Actuarial Equivalence of such Member's Disability Pension through a transfer to the Retired Member's Reserve Account.

## (b) Member Reserve Account

This account is the aggregation of each Member's Contribution Accounts and subaccounts and such monies are held in Trust for each Member. The monies are used to provide for the applicable Pension benefits provided to Members by their respective Municipality through transfers to the Retired Member's Reserve Account.

# (c) Municipal Reserve Account

This account is the aggregation of each Municipality's accounts and subaccounts and such monies are held in Trust for each Municipality. The monies are used to provide for the applicable Pension benefits provided to Members by each such applicable Municipality through transfers to the Retired Member's Reserve Account.

## (d) Retired Members' Reserve Account

This account is the aggregation of monies transferred from the other Reserve Accounts and held in Trust for each Retiree. The monies are used to provide for the applicable Pension benefits provided to Members by each such applicable Municipality.

- 1.80 <u>Service</u> means any period of time the Employee is in the employ of the Municipality, including any period the Employee is on a Leave of Absence authorized by the Municipality up to one (1) year, unless a longer period is required by law to be counted as Service. Notwithstanding any provision of this Plan to the contrary, contributions, benefits, Vesting, Eligibility and Credited Service with respect to qualified military service will be provided in accordance with USERRA and Code Section 414(u) and with respect to FMLA Leave will be provided in accordance with the Family and Medical Leave Act of 1993.
- 1.81 Service Connected Disability means total and permanent disability of a Member prior to eligibility for superannuation retirement resulting from a condition arising out of and incurred in the course of his employment, and which is compensable under the applicable provisions of the act of June 2, 1915 (P.L. 736, No. 338), known as "The Pennsylvania Workmen's Compensation Act," or the act of June 21, 1939 (P.L. 566, No. 284), known as "The Pennsylvania Occupational Disease Act." See Addendum A.
- **1.82** Service Increment Benefit means a supplemental benefit payable to a Member who has twenty-six (26) or more years of Credited Service as provided in the Municipality's Adoption Agreement.
- **1.83** Severance from Credited Service means an Employee's Termination of Employment or change in employment status (via election, appointment or otherwise) such that he is no longer an Eligible Employee under the Plan.
- 1.84 <u>Single Coverage Member</u> means a Member who shall become a Member of the System on or before the last date permitted by the municipality employing him for statement of preference concerning social security coverage and who either shall have filed with the Municipality a written statement that he does not elect social security coverage under any agreement with the Federal Secretary of Health, Education and Welfare entered into by the Commonwealth of Pennsylvania, or shall not have filed with the municipality any written statement.
- **Spouse or Surviving Spouse** means the person to whom the Member is legally married for Federal tax purposes.
- **1.86** Stability Period means the calendar year during which Regular Interest will be in effect.
- **1.87** Superannuation Retirement Date means the date of the Member's Retirement after becoming eligible for a Superannuation Retirement Pension as specified in Section 5.01 of the Municipality's Adoption Agreement.
- **1.88** Superannuation Retirement Pension means, with respect to a Member, the benefit described in Section 5.02 of the Plan.
- **1.89 Survivor Annuitant** means any person who has been named by a Member under a joint and survivor annuity option to receive an annuity upon the death of such Member. The term shall also

- include a spouse, or child(ren) if applicable, for plans that have elected a life and 50% spouse benefit or life and 100% spouse benefit as specified in Section 10.01 of the Base Plan Document.
- 1.90 System means the Pennsylvania Municipal Retirement System, an agent multiple employer retirement system for purposes of Government Accounting Standards Board requirements created pursuant to the PMRL. The above notwithstanding, this Base Plan document and accompanying Adoption Agreements are approved by the Internal Revenue Service pursuant to Rev. Proc. 2015-36 solely for the purposes of establishing and maintaining an individual Plan by an adopting Municipality.
- **Termination Contribution** means an amount transferred to this Plan pursuant to an Eligible Rollover Distribution (as defined in Article X) of a Member's account under a Prior Plan of the Municipality or the predecessor to the Municipality.
- 1.92 <u>Termination of Employment, Terminate Employment, Termination, or Terminated</u> means a termination of employment with the Municipality, including Retirement, resignation, discharge, and death except as otherwise provided by the Municipality as a Leave of Absence or any other leave of absence regulated by federal or state law. Unless otherwise provided by law, if an Employee on a Leave of Absence fails to return to active employment upon expiration of the Leave of Absence, the Employee will be considered Terminated as of the last day worked immediately preceding the Leave of Absence.
- **1.93** Transfer Contribution means an amount transferred to this Plan directly from a Member's account under an Eligible Retirement Plan (as defined in Article X) sponsored by the Municipality or the predecessor to the Municipality.
- 1.94 <u>Trust</u> means the agreement setting forth the terms and conditions with respect to the administration, investment and distribution of the Fund by the Board.
- **1.95 USERRA** means the Uniform Services Employment and Reemployment Rights Act of 1994.
- **1.96** <u>Vested</u> means that portion of a Member's interests in this Plan that is vested pursuant to Article XIII or Section 16.06, if applicable.
- 1.97 <u>Vesting Service</u> means the measurement of a Member's Service that is used to determine the Member's Vested Accrued Benefit, and if the Member meets Service related requirements for all other benefits provided in the Plan. Vesting Service shall be measured from the Member's Employment Commencement Date or Reemployment Commencement Date and each anniversary thereof. Vesting Service shall not include the Period of Service for which the Employee was considered an Ineligible Employee.
  - (a) Service shall be determined, as specified in each Municipality's Adoption Agreement, by either the:
    - (i) Elapsed Time Method or
    - (ii) Hours of Service Method.
  - (b) A Municipality shall elect in its Adoption Agreement to include or exclude:
    - (i) Service before the Original Plan Effective Date, or
    - (ii) Service before the PMRS Restatement Plan Date, or
    - (iii) unused sick leave and/or unused annual leave in accordance with the Municipality's personnel policies.

#### ARTICLE II: MEMBERSHIP

# 2.01 Initial Membership

## (a) Mandatory Membership

Each Eligible Employee in a category selected for mandatory membership under Section 1.35 of the Adoption Agreement shall become a Member in the Plan on the Plan Entry Date (if actively employed on that date) coincident with or immediately following the date on which he meets the eligibility conditions selected by the Municipality in its Adoption Agreement.

# (b) Optional Membership

- (i) Each Eligible Employee in a category selected for optional membership under Section 1.35 of the Adoption Agreement must elect to become a Member by filing a written election with the Municipality within ninety (90) days of notification by the Municipality of membership eligibility. Such Eligible Employee shall become a Member effective on the date of filing of the written election and membership shall be prospective from such date. The election, or failure to elect membership shall be irrevocable for the duration of the Eligible Employee's service in the category of optional membership.
- (ii) The above notwithstanding, if the Plan does not require pre-tax pick-up contributions pursuant to Code Section 414(h), then an Eligible Employee for optional membership may elect to become a Member by filing a written election with the Municipality within one year of notification by the Municipality.

# (c) <u>Ineligible Employee</u>

An Ineligible Employee identified in the Municipality's Adoption Agreement shall not be eligible to become a Member in the Plan.

## 2.02 Membership Upon Reemployment

Upon an Employee's Reemployment Commencement Date, the Eligible Employee shall have the following status in the Plan based on his status as of his most recent Termination of Employment:

- (a) If the Eligible Employee was a Member, he shall reenter the Plan as a Member on his Reemployment Commencement Date.
- (b) If the Eligible Employee had satisfied the Plan's eligibility conditions but had not become a Member, he shall become a Member on the next Plan Entry Date after his Reemployment Commencement Date
- (c) If the Eligible Employee had not satisfied the Plan's eligibility conditions, he shall receive all previous Eligibility Service and shall become a Member on the Plan Entry Date coincident with or immediately following his satisfying the Plan's eligibility conditions.
- (d) The above notwithstanding, the Eligible Employee's position upon reemployment (either optional membership or mandatory membership) shall determine the conditions of membership upon such reemployment.

# 2.03 Membership Upon Change in Employment Status

If an Eligible Employee has a change in employment status, the Period of Service when the Employee was an Ineligible Employee shall be not included in determining Eligibility Service.

- (a) If the Employee had not previously satisfied the Plan's eligibility conditions, he shall become a Member on the Plan Entry Date coincident with or immediately following his satisfying the Plan's eligibility conditions as an Eligible Employee.
- (b) If, during one or more previous periods of Service in which he was an Eligible Employee, the Employee had previously satisfied the Plan's eligibility conditions but had not become a Member, he shall become a Member on the next Plan Entry Date coincident with or immediately after his most recent change in employment status.
- (c) If, during one or more previous periods of Service in which he was an Employee, the Employee was a Member, he shall reenter the Plan as a Member on date of his change in employment status.

Under no circumstances shall a person who is not an Eligible Employee, as defined by the Adoption Agreement, become a Member unless he is an Eligible Employee on the Plan Entry Date.

#### ARTICLE III: MUNICIPALITY CONTRIBUTIONS

# 3.01 Determination of Contribution

The Board will actuarially determine the normal cost of the benefits provided for in a Municipality's Plan and any amortization of unfunded liability associated with the actuarial experience of such benefits which shall be contributed annually by the Municipality. Additionally, the Board shall determine and the Municipality shall be charged an additional amount to be contributed annually toward a reserve account for any disability benefit if disability benefits are elected by the Municipality under the terms of its Adoption Agreement.

## 3.02 Amount

Unless otherwise provided in Article IV of its Adoption Agreement, the Municipality shall make the contributions required to fund the Municipality's cost of the benefits provided to its Members as set forth in Section 5.02 of the Plan.

# 3.03 Time of Payment of Contribution

## (a) Defined Benefit Plan

Unless otherwise provided in the Municipality's Adoption Agreement, the Municipality shall pay its contribution for each Plan Year in one (1) annual payment. The Municipality must make its contribution to the Board no later than December 31 of the year for which the contribution was required.

## (b) Cash Balance Plan

Unless otherwise provided in the Municipality's Adoption Agreement, the Municipality shall pay its contributions for each Plan Year in the same manner and frequency as it remits Member contributions in accordance with Section 4.04 of the Plan and Section 4.02 of the Municipality's Adoption Agreement. The above notwithstanding, the Municipality must complete its aggregate requirement contributions to the Board no later than December 31 of the year for which the contributions were required.

#### **ARTICLE IV: MEMBER CONTRIBUTIONS**

# 4.01 Pre-Tax Pick-Up Contributions

If the Municipality elects the application of this Section in its Adoption Agreement, it shall contribute to the Plan, to the credit of each Member, the amount of the required Member contribution (as a percentage of Compensation as specified in the Adoption Agreement) to fund the Member's cost of the benefit provided to the Member as set forth in Section 5.02(a) or (b) as applicable. The contributions are mandatory and no Member shall be entitled under any circumstances to receive such contributions in cash in lieu of having them contributed to the Trust by the Municipality in accordance with the preceding sentence. Such contributions shall be made pursuant to Code Section 414(h) and shall be treated as Municipality contributions in determining their federal income tax treatment under the Code.

## 4.02 After-Tax Member Contributions

#### (a) Required After-Tax Contributions

If the Municipality elects the application of this Section in its Adoption Agreement, it shall require a Member to make after-tax contributions by payroll deduction to the Trust. The Board shall allocate and credit such contribution made for a particular Plan Year to the Member's Required After-Tax Contribution Account as soon as administratively possible. The Board may establish appropriate procedures it deems necessary to facilitate Member payroll deduction contributions. Required after-tax contributions shall be allocated to the cost of the Member's benefit as set forth in Section 5.02(a) or (b) as applicable.

## (b) Optional After-Tax Contributions

If the Municipality elects the application of this Section in its Adoption Agreement, it shall allow a Member to elect after-tax contributions by payroll deduction to the Trust. The Board shall allocate and credit such contribution made for a particular Plan Year to the Member's Optional After-Tax Contribution Account as soon as administratively possible. The Board may establish appropriate procedures it deems necessary to facilitate Member payroll deduction contributions and the Member's elected use of such contributions. Optional after-tax contributions may be (i) utilized for the purchase of Permissive Service Credit as provided in Section 11.07; or (ii) allocated to the Member's Hypothetical Account Balance pursuant to Section 5.06.

#### 4.03 Other Member Contributions

# (a) Service Purchase Contributions

If the Municipality elects in its Adoption Agreement to allow Members to purchase transferred, prior or additional service in accordance with Article XI of the Plan, the Member may make contributions to purchase such service. For the purchase of such service, the Plan may accept payment directly from the Member or through a plan-to-plan transfer or rollover contribution from any other allowable qualified retirement plan, Section 401(k) plan, Section 457(b) governmental deferred compensation plan or a Section 403(b) tax sheltered annuity. Any Service Purchase Contributions received from the sources specified in this subsection and used for service purchases shall be allocated to the Member Service Purchase Account.

#### (b) Transferred Contributions

If the Municipality elects in its Adoption Agreement to transfer funds from one or more of the Municipality's Prior Plans pursuant to a merger of such Prior Plan with this Plan, such funds may be: (i) utilized for the purchase of Permissive Service Credit as provided in Section 11.07; or (ii) allocated to a Member's Hypothetical Account Balance pursuant to

Section 5.06. Any such Transferred Contributions shall be allocated to the Member's Transferred Contribution Account.

# (c) <u>Terminated Contributions</u>

If the Municipality elects in its Adoption Agreement to permit Members to transfer funds from one or more of the Municipality's Prior Plans where such plans have been previously terminated by the Municipality, such funds may be: (i) utilized for the purchase of Permissive Service Credit as provided in Section 11.07; or (ii) allocated to a Member's Hypothetical Account Balance pursuant to Section 5.06. Any Terminated Contributions shall be allocated to the Member's Terminated Contribution Account.

# 4.04 <u>Time and Manner of Contribution</u>

Contribution deductions from Member Compensation as specified in Sections 4.01 and 4.02 shall be made on a payroll basis and unless otherwise elected in the Adoption Agreement, the Municipality shall remit such contributions as of each payroll period as soon as administratively possible. Other Member contributions identified in Section 4.03 shall be remitted to the System as soon as administratively possible by the Municipality or representative acting on its behalf and, if applicable, by the Member or representative acting on its behalf.

## 4.05 <u>Allocation of Contributions</u>

The Board shall establish and maintain the Member Contribution Accounts and subaccounts to which all Member contributions shall be allocated. Each Member Contribution Account shall be credited with contributions, rollovers, transfers and earnings allocated to such Account if provided under the Adoption Agreement and debited with distributions made during the Plan Year.

# 4.06 <u>Interest on Member Contribution Account</u>

The Member Contribution Account described in this Article of the Plan shall be credited with Regular Interest.

# 4.07 Refund of Member Contribution Account

A Member or Beneficiary shall receive a refund or withdrawal of his Member Contribution Account if:

- (a) the Member Terminates Employment and, at the time of such Termination, does not have sufficient vesting Service to qualify for a vested Accrued Benefit in accordance with the vesting schedule specified in Section 8.05 of the Adoption Agreement, or
- (b) the Member or Beneficiary is receiving benefits under the Plan and dies before receiving Pension benefit payments in an amount equal to or greater than the Member Contribution Account, and no additional Pension benefits are due, then the Beneficiary (or estate, if no Beneficiary) shall receive the amount remaining in the Member Contribution Account, or
- (c) the Member Terminates Employment and, at the time of such Termination, requests the refund of his Member Contribution Account in lieu of retaining an Accrued Benefit, or
- (d) the Member dies before receiving any benefits under the Plan and the value of the death benefit payable to the Beneficiary is equal to or less than the Member Contribution Account, then the Beneficiary (or estate, if no Beneficiary) shall receive the amount remaining in the Member Contribution Account and no additional death benefits will be paid, or
- (e) an individual has been determined to be an Ineligible Employee and is ineligible to be a Member in the Plan.

Distribution of the Member Contribution Account shall be made only in a lump sum and for no less than 100% of the Member Contribution Account including interest earnings. Upon distribution of the Member Contribution Account, the Member or Beneficiary shall have no Accrued Benefit under the Plan.

## 4.08 Repayment of Member Contribution Account

#### (a) Restoration of Service

A Member who is reemployed with the Municipality after receiving a refund of amounts allocated to his Member Contribution Account:

- (i) shall have his Eligibility and Vesting Service restored in accordance with Article XI of the Plan, and
- (ii) shall have his Credited Service and any previous Accrued Benefits restored by repaying the Board the entire amount distributed under Section 4.07 plus interest at the Regular Interest rate. Interest shall begin on the first day of the month following the month of the previously refunded Member Contribution Account and shall end on the last day of the month preceding such repayment.

# (b) Repayment Requirements

Unless otherwise provided in the Municipality's Adoption Agreement, the Member shall not:

- (i) repay the Member Contribution Account balance and interest later than ninety (90) days after the Member's Reemployment Commencement Date and
- (ii) repay less than 100% of the previously refunded Member Contribution Account plus interest.
- (iii) Upon 100% repayment, the Member shall have his Member Contribution Account restored along with the applicable Credited Service under Section 5.02.
- (iv) Upon less than 100% repayment, the Member shall have the respective subaccounts of his Member Contribution Account restored in the following order:
  - (A) Pick-up Contribution Account;
  - (B) Required After-Tax Contribution Account;
  - (C) Service Purchase Account (if any);
  - (D) Optional After-Tax Contribution Account (if any);
  - (E) Transferred Contribution Account (if any); and
  - (F) Terminated Contribution Account (if any).

The full amount of refund or withdrawal, plus interest, shall be restored to each sub account before restoration of the next sequential sub account. Credited Service under Section 5.02 attributed to such repaid contribution shall be determined upon completion of the repayment.

The Plan may accept any such repayment directly from the Member or through a plan-to-plan transfer from any other qualified retirement plan, Section 401(k) plan, Section 457(b) plan or a Section 403(b) tax sheltered annuity.

#### ARTICLE V: SUPERANNUATION RETIREMENT PENSION

# 5.01 Offering of Superannuation Retirement Pension

A Member who Retires on or after satisfaction of the conditions set forth in Section 5.01 of the Adoption Agreement, shall receive a Superannuation Retirement Pension.

## 5.02 Amount of Superannuation Retirement Pension

## (a) <u>Defined Benefit Plan</u>

Subject to the maximum permissible amount limitations in Article XII, a Member's Superannuation Retirement Pension shall equal his Accrued Benefit. The Member's Accrued Benefit shall be the sum of:

- (i) his Basic Benefit as determined under Adoption Agreement; and
- (ii) his Service Increment Benefit if such benefit is authorized under the Adoption Agreement.

The Superannuation Retirement Pension shall be expressed in the normal form of benefit as elected by the Municipality in Section 10.01 of the Adoption Agreement.

## (b) Cash Balance Plan

Subject to the maximum permissible amount limitations of Article XII, a Member's Superannuation Retirement Pension shall be his Accrual Benefit as calculated pursuant to Section 5.06. The Superannuation Retirement Pension shall be expressed in the normal form of benefit as elected by the Municipality in Section 10.01 of the Adoption Agreement.

# 5.03 Computation and Payment of Superannuation Retirement Pension

# (a) Computations

The Superannuation Retirement Pension shall be computed by the Board in the form of benefit selected by the Member. A Member shall select the form of benefit for his Superannuation Retirement Pension as either the normal form as elected by the Municipality in Section 10.01 of its Adoption Agreement or any eligible optional form as elected by the Municipality in Section 10.02 of its Adoption Agreement. If no optional form of benefit is selected by the Member, the Pension benefit shall be paid in the normal form.

# (b) Payments

Payments shall begin no earlier than the Member's Superannuation Retirement Date and begin no later than the date specified in Section 10.04(b) of the Plan. Between the dates a Member is first eligible to receive Superannuation Retirement Pension and the Mandatory Commencement of Benefits date specified in Section 10.04(b), a Member must apply for benefits to commence. The Member's Benefit Commencement Date shall be the first day of the month coincident with or next following the later of (i) Member's Termination of Employment following the Superannuation Retirement Date; (ii) the date specified by the Member in his or her application, or (iii) the date the application is approved by the Board.

Payments for an annuity form of benefit shall continue until the last scheduled payment coincident with the month of the Member's death or, if applicable, the month of the death of the Survivor Annuitant.

The above notwithstanding, if a Member's Retirement arises due to a Severance from Credited Service, and the Member remains employed by the Municipality, payment shall

begin no earlier than the Member's Normal Retirement Age as set forth in Treasury Regulation § 1.401(a)-1(b).

# 5.04 <u>Involuntary Lump Sum Payment of Superannuation Retirement Pension</u>

If elected by the Municipality in its Adoption Agreement, a lump sum payment shall be made for a Superannuation Retirement Pension to Members, with the Member's consent, if the lump sum Actuarial Equivalent of the Member's vested Accrued Benefit is less than or equal to \$5,000.

However, effective as of January 1, 2006, if the mandatory distribution is greater than \$1,000 but less than \$5,000, and the Member does not elect to have such distribution paid directly to an Eligible Retirement Plan, specified by the Member in a direct rollover or to receive the distribution directly, then the Board will (i) pay the distribution in a direct rollover to an individual retirement plan designated by the Board; or (ii) retain the distributable amount until the Member elects to have such benefit paid provided, however, that such benefit shall commence on or before the Required Beginning Date as set forth under Section 10.04(b).

If such a lump sum payment is made or transferred to an Individual Retirement Account or an Eligible Retirement Plan, the Member's vested Accrued Benefit shall be disregarded and he shall not be entitled to any other Pension benefits under the Plan.

## 5.05 Late Retirement

Except as provided in Article X, a Member shall receive Credited Service for Service completed after the Member's Superannuation Retirement Date, until subsequent Termination of Employment.

## 5.06 Cash Balance Superannuation Retirement Pension

#### (a) Accrued Benefit

- (i) A Member's Accrued Benefit, as of any determination date (on or prior to the Superannuation Retirement Date), means a lifetime annuity in the normal form of benefits as described in Section 10.01 of the Plan commencing at a Participant's Superannuation Retirement Date, calculated by projecting the Member's Hypothetical Account Balance to Superannuation Retirement Age with interest at the Cash Balance Interest Crediting Rate in effect at the date of determination, and converting the projected account to an Actuarial Equivalent benefit payable in the normal form at the Member's Benefit Commencement Date.
- (ii) In the event of a change in the assumptions defined under Section 1.04, Actuarial Equivalence/ or Actuarial Equivalent such that the change would result in a lower projected Superannuation Retirement Pension, the Accrued Benefit will be determined as the last day of the Stability Period immediately preceding the Stability Period in which the assumption change takes effect. In no event will a Member's accrued benefit be less than the Accrued Benefit defined in this subsection

# (b) Establishment of Hypothetical Account Balance.

A Hypothetical Account Balance shall be established and maintained for each Member. Additions to and reductions in the Hypothetical Account Balance shall be made in accordance with the provisions set forth below. This Hypothetical Account Balance shall be a hypothetical account for bookkeeping purposes only and neither the maintenance nor the adding of credits thereto shall be construed as an allocation of assets of the Plan to, or a segregation of such assets in, any such Hypothetical Account Balance, or otherwise creating a right for any individual to receive specific assets of the Plan. Benefits provided under the

Plan shall be paid from the general assets of the Trust in the amounts, in the forms, and at the times provided, under the terms of the Plan. When applying any statutory or Plan limitation and/or minimum benefit that is expressed in terms of an annuity to the benefit derived from the Hypothetical Account Balance, the limit shall be applied to the annuity derived from the Hypothetical Account Balance that is payable at the time and in the form corresponding to the Plan limitation or minimum benefit, determined under the terms of the Plan.

## (c) Cash Balance Contribution Credits

As of the last day of each Plan Year, a Cash Balance Contribution Credit amount as set forth in the Adoption Agreement shall be credited to such Member's Hypothetical Account. If the Member is employed for a period of less than the full Plan Year, the Cash Balance Contribution Credit shall be equal to the ratio of (i) the number of days in which the Member is employed in the Plan Year; to (ii) 365.

## (d) Cash Balance Interest Credits

At the end of each Cash Balance Interest Credit Period, a Cash Balance Interest Credit shall be credited to the Hypothetical Account Balance. The Cash Balance Interest Credit shall be calculated by multiplying the balance in the Member's Hypothetical Account Balance at the beginning of the Cash Balance Interest Credit Period by the Cash Balance Interest Crediting Rate applicable for such Cash Balance Interest Credit Period, based upon the Stability Period and the Lookback Month that applies for the Cash Balance Interest Credit Period. No Cash Balance Interest Credits shall accrue to any portion of the Hypothetical Account Balance after the annuity starting date that applies to that portion.

## (e) Cash Balance Interest Credit Period

The Cash Balance Interest Credit Period shall be each calendar quarter.

#### ARTICLE VI: EARLY RETIREMENT PENSION

# 6.01 Offering of Early Retirement Pension

The Municipality shall elect in its Adoption Agreement whether to offer an Early Retirement Pension. If the Municipality elects to offer an Early Retirement Pension, a Member who satisfies the eligibility criteria specified in Section 6.02 of its Adoption Agreement and who Retires on or after the Early Retirement Date, but before the Superannuation Retirement Date, may elect to receive an Early Retirement Pension.

## 6.02 Eligibility for Early Retirement Pension

If the Municipality elects to provide an Early Retirement Pension, its Adoption Agreement shall specify the following eligibility criteria:

- (a) The minimum age, if any, for a Member to be eligible for an Early Retirement Pension,
- (b) The minimum years of Vesting Service, if any, required to be completed to be eligible for an Early Retirement Pension, and
- (c) The minimum age and years of Vesting Service combined (expressed as a single cumulative number), if any, required to be completed to be eligible for an Early Retirement Pension,

The Municipality may elect to provide for more than one method of determining Member eligibility for the Early Retirement Pension under (a), (b) and (c).

If a Member must satisfy a minimum age and Vesting Service requirement for an Early Retirement Pension, a Member who has a Termination of Employment after satisfying the Vesting Service requirement but not the minimum age requirement may elect to receive an Early Retirement Pension upon satisfying the minimum age requirement.

# 6.03 Amount of Early Retirement Pension

Subject to the maximum permissible amount limitations of Article XII of the Plan, if the Municipality elects to provide an Early Retirement Pension, it shall equal the Member's vested Accrued Benefit or the Actuarial Equivalent of the vested Accrued Benefit as of the Early Retirement Date.

The formula specified by the Municipality in Section 5.02 of the Adoption Agreement shall be used to calculate the Early Retirement Pension in the normal form of benefit as elected by the Municipality in Section 10.01 of the Adoption Agreement. The above notwithstanding, the Early Retirement Pension payable from a Cash Balance Plan shall be the Actuarial Equivalent of his Hypothetical Account Balance determined as of the Benefit Commencement Date.

The Early Retirement Pension shall be adjusted by any reductions elected by the Municipality in its Adoption Agreement for a reduced Early Retirement Pension when the Member's Benefit Commencement Date is prior to his Superannuation Retirement Date.

# 6.04 Computation and Payment of Early Retirement Pension

#### (a) Computations

The Early Retirement Pension shall be computed by the Board in the form of benefit selected by the Member. A Member shall select the form of benefit for his Early Retirement Pension as either the normal form as elected by the Municipality in Section 10.01 of its Adoption Agreement or any eligible optional form as elected by the Municipality in Section 10.02 of

its Adoption Agreement. If no optional form of benefit is selected by the Member, the Pension benefit shall be paid in the normal form.

## (b) Payments

Payments shall begin no earlier than the Early Retirement Date and begin no later than the date specified in Section 10.04(b) of the Plan. Between the dates a Member is first eligible to receive the Early Retirement Pension and the Mandatory Commencement of Benefits date specified in Section 10.04(b), a Member must apply for benefits to commence. The Member's Benefit Commencement Date shall be the first day of the month coincident with or next following the later of (i) the Member's Termination of Employment following Early Retirement Date; (ii) the date the application is approved by the Board; or (iii) the date specified by the Member in his or her application. The above notwithstanding, if a Member's Retirement arises due to a Severance from Credited Service, and the Member remains employed by the Municipality, payment shall begin no earlier than the Member's Normal Retirement Age as set forth in Treasury Regulation § 1.401(a)-1(b).

Payments for an annuity form of benefit shall continue until the last scheduled payment coincident with the month of the Member's death or, if applicable, the month of the death of the Survivor Annuitant.

# 6.05 <u>Limited Offering of Early Retirement Pension Under Alternative Eligibility Requirements</u>

The Municipality may amend this Plan to provide for different eligibility requirements for an Early Retirement Pension as part of a bona fide retirement incentive program ("Program"). Prior to the implementation of any such Program, the Municipality shall:

- (a) submit to the System a proposed addendum as an amendment to its Adoption Agreement specifying the proposed Program's terms and conditions as tentatively approved by the Municipality's governing authority; and
- (b) provide a draft of the disclosure documents to be provided to eligible Members explaining the Program and an illustration of the Program's impact on a typical Member; and
- (c) provide an explanation of the Municipality's reason for the Program including the expected Municipality benefits; and
- (d) request the System to complete an actuarial cost study of the proposed Program.

Upon completion of the actuarial cost study, the System shall provide the Municipality the cost study results and may request additional information from the Municipality. If the Municipality elects to amend the Plan to implement the Program, it shall provide the System with an ordinance or resolution (i) authorizing an amendment of the Plan in the form of a proposed addendum; (ii) a copy of the disclosure documents for each eligible Member; and (iii) identifying the Program's final terms and conditions, the estimated annual and total additional costs to the Municipality's Plan and any other information requested by the System.

Such Program shall be permitted only upon a determination by the Board that (i) additional benefits thereunder, when added to all other benefits provided under the plan do no exceed the limitation on benefits set forth under Code Section 415(b); and (ii) the Program does not otherwise violate any tax qualification provisions of the Code.

#### ARTICLE VII: DISABILITY PENSION

# 7.01 Offering of Disability Pension

The Municipality may elect in its Adoption Agreement to provide a disability benefit under the Plan. If the Municipality elects to provide such benefit, an active Member who, prior to satisfying the requirements for a Superannuation Retirement Pension, becomes Disabled and who satisfies the eligibility criteria established in Section 7.02 of the Municipality's Adoption Agreement shall receive a Disability Pension.

## 7.02 Eligibility for Disability Pension

If the Municipality elects to provide a disability benefit, the Municipality shall also elect the eligibility criteria for Member submission of an application for a non-service connected Disability Pension. Where the disability of a Member is determined to be a Service Connected Disability, no minimum period of service, nor minimum age, shall be required for eligibility. In addition to any other criteria elected by the Municipality which shall be non-discretionary and applied on a uniform basis, a Member may become eligible for a Disability Pension only if he is an Eligible Employee on his Disability Date.

# 7.03 Amount of Disability Pension

Subject to the Maximum Permissible Dollar Limitations of Article XII of the Plan, if the Municipality elects to provide a Disability Pension, its Adoption Agreement shall establish the amount of a Member's Disability Pension including any adjustments or offsets for payment made to the Member for the same disability. The Municipality shall be allowed to establish a different benefit amount if the Member is determined to have a Service Connected Disability.

The above notwithstanding, the Disability Pension payable from a Cash Balance Plan shall be the Actuarial Equivalence of his Hypothetical Account Balance determined as of the Benefit Commencement Date.

#### 7.04 Computation and Payment of Disability Pension

## (a) Benefit Commencement Date

- (i) The Benefit Commencement Date shall be no earlier than the later of the first day of the month coincident with or next following the Member's Termination of Employment; or
- (ii) the first day of the first month coincident with or next following the Disability Date as determined by the Board.

# (b) <u>Program Computations</u>

The Disability Pension shall be computed by the Board in the Single Life form of benefit.

# (c) Payments

- (i) Shall be in accordance with Article X of the Plan.
- (ii) Payments shall continue until earlier of:
  - (A) the last scheduled payment coincident with the month the Member has been determined to no longer be Disabled, or
  - (B) the last scheduled payment coincident with the month of the Member's death.

# 7.05 Recovery from Disability

If a Member recovers from Disability and is reemployed as an Employee under the Plan, the Member's Credited Service shall be restored up to the Benefit Commencement Date of his Disability Pension. The Member then shall commence to accrue benefits under the Plan based upon his Credited Service before the Benefit Commencement Date of his Disability Pension and after his Reemployment Commencement Date.

# 7.06 Application and Continuing Evidence of Total Disability

- (a) A Member may, upon application or on application of one acting on the Member's behalf, or upon application of a responsible official of the Municipality, be retired by the Board on a disability retirement following a review of the application and any supporting medical records and other documentation submitted with the application certifies to the Board that the Member is unable to engage in any gainful employment and that said Member ought to be retired.
- (b) The Board may require a Disabled Member to undergo medical examination by a physician designated by the Board at least once per year prior to the Member's Superannuation Retirement Date. Such examination shall be made at the place of residence of the Member or other place mutually agreed upon. Should the physician report and certify to the Board that such Disabled Member is no longer physically or mentally incapacitated for the performance of duty and is able to engage in a gainful occupation, then his disability retirement allowance shall be discontinued, and in lieu thereof an early involuntary retirement allowance shall at that time be granted as if such person had been retired not voluntarily, if such provision is selected by the Municipality in Section 6.01 of the Adoption Agreement and if the Member shall have had the required number of years of total service as stated in the Adoption Agreement.
- (c) Should a Disabled Member refuse to submit to a medical examination as provided above, his Disability Pension shall be discontinued until the withdrawal of such refusal, and should such refusal continue for one year, then all his right in and to any disability retirement allowance or for early involuntary retirement allowance shall be forfeited.
- (d) Should a Disabled Member die before the amount payable under this section equals the total amount allocated to his Member Contribution Account, plus the balance in the Member's Excess Investment Account as of the Effective Date of Disability Retirement, the Board shall pay to the Beneficiary (if living, or if the Beneficiary predeceased the Disabled Member, or no Beneficiary was named, then to the Disabled Member's estate) an amount equal to the difference between such amounts paid under this Section and the Member Contribution Account and the balance in the Member's Excess Investment Account. If such difference is less than one hundred dollars (\$100) and no letters have been taken out on the estate within six (6) months after the Disabled Member's death, such difference may be paid to the undertaker or to any person or Municipality who or which shall have paid the claim of the undertaker.

#### ARTICLE VIII: DEFERRED VESTED PENSION

# 8.01 Offering of Deferred Vested Pension

A Member who meets the eligibility criteria specified in Section 8.02 shall receive a Deferred Vested Pension.

## 8.02 Eligibility for Deferred Vested Pension

A Member is eligible to receive a Deferred Vested Pension provided that: (a) the Member has completed the minimum amount of Vesting Service specified in Section 8.05 of the Municipality's Adoption Agreement; and (b) the Member Terminated Employment but has not yet Retired.

# 8.03 Amount of Deferred Vested Pension

The Member's Deferred Vested Pension shall equal his Vested Accrued Benefit as of the date of his Termination of Employment.

#### 8.04 Computation and Payment of Deferred Vested Pension

## (a) Computations

The Deferred Vested Pension shall be computed by the Board in the form of benefit selected by the Member. A Member shall select the form of benefit for his Deferred Vested Pension as either the normal form as elected by the Municipality in Section 10.01 of its Adoption Agreement or any eligible optional form as elected by the Municipality in Section 10.02 of its Adoption Agreement. If no optional form of benefit is selected by the Member, the Pension benefit shall be paid in the normal form.

#### (b) Payments

Payments shall begin no earlier than the Member's Early Retirement Date and begin no later than the date specified in Section 10.04(b) of the Plan. Between the dates a Member is first eligible to receive his Early Retirement Pension and the Mandatory Commencement of Benefits date specified in Section 10.04(b), a Member must apply for benefits to commence; provided however, the Member's Benefit Commencement Date shall be the first day of the month coincident with or next following the later of (1) the Member's Termination from Employment; or (2) the date the application is approved by the Board.

Payments for an annuity form of benefit shall continue until the last scheduled payment coincident with the month of the Member's death or, if applicable, coincident with the month of the death of the Survivor Annuitant.

#### (c) Voluntary Lump Sum Payment of Deferred Vested Pension

Notwithstanding the provisions of paragraphs (a) and (b), if elected by the Municipality in its Adoption Agreement, a lump sum payment shall be made for a Deferred Vested Pension to Members, at any time on or after the Member's Termination of Employment with the Member's consent, if the lump sum Actuarial Equivalent of the Member's vested Accrued Benefit is less than \$5,000.

If such a lump sum payment is made or transferred to an Individual Retirement Account or an Eligible Retirement Plan, the Member's vested Accrued Benefit shall be disregarded and he shall not be entitled to any other benefits under the Plan.

#### (d) Involuntary Lump Sum Payment of Deferred Vested Pension

Notwithstanding the provisions of paragraphs (a), (b) and (c), if elected by the Municipality in its Adoption Agreement, a lump sum payment shall be made for a Deferred Vested Pension

to Members, at any time on or after the Member's Termination of Employment and without the Member's consent, if the lump sum Actuarial Equivalent of the Member's vested Accrued Benefit is less than or equal to \$5,000 or such other higher amount as the Board may approve from time to time.

However, effective as of January 1, 2006, if the mandatory distribution is greater than \$1,000, and the Member does not elect to have such distribution paid directly to an Eligible Retirement Plan, specified by the Member in a direct rollover or to receive the distribution directly, then the Board will (i) pay the distribution in a direct rollover to an individual retirement plan designated by the Board; or (ii) retain the distributable amount until the Member elects to have such benefit paid provided, however, that such benefit shall commence on or before the Required Beginning Date set forth under Section 10.04(b).

If such a lump sum payment is made or transferred to an Individual Retirement Account or an Eligible Retirement Plan, the Member's vested Accrued Benefit shall be disregarded and he shall not be entitled to any other Pension benefits under the Plan.

## 8.05 Vesting Schedule

A Member's Accrued Benefit derived from Municipality contributions shall be one hundred percent (100%) vested:

- (a) on and after his Superannuation Retirement Date (if employed on or after that date),
- (b) if his employment Terminates as a result of:
  - (i) death, provided the member has satisfied the death benefit eligibility criteria elected by the Municipality in Section 9.01 of its Adoption Agreement; or
  - (ii) Disability, provided the member has satisfied the Disability benefit eligibility criteria elected by the Municipality in Section 7.01 of its Adoption Agreement;
- (c) if there is a complete termination of the Plan, but only to the extent the benefits are funded.

For all other circumstances other than those applicable above, a Member shall receive a vested percentage of his Accrued Benefit derived from Municipality contributions equal to the percentage of completed vesting service designated in the Municipality's Adoption Agreement.

#### **8.06** Member Contribution Forfeitability

A Member shall be one hundred percent (100%) vested at all times in the amounts allocated to his Member Contribution Account.

#### ARTICLE IX: DEATH BENEFITS

# 9.01 Pre-Retirement Death Benefit

## (a) Eligibility and Amount

If a Member dies while an Employee of the Municipality, the Member's designated Beneficiary shall be entitled to an immediate death benefit equal to Actuarial Equivalent of his Vested Accrued Benefit, provided the Member, prior to his death, satisfied the death benefit eligibility criteria elected by the Municipality in Section 9.01 of its Adoption Agreement.

If the Member fails to satisfy the death benefit eligibility criteria prior to his death, the designated Beneficiary shall receive the value of the Member Contribution Account balance valued as of the Member's date of death in a single lump sum payment.

# (a) Optional Forms of Payment

Upon satisfying the death benefit eligibility criteria, a Member shall have the option of filing with the System an application to specify a form of payment for the distribution of the death benefit. The available forms of payment shall be the forms of benefit elected by the Municipality in Section 10.02 of its Adoption Agreement.

If the Member fails to specify a form of payment upon satisfying the death benefit eligibility criteria, it will be deemed that the Member has specified the optional form of benefit identified as the "Single Life Guaranteed Present Value" and shall be paid to the designated Beneficiary in a single lump-sum payment or, in accordance with the limitations found in Section 10.04 of the Plan, an Actuarial Equivalent period certain annuity.

# 9.02 Deferred Vested Death Benefit

# (a) Eligibility and Amount

If a Deferred Vested Member eligible to receive a Deferred Vested Pension dies prior to the Benefit Commencement Date, the Member's designated Beneficiary shall be entitled to a death benefit equal to Actuarial Equivalent of his Vested Accrued Benefit, provided the Member, prior to his death, satisfied the death benefit eligibility criteria elected by the Municipality in Section 9.01 of its Adoption Agreement.

If the Member fails to satisfy the death benefit eligibility criteria prior to his death, the designated Beneficiary shall receive the value of the Member Contribution Account balance valued as of the Member's date of death less any benefit payment paid to or on behalf of the Member. Such amount shall be paid in a single lump sum payment.

## (b) Optional Forms of Payment

Upon satisfying the death benefit eligibility criteria, a Member shall have the option of filing with the System an application to specify a form of payment for the distribution of the death benefit. The available forms of payment shall be the forms of benefit elected by the Municipality in Section 10.02 of its Adoption Agreement.

If the Member fails to specify a form of payment upon satisfying the death benefit eligibility criteria, it will be deemed that the Member has specified the optional form of benefit identified as the "Single Life Guaranteed Present Value" and shall be paid to the designated Beneficiary in a single lump-sum payment or, in accordance with the limitations found in Section 10.04 of the Plan, an Actuarial Equivalent period certain annuity.

## 9.03 Post Retirement Death Benefit

If a Member's death occurs after the Benefit Commencement Date and there remains no further benefit payment obligations based on the Member's elected optional form of benefit, the designated Beneficiary shall be entitled to an amount equal to the Member's Contribution Account as of the Benefit Commencement Date less the total accumulated benefit payment amounts made to the Member or to the applicable recipients designated by the Member; otherwise no death benefit is payable.

## 9.04 Minimum Death Benefit

The minimum death benefit shall be a refund of the Member's Contribution Account valued at the time of the Member's death less any benefit payments paid to or on behalf of the Member.

# 9.05 Limitation on Death Benefit

Notwithstanding anything in this Article to the contrary, death benefits shall not be paid in excess of the amount that would be considered an incidental death benefit as defined under Treasury Regulation § 1.401-1(b)(1)(i).

# ARTICLE X: PAYMENT OF ACCRUED BENEFIT OPTIONAL FORMS OF PAYMENT

#### **10.01** Normal Form of Benefit

The Municipality shall elect in its Adoption Agreement to provide one of the following the normal form of benefits that shall be used for the computation and payment of pension benefits.

## (a) Single Life

An immediate monthly benefit payment payable for the life of the Member.

#### (b) Life and 50% Spouse Benefit

An immediate monthly benefit payment payable for the life of the Member and the life of the Spouse. Upon the Member's death, the Spouse at the time of the Member's death shall receive a survivor's annuity equal to fifty percent (50%) of the monthly benefit amount being received by the Member immediately preceding his death and such survivor annuity shall be paid for the remainder of the Spouse's life. If the Member has no Spouse at the time of death, and the Survivor Annuitant is one or more of Member's children under the age of 18, (or, if in college, under the age of 23), the survivor's annuity shall be divided and paid equally to each child until such child attains age 18 (or, if in college, age 23).

# (c) Life and 100% Spouse Benefit

An immediate monthly benefit payment payable for the life of the Member and the life of the Spouse. Upon the Member's death, the spouse at the time of the Member's death shall receive a survivor's annuity equal to one hundred percent (100%) of the monthly benefit amount being received by the Member immediately preceding his death and it shall be paid for the remainder of the Spouse's life. If the Member has no Spouse at the time of death, and the Survivor Annuitant is one or more of Member's children under the age of 18, (or, if in college, under the age of 23), the survivor's annuity shall be divided and paid equally to each child until such child attains age 18 (or, if in college, age 23).

#### 10.02 Optional Forms of Benefit

Subject to the limitations of Section 12.08 of the Plan, if the Member selects an optional form of benefit, the Member shall receive the Actuarial Equivalent of the Vested Accrued Benefit payable at Early Retirement or Superannuation Retirement Date.

## (a) Required Optional Forms of Benefit

The Municipality shall provide the following optional forms of benefit for the computation and payment of monthly benefits:

## (i) Single Life with Minimum Net Present Value

A monthly benefit payable for the life of the Member; provided however, that the total benefit payments shall not be less than the present value of the Member's Vested Accrued Benefit ("NPV") as of the Benefit Commencement Date. Should the Member die prior to receiving benefit payments at least equal to the NPV, the designated beneficiary shall receive the remaining amount.

#### (ii) Life and 50% Survivor Annuitant Benefit

A monthly benefit payable for the life of the Member. Upon the Member's death, the Survivor Annuitant shall receive fifty percent (50%) of the monthly benefit amount being received by the Member immediately preceding his death for the remainder of the Survivor Annuitant's life.

## (iii) Life and 100% Survivor Annuitant Benefit

A monthly benefit payable for the life of the Member. Upon the Member's death, the Survivor Annuitant shall receive one hundred percent (100%) of the monthly benefit amount being received by the Member immediately preceding his death for the remainder of the Survivor Annuitant's life.

# (b) Employer Elective Optional Forms of Benefit

The Municipality may elect in its Adoption Agreement to provide the following additional optional forms of benefit for the computation and payment of monthly benefits:

# (i) Single Life with 10 Years Certain

A monthly benefit payable for the life of the Member; provided however; that the total number of monthly benefit payments made shall not be less than 120. Should the Member die prior to receiving 120 monthly benefit payments, the designated Beneficiary shall receive the remaining monthly benefit payments.

# (ii)Single Life with 20 Years Certain

A monthly benefit payable for the life of the Member; provided however; that the total number of monthly benefit payments made shall not be less than 240. Should the Member die prior to receiving 240 monthly benefit payments, the designated Beneficiary shall receive the remaining monthly benefit payments.

## (iii) Lump Sum Payment with Reduced Monthly Benefit

A lump sum payment of the Member's Contribution Account at his Benefit Commencement Date combined with a reduced monthly benefit based on the Member's election of one of remaining available optional forms of benefit. The Member may elect to withdraw less than the full amount of the Member Contribution Account. The reduced monthly benefit shall be the Actuarial Equivalent of the Member's Vested Accrued Benefit less the lump sum payment of the Member's Contribution Account.

The Member may select in writing one of the permitted optional forms of benefit prior to his Benefit Commencement Date. This optional form selection shall become irrevocable upon deposit of the Member's first benefit payment, including lump sum payments. If a Member fails to elect one of the optional forms of benefit as of his Benefit Commencement Date, the System shall deem the Member as having elected the Normal Form of Benefit specified in Section 10.01 of the Municipality's Adoption Agreement.

## 10.03 Cost of Living Adjustment

If elected by the Municipality in its Adoption Agreement, eligible Plan Members shall receive a cost of living adjustment applied to eligible benefit payments in an amount provided in the Municipality's Adoption Agreement as follows:

## (a) Municipality Discretionary Percentage Rate

The Municipality may provide from time to time a specified cost of living adjustment to be applied to eligible benefit payments. Any such cost of living adjustment does not obligate the Municipality to provide future cost of living adjustments. Each such adjustment shall be specified as a fixed percentage rate to be applied equitably to all eligible benefit payments and shall be adopted by the Municipality as an amendment to the Plan in accordance with the provisions of Article XVI.

Any Municipality discretionary adjustment authorized and adopted shall not exceed the percentage change calculated in accordance with Subsection (c) below. If the discretionary adjustment percentage rate exceeds the percentage change calculated in accordance with Subsection (c) below, the cost of living adjustment shall be adjusted to the to the rate calculated in Subsection (c) below.

## (b) Fixed Percentage Rate

If a fixed percentage cost of living adjustment has been elected by the Municipality in the Adoption Agreement, the Municipality will provide for a cost of living adjustment based on a specified fixed percentage rate to be applied equitably to all eligible benefit payments.

Any fixed percentage rate adjustment authorized and adopted shall not exceed the percentage change calculated in accordance with Subsection (c) below. If the fixed percentage rate specified the Adoption Agreement exceeds the percentage change calculated in accordance with Subsection (c) below, the cost of living adjustment shall be adjusted to the rate calculated in Subsection (c) below.

## (c) Adjustable Percentage Rate based on the Consumer Price Index

The Municipality may provide for a cost of living adjustment based on the United States Bureau of Labor Statistics' Consumer Price Index. The index series used shall be the Consumer Price Index for All Urban Consumers (CPI-U) for the U.S. City Average for All Items, 1982-84=100.

- (i) Such increases shall be a percentage to be applied to the existing benefit payment amount
- (ii) The specific percentage change shall be calculated by:
  - (A) averaging the monthly index amounts for the final three months of each calendar year immediately preceding the year in which the increase is to be applied (the "Second Year") and averaging the monthly index for the final three months of the calendar year preceding the Second Year (the "First Year"); and
  - (B) Subtracting the First-Year average from the Second-Year average; and
  - (C) Dividing the result of subsection (B) by the First-Year average.
- (iii) If the result in subsection (ii)(C) is less than zero, no adjustment shall be made for the applicable Plan year.
- (iv) If the result in subsection (ii)(C) is greater than zero, the percentage amount, rounded down to one decimal place, shall be applied to all eligible benefit payments.

Cost of living adjustments shall be applied to eligible benefit payments beginning with the payments dated April 1 in the applicable Plan Year.

# (d) Maximum Cost of Living Adjustment

Notwithstanding anything in this Section, in no event shall a cost of living adjustment cause annual payment to a Member or Beneficiary to exceed (i) seventy-five percent (75%) of the Member's Final Average Salary; or (ii) one hundred thirty percent (130%) of the Member's or Beneficiary's annual Pension payment; each determined as of the Member's Date of Termination of Service.

For the purposes of this Section 10.03, "eligible benefit payments" shall mean payments to eligible Members that were disbursed or required to be disbursed by December 31 of the Plan Year immediately preceding the Plan Year in which the cost of living adjustment is to be applied. For the purposes of this Section 10.03, "eligible Plan Members" shall mean

Members receiving benefit payments for an Early or Superannuation Retirement Pension and, if elected by the Municipality in its Adoption Agreement, Members receiving a Disability or Deferred Vested Pension or Survivor Annuitants receiving a joint and survivor annuity.

#### 10.04 Commencement of Benefits/Payment Schedules

## (a) Entitlement to Payments

The Member is entitled to commence the payment of his Superannuation or Early Retirement on the first day of the first month coincident with or following his Termination of Employment following his Superannuation or Early Retirement Date.

# (b) Mandatory Commencement of Benefits – Required Beginning Date

In no event shall the Board commence the payment of a Member's Superannuation Retirement Pension, Early Retirement Pension or Deferred Vested Pension later than the first day of April in the calendar year following the later of:

- (i) the calendar year in which the Member attains age 70-1/2, or
- (ii) the calendar year in which the Member Terminates Employment as required by Code Section 401(a)(9).

## (c) Benefit Payments to Beneficiaries After Member's Death

- (i) If Pension benefit payments begin prior to the Member's death, the remaining Vested Accrued Benefit will be distributed to his Beneficiary at least as rapidly as under the method of benefit payments being used as of the date of the Member's death.
- (ii) If the Member dies after application to the Board for the commencement of benefits but prior to the Benefit Commencement Date, the Member's Beneficiary, based on the Actuarial Equivalence of a single, lump sum payment calculated as of the Member's date of death, shall receive the greater of:
  - (A) the remaining vested Accrued Benefit, or
  - (B) the total of any Death Benefits described in Article IX as elected by the Municipality in its Adoption Agreement.
- (iii) If the Member dies before his Benefit Commencement Date the following rules apply:
  - (A) If the Member's Spouse is the sole Beneficiary, distribution must begin by December 31 of the calendar year immediately following the calendar year in which the Member dies or by December 31 of the calendar year in which the Member would have attained age 70½, if later.
  - (B) If the Member's Spouse is not the sole Beneficiary, then distribution must begin by December 31 of the calendar year immediately following the calendar year in which the Member dies.
  - (C) If there is no designated Beneficiary as of September 30 of the year following the year of the Member's death, the Member's entire benefit must be distributed by December 31 of the calendar year containing the fifth anniversary of the Member's death.
  - (D) If the Member's Spouse is the sole Beneficiary and the Spouse dies after the Member but before distributions to the Spouse begin, this Section 10.04(c)(iii) (other than Section 10.04(c)(iii)(A)) will apply as if the Spouse were the Member.

## (d) Conformance to Code Section 401(a)(9)

- (i) All distributions will be made in accordance with Code Section 401(a)(9), the regulations promulgated under Code Section 401(a)(9) and any other provisions reflecting the requirements of Code Section 401(a)(9) and prescribed by the Internal Revenue Service; and the terms of the Plan reflecting the requirements of Code Section 401(a)(9) override the distribution options (if any) in the Plan which are inconsistent with those requirements.
- (ii) Effective January 1, 2006, unless the Member's benefit is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the Required Beginning Date, as of the first distribution calendar year distributions will be made in accordance with sections (e), (f) and (g) of this Section 10.04. If the Member's benefit is distributed in the form of an annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code Section 401(a)(9) and the Treasury regulations.
- (iii) Notwithstanding any other provision in the Plan to the contrary, distributions from the Plan will be made in accordance with a good faith interpretation of Code Section 401(a)(9) and the regulations thereunder as applicable to governmental plans with the meaning of Code Section 414(d) and shall be implemented in accordance with the grandfathering provisions of such regulations applicable to annuity option distributions in effect on April 17, 2001.

# (e) Determination of Amount to be Distributed Each Year

(i) Annuity Distributions

If the Member's benefit is paid in the form of annuity distributions under the Plan, payments under the annuity will satisfy the following requirements:

- (A) the annuity distributions will be paid in periodic payments made at intervals not longer than one year;
- (B) the distribution period will be over a life (or lives) or over a period certain not longer than the period described in Section 10.04 (f) or 10.04 (g);
- (C) Once payments have begun over a period certain, the period certain will not be changed even if the period certain is shorter than the maximum permitted; and
- (D) Payments will either be non-increasing or increase only as follows:
  - 1. by an annual percentage increase that does not exceed the annual percentage increase in a cost-of-living index that is based on prices of all items and issued by the Bureau of Labor Statistics;
  - 2. to the extent of the reduction in the amount of the Member's payments to provide for a survivor benefit upon death, but only if the Beneficiary whose life was being used to determine the distribution period described in Section 10.04(f) dies;
  - 3. to provide cash refunds of employee contributions upon the Member's death;
  - 4. to pay increased benefits that result from the allocation of Excess Interest pursuant to Section 15.10; or
  - 5. to pay increased benefits that result from a Plan amendment.

## (ii) Amount Required to be Distributed by Required Beginning Date

The amount that must be distributed on or before the Member's Required Beginning Date (or, if the Member dies before distributions begin, the date distributions are required to begin under section (c)(iii)(A) or (B)) is the payment that is required for one payment interval. The second payment need not be made until the end of the next payment interval even if that payment interval ends in the next calendar year. Payment intervals are the periods for which payments are received, e.g., bi-monthly, monthly, semi-annually, or annually. All of the Member's benefit accruals as of the last day of the first distribution calendar year will be included in the calculation of the amount of the annuity payments for payment intervals ending on or after the Member's Required Beginning Date.

#### (iii) Additional Accruals After First Distribution Calendar Year

Any additional benefits accruing to the Member in a calendar year after the first distribution calendar year will be distributed beginning with the first payment interval ending in the calendar year immediately following the calendar year in which such amount accrues.

## (f) Requirements For Annuity Distributions That Commence During Member's Lifetime

## (i) Joint Life Annuities Where the Beneficiary Is Not the Member's Spouse

If the Member's benefit is being distributed in the form of a joint and survivor annuity for the joint lives of the Member and a non-spouse Beneficiary, annuity payments to be made on or after the Member's Required Beginning Date to the Beneficiary after the Member's death must not at any time exceed the applicable percentage of the annuity payment for such period that would have been payable to the Member using the table set forth in Q&A-2 of section 1.401(a)(9)-6 of the Treasury regulations. If the form of distribution combines a joint and survivor annuity for the joint lives of the Member and a non-spouse Beneficiary and a period certain annuity, the requirement in the preceding sentence will apply to annuity payments to be made to the Beneficiary after the expiration of the period certain.

## (ii) Period Certain Annuities

Unless the Member's Spouse is the sole designated Beneficiary and the form of distribution is a period certain and no life annuity, the period certain for an annuity distribution commencing during the Member's lifetime may not exceed the applicable distribution period for the Member under the Uniform Lifetime Table set forth in section 1.401(a)(9)-9 of the Treasury regulations for the calendar year that contains the Benefit Commencement Date. If the Benefit Commencement Date precedes the year in which the Member reaches age 70, the applicable distribution period for the Member is the distribution period for age 70 under the Uniform Lifetime Table set forth in section 1.401(a)(9)-9 of the Treasury regulations plus the excess of 70 over the age of the Member as of the Member's birthday in the year that contains the Benefit Commencement Date. If the Member's Spouse is the Member's sole Beneficiary and the form of distribution is a period certain and no life annuity, the period certain may not exceed the longer of the Member's applicable distribution period, as determined under this section (f)(ii), or the joint life and last survivor expectancy of the Member and the Member's Spouse as determined under the Joint and Last Survivor Table set forth in section 1.401(a)(9)-9 of the Treasury regulations, using the Member's and Spouse's attained ages as of the Member's and Spouse's birthdays in the calendar year that contains the Benefit Commencement Date.

# (g) <u>Requirements For Minimum Distributions Where Member Dies Before Date Distributions</u> Begin

## (i) Member Survived by Designated Beneficiary

If the Member dies before the date distribution of his or her benefit begins and there is a Beneficiary, the Member's entire benefit will be distributed, beginning no later than the time described in section (c)(iii)(A) or (B), over the life of the designated Beneficiary or over a period certain not exceeding:

- (A) unless the Benefit Commencement Date is before the first distribution calendar year, the life expectancy of the Beneficiary determined using the Beneficiary's age as of the Beneficiary's birthday in the calendar year immediately following the calendar year of the Member's death; or
- (B) if the Benefit Commencement Date is before the first distribution calendar year, the life expectancy of the Beneficiary determined using the Beneficiary's age as of the Beneficiary's birthday in the calendar year that contains the Benefit Commencement Date.

## (ii) No Beneficiary

If the Member dies before the date distributions begin and there is no Beneficiary as of September 30 of the year following the year of the Member's death, distribution of the Member's entire benefit will be completed by December 31 of the calendar year containing the fifth anniversary of the Member's death.

# (iii) Death of Spouse Before Distributions to Begin

If the Member dies before the date distribution of his or her benefit begins, the Member's Spouse is the Member's sole Beneficiary, and the Spouse dies before distributions to the Spouse begin, this Section 10.04(g) will apply as if the Spouse were the Member, except that the time by which distributions must begin will be determined without regard to section (c)(iii)(A).

#### (h) Definitions

(i) Designated Beneficiary

The individual who is designated as the Beneficiary under Section 1.09 of the Plan and is the designated Beneficiary under Code Section 401(a)(9) and section 1.401(a)(9)-4 of the Treasury regulations.

(ii) Distribution Calendar Year

A calendar year for which a minimum distribution is required.

For distributions beginning before the Member's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Member's Required Beginning Date. For distributions beginning after the Member's death, the first distribution calendar year is the calendar year in which distributions are required to begin pursuant to section 10.04(c)(iii).

## (iii) Life Expectancy

Life expectancy as computed by use of the Single Life Table in Section 1.401(a)(9)-9 of the Treasury regulations.

## (iv) Required Beginning Date

The date specified in section 10.04(b) of the Plan.

## 10.05 Member Continuing Employment after Normal Retirement Age

The Municipality may elect in its Adoption Agreement to permit In-service Distributions. If permitted and a Member elects to commence In-service Distribution, the Member's Service and Compensation earned after commencement of benefit payments shall not thereafter be counted as Credited Service or otherwise increase the Member's Accrued Benefit.

#### 10.06 Reemployment of Member after Benefit Commencement Date

The Municipality shall elect in its Adoption Agreement to allow a retired Member, upon reemployment to (i) elect to either continue or cease receiving his Early or Superannuation Retirement Pension; or (ii) be required to suspend receipt of his Early or Superannuation Retirement Pension.

- (a) If permitted in the Municipality's Adoption Agreement and if the Member elects to continuing receiving his Retirement Benefit, the Member's Service and Compensation earned after the Reemployment Commencement Date shall not thereafter be counted as Credited Service or otherwise increase the Member's Accrued Benefit.
- (b) If the Municipality's Adoption Agreement requires the Reemployed Retired Member suspend receipt of his Retired Benefit; or if the Member elects to suspend receipt of his Retirement Benefit, the Member's Service and Compensation earned after the Reemployment Commencement Date shall thereafter be counted as Credited Service to increase the Member's Accrued Benefit.
- (c) The Municipality is required to notify the Board immediately of the reemployment status of any Retired Member.

#### 10.07 Rollovers

#### (a) General Rule

Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Member's election under this Section, a Member may elect, at the time and in the manner prescribed by the Board, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Member, in a direct rollover.

## (b) Definitions

## (i) Eligible Rollover Distribution

An Eligible Rollover Distribution is any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include: (A) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Member or the joint lives (or joint life expectancies) of the Member and the Member's designated Beneficiary, or for a specified period often (10) years or more; (B) any distribution to the extent such distribution is required under Code Section 401(a)(9); and (C) the portion of any distribution that is a hardship distribution under Code Section 401(k). A Distributee may not elect a direct rollover with respect to an Eligible Rollover Distribution during the Plan Year that is less than \$200. If the Distributee elects to have only a portion of an Eligible Rollover Distribution paid to an Eligible Retirement Plan, that portion must be equal to at least \$500. Notwithstanding anything contained herein to the contrary, the portion of a distribution that is not includible in the gross income because it represents after-tax amounts shall constitute an Eligible Rollover Distribution, but before January 1, 2007, such portion may be transferred only to an individual retirement account or annuity described in Code

Section 408(a) or (b), or to a qualified defined contribution plan described in Code Section 401(a) or 403(a) that agrees to separately account for amounts so transferred, including separately accounting for the portion of the distribution that is includible in gross income and the portion that is not so includible. Effective for distributions on and after January 1, 2007, after-tax amounts also may be rolled over to a defined benefit plan described in Code Section 401(a) or an annuity described in Code Section 403(b), provided in either case that the recipient agrees to separately account for amounts so transferred, including separately accounting for the portion of the distribution that is includible in gross income and the portion that is not so includible.

## (ii) Eligible Retirement Plan

An Eligible Retirement Plan is, in the case of an Eligible Rollover Distribution to the Member or Member's Surviving Spouse, an individual retirement account described in Code Section 408(a), an individual retirement annuity described in Code Section 408(b), an annuity plan described in Code Section 403(a), a qualified trust described in Code Section 401(a), an annuity contract described in Code Section 403(b) that accepts the Member's Eligible Rollover Distribution and an eligible plan under Code Section 457(b) which is maintained by a state, political subdivision or agency or instrumentality of a state and which agrees to separately account for amounts transferred to such plan from this Plan, provided that such account, annuity, contract, plan or trust accepts the Distributee's Eligible Rollover Distribution. Effective as of January 1, 2008, "eligible retirement plan" shall include a Roth IRA established under Code Section 408A.

Effective as of January 1, 2010, in the case of an Eligible Rollover Distribution to a Beneficiary other than the Member's Surviving Spouse, "eligible retirement plan" shall mean: (i) an individual retirement account described in Code Section 408(a), (ii) an individual retirement annuity described in Code Section 408(b), or (iii) as of January 1, 2008, a Roth IRA established under Code Section 408A which shall be treated as an inherited IRA.

## (iii) Distributee

A Distributee includes the Member, former Member, Surviving Spouse or effective January 1, 2010, a nonspouse Beneficiary of the Member.

## (iv) Direct Rollover

A Direct Rollover is a payment by the Plan to the Eligible Retirement Plan specified by the Member.

# ARTICLE XI: MISCELLANEOUS PROVISIONS AFFECTING THE CREDITING OF SERVICE

## 11.01 Service Upon Reemployment

The Municipality, in its Adoption Agreement, shall specify the maximum number of consecutive one-year Breaks in Service a reemployed Employee may incur in order for the Employee to have his Eligibility Service, Vesting Service and Credited Service restored from his prior employment with the Municipality. An Employee who is reemployed by the Municipality after incurring the number of consecutive one year Breaks in Service equal to or greater than that elected by the Municipality in its Adoption Agreement, shall not have restored any prior Credited Service, Eligibility Service and Vesting Service.

All restored Credited Service, Eligibility Service and Vesting Service shall apply to the Plan provisions in effect under the Plan at the time the Member is reemployed and again participates in the Plan.

# (a) Reemployment Having an Existing Vested Accrued Benefit

- (i) If, at an Employee's Reemployment Commencement Date, the Employee has a Vested Accrued Benefit under the Plan and prior Service is restored, his benefit at Benefit Commencement Date shall be the greater of his Vested Accrued Benefit at his Reemployment Commencement Date or his Vested Accrued Benefit as of his Benefit Commencement Date.
- (ii) If, at the Date of Reemployment, the Employee has a vested Accrued Benefit under the Plan and prior Service is not restored, his benefit at Benefit Commencement Date shall be the sum of his vested Accrued Benefit at his Reemployment Commencement Date and his vested Accrued Benefit for subsequent periods of Plan membership as of his Benefit Commencement Date.

#### 11.02 Service Prior to Original Plan

The Municipality may elect in its Adoption Agreement to grant some or all of an Employee's Service with the Municipality prior to the Original Plan Effective Date as Credited Service.

- (a) The Municipality may require the Member to purchase such Service in amounts necessary to reimburse the Plan for up to the total cost of such Service to the Plan; provided, however, any such purchase cost shall not exceed an amount equal to the Actuarial Equivalent present value of the additional Accrued Benefit attributable to such purchased Service as of the date of purchase. The Plan may accept payment directly from the Member or through a plan-to-plan transfer from any other qualified retirement plan, Section 401(k) plan, Section 457(b) deferred compensation plan or a Section 403(b) tax sheltered annuity.
- (b) If Members are required to purchase prior Credited Service, the Member shall have the option to decline any such Service (rather than received such Credited Service as an automatic grant).

#### 11.03 **Qualified Military Service**

## (a) Definitions.

For purposes of this Section, the following definitions shall apply:

(i) <u>Differential Wage Payments</u> means, effective January 1, 2009, any payments that are made by the Municipality to a Member for any period during which the Member is performing Military Service for more than 30 days and represents all or part of the

- Compensation that the Member would have received from the Municipality if he were performing services for the Municipality.
- (ii) <u>Military Service</u> means the period of a Member's active duty for training and service in the Army, Navy, Air Force, Coast Guard or Marines of the United States of America, the Army National Guard and the Air National Guard when engaged in active duty for training, inactive duty training, or full-time National Guard duty, the commissioned corps of the Public Health Service, and any other category of persons designated by the President in time of war or national emergency.
- (iii) **Qualified Military Service** means Military Service during which the Member is entitled to reemployment rights under Chapter 43 Title 38 of the United State Code.
- (iv) <u>USERRA</u> means the Uniformed Services Employment and Reemployment Rights Act of 1994.
- (b) <u>USERRA</u>. Notwithstanding any provision of the Plan to the contrary, contributions, benefits and service credit with respect to Qualified Military Service will be provided in accordance with Code Section 414(u). Plan Members who are reemployed in accordance with the requirements of USERRA shall be treated as not having a Break in Service during such Qualified Military Service and such periods of Qualified Military Service shall be counted for Vesting Service and Credited Service except as provided below.
- (c) <u>Reemployment.</u> The Municipality shall elect in its Adoption Agreement whether a Plan Member must make-up any required Member Contributions in order to receive Credited Service under the Plan for the period of Qualified Military Service.
- (d) Make-up Member Contributions. Except as provided in subsection (i) below, a Member who is required to make-up Member Contributions in accordance with this Section must do so within the time period the begins on the Employee's Reemployment Commencement Date and ends on the date that is earliest of (i) three (3) times the period of Qualified Military Service; (ii) five (5) Years; or (iii) Termination of Employment. The Member may make-up all or a portion of any required Member Contributions and shall receive the amount of Credited Service that is directly proportionate to the percentage of Member Contributions that are made-up. The Member shall designate the Plan Year to which such Employee Contributions relate. The Municipality shall grant Credited Service based on the made-up Member Contributions in the time and manner as such Credited Service is accrued by active Members. Such make-up Member Contributions can be paid to the Plan in a lump sum, in installments or by payroll deduction.

## (e) Death During Military Service

- (i) <u>Deemed Return to Employment</u>. If a Member dies during a period of Qualified Military Service, the Member shall be treated as having returned to employment with the Municipality on the day before his death and died the next day for purposes of any survivor benefits including pre-retirement survivor benefits and any accelerated vesting. Such Member shall receive Vesting Service for the period of Qualified Military Service.
- (ii) <u>Credited Service</u>. The Municipality shall elect in its Adoption Agreement whether to grant Credited Service to any Member who dies during a period of Qualified Military Service. If the Municipality elects to grant such Credited Service and the Plan requires Member Contributions in order for a Member to receive Credited Service, the Member shall be deemed to have made the maximum amount of required Member Contributions during the period of Qualified Military Service.

## (f) <u>Disability during Qualified Military Service.</u>

The Municipality shall elect in its Adoption Agreement whether to grant Credited Service and Vesting Service if a Member becomes Disabled during a period of Qualified Military Service. If the Municipality elects to grant Credited Service to such Members, the Municipality shall also credit Vesting Service to the Member for such period. The Municipality may elect to credit Vesting Service without crediting Credited Service. However, if the Municipality elects to grant such Credited Service and the Plan requires Member Contributions in order for a Member to receive Credited Service, the Member shall be deemed to have made the maximum amount of required Member Contributions during the period of Qualified Military Service.

# (g) Compensation.

For purposes of this Section, a Member's Compensation during the period of Qualified Military Service shall be treated as equivalent to the Compensation he or she would have received during such period but for the period of Qualified Military Service. Such determination shall be based on the rate of pay the Employee would have received during that time; provided however if the Compensation the Employee would have received is not reasonably certain, Compensation for this purpose shall equal the Employee's average Compensation during the 12 months period immediately preceding the Qualified Military Service (or, if shorter, the period of employment immediately preceding the Qualified Military Service).

## (h) <u>Intervening Military Service</u>

- (i) Any Member who enters the uniformed services as defined by USERRA and returns to service as a Member within the authorized time period of the law, shall have the authorized time spent in such service credited to the Member's employment record for pension or retirement benefits if the Member makes the required member contributions.
- (ii) The amount due from the Member shall be certified by the System and shall be paid by the Member in accordance with applicable law.
- (iii) A Member may purchase credit for intervening military service only if discharge or separation from the service was granted under other than dishonorable conditions. A Member may not purchase military credit for any service that is covered by another retirement system administered and wholly or partially paid for by any other government agency or private employer.

# (i) Non-Intervening Military Service

An Active Member may purchase credit for other than intervening military service performed for the armed forces of the United as defined by USERRA, for a period not to exceed five (5) years, provided the Member has completed five (5) years of service to the Municipality subsequent to such military service. An Active Member may file an application with the Board for permission to purchase credit for non-intervening military service upon completion of five (5) years of subsequent service to the Municipality. The cost of such purchase shall not exceed an amount equal to the Actuarial Equivalent present value of the additional Accrued Benefit attributable to such purchased service as of the date of such purchase.

(i) The rate of interest charged a Member on purchase of credit for non-intervening military service shall be the Regular Interest in effect on the date of the Member's application.

(ii) An Active Member may purchase credit for non-intervening military service only if discharge or separation from the service was granted under other than dishonorable conditions. A Member may not purchase military credit for any service that is covered by another retirement system administered and wholly or partially paid for by any other government agency or private employer.

# (j) <u>Differential Wage Payments</u>

If the Municipality pays Differential Wage Payments to the Member while on Qualified Military Leave, such payments shall be treated as Compensation under the Plan. Such Differential Wage Payments shall also be treated as Compensation for purposes of Code Section 415 and Section 12.11 of the Plan.

## 11.04 Transfer of Service and Assets from Another System Plan

The Municipality may elect in its Adoption Agreement to allow Eligible Employees who have previously been Members of another System Plan to transfer Credited Service and Vesting Service and assets from other System plans to the Municipality's Plan subject to certain terms and conditions.

# (a) Limitations

- (i) The Employee shall be an Eligible Employee with the Municipality within one (1) year of Termination of Employment from the municipality for which the service is being transferred
- (ii) At the time the Employee elects to transfer service, the Employee shall be an Eligible Employee under the Plan but is not required to be a Member of the Plan at the time the Employee elects to transfer service. If not a Plan Member at the time the Employee elects to transfer service, the Employee shall subsequently become a Member after satisfying the Plan's eligibility requirements.
- (iii) The Employee shall affirmatively elect to transfer service no later than the one (1) year period specified in subparagraph (i). If however, the Employee's Employment Commencement Date is within ninety (90) days of the expiration of such one (1) year period, he will have ninety (90) days from his Employment Commencement to provide the affirmative election to transfer service.
- (iv) The affirmative election shall be made in the form and substance approved by the Board which shall (A) identify the Plan from which the Credited Service is to be transferred, and (B) specify the amount of Credited Service to be transferred. Failure to affirmatively elect in such manner within the required time limit shall prohibit the transfer of service under this Section.

# (b) Transfer of Applicable Vesting Service and Credited Service

- (i) Vesting Service will be transferred in an amount equal to the Vesting Service earned by the Employee with the previous Municipality as of his Termination of Employment with the previous Municipality.
- (ii) Credited Service will be transferred in an amount equal to the Credited Service earned by the Employee with the previous Municipality as of his Termination of Employment with the previous Municipality. A System Policy Statement that reflects the differences in the Actuarial Equivalent present values of the Accrued Benefit when the Accrued Benefit is calculated will be issued to the Member.

# (c) Transfer of Applicable Assets

- (i) Assets in the Member's Member Contribution Account of the previous Municipality shall be transferred to equivalent Member Contribution Accounts in the Municipality Plan in an amount equal to the amounts as of the date of transfer. If an appropriate Member Contribution Account type has not been established within the Municipality Plan, such an account type shall be established if needed for the appropriate accounting of any transferred Member Contributions.
- (ii) Assets in the Municipality Contribution Account from the previous Municipality shall be transferred to equivalent Municipality Contribution Account in an amount equal to the Actuarial Equivalent present value of Member's Accrued Benefit in the previous Municipality plan less the transferred amounts of the Member's Member Contribution Account referenced in paragraph (c)(i) above.
- (iii) Any transfer of service under this Section shall be treated as a purchase of Permissive Service Credit and subject to the provision of Code Section 415(n) as set forth in Section 11.09 below.

# (d) Calculation of Benefits from Transferred Service and Assets

After the transfer of service and assets, the calculation of benefits based on the transferred service and assets shall be based on the Plan provisions of the Municipality receiving such transfer. Transferred assets shall be applied to the cost of purchasing Permissive Service Credit (including Non-Qualified Service Credit) pursuant to Section 11.07 and 11.08. If, however, the assets to be transferred exceed the amount necessary for the purchase of Permissive Service Credit, such excess shall be allocated as follows:

- (i) If the Plan is a Cash Balance Plan, any excess shall be allocated to the Member's Hypothetical Account Balance pursuant to Section 5.06.
- (ii) If the Plan is a Defined Benefit Plan, the Municipality shall transfer such excess to a Cash Balance Plan established and maintained by the Municipality. If the Municipality does not maintain a Cash Balance Plan, the excess amounts shall be returned to the Member.

# 11.05 Purchase of Service Credited Under Prior Plan of the Municipality

# (a) <u>Transfer Contributions</u>

If elected by the Municipality in the Adoption Agreement, a Member may elect to utilize Transfer Contribution amounts to purchase and convert Prior Plan Service to Credited Service under this Plan. The cost of such conversion shall not exceed an amount equal to the Actuarial Equivalent present value of the additional Accrued Benefit attributable to such purchased service as of the date of such purchase. The amount of Prior Plan Service to be purchased and converted to Credited Service shall be determined at the discretion of the Member subject to the following limitations:

- (i) Transfer Contribution amounts cannot be used to purchase and convert years of Credited Service in excess of the Member's years of Prior Service;
- (ii) The Member may determine the amount of Prior Service to be purchased and converted to Credited Service provided, however that Prior Service shall be purchased and converted to Credited Service in increments authorized in the election notice;
- (iii) Transfer Contribution amounts cannot be used to purchase Prior Service that results in the Member receiving a duplicate benefit under this Plan and the Prior Plan for the same service:

- (iv) The election to convert Prior Service to Credited Service by purchase shall be made in accordance with the election notice; and
- (v) The election to convert Prior Service to Credited Service by purchase shall be irrevocable.

#### (b) Termination Contributions

If elected by the Municipality in the Adoption Agreement, Termination Contributions amounts shall be used to purchase and convert Prior Plan Service to Credited Service under this Plan. The cost of such conversion shall not exceed an amount equal to the Actuarial Equivalent present value of the additional Accrued Benefit attributable to such purchased Service on the date of purchase. The Prior Plan Service to be purchase and converted to Credited Service shall be determined as follows:

- (i) Termination Contribution amounts cannot be used to purchase and convert years of Credited Service in excess of the Member's years of Prior Service;
- (ii) Termination Contribution amounts shall be utilized to purchase and convert the maximum increment of Credited Service available for such amounts;
- (iii) Termination Contribution amounts cannot be used to purchase Prior Service that results in the Member receiving a duplicate benefit under this Plan and the Prior Plan for the same service; and
- (iv) The purchase and conversion of Prior Service to Credited Service shall be irrevocable.
- (c) Any purchase of service under this Section shall be treated as a purchase of Permissive Service Credit and subject to the provisions of Code Section 415(n) as set forth in Section 11.07 and 11.08 below. If, however, such Termination Contributions or Transfer Contributions exceed the amount necessary for the purchase of Permissive Service Credit, the excess shall be allocated to the Member's Hypothetical Account Balance under the Cash Balance Plan maintained by the Municipality.

#### 11.06 Transfer of Member Loans

Upon joining the System, a Municipality may elect in the Adoption Agreement to have Member loans transferred from the Municipality's Original Plan or Prior Plan to this Plan with other assets transferred from the Original Plan or Prior Plan. Such transfer shall be permitted only upon a determination by the Board that the loans to be transferred satisfy the provisions of Code Section 72(p) and Treas. Reg. 1.72(p)-1. In the event the Member fails to repay the outstanding loan in accordance with the loan agreement, then the outstanding loan amount shall be treated as a taxable distribution.

# 11.07 Permissive Service Credit

A Municipality may elect in the Adoption Agreement to allow a Member to purchase certain types of Permissive Service Credit pursuant to the requirements of Section 415(n) of the Code as set forth in Section 11.09.

# (a) Limitations

- (i) The Municipality shall elect in its Adoption Agreement the maximum amount of Permissive Service Credit a Member may purchase.
- (ii) The Municipality shall elect in its Adoption Agreement the types of Permissive Service Credit the Member may purchase.

(iii) A Member may not purchase Permissive Service Credit for any service that is covered by another retirement system administered by an employer for which Permissive Service Credit is otherwise allowed to be purchased.

# (b) Cost

The cost to purchase Permissive Service Credit shall be an amount equal to the Actuarial Equivalence attributable to such service as of the service purchase date.

Subject to the Code, the Municipality may also limit the type and amount of Permissive Service Credit in the Adoption Agreement.

# 11.08 Nonqualified Service Credit

A Municipality may elect in the Adoption Agreement to allow a Member to purchase Nonqualified Service Credit pursuant to the requirements of Section 415(n) of the Code as set forth in Section 11.09. The cost to purchase Nonqualified Service Credit shall be an amount equal to the Actuarial Liability attributable to such service as of the service purchase date.

# 11.09 Service Purchases under Code Section 415(n)

- (a) If a Member makes one or more contributions to purchase Permissive Service Credit or Nonqualified Service Credit under the Plan, then the requirements of Section 415(n) of the Code will be treated as satisfied only if:
  - (i) the requirements of Section 415(b) are met, determined by treating the Accrued Benefit derived from all such contributions as an annual benefit for purposes of Section 415(b) of the Code; or
  - (ii) the requirements of Section 415(c) of the Code are met, determined by treating all such contributions as annual additions for purposes of Code Section 415(c).

## (b) Permissive Service Credit

- (i) For purposes of this Plan, the term "Permissive Service Credit" means service credit:
  - (A) recognized by the Plan for purposes of calculating a Member's benefit under the Plan:
  - (B) that the Member has not received under the Plan; and
  - (C) that the Member may receive only by making a voluntary additional contribution, in an amount determined under the Plan, that does not exceed the amount necessary to fund the benefit attributable to the service credit.

The term may include service credit for periods for which there is no performance of service and may include service credited in order to provide an increased benefit for service credit a member is receiving under the plan.

- (ii) The Plan will fail to meet the requirements of this Section if:
  - (A) more than five (5) years of "Nonqualified Service Credit" as defined below are taken into account for purposes of this Section; or
  - (B) any Nonqualified Service Credit is taken into account under this Section before the Member has at least five years of Membership under the Plan.
- (iii) The term "Nonqualified Service Credit" means Permissive Service Credit other than that allowed with respect to:

- (A) service (including parental, medical, sabbatical and similar leave) as an employee of the Government of the United States, any state or political subdivision of the United States, or any agency or instrumentality of any of the foregoing (other than military service or service for credit that was obtained as a result of a repayment described in Code Section 415(k)(3));
- (B) service (including parental, medical, sabbatical and similar leave) as an employee (other than as an employee described in subsection (e)(1)) of an education organization described in Code section 170(b)(1)(A)(ii) that is a public, private or sectarian school that provides elementary or secondary education (through grade 12), or a comparable level of education, as determined under the applicable law of the jurisdiction in which the service was performed;
- (C) service as an employee of an association of employees who are described in subparagraph (A) above; or
- (D) military service (other than qualified military service under IRC section 414(u)) recognized by the Plan.

In the case of service described in subparagraphs (A), (B) and (C) above such service will be nonqualified service if recognition of the service would cause a Member to receive a pension benefit for the same service under more than one plan.

- (iv) In the case of a Transfer Contribution to which Code Section 403(b)(13)(A) or 457(e)(17)(A) applies (without regard to whether the transfer is made between plans maintained by the same Municipality):
  - (A) the limitations of paragraph (i) will not apply in determining whether the transfer is for the purchase of Permissive Service Credit; and
  - (B) the distribution rules of the Plan will apply to the Transfer Contribution and any benefits attributable to those amounts.

# ARTICLE XII: MISCELLANEOUS PROVISIONS AFFECTING THE PAYMENT OF BENEFITS

# 12.01 Non-Duplication of Benefits

If the Board distributes any part or all of a Member's Accrued Benefit to him and the Member is later rehired by the Municipality before or after Retirement, the Board shall compute the Member's Accrued Benefit by taking into account all of the Member's Credited Service. However, the Board shall offset the Member's Accrued Benefit so computed by the Member's Accrued Benefit attributable to any distribution described in Article X of the Plan and by any Accrued Benefit disregarded under Section 8.04 of the Plan.

# 12.02 Form of Payment of Benefits

Benefit payments, including lump sum payments, will be made by direct deposit to a financial institution specified by the Member. Exceptions may be granted upon request for Members who do not have an account at a financial institution.

#### 12.03 Merger of Plan

Neither the Municipality nor the Board shall consent to, or be a party to, any merger or consolidation of the Plan with another plan, or to a transfer of assets or liabilities to another plan, unless immediately after the merger, consolidation or transfer, the surviving Plan provides each Member a benefit equal to or greater than the benefit each Member would have received had the Plan terminated immediately before the merger or consolidation or transfer. However, the Board possesses the specific authority to enter into a merger agreement or a direct transfer of assets agreements with the trustees of other retirement plans described in Code Section 401(a) and to accept the direct transfer of plan assets, or to transfer plan assets, as a party to any such agreement.

The Board may accept a direct transfer of plan assets on behalf of an Employee prior to the date the Employee satisfies the Plan's eligibility condition(s). If the Board accepts such a direct transfer of plan assets, the Board shall treat the Employee as a Member for all purposes of the Plan except the Employee may not make Member Contributions under Sections 4.01 or 4.02 of the Plan, nor shall the Employee accrue benefits, including any minimum Superannuation Retirement Pension provided for, until he actually becomes a Member in the Plan.

# 12.04 Payments to Minors or Legally Incompetent Persons

Whenever any benefit is to be paid to or for the benefit of any person who is a minor or determined to be incompetent by qualified medical advice, the Board shall cause the benefit to be paid to the person having custody of the minor or incompetent or to the legal guardian or custodian appointed by the Member or a court of competent jurisdiction.

# 12.05 Unclaimed Payments

The Board shall take reasonable steps, as appropriate, to locate a Member entitled to Plan benefits. Such steps shall include: (i) use of certified mail; (ii) verification through related Plan records and, as appropriate, Municipality records (such as group health plans); (iii) contact of Member's designated Beneficiaries; and (iv) use of free electronic search tools including internet search engines, public record databases (such as those for licenses, mortgages and real estate taxes), obituaries and social media.

If the above attempts to locate a Member have failed, the Board may direct that the payment and all remaining payments otherwise due to the Member allocated to an appropriate Reserve Account at the discretion of the Board. If the Member later notifies the Board of his whereabouts and

requests the payments due to him, an amount equal to the undistributed amount to be paid to him as soon as administratively feasible.

# 12.06 Assignment or Alienation

A Member, Survivor Annuitant or Beneficiary shall not anticipate, assign or alienate (either at law or in equity) any benefit provided under the Plan, and the Board shall not recognize any such anticipation, assignment or alienation, subject to the following exceptions (a) federal tax liens or other applicable garnishment under federal law, (b) an assignment of Plan benefits for the provision of health care premiums, or (c) an Approved Domestic Relations Order, or (d) a trustee-to-trustee transfer of a Member's accrued benefit in accordance with Section 12.02 of the Plan. Furthermore, a benefit under the Plan is not subject to attachment, garnishment, levy, execution or other legal or equitable process.

# 12.07 No Decrease in Benefits by Change in Social Security

In the case of a Member or Beneficiary who is receiving benefits under this Plan or a Member who has Terminated Employment with the Municipality and has a vested Accrued Benefit under this Plan, any increase in the taxable wage base or the benefit level payable under Title II of the Social Security Act shall not affect the way benefits are payable under this Plan to such Member or Beneficiary. The Plan does not permit the recalculation of any benefits accrued before the Termination of Employment of a Member on the basis of changes in Social Security benefit levels or the taxable wage base in effect after reemployment with the Municipality.

# 12.08 Limitation on Benefits

#### (a) General Rule

The annual benefit under this Plan payable to a Member at any time shall not exceed the maximum permissible amount. "Maximum permissible amount" shall mean \$160,000, such limitation to be adjusted automatically as determined by the Internal Revenue Service for each calendar year, with the new limitation to apply to limitation years ending within the calendar year of the date of the adjustment.

#### (a) Annual Benefit

#### (i) In General

For purposes of subsection (a), the term "annual benefit" means a benefit payable annually in the form of a straight life annuity (with no ancillary benefits) under a plan to which employees do not contribute and under which no rollover contributions (as defined in Code sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3), and 457(e)(16)) are made.

# (ii) Adjustment for Certain Other Forms of Benefit

If the benefit under the Plan is payable in any form other than the form described in paragraph (i), or if the Member contributes to the plan or make rollover contributions (as defined in sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3), and 457(e)(16)), the determinations as to whether the limitation described in subsection (a) has been satisfied shall be made, in accordance with regulations by adjusting such benefit so that it is equivalent to the benefit described in subparagraph (i). For purposes of this subparagraph, any ancillary benefit which is not directly related to retirement income benefits shall not be taken into account; and that portion of any joint and survivor annuity which constitutes a qualified joint and survivor annuity (as defined in section 417) shall not be taken into account.

(iii) Adjustment to \$160,000 Limit Where Benefit Begins Before Age 62

If the retirement income benefit under the plan begins before age 62, the determination as to whether the \$160,000 limitation set forth above has been satisfied shall be made, in accordance with regulations prescribed by the Internal Revenue Service, by reducing the limitation so that such limitation (as so reduced) equals an annual benefit (beginning when such retirement income benefit begins) which is equivalent to a \$160,000 annual benefit beginning at age 62.

(iv) Adjustment to \$160,000 Limit Where Benefit Begins After Age 65

If the annual benefit under the Plan begins after age 65, the determination as to whether the \$160,000 limitation set forth above has been satisfied shall be made, in accordance with regulations, by increasing the limitation so that such limitation (as so increased) equals an annual benefit (beginning when such retirement income benefit begins) which is equivalent to a \$160,000 annual benefit beginning at age 65.

- (v) Limitation on Certain Assumptions
  - (A) For purposes of adjusting any limitation under paragraph (iii) and, for purposes of adjusting any benefit under subparagraph (ii), the interest rate assumption shall not be less than the greater of 5 percent or the rate specified in the Plan.
  - (B) For purposes of adjusting any limitation under paragraph (iv), the interest rate assumption shall not be greater than the lesser of 5 percent or the rate specified in the Plan.
  - (C) For purposes of this Section, no cost of living adjustment under Code Section 415(d)(1) shall be taken into account before the year for which such adjustment first takes effect
  - (D) For purposes of adjusting any benefit or limitation under paragraph (ii), (iii) or (iv), the mortality table used shall be the applicable mortality table (within the meaning of Code Section 417(e)(3)(B)).
- (vi) Special Limitation for Qualified Police or Firefighters

In the case of a qualified participant, paragraph (iii) of this subsection shall not apply.

(vii) Qualified Participant Defined

For purposes of paragraph (vi), the term "qualified participant" means a participant with respect to whom the period of service taken into account in determining the amount of the benefit under such defined benefit plan includes at least 15 years of service of the participant:

- (A) as a full-time employee of any police department or fire department which is organized and operated by the State, Indian tribal government (as so defined), or any political subdivision maintaining such defined benefit plan to provide police protection, firefighting services, or emergency medical services for any area within the jurisdiction of such State, Indian tribal government (as so defined), or any political subdivision, or
- (B) as a member of the Armed Forces of the United States.
- (viii) Exemption for Survivor and Disability Benefits.

Paragraph (iii) of this subsection and subsection (v) shall not apply to:

- (A) income received from a governmental plan (as defined in section 414(d)) as a pension, annuity, or similar allowance as the result of the recipient becoming disabled by reason of personal injuries or sickness, or
- (B) amounts received from a governmental plan by the beneficiaries, survivors, or the estate of an employee as the result of the death of the employee.

# (b) Total Annual Benefits Not in Excess of \$10,000

Notwithstanding the preceding provisions of this Section 12.09, the benefits payable with respect to a Member under the Plan shall be deemed not to exceed the limitation of this Section if:

- (i) the retirement benefits payable with respect to such Member under such plan and under all other defined benefit plans of the Municipality do not exceed \$10,000 for the Plan Year, or for any prior Plan Year, and
- (ii) the Municipality has not at any time maintained a defined contribution plan in which the Member participated.
- (c) Reduction for Membership of less than 10 years
  - (i) Dollar Limitation

In the case of a Member who has less than 10 years of Membership in the Plan, the limitation referred to in subsection (a)(i) shall be the limitation determined under such subsection (without regard to this subsection) multiplied by a fraction:

- (A) the numerator of which is the number of years (or part thereof) of participation in the defined benefit plan of the employer, and
- (B) the denominator of which is 10.
- (ii) Limitation on Reduction

In no event shall paragraph (i) reduce the limitations referred to in subsection (a) to an amount less than 1/10 of such limitation (determined without regard to this paragraph).

# (d) Plan Aggregation

If the Municipality maintains both a Defined Benefit Plan and Cash Balance Plan, the limits set forth in this Section shall apply to the aggregate Accrued Benefit of such plans.

#### ARTICLE XIII: ADMINISTRATIVE PROVISIONS

# 13.01 General Powers of the Board

In the administration of the System, the Board has all power granted to it under Section 104 of the PMRL, as set forth in Addendum A, including but not limited to:

- (a) Entering into agreements for professional services, including but not limited to actuarial, investment and medical as it deems advisable;
- (b) Keep in convenient form such data as shall be deemed necessary for actuarial valuation purposes;
- (c) From time to time, through its Actuary, make an actuarial investigation into the mortality and service experience of the Members and Annuitants and of the various accounts created by 16 Pa. Code §81.7 (see Addendum A);
- (d) Adopt for the Base Plan one or more mortality tables and such other tables as shall be deemed necessary;
- (e) Amend the Base Plan Document as necessary and appropriate to comply with the Code or adopt administrative changes;
- (f) Adopt Policy Statements as necessary to administer the System and establish rules regarding the operation of the System. Policy Statements shall remain in full force and effect until superseded or otherwise revoked by the Board;
- (g) Certify annually the amount of appropriation which the Municipality shall pay into the Fund, which amounts shall be based on estimates furnished by the Actuary;
- (h) Prepare and distribute annual statements of accounts to each of the Active Members and annuitants showing the contributions made during the year, the interest earned and the total balance standing in the Member's Contribution Account at the end of the year; and
- (i) Perform such other functions as are required for the execution of any other Federal and State law and to administer and interpret this Plan to ensure that the System is maintained as tax qualified under the IRC.

The Board shall discharge its duties and obligations above solely in the interest of the participants and beneficiaries and with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person, acting in a like capacity and familiar with such matters, would use in the conduct of an enterprise of a like character and with like aims.

# 13.02 Policy Statements

- (a) All Policy Statement set forth in Addendum B, shall be deemed incorporated herein by reference regardless of its adoption date and shall be binding on the Municipality to the extent applicable.
- (b) All Policy Statement may be retitled, amended, or supplemented from time to time by written notice to the Municipality.
- (c) The Board shall provide written notice, which may be email, to the Municipality within ninety (90) days of enactment of any new Policy Statement.
- (d) The above notwithstanding, a Policy Statement shall not be applicable to the extent inconsistent with the Base Plan Document.

# 13.03 Plan Withdrawal

- (a) The Municipality may, for good cause, file an application with the Board for permission to withdraw from the system if it meets all of the following requirements:
  - (i) The Municipality has been enrolled in the System for a period of at least five years.
  - (ii) The Municipality has met all of its financial obligations to the System.
  - (iii) The Municipality has passed an ordinance or resolution signifying its intention to withdraw from the System.
  - (iv) The Municipality has certified to the Board that an affirmative vote approving withdrawal from the System had been obtained from at least seventy-five percent (75%) of all Active Members, Inactive Members, vested Members, Annuitants, and Survivor Annuitants after providing such Members with the information specified in Section 13.03(e) of the Base Plan Document.
  - (v) The Municipality has acknowledged its responsibility to assume and provide for all future benefit payments to the existing Active Members, Inactive Members, vested Members, Annuitants, Survivor Annuitants, and their Beneficiaries effective upon the withdrawal.
  - (vi) The application has specified a date for the withdrawal to become effective, provided if there are Annuitants and/or Survivor Annuitants in active pay status the effective date is the first day of a month no earlier than the month after which the Board is scheduled to take action on the withdrawal application.
- (b) In the event that the Municipality elects to withdraw from the System, a Member's basic benefit under this Plan shall be immediately vested to the extent funded as of the effective date of the withdrawal.
- (c) In the event the Municipality withdraws from the System, the Municipality shall only be entitled to the assets credited to the Municipal Account and the Members' Accounts in accordance with the provisions of the PMRL and Policy Statements. Assets that are actuarially determined by the Board's Actuary to be matched to the Municipality's Annuitants and Survivor Annuitants will also be returned to the Municipality in accordance with the PMRL and Policy Statements.
- (d) If the Municipality intends to continue the Plan in substantially the same form with a successor to the System, the Board shall determine the amount of Plan assets attributable to the Municipality in accordance with policies and procedures established by the Board and provided in the Trust Agreement or a separate written instrument. After the determination of the assets attributable to the Municipality, the Board, with consent of the Municipality, may apply the assets of the portion of the Trust attributable to the Municipality to either:
  - (i) Purchase deferred paid-up annuities for all vested accrued benefits of all Plan Members;
  - (ii) Purchase deferred paid-up annuities for all vested accrued benefits of Plan Members who are former Employees of the Municipality;
  - (iii) Purchase deferred paid-up annuities for all accrued benefits of retired Employees of the Municipality; or
  - (iv) Not purchase any deferred paid-up annuities.

Any remaining assets of the Trust attributable to the Municipality, shall, after satisfaction of the requirements of this Section, be transferred to the successor trustee designated by the Municipality, as soon as practical after the later of the December 31st or ninety (90) days following the date of determination of any such remaining assets.

- (e) Prior to the Board transferring any assets to a successor to the System, the Municipality shall warrant and certify to the Board, in a manner satisfactory to the Board, the following:
  - (i) The successor trustee has been duly appointed by the Municipality;
  - (ii) The plan document pursuant to which the Plan will continue to be maintained, is qualified under Code Section 401(a);
  - (iii) The assets of the Plan will continue to be held by the successor trustee for the exclusive benefit of Plan Members and their Beneficiaries;
  - (iv) The successor plan document does not reduce the Accrued Benefit, vested percentage or existing benefit payment options of any Plan Member;
  - (v) The Municipality has met all minimum funding requirements applicable to the Plan in accordance with the Board's funding policy; and
  - (vi) To the extent permitted by federal, state or local law, the Municipality agrees to indemnify the Board from and against any loss, liability or claim arising out of the Municipality's maintenance of the Plan from and after the date of withdrawal from the Trust.

#### **13.04** Notice

Any notice, demand, direction, instruction, or other communication required or permitted ("Notice") to the Board hereunder shall be confirmed in writing and shall be sufficiently given for all purposes when sent by:

- (a) registered U.S. Postal Service delivery, postage prepaid, or
- (b) facsimile, with a copy sent by 1<sup>st</sup> Class U.S. Postal Service delivery (provided that if the date of dispatch is not a working day, the facsimile shall be deemed to have been received at the opening of business of the addressee on the next working day),
- (c) email to any party at the email address below, or
- (d) delivering the same in person to any party at the applicable address, or
- (e) any other method or other address as may be designated in writing from time to time by the parties:

# PENNSYLVANIA MUNICIPAL RETIREMENT BOARD

Attn: Secretary

For Registered U.S. Postal Service delivery; for overnight/courier service delivery; and for in-person delivery:

1721 N. Front Street, 3rd Floor Harrisburg, PA 17102-2315

For 1st Class U.S. Postal Service delivery:

P.O. Box 1165

Harrisburg, PA 17108-1165

Fax: (717) 783-8363

Email: RA-Staff@state.pa.us

Any Notice to the Municipality shall be as provided in the Adoption Agreement.

#### 13.05 Information to Board

The Municipality shall supply current information to the Board as to the name, date of birth, Employment Commencement Date, annual Compensation, Leaves of Absences, Vesting, Eligibility, and Credited Service and date of Termination of Employment of each Employee who is, or who will be eligible to become, a Member under the Plan, together with any other information which the Board considers necessary. The Municipality's records as to the current information the Municipality furnishes to the Board shall be conclusive as to all persons.

The following documents will be accepted as documentation of proof of an individual's date of birth in the following order of preference: birth certificate, baptismal certificate showing the date of birth, selective service record, armed forces discharge papers, passport, Pennsylvania Department of Transportation-issued photo ID (need not be current), military ID (DD Form 2) or its equivalent, school records, naturalization record, alien registration records, other records as may be submitted by the Member which are acceptable to the Board.

# 13.06 Liability

The Municipality shall hold and save harmless the Commonwealth, the Board, its officers, agents and employees and the System from any and all claims, demands, actions or liability of any nature based upon or arising out of any services performed by its employees, servants and agents under this Plan in a manner which is clearly contrary to the written direction of the System or the Board and shall defend any and all actions brought against the System or the Board based upon such claims or demands.

## 13.07 Amendment to Vesting Schedule

Although the Municipality reserves the right to amend the vesting schedule in the Adoption Agreement at any time, the Municipality shall not apply the amended vesting schedule to reduce the Vested percentage of any Member's Accrued Benefit derived from Municipality contributions (determined as of the later of the date the Municipality adopts the amendment, or the date the amendment becomes effective) to a percentage less than the Vested percentage computed under the Plan without regard to the amendment.

#### 13.08 Information to Municipalities

The Board shall, with the aid of the Actuary, prepare a circular of information relative to the system, showing estimates of the costs thereof to municipalities and Members, including costs of administration, the benefits to be derived therefrom, the methods of administration by the Board and the municipality, and such other information as may be deemed appropriate. Such circular of information shall be furnished to municipalities upon request.

#### ARTICLE XIV: MEMBER ADMINISTRATIVE PROVISIONS

# 14.01 Beneficiary Designation

Any Member may from time to time designate, in writing or via secure electronic communication, any person or persons, continentally or successively, to whom the Board shall pay various death benefits provided under the Plan on event of his death. The Board shall prescribe the form or forms, for the designation of Beneficiaries and, upon the Member's filing the form with the Municipality or Board, it effectively shall revoke all designations filed prior to that date by the same Member. Beneficiary designations may be made and/or maintained electronically, if the Municipality or the Board has established a method that is reasonably calculated to provide accurate results.

# 14.02 **No Beneficiary Designation**

If a Member fails to name a Beneficiary in accordance with Section 14.01 of the Plan, or if the Beneficiary named by a Member predeceases him or dies before complete distribution of all benefits payable under the Plan, then the Board shall pay such benefits to the Member's estate or next of kin under 20 Pa.C.S. § 3101 (relating to payments to family and funeral directors (see Addendum A) to the extent applicable.

If payment cannot be made as specified above, the Member's benefits shall remain a part of the Municipality's pension assets until his payment can be processed pursuant to this Section 14.02.

# 14.03 Personal Data to Board

Each Member and each Beneficiary of a deceased Member must furnish to the Board such evidence, data or information as the Board considers necessary or desirable for the purpose of administering the Plan. The provisions of this Plan are effective for the benefit of each Member upon the condition precedent that each Member will furnish promptly full, true and complete evidence, data and information when requested by the Board, provided the Board shall advise each Member of the effect of his failure to comply with its request.

#### 14.04 Information Available

Any Member in the Plan or any Beneficiary may examine copies of the Plan, latest annual report, any bargaining agreement, Adoption Agreement, or any other instrument under which the Plan was established or is operated. The Board will maintain all of the items listed in this Section 14.04 in its offices, or in such other place or places as it may designate from time to time in order to comply with all applicable regulations, for examination during reasonable business hours. Upon the written request of a Member or Beneficiary the Board shall furnish him with a copy of any item listed in this Section 14.04. The Board may make a reasonable charge to the requesting person for the copy so furnished. The Board may provide Members with any information required under any applicable federal or state law via electronic communication, provided the electronic communication is not prohibited under such laws and the method of electronic communication is reasonably calculated to provide accurate results. A Beneficiary's right to (and the Board's duty to provide to the Beneficiary) information or data concerning the Plan, shall not arise until the Beneficiary first becomes entitled to receive a benefit under the Plan.

#### 14.05 Benefits Claims Procedures

(a) The Board is statutorily vested with the exclusive jurisdiction to administer and interpret this Plan. The parties hereby waive any claim or defense that forum before the Board is not convenient or proper.

- (b) If a Member and/or Municipality believe that a right or benefit under this Plan has been improperly denied, a written request for a final determination must be filed with the Board.
- (c) If the Member and/or Municipality disagrees with the final determination reached by the Board. The Member and/or the Municipality has the right to file an appeal and request an administrative hearing before the Board. See 1 Pa. Code §§ 31.11, 31.15, 33.31, 33.34, and 35.20. See Addendum A. The appeal and request for an administrative hearing must include a statement of the facts that forms the basis of the requested relief and all points of law that support the claim.
- (d) An administrative hearing is a process in which evidence is presented before an independent hearing examiner who makes a recommendation to the Board, following which the Board renders a final adjudication. All administrative hearings are conducted at the Board's headquarters in Harrisburg, Pennsylvania. The Board's final adjudication is subject to an appeal before the Commonwealth Court of Pennsylvania.
- (e) All appeals and requests for administrative hearings must be made in writing within 30 days of the date of the final determination and sent to:

# Appeal Docket Administrator Pennsylvania Municipal Retirement Board 5 North Fifth Street Harrisburg, PA 17110

(f) If the appeal and request for an administrative hearing is not received by the Board within the thirty (30) day period, the Member and/or Municipality will no longer have the right to administratively appeal the denial of its request to the Board.

# 14.06 Filings with the Board

# (a) Receipt

Any filing, election or notice required to be made under this Plan by a Member must be made by written statement, duly attested, and filed in the office of the Board or deposited in the United States mail, addressed to the Board.

# (b) <u>Timeliness</u>

The timeliness of any filing, election or notice required to be made under this Plan to the Board by a Member shall be governed by the actual receipt of the filing, election or notice rather than the date of mailing.

## (c) Municipal Certifications

- (i) In the event the Board receives an application for a benefit from a Member in which certification by the Municipality with regard to the data used to calculate such benefit is needed but such certification has not been made by the Municipality, the Board shall issue a formal demand to the Municipality for such certification of the necessary data.
- (ii) Any certification required to be made by the Municipality with regard to any application for benefits by a Member under this Plan shall be made by the Municipality within forty-five (45) days following the date of the Board's formal demand.
- (iii) Failure of the Municipality to provide such requested certification within the forty-five (45) day period shall result in the Board utilizing the most current reliable data maintained by the Board for the Member.

#### ARTICLE XV: PENNSYLVANIA MUNICIPAL RETIREMENT FUND

# 15.01 General

The Pennsylvania Municipal Retirement Fund shall consist of (i) contributions by the Municipalities; (ii) contributions by Members through the employing Municipality; (iii) transfers made from municipal retirement plans or pension systems that are tax qualified under Section 401(a) of the Code; and (iv) investment earnings.

# 15.02 **Separate Accounts**

The Board shall keep separate accounts for each Municipality, except the Disability Reserve Account and the Retired Member's Reserve Account which shall be maintained as pooled accounts, as described in 16 Pa. Code §81.7 (attached in Addendum A). The Municipality and the Members shall be liable to the Board for the amount of contributions required to cover the cost of the Vested Accrued Benefit and other benefits payable to such Members.

# 15.03 <u>Disability Pension</u>

Upon the granting of a Disability Retirement Pension, there shall be transferred to Retired Member's Reserve Account the amount of the Disabled Annuitant's Member's Contribution Account, together with an amount from the Municipal Contribution Account, equal to the amount of the Actuarial Equivalent of the Member's Vested Accrued Benefit as of the Benefit Commencement Date of the Disability Retirement Pension. If the total amount transferred from the applicable Member Contribution Account and Municipal Account is less than the Actuarial Equivalent of the Disability Retirement Pension, the difference in amount shall be transferred from the Disability Reserve Account to the Retired Member's Reserve Account.

# 15.04 Superannuation, Early and Deferred Vested Pension

Upon the granting of a Superannuation, Early or Deferred Vested Pension, there shall be transferred to the Retired Member's Reserve Account the amount of the Member's Account, plus an amount from the Municipal Account such that the sum will equal the Actuarial Equivalent of the Superannuation, Early or Deferred Vested Pension.

# 15.05 Municipal Liability

The Regular Interest charges payable and the creation and maintenance of the necessary reserves for the payment of the benefits, as to the Municipality and its Members are hereby made the obligation of the Municipality in accordance with PMRL Section 111 and shall not otherwise attach to those assets of the Fund attributable to any other municipality and its members. See Addendum A

# 15.06 Exclusive Benefit

The Fund is a trust, and the assets of the Plan are held in trust. No part of the assets of the Plan shall be used for or diverted to purposes other than for the exclusive benefit of the Members, their Survivor Annuitant or Beneficiaries prior to the satisfaction of all liabilities of the Plan with respect to them. The assets of the Fund shall be used only to pay:

- (a) Benefits to Members in accordance with the Plan; and
- (b) Necessary and appropriate expenses of the Plan.

# 15.07 <u>Custody of the Fund</u>

All moneys and securities in the Fund shall be placed in the custody of the State Treasurer as is required by PMRL Section 109. See Addendum A.

# 15.08 Management and Investment of the Fund

The members of the Board shall be trustees of the Fund, and shall have the exclusive management of said Fund, with full power to invest the moneys therein, subject to the terms, conditions, limitations and restrictions imposed by law upon fiduciaries. The Board shall have power to hold, purchase, sell, assign, transfer and dispose of any securities and investments in said Fund, as well as the proceeds of such investments, and of the money belonging to such Fund.

The members of the Board shall discharge its duties and obligations solely in the interest of the participants and beneficiaries and with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person, acting in a like capacity and familiar with such matters, would use in the conduct of an enterprise of a like character and with like aims. For the avoidance of doubt, nothing in this Section 15.08 shall prevent the Board from contractually requiring plan fiduciaries to perform services in accordance with a heightened fiduciary standard.

# 15.09 Allocation of Regular Interest

The Board shall annually allocate Regular Interest to the credit of each Member's Contribution Account, Municipal Account, and the Retired Member's Reserve Account.

# 15.10 Allocation of Excess Interest

- (a) The Board may, after deducting money to pay for Regular Interest and any appropriate expenses of each Plan, award Excess Interest to the Municipality in accordance with the terms of this Plan.
- (b) In the event that the Municipality's funding status in relation to the benefit structure set forth under this Plan is less than 95% funded as of the most recent actuarial report filed pursuant to the Municipal Pension Plan Funding Standard and Recovery Act, 53 P.S. § 895.101 et seq., (See Addendum A) all Excess Interest awarded by the Board shall be allocated to the Municipal Account.
- (c) In the event that the Municipality's funding status in relation to the benefit structure set forth under this Plan is 95% funded or more as of the most recent actuarial report filed pursuant to the Municipal Pension Plan Funding Standard and Recovery Act, 53 P.S. § 895.101 et seq., (See Addendum A) Excess Interest awarded to the Municipality shall be allocated as elected by the Municipality in its Adoption Agreement.

#### ARTICLE XVI: AMENDMENT

# 16.01 Amendment by Board

The Base Plan Document may be amended at any time and from time to time, in the sole discretion of the Board as sponsor of the Base Plan, by a written instrument. Each amendment shall state the date to which it is either retroactively or prospectively effective. Any amendment which is required by the Internal Revenue Service in order for the Plan or Trust to qualify or continue to be qualified under the applicable provisions of the Code, or which in the judgment of the Board is necessary or appropriate to such qualification or continued qualification, may be made effective retroactively. Each Municipality, by executing an Adoption Agreement, shall thereby delegate to the Board the power to amend its Plan and shall thereby be deemed to have assented to any such amendment. Municipalities will be notified in writing of any Plan amendments made by the Board.

# 16.02 **Amendment by Municipality**

The Municipality may amend its Plan from time to time under the circumstances provided below. Each Amendment shall specify its effective date and unless otherwise specified, Amendments are assumed to be applied prospectively from its effective date. The above notwithstanding, no Amendment shall be effective until approved as to form and content by the Board. A resolution, in absence of a completed and approved Adoption Agreement amendment, shall not constitute an amendment of the Plan.

# (a) Adoption Agreement Amendment

The Municipality may amend its Plan by changing its election of options in the Adoption Agreement, in the manner it deems necessary or advisable. The Municipality shall identify to the Board any information related to the changes the Board may require.

## (b) Application of Amendments

An Amendment shall be considered to be of "General Application" or "Limited Application" as described below:

#### (i) General Application

Amendments of General Application, as of the Effective Date of the Amendment shall apply to:

- (A) all current and future Eligible Employees; and
- (B) all past and future Eligibility Service, Vesting Service and Credited Service.

#### (ii) Limited Application

Amendments of Limited Application shall be all other Amendments not of General Application. Amendments of Limited Application shall, at a minimum, include:

- (A) the specific subset of Employees, as identified in the Adoption Agreement, if applicable, which will be affected; and
- (B) the specific Service, if applicable, which will be affected.

# (c) Impact on Tax-Qualified Status

For purposes of reliance on the advisory letter issued pursuant to Rev. Proc. 2015- 36, the Board will no longer have the authority to amend the Plan on behalf of a Municipality (i) as of the date of the adoption of a Municipality amendment to the Plan to incorporate a feature described in Section 16.03 of Rev. Proc. 2015-36: or (ii) as of the date the IRS notifies the

Board that the Plan is being treated as an individually designed plan pursuant to Section 24.03 of Rev. Proc. 2015-36.

# 16.03 Limitations on Amendments

No amendment shall be made that would jeopardize the qualified status of the Plan.

No amendment shall authorize or permit any portion of the Trust Fund (other than the part which is required to pay investment or administration expenses) to be used for or diverted to purposes other than for the exclusive benefit of the Members or their Beneficiaries.

No amendment shall have the effect of decreasing a Member's Accrued Benefit, including a change in the actuarial assumptions or in the Compensation levels used to determine a Member's Superannuation Retirement Pension.

No amendment shall affect the rights, duties, or responsibilities of the Board without the written consent of the Board. Additionally, no such amendment shall be enacted or approved that impairs or is otherwise violative of a Member's rights and protections under the Pennsylvania Constitution.

No amendment shall take effect without the consent and approval of the Board.

# 16.04 <u>Termination or Freeze by Municipality</u>

By adopting the Plan, the Municipality represents that the Plan is intended to be a permanent and continuing program for providing benefits to the Members therein. The Municipality shall have the right to terminate its Plan upon not less than ninety (90) days advance written notice to the Board. A termination of the Municipality's Plan shall be effective as of the last day of the Plan Year, which follows receipt of the notice of termination (unless the Municipality and Board agree upon a different date).

#### (a) Complete Plan Termination

If Municipality intends to completely terminate the Plan, the Board shall determine the amount of Plan assets attributable to the Municipality in accordance with the terms of the Plan and Trust. The Board may use the Trust assets attributable to the Municipality to purchase deferred paid-up annuities for the benefits described in (b)(i) through (iii) below.

# (b) Annuities Purchased

If any annuities are purchased by the Board in accordance with subsections (a) or (b) hereof, at the option of the Board, such annuities shall be either (i) held by the Board, (ii) transferred to a successor trustee, or (iii) transferred to the applicable Employee or Beneficiary.

# 16.05 Distribution Upon Termination of Trust

If the Municipality terminates the Plan, the Board shall determine the amount of Plan assets attributable to the Municipality in accordance with the terms of this Plan and Trust. The Board shall allocate assets of the Plan among the Members and Beneficiaries according to the following priorities:

- (a) Benefits payable as an annuity, in the case of the benefit of a Member or Beneficiary which was in pay status as of the termination date of the Plan, each such benefit, based on the provisions of the Plan under which such benefit would be the least;
- (b) All other Vested benefits under the Plan; and

(c) Any other benefits under the Plan.

If assets are insufficient to provide all benefits under the Plan, the Board shall allocate such assets to satisfy obligations within each category by order of priority. If assets are insufficient to provide all benefits under a priority category, the Board shall allocate assets to Members within that category in the ratio which each Member's total benefit bears to the total benefits of all Members within that category.

# 16.06 Full Vesting on Termination

Notwithstanding any other provision of this Plan to the contrary, upon either full or partial termination of the Plan or the discontinuance of contributions or benefit accruals under the Plan (i.e. a freeze), under Sections 16.04 or 16.05 of the Plan, the Accrued Benefit of those Members, Beneficiaries, Inactive Members and joint annuitants affected shall become one hundred percent (100%) vested and Vested to the extent funded.

#### ARTICLE XVII: MISCELLANEOUS PROVISIONS

# 17.01 Use of Trust Fund

The terms of the Plan shall govern the establishment of the Trust Fund from which the benefits provided by the Plan shall be paid. All contributions paid over to the Board shall be invested in accordance with the terms of the Plan and Trust.

# 17.02 Amount of Municipality Contributions

Each Municipality shall contribute to the Trust Fund such amounts as are necessary to fund its respective benefits under the Plan, and shall contribute such additional amounts as the Board (based on the recommendation of the Actuary and Board) deem necessary or desirable to maintain the actuarial soundness of the Plan. The Board may establish a formal funding policy for this purpose.

# 17.03 <u>Use of Forfeitures</u>

Forfeitures and investment income attributable to contributions shall be used to reduce Municipality contributions and shall not be used hereunder to increase the benefit of any person.

# 17.04 Payment of Contributions

Contributions shall be paid by the Municipality to the Board or the Trust Fund manager. The Board or Trust Fund Manager shall provide each Municipality instructions regarding the payment of contributions, including: the schedule for paying contributions; the deadline for paying contributions; the amount of contributions due; and to whom contributions shall be sent.

# 17.05 Contingent Nature of Municipality Contributions

Contributions made by each Municipality are hereby made expressly contingent on the maintenance of a qualified status by the Plan for the year with respect to which such contribution is made.

# 17.06 Form of Municipality Contribution

Municipalities may pay their contributions to the Board or Trust Fund manager in cash.

## 17.07 Exclusive Benefit

Except as otherwise provided herein, the Municipality shall have no beneficial interest in any asset of the Trust or Trust Fund and no part of any asset in the Trust or Trust Fund shall ever revert to or be repaid to a Municipality, either directly or indirectly; prior to the satisfaction of all liabilities with respect to the Members and their Beneficiaries under the Plan. At no time shall any part of the corpus or income of the Trust Fund, or any asset of the Trust, be (at any time) used for or diverted to purposes other than the exclusive benefit of the Members or their Beneficiaries or to pay necessary and appropriate expenses of the System.

# 17.08 Evidence

Anyone required to give evidence under the terms of the Plan may do so by certificate, affidavit, document or other information which the person to act in reliance may consider pertinent, reliable and genuine, and to have been signed, made or presented by the proper party or parties. The Board shall be fully protected in acting and relying upon any evidence described under the immediately preceding sentence.

# 17.09 Waiver of Notice

Any person entitled to notice under the Plan may waive the notice.

# 17.10 Successors

The Plan shall be binding upon all persons entitled to benefits under the Plan, their respective heirs and legal representatives, upon the Municipality, its successors and assigns, and upon the Board and its successors.

# 17.11 Word Usage

Words used in the masculine shall apply to the feminine where applicable, and wherever the context of the Municipality's Plan dictates, the plural shall be read as the singular and the singular as the plural.

# 17.12 Intent of Tax Qualification

In an intent to maintain tax qualification, the applicable tax-qualification provisions of the Code shall control over any conflicting provision of state law.

# 17.13 Employment Not Guaranteed

Nothing contained in this Plan, or any modification or amendment to the Plan, or in the creation of any account, or the payment of any benefit, shall give any Employee, Member, or Beneficiary any right to continue employment, any legal or equitable right against the Municipality, or an Employee of the Municipality, or against the Board, or its agents or employees or against the Board, except as expressly provided by the Plan or by a separate agreement.

# 17.14 Entire Agreement

Except as expressly provided herein, this Plan supersedes all prior contracts and undertakings, written or oral, between the same parties concerning the same subject matter. The above notwithstanding, in no event shall any provision of this Plan reduce, diminish or otherwise impair any right of a Member Beneficiary or Survivor Annuity under any plan or contract for such benefits accrued as of the date of execution of this Plan.

## 17.15 Mistake in the Record

Should any change or mistake in records result in any Member, Beneficiary or Survivor Annuitant receiving from the System more or less than the individual would have been entitled to receive had the records been correct, then regardless of the intentional or unintentional nature of the error and upon the discovery of such error, the Board shall correct the error and so far as practicable adjust the payments which may be made for and to such person in such a manner that the actuarial equivalent of the benefit to which he was correctly entitled shall be paid.

# 17.16 Counterparts

This Plan may be executed in any number of separate counterparts, each of which shall be deemed an original, but the several counterparts shall together constitute but one and the same instrument.

#### 17.17 Severability

If any one or more of the covenants, agreements, provisions, or terms of this Plan shall be held contrary to any express provision of applicable law then such covenants, agreements, provisions, or terms shall be deemed severable from the remaining covenants, agreements, provisions, or terms

of this Plan and shall in no way affect the validity or enforceability of the remainder of this Plan or the rights of the parties hereto.

# 17.18 No Presumption Based upon Draftsmanship

For purposes of the interpretation and construction of this Plan and for the purpose of resolving any ambiguity herein, no inference or presumption shall be accorded to any party hereto based upon draftsmanship.

# 17.19 Headings

The headings and captions in this Plan are for convenience and reference purposes only and shall not be construed or deemed to explain, modify, amplify, or aid in the interpretation, construction, or meaning of the provisions hereof.

# ADDENDUM A – APPLICABLE STATE LAWS

Sections 101, 103, 104, 107, 109, 111, 401 and 402 of the Pennsylvania Municipal Retirement Law 53 P.S. § 881.101 et. seq., ("PMRL")

Enforcement Officers Disability Benefits Law, 53 P.S. § 637

16 Pa. Code § 81.7

Public Employee Pension Forfeiture Act (P.L. 752, No. 140)

20 Pa.C.S. § 3101

1 Pa. Code §§ 31.11, 31.15, 33.31, 33.34, and 35.20

# **ADDENDUM B – PMRS POLICY STATEMENTS**

# **Subject**

**Actuary Tables** 

Allocation of Excess Interest

Disability Reserve

**Excess Interest Award Process** 

Overpayment

Permanent Employment

Plan Withdrawal Procedure

Portability-Individual Member

Portability-Plan Spin-offs

Refund- Plan Withdrawal

Treatment of Member Contributions

# **ADDENDUM A**

#### 1 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

#### Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.101 (2016)

NOTICE: As to repeal of this section where it relates to townships of the first class, see § 3501 of act 1931, June 24, P.L. 1206, art. XXXV. As to repeal of this section where it relates to townships of the second class, see § 2101 of act 1933, May 1, P.L. 103, art. XXI (renumbered § 2201, art. XXII) and § 49 of act 1947, July 10, P.L. 1481. As to repeal of this section where it relates to boroughs, see § 1 of act 1915, May 14, P.L. 312, ch. XIII, art. 1. As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

## § 881.101. Short title

This act shall be known and may be cited as the "Pennsylvania Municipal Retirement Law."

# 2 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

#### Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.103 (2016)

NOTICE: As to repeal of this section where it relates to the conferrance of powers or the imposition of any duties on

townships, see § 1500 of act 1917, July 14, P.L. 840. As to repeal of this section where it relates to the conferrance of powers or the imposition of any duties on boroughs, see § 1 of act 1915, May 14, P.L. 312, ch. XIII, art. I. As to repeal of this section as it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

#### § 881.103. Pennsylvania Municipal Retirement Board

% A Pennsylvania Municipal Retirement Board is hereby created, which shall consist of the State Treasurer, Secretary of the Commonwealth, six municipal elected officials or employes of different classes of municipalities which have joined the system, one retired member of the system who is receiving a retirement allowance, one municipal fireman employed by a municipality which has joined the system and one municipal policeman employed by a municipality which has joined the system. The nine latter members shall be appointed by the Governor from among nominations made by the County Commissioners Association, the Pennsylvania League of Cities, the Pennsylvania Association of Township Commissioners, the Pennsylvania State Association of Township Supervisors, the Pennsylvania State Association of Boroughs and the Pennsylvania Municipal Authorities Association, and associations representing municipal firemen and municipal police, to serve for a term of four years each and until their successors are appointed and qualified. Appointments of members made by the Governor shall not require the advice and consent of the Senate. The two municipal employe members of the Municipal Employes' Retirement Board, appointed by the Governor from among nominations made by various associations of county and municipal officers, who are serving on the effective date of this act, shall continue to serve as members of the Pennsylvania Municipal Retirement Board until the expiration of their respective terms.

A chairman and vice chairman of the board shall be elected by the board every year at the January meeting of the board and the chairman and vice-chairman may succeed themselves for the appointed term of four years.

Vacancies happening from among members appointed from among the nominations made by the associations shall be filled by the appointment of a successor for a full term of four years.

No appointed member shall serve more than two consecutive full terms.

Each member of the board shall take an oath of office that he will diligently and honestly administer the affairs of the board, and that he will not knowingly violate or wilfully permit to be violated any of the provisions of this act.

A quorum of the board shall consist of six members.

## 3 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)

Copyright © 2016

All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*
\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.104 (2016)

NOTICE: As to repeal of this section where it relates to the conferrance of powers or the imposition of any duties on

townships, see § 1500 of act 1917, July 14, P.L. 840. As to repeal of this section where it relates to the conferrance of powers or the imposition of any duties on boroughs, see § 1 of act 1915, May 14, P.L. 312, ch. XIII, art. I. As to repeal of this section as it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

#### § 881.104. General powers of the Board

#### The board shall:

- (1) Appoint a secretary and an assistant secretary. The secretary, with the prior approval of the board, shall appoint the clerical and other employes of the board, whose positions, excluding the secretary's and assistant secretary's, shall be under the classified service provisions of the act of August 5, 1941 (P.L. 752, No. 286), as amended and the secretary shall fill future vacancies in accordance with such provisions. The compensation of all persons so appointed shall be fixed by the board and shall be consistent with the standards established by the Executive Board of this Commonwealth;
- (2) Contract for professional services, including but not limited to actuarial, investment and medical as it deems advisable;
  - (3) Keep in convenient form such data as shall be deemed necessary for actuarial valuation purposes;
- (4) From time to time, through its actuary, make an actuarial investigation into the mortality and service experience of the contributors and annuitants and of the various accounts created by this act;
  - (5) Adopt for the system one or more mortality tables and such other tables as shall be deemed necessary;
  - (6) Deleted by 1980, May 17, P.L. 135, No. 50, § 3, eff. in 60 days.
- (7) Certify annually the amount of appropriation which each municipality shall pay into the retirement fund, which amounts shall be based on estimates furnished by the actuary, and shall be credited to the municipal account of the fund;
- (8) Prepare and publish annually a financial statement showing the condition of the fund and the various accounts thereof, and setting forth such other facts, recommendations and data as may be of use in the advancement of knowledge concerning the Pennsylvania Municipal Retirement System, and furnish a copy thereof to each municipality which has joined the system, and to such persons as may request copies thereof;
  - (9) Keep a record of all its proceedings, which will be open to inspection by the public;
- (10) From time to time, with the advice of the Attorney General and the actuary, to adopt and promulgate such rules and regulations as may be required for the proper administration of the fund created by this act and the several accounts thereof; to establish procedures which would permit an eligible retired public safety officer to pay qualified accident, health or long-term care insurance premiums as permitted under section 402(l) of the Internal Revenue Code; and to provide for the transaction of the business of the board;
- (11) Be authorized to approve any optional retirement plan for municipal employes, municipal firemen or municipal police, with any municipality as long as it is actuarially sound and benefits under the plan are not in excess of or member's minimum contribution rates are not less than those provided in other existing retirement laws pertaining to that class of municipality; except to the extent that excess investment earnings are allocated to provide for additional pension benefits or member accruals as otherwise provided in this law;
- (12) Prepare and distribute annual statements of accounts to each of the active members of the system, showing the contributions made during the year, the interest earned and the total balance standing in the member's account at the

end of the year;

- (12.1) In accordance with the plan contract, provide cost-of-living increases from a plan's share of excess investment earnings to those members of such municipalities who have already retired. Such allocations shall be made, with the advice of the actuary, on a fully funded basis employing actuarial assumptions which reflect the nature of the liability. An award of excess interest shall not be made under this paragraph if the plan is less than 95% funded as of the plan's most recent filing of the actuarial report required under the Municipal Pension Funding Standard and Recovery Act.
- (12.2) In accordance with the plan contract, the board shall allocate excess investment earnings for active members of such municipalities by applying such allocation to member contributions. To the extent that additional liabilities may accrue as a result of such allocation, the actuary shall employ actuarial assumptions, on a fully funded basis, to accurately reflect the nature of the liability generated therefrom. An award of excess interest shall not be made under this paragraph if the plan is less than 95% funded as of the plan's most recent filing of the actuarial report required under the Municipal Pension Funding Standard and Recovery Act.
  - (12.3) Promulgate regulations required for the proper administration of the DROP.
- (12.4) Establish a subsidiary DROP participant account for each DROP participant, provide for the separate investment and funding of each account and establish a monthly rate of interest to credit the money deposited in each account.
- (13) Perform such other functions as are required for the execution of the provisions of this act and of other Federal and State law and administer and interpret this act to ensure that the system is maintained as tax qualified under the Internal Revenue Code.

#### 4 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.107 (2016)

NOTICE: As to repeal of this section where it relates to boroughs, see § 3301 of act 1927, May 4, P.L. 519, art. XXXIII. As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

§ 881.107. Election by municipalities to join retirement system

% Any municipality may elect, by ordinance or resolution adopted by the tax levying body, or in the case of municipal authorities by the board of such municipal authority to join the system: Provided, however, That any municipality electing coverage under the provisions of Article II of this act must have first placed its municipal employes in so far as they are eligible under the Federal Social Security Act.

Any municipality, by action of its tax levying body, may and upon petition of electors equal to at least five per cent of the registered electors of the municipality, shall, submit the question of joining such system to the voters of the municipality at any municipal or general election, in the same manner as other questions are submitted to the electors under the election code of the Commonwealth. If the majority of the electors voting on the question vote in favor thereof, the tax levying body shall adopt an ordinance or resolution electing to join such system. If the electors vote against joining the system, then no further action shall be taken in the municipality for a period of two years.

A duly certified copy of any such ordinance or resolution electing to join the system shall be filed with the board.

#### 5 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

#### Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.109 (2016)

NOTICE: As to repeal of this section where it relates to boroughs, see § 3301 of act 1927, May 4, P.L. 519, art. XXXIII. As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

# § 881.109. Custody of and payments from fund

All moneys and securities in the fund shall be placed in the custody of the State Treasurer for safekeeping, and all payments on account of retirement allowances shall be made on requisition signed by the chairman and secretary of the board.

#### 6 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)

Copyright © 2016

All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article I. General Provisions

#### Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.111 (2016)

NOTICE: As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII.

# § 881.111. Municipal guarantee

% The regular interest charges payable, the DROP interest charges payable and the creation and maintenance of the necessary reserves for the payment of the municipal and member's annuities, as to any municipality in accordance with this act, are hereby made the obligation of that municipality.

In the case of the failure of a municipality to make payments as required by this act, the Commonwealth shall withhold payment to the municipality of any funds to which the municipality may be entitled for pension purposes. The board may recover any sums due to the fund by suit at law, or other appropriate remedy.

# 7 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article IV. Optional Retirement Plans

#### Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.401 (2016)

NOTICE: As to repeal of this section where it relates to boroughs, see § 1 of act 1915, May 14, P.L. 312, ch. XIII, art. I. As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII. See also act 1899, April 28, P.L. 74 and § 42 of act 1901, June 4, P.L. 364.

§ 881.401. Purpose

This article shall provide for the enrollment of those municipalities in the Pennsylvania Municipal Retirement System which want to offer retirement benefits to their employes different from those available under Article II and Article III of this act. It shall also provide for increasing member benefits for municipalities formerly enrolled under the provisions of Article II and Article III of this act.

#### 8 of 8 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)

Copyright © 2016

All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8A. Pennsylvania Municipal Retirement Law
Article IV. Optional Retirement Plans

Go to the Pennsylvania Code Archive Directory

53 P.S. § 881.402 (2016)

NOTICE: As to repeal of this section where it relates to boroughs, see § 1 of act 1915, May 14, P.L. 312, ch. XIII, art. I. As to repeal of this section where it relates to third class cities, see § 4701 of act 1931, June 23, P.L. 932, art. XLVII. See also act 1899, April 28, P.L. 74 and § 42 of act 1901, June 4, P.L. 364.

#### § 881.402. Existing local retirement systems and compulsory and optional membership

Where a municipality elects to join the system established by this act, and is then maintaining a retirement or pension system or systems covering its employes in whole or in part, those employes so covered, and employes thereafter eligible to join such pension system, shall not become members of the retirement system established by this act, unless at the time the municipality elects to join the system, the members of each such existing retirement or pension system shall, by the affirmative vote of seventy-five per cent of all the members of each pension system, elect to be covered by the retirement system established by this act. At any time thereafter, within a period of three years after the municipality has elected to join the system, but not thereafter, the members of an existing retirement or pension system may, in like manner, elect to join the system established by this act. In any such case, provisions may be made for the transfer of moneys and securities in its retirement or pension fund or funds, in whole or in part, to the fund established by this act. Securities so transferred shall be only those acceptable to the board. Securities not so acceptable shall be converted into cash, and said cash transferred to the fund created by this act. In any such transfer, provision shall be made to credit the accumulated deductions of each member, at least the amount he has paid into the retirement or pension system of the municipality, which moneys shall be credited against the prior service contributions of such member, or a municipality may turn over to the retirement system created by this act any existing local pension system on a completely funded basis, as to pensioners and pension credits of members related to prior service to the date of transfer, or on a partially funded basis if the municipality pays annually into the retirement system amounts sufficient to completely liquidate the municipality's liability for prior service within a period not to exceed thirty years.

No liability, on account of retirement allowances or pensions being paid from any retirement or pension fund of the municipality, shall attach against the fund, except as provided in the agreement, making a transfer of an existing system

in accordance with this section. The liability to continue payment of pensions not so transferred shall attach against the municipality, which shall annually make appropriations from its tax revenues sufficient to pay the same. In cases where workers covered by an existing retirement or pension system elect to join the system created by this act, the election to join shall be deemed to have been made at the time the municipality elected to join the system, and the liabilities of the municipality shall be fixed accordingly.

If a municipality elects to join the system under the provisions of this Article IV, it shall first negotiate a contract with the board, acceptable to both the municipality and the board, which shall set forth all the specific details of municipal and member contribution rates and benefits. The municipality shall then pass an ordinance or resolution electing to join the system, and confirming the terms of the contract by reference thereto. Separate contracts and separate resolutions shall be executed for each class of employes, namely municipal employes, municipal firemen and municipal police in those cases where the municipality elects to bring more than one class of its employes into the system.

When a municipality elects to enroll its municipal employes into the system, then each officer other than elected officers, and each municipal employe thereof, employed on a full-time basis, shall be required to become a member of the system. Each municipality shall determine whether membership in said system for elected officials and employes hired on a temporary, seasonal or part-time basis shall be compulsory, optional or prohibited. Where membership may be optional with an elected officer or an employe hired on a temporary, seasonal or part-time basis, an election to join the system must be made within one year after the municipality elected to join the system or within one year after the officer or temporary, seasonal or part-time employe first entered the service of the municipality. Officers and employes paid only on a fee basis shall not be eligible to join the system.

When a municipality elects to enroll its municipal firemen or its municipal police into the system, then each municipal firemen or each municipal policeman, as defined in section 102 of this act, shall be required to become a member of the system.

When a municipality has established a policy of placing new employes on a probationary status it may elect to refrain from enrolling such employes into the system for a period of up to one year from the date the probationary employe first entered the service of the municipality. In such cases service credits shall not be earned by the employe for probationary time served prior to enrollment. Notwithstanding any other provision herein, the board may, in its discretion, entertain a request from a municipality to join the system established by this act for those employes who are excluded from local pension plan coverage by virtue of the collective bargaining process or otherwise. The request to join the system must be accompanied by an affirmative vote of no less than three-fourths of those employes not covered by the local pension plan. The benefits to be established may be in accordance with the provisions of this article or any other relevant pension law covering that class of municipality. The other requirements of this section for joining this system shall be observed.



#### PENNSYLVANIA ADMINISTRATIVE CODE

Commonwealth of Pennsylvania
Pennsylvania Codes
Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.
All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

TITLE 16. COMMUNITY AFFAIRS
PART III. MUNICIPAL RETIREMENT BOARD
CHAPTER 81. GENERAL PROVISIONS

Go to the Pennsylvania Administrative Code Archive Directory

16 Pa. Code § 81.7 (2016)

#### § 81.7. Retirement funds and accounts

- (a) The Board will consolidate for investment purposes the assets of the various plans. The Board will account separately for each plan's assets in a municipal account and each individual active member, inactive member and vested member's assets in a member's account. The Board maintains pooled accounts for retired members (the Retired Members' Reserve Account) and for the funding of disability benefits (the disability reserve).
- (b) When a municipality withdraws the administration of its plan from the System, the municipality shall only be entitled to the assets credited to the plan's municipal account and the plan's members' accounts in accordance with the provisions of the law. Assets that are actuarially determined by the Board's actuary to be matched to a withdrawing plan's retired members as of the effective date of withdrawal will also be returned to the plan in accordance with the law provided there are sufficient funds in the retired member's reserve account to meet the actuarially determined liability of all retired members of the System, as of the date of withdrawal; otherwise payment shall be on a prorated basis.

#### SOURCE:

The provisions of this § 81.7 amended June 23, 2000, effective June 24, 2000, 30 Pa.B. 3168. Immediately preceding text appears at serial page (260163).

**HIERARCHY NOTES:** 

Part Note

#### 1 of 1 DOCUMENT

Pennsylvania Statutes, Annotated by LexisNexis(R)

Copyright © 2016

All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 53. Municipal and Quasi-Municipal Corporations
Part I. General Municipal Law
Chapter 8. Public Officers and Employees
Article IV. Compensation

Go to the Pennsylvania Code Archive Directory

53 P.S. § 637 (2016)

#### § 637. Enforcement officer disability benefits

(a) Any member of the State Police Force, any enforcement officer or investigator employed by the Pennsylvania Liquor Control Board, and the parole agents, enforcement officers and investigators of the Pennsylvania Board of Probation and Parole, Capitol Police officers, correction employes employed by the Department of Corrections, whose principal duty is the care, custody and control of inmates, psychiatric security aides employed by the Department of Public Welfare and the Department of Corrections, whose principal duty is the care, custody, and control of the criminally insane, drug enforcement agents of the Office of Attorney General whose principal duty is the enforcement of the drug laws of the Commonwealth, special agents of the Office of Attorney General whose principal duty is the enforcement of the criminal laws of the Commonwealth, any member of the Delaware River Port Authority Police, any policeman, fireman or park guard of any county, city, borough, town or township, or any sheriff or deputy sheriff who is injured in the performance of his duties including, in the case of firemen, duty as special fire police, and by reason thereof is temporarily incapacitated from performing his duties, shall be paid by the Commonwealth of Pennsylvania if a member of the State Police Force or an enforcement officer or investigator employed by the Pennsylvania Liquor Control Board or the parole agents, enforcement officers and investigators of the Pennsylvania Board of Probation and Parole, Capitol Police officers, correction employes employed by the Department of Corrections, whose principal duty is the care, custody and control of immates, psychiatric security aides employed by the Department of Public Welfare and the Department of Corrections, whose principal duty is the care, custody, and control of the criminally insane, drug enforcement agents of the Office of Attorney General whose principal duty is the enforcement of the drug laws of the Commonwealth, special agents of the Office of Attorney General whose principal duty is the enforcement of the criminal laws of the Commonwealth, or by the Delaware River Port Authority if a member of the Delaware River Port Authority Police or by the county, township or municipality, by which he is employed, his full rate of salary, as fixed by ordinance or resolution, until the disability arising therefrom has ceased. All medical and hospital bills, incurred in connection with any such injury, shall be paid by the Commonwealth of Pennsylvania or by the Delaware River Port Authority or by such county, township or municipality. During the time salary for temporary incapacity shall be paid by the Commonwealth of Pennsylvania or by the Delaware River Port Authority or by the county, city, borough, town or township, any workmen's compensation, received or collected by any such employe for such period, shall be turned over to the Commonwealth of Pennsylvania or to the Delaware River Port Authority or to such county, city, borough, town or township, and paid into the treasury thereof, and if such payment shall not be so made by the employe the amount so due the Commonwealth of Pennsylvania, the Delaware River Port Authority or the county, city, borough,

town or township shall be deducted from any salary then or thereafter becoming due and owing.

- (b) In the case of the State Police Force, enforcement officers and investigators employed by the Pennsylvania Liquor Control Board and the parole agents, enforcement officers and investigators of the Pennsylvania Board of Probation and Parole, Capitol Police officers, correction employes employed by the Department of Corrections, whose principal duty is the care, custody and control of inmates, psychiatric security aides employed by the Department of Public Welfare and the Department of Corrections whose principal duty is the care, custody, and control of the criminally insane, drug enforcement agents of the Office of Attorney General whose principal duty is the enforcement of the drug laws of the Commonwealth, special agents of the Office of Attorney General whose principal duty is the enforcement of the criminal laws of the Commonwealth, members of the Delaware River Port Authority Police, salaried policemen and firemen, and sheriffs and deputy sheriffs who have served for four consecutive years or longer, diseases of the heart and tuberculosis of the respiratory system, contracted or incurred by any of them after four years of continuous service as such, and caused by extreme overexertion in times of stress or danger or by exposure to heat, smoke, fumes or gases, arising directly out of the employment of any such member of the State Police Force, enforcement officer, investigator employed by the Pennsylvania Liquor Control Board, guard or enforcement officer employed by the Department of Corrections or parole agent, enforcement officer or investigator of the Pennsylvania Board of Probation and Parole, Capitol Police officers, correction employes employed by the Department of Corrections whose principal duty is the care, custody and control of inmates, psychiatric security aides employed by the Department of Public Welfare and the Department of Corrections whose principal duty is the care, custody, and control of the criminally insane, drug enforcement agents of the Office of Attorney General whose principal duty is the enforcement of the drug laws of the Commonwealth, special agents of the Office of Attorney General whose principal duty is the enforcement of the criminal laws of the Commonwealth, member of the Delaware River Port Authority Police, policeman or fireman, or sheriff or deputy sheriff shall be compensable in accordance with the terms hereof; and unless any such disability shall be compensable under the compensation laws as having been caused by accidental injury, such disability shall be compensable as occupational disease disabilities are presently compensable under the compensation laws of this Commonwealth. It shall be presumed that tuberculosis of the respiratory system contracted or incurred after four consecutive years of service was contracted or incurred as a direct result of employment.
- (c) In the case of any person receiving benefits pursuant to this act, the statutes of limitations set forth in sections 306.1, 315, 413, and 434 of the act of June 2, 1915 (P.L. 736, No. 338), known as the "Workers' Compensation Act," shall not begin to run until the expiration of the receipt of benefits pursuant to this act.
- (d) All payments herein required to be made by the Commonwealth of Pennsylvania on account of any member of the State Police Force shall be made from moneys appropriated to the Pennsylvania State Police, and any payments required to be made on account of any enforcement officer or investigator employed by the Pennsylvania Liquor Control Board shall be from appropriations out of the State Stores Fund, any payments required to be made on account of any parole agent, enforcement officer or investigator employed by the Pennsylvania Board of Probation and Parole shall be from moneys appropriated to the Pennsylvania Board of Probation and Parole, any payments required to be made on account of Capitol Police officers shall be made from moneys appropriated to the Department of General Services, any payments required to be made on account of any correction employe shall be made from moneys appropriated to the Department of Corrections, any payments required to be made on account of any psychiatric security aides shall be made from moneys appropriated to the Department of Public Welfare or the Department of Corrections where appropriate, and any payments required to be made on account of any drug enforcement agent or special agents shall be made from moneys appropriated to the Office of Attorney General.

### 1 of 5 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 43. Labor
Chapter 22. Public Employee Pension Forfeiture Act

Go to the Pennsylvania Code Archive Directory

43 P.S. § 1311 (2016)

§ 1311. Short title

This act shall be known and may be cited as the "Public Employee Pension Forfeiture Act."

#### 2 of 5 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 43. Labor
Chapter 22. Public Employee Pension Forfeiture Act

Go to the Pennsylvania Code Archive Directory

43 P.S. § 1312 (2016)

NOTICE: As to suspension of this section for the purpose of considering Act 1984, Dec. 19, P.L. 1191, No. 226, and all amendments to it, see § 8 of said act.

# § 1312. Definitions

% The following words and phrases when used in this act shall have, unless the context clearly indicates otherwise, the meanings given to them in this section:

"CRIMES RELATED TO PUBLIC OFFICE OR PUBLIC EMPLOYMENT." Any of the criminal offenses as set forth in the following provisions of Title 18 (Crimes and Offenses) of the Pennsylvania Consolidated Statutes or other enumerated statute when committed by a public official or public employee through his public office or position or when his public employment places him in a position to commit the crime:

Any of the criminal offenses set forth in Subchapter B of Chapter 31 (relating to definition of offenses) when the criminal offense is committed by a school employee as defined in 24 Pa.C.S. § 8102 (relating to definitions) against a student.

Section 3922 (relating to theft by deception) when the criminal culpability reaches the level of a misdemeanor of the first degree or higher.

Section 3923 (relating to theft by extortion) when the criminal culpability reaches the level of a misdemeanor of the first degree or higher.

Section 3926 (relating to theft of services) when the criminal culpability reaches the level of a misdemeanor of the first degree or higher.

Section 3927 (relating to theft by failure to make required disposition of funds received) when the criminal culpability reaches the level of a misdemeanor of the first degree or higher.

Section 4101 (relating to forgery).

Section 4104 (relating to tampering with records or identification).

Section 4113 (relating to misapplication of entrusted property and property of government or financial institutions) when the criminal culpability reaches the level of misdemeanor of the second degree.

Section 4701 (relating to bribery in official and political matters).

Section 4702 (relating to threats and other improper influence in official and political matters).

Section 4902 (relating to perjury).

Section 4903(a) (relating to false swearing).

Section 4904 (relating to unsworn falsification to authorities).

Section 4906 (relating to false reports to law enforcement authorities).

Section 4909 (relating to witness or informant taking bribe).

Section 4910 (relating to tampering with or fabricating physical evidence).

Section 4911 (relating to tampering with public records or information).

Section 4952 (relating to intimidation of witnesses or victims).

Section 4953 (relating to retaliation against witness, victim or party).

Section 5101 (relating to obstructing administration of law or other governmental function).

Section 5301 (relating to official oppression).

Section 5302 (relating to speculating or wagering on official action or information).

Article III of the act of March 4, 1971 (P.L. 6, No. 2), known as the "Tax Reform Code of 1971."

In addition to the foregoing specific crimes, the term also includes all criminal offenses as set forth in Federal law substantially the same as the crimes enumerated herein.

"POLITICAL SUBDIVISION." Any county, city, borough, incorporated town, township, school district, vocational school district, intermediate unit, municipal authority, home rule, optional plan or optional charter municipality, and any agencies, boards, commissions, committees, departments, instrumentalities, or entities thereof designated to act in behalf of a political subdivision either by statute or appropriation.

"PUBLIC OFFICIAL" or "PUBLIC EMPLOYEE." Any person who is elected or appointed to any public office or employment including justices, judges and justices of the peace and members of the General Assembly or who is acting or who has acted in behalf of the Commonwealth or a political subdivision or any agency thereof including but not limited to any person who has so acted and is otherwise entitled to or is receiving retirement benefits whether that person is acting on a permanent or temporary basis and whether or not compensated on a full or part-time basis. This term shall not include independent contractors nor their employees or agents under contract to the Commonwealth or political subdivision nor shall it apply to any person performing tasks over which the Commonwealth or political subdivision has no legal right of control. However, this term shall include all persons who are members of any retirement system funded in whole or in part by the Commonwealth or any political subdivision. For the purposes of this act such persons are deemed to be engaged in public employment.

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 43. Labor
Chapter 22. Public Employee Pension Forfeiture Act

# Go to the Pennsylvania Code Archive Directory

43 P.S. § 1313 (2016)

NOTICE: As to termination of this commission on December 31, 2001, in the manner provided for by the act of December 22, 1981 (P.L. 508, No. 142), known as the Sunset Act, see § 2 of Act 1991, Dec. 20, P.L. 393, No. 42.

### § 1313. Disqualification and forfeiture of benefits

- (a) Notwithstanding any other provision of law, no public official or public employee nor any beneficiary designated by such public official or public employee shall be entitled to receive any retirement or other benefit or payment of any kind except a return of the contribution paid into any pension fund without interest, if such public official or public employee is convicted or pleads guilty or no defense to any crime related to public office or public employment.
- (b) The benefits shall be forfeited upon entry of a plea of guilty or no defense or upon initial conviction and no payment or partial payment shall be made during the pendency of an appeal. If a verdict of not guilty is rendered or the indictment or criminal information finally dismissed, then the public official or public employee shall be reinstated as a member of the pension fund or system and shall be entitled to all benefits including those accruing during the period of forfeiture if any. Such conviction or plea shall be deemed to be a breach of a public officer's or public employee's contract with his employer.
- (c) Each time a public officer or public employee is elected, appointed, promoted, or otherwise changes a job classification, there is a termination and renewal of the contract for purposes of this act.
- (d) The appropriate retirement board may retain a member's contributions and interest thereon for the purpose of paying any fine imposed upon the member of the fund, or for the repayment of any funds misappropriated by such member from the Commonwealth or any political subdivision.
- (e) Notwithstanding any other provision of this act, the State Employees' Retirement Board shall not disburse any funds to any person who has forfeited their right to benefits until the Auditor General and the Attorney General have determined and certified that there has been no loss to the Commonwealth as a result of the conduct that resulted in forfeiture of benefits. If there is a loss to the Commonwealth, the board shall pay the amount of the loss to the State Treasurer from the member's contributions and the interest thereon.

4 of 5 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 43. Labor
Chapter 22. Public Employee Pension Forfeiture Act

### Go to the Pennsylvania Code Archive Directory

43 P.S. § 1314 (2016)

# § 1314. Restitution for monetary loss

- (a) Whenever any public official or employee who is a member of any pension system funded by public moneys is convicted or pleads guilty or pleads no defense in any court of record to any crime related to a public office or public employment, the court shall order the defendant to make complete and full restitution to the Commonwealth or political subdivision of any monetary loss incurred as a result of the criminal offense.
- (b) If the court fails to order such restitution the Commonwealth, through the Attorney General, or a political subdivision shall petition the court pronouncing sentence for an order establishing the amount of restitution due it. If the court does not have authority to order restitution, the Commonwealth or the political subdivision shall bring an original action for restitution.
- (c) Notwithstanding any law or provision of law exempting the pension account or benefits of any public official or public employee from garnishment or attachment, whenever the court shall order restitution or establish the amount of restitution due after petition, all sums then credited to the defendant's account or payable to the defendant including the contributions shall be available to satisfy such restitution order.
- (d) The retirement board, administrator of the pension fund or employer of the defendant, upon being served with a copy of the court's order, shall pay over all such pension benefits, contributions or other benefits to the extent necessary to satisfy the order of restitution.

### 5 of 5 DOCUMENTS

Pennsylvania Statutes, Annotated by LexisNexis(R)
Copyright © 2016
All rights reserved.

\*\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-101 \*\*

\*\* P.S. documents are current through 2016 Regular Session Acts 1-81 \*\*

Pennsylvania Statutes
Title 43. Labor
Chapter 22. Public Employee Pension Forfeiture Act

Go to the Pennsylvania Code Archive Directory

43 P.S. § 1315 (2016)

All other acts or parts of acts inconsistent with the provisions of this act are repealed to the extent of their inconsistency.

# Pennsylvania Statutes, Annotated by LexisNexis(R) Copyright © 2016 All rights reserved.

\* Pa.C.S. documents are current through 2016 Regular Session Acts 1-174 with the exception of Act 170 \* P.S. documents are current through 2016 Regular Session Acts 1-81 \*

Pennsylvania Consolidated Statutes
Title 20. Decedents, Estates and Fiduciaries
Chapter 31. Dispositions Independent of Letters; Family Exemption; Probate of Wills and Grant of Letters
Subchapter A. Dispositions Independent of Letters

### Go to the Pennsylvania Code Archive Directory

20 Pa.C.S. § 3101 (2016)

### § 3101. Payments to family and funeral directors.

### (a) Wages, salary or employee benefits. --

Any employer of a person dying domiciled in this Commonwealth at any time after the death of the employee, whether or not a personal representative has been appointed, may pay wages, salary or any employee benefits due the deceased in an amount not exceeding \$ 5,000 to the spouse, any child, the father or mother, or any sister or brother (preference being given in the order named) of the deceased employee. Any employer making such a payment shall be released to the same extent as if payment had been made to a duly appointed personal representative of the decedent and he shall not be required to see to the application thereof. Any person to whom payment is made shall be answerable therefor to anyone prejudiced by an improper distribution.

#### (b) Deposit account. --

Any bank, savings association, savings and loan association, building and loan association, credit union or other savings organization, at any time after the death of a depositor, member or certificate holder, shall pay the amount on deposit or represented by the certificate, when the total standing to the credit of the decedent in that institution does not exceed \$ 10,000, to the spouse, any child, the father or mother or any sister or brother (preference being given in the order named) of the deceased depositor, member or certificate holder, provided that a receipted funeral bill or an affidavit, executed by a licensed funeral director which sets forth that satisfactory arrangements for payment of funeral services have been made, is presented. Any bank, association, union or other savings organization making such a payment shall be released to the same extent as if payment had been made to a duly appointed personal representative of the decedent and it shall not be required to see to the application thereof. Any person to whom payment is made shall be answerable therefor to anyone prejudiced by an improper distribution.

## (c) Patient's care account. --

When the decedent was a qualified recipient of medical assistance from the Department of Public Welfare, the facility in which he was a patient may make payment of funds, if any, remaining in the patient's care account, for the decedent's burial expenses to a licensed funeral director in an amount not exceeding \$ 10,000 whether or not a personal representative has been appointed. After the payment of decedent's burial expenses, the facility may pay the balance of decedent's patient's care account, as long as the payments including the payment for burial expenses does not exceed \$ 10,000, to the spouse, any child, the father or mother or any sister or brother (preference being given in the order

named) of the deceased patient. Any facility making such a payment shall be released to the same extent as if payment had been made to a duly appointed personal representative of the decedent and it shall not be required to see to the application thereof. Any licensed funeral director or other person to whom payment is made shall be answerable therefor to anyone prejudiced by an improper distribution.

### (d) Life insurance payable to estate. --

Any insurance company which upon the death of an individual residing in this Commonwealth owes his estate a total amount of \$11,000 or less under any policy of life, endowment, accident or health insurance, or under any annuity or pure endowment contract, may at any time after 60 days following his death pay all or any part of that amount to the spouse, any child, the father or mother or any sister or brother of the decedent (preference being given in the order named) provided that at the time of the payment no written claim for that money has been received at the office of the company specified in the policy or contract for the receipt of claims from any duly appointed personal representative of the decedent. Any insurance company making any payment in accordance with this section to an adult may rely on the affidavit of any of the persons named in this subsection concerning the existence and relationship of these persons and shall be released to the same extent as if payment had been made to a duly appointed personal representative of the decedent and the insurance company shall not be required to see to the application thereof. Any person to whom payment is made shall be answerable therefor to anyone prejudiced by an improper distribution.

# (e) Unclaimed property.

- (1) In any case where property or funds owned by an individual who has died a resident of this Commonwealth have been reported to the Commonwealth and are in the custody of the State Treasurer as unclaimed or abandoned property, the State Treasurer, at any time after the death of the individual, shall be authorized under this section to distribute the property or to pay the amount being held in custody where all of the following conditions are present:
  - (i) The amount of the funds or the value of the property is \$ 11,000 or less.
- (ii) The person claiming the property or the funds is the surviving spouse, child, mother or father, or sister or brother of the decedent, with preference given in that order.
- (iii) A personal representative of the decedent has not been appointed or five years have lapsed since the appointment of a personal representative of the decedent.
- (2) Upon being presented with a claim for property owned by a decedent, the State Treasurer shall require the person claiming the property to provide all of the following prior to distributing the property or paying the amount held in custody:
  - (i) A certified death certificate of the owner.
- (ii) A sworn affidavit under the penalties of 18 Pa.C.S. § 4904 (relating to unsworn falsification to authorities) setting forth the relationship of the claimant to the decedent, the existence or nonexistence of a duly appointed personal representative of the decedent and any other persons that may be entitled under this section to make a claim to the decedent's property.
- (iii) Other information determined by the State Treasurer to be necessary in order to distribute property or pay funds under this section to the proper person.
- (3) If the State Treasurer determines the claimant to be a person entitled to claim property of a decedent owner, the State Treasurer shall pay or distribute such property to the claimant and shall thereby be released to the same extent as if payment or distribution had been made to a duly appointed personal representative of the decedent and shall not be required to oversee the application of the payments made. Any claimant to whom payment is made shall be answerable

therefore to anyone prejudiced by an improper distribution or payment.

#### 1 of 5 DOCUMENTS

# PENNSYLVANIA ADMINISTRATIVE CODE Commonwealth of Pennsylvania Pennsylvania Codes Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.

All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

# TITLE 1. GENERAL PROVISIONS PART II. GENERAL RULES OF ADMINISTRATIVE PRACTICE AND PROCEDURE

# CHAPTER 31. PRELIMINARY PROVISIONS SUBCHAPTER B. TIME

Go to the Pennsylvania Administrative Code Archive Directory

1 Pa. Code § 31.11 (2016)

§ 31.11. Timely filing required

Pleadings, submittals or other documents required or permitted to be filed under this part, the regulations of the agency or any other provision of law shall be received for filing at the office of the agency within the time limits, if any, for the filing. The date of receipt at the office of the agency and not the date of deposit in the mails is determinative.

# 2 of 5 DOCUMENTS

PENNSYLVANIA ADMINISTRATIVE CODE
Commonwealth of Pennsylvania
Pennsylvania Codes
Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.
All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

TITLE 1. GENERAL PROVISIONS
PART II, GENERAL RULES OF ADMINISTRATIVE PRACTICE AND PROCEDURE

CHAPTER 31. PRELIMINARY PROVISIONS SUBCHAPTER B. TIME

Go to the Pennsylvania Administrative Code Archive Directory

1 Pa. Code § 31.15 (2016)

### § 31.15. Extensions of time

- (a) Extensions of time shall be governed by the following:
- (1) Except as otherwise provided by law, whenever by these rules or by a regulation or order of an agency, or a notice given thereunder, an act is required or allowed to be done at or within a specified time, the time fixed or the period of time prescribed may, by the agency head or the presiding officer, for good cause be extended upon motion made before expiration of the period originally prescribed or as previously extended; and upon motion made after the expiration of the specified period, the act may be permitted to be done where reasonable grounds are shown for the failure to act.
- (2) Requests for the extension of time in which to file briefs shall be filed at least 5 days before the time fixed for filing such briefs.
- (b) Except as otherwise provided by law, requests for continuance of hearings or for extension of time in which to perform an act required or allowed to be done at or within a specified time by these rules or a regulation or order of an agency, shall be by motion in writing, timely filed with the agency, stating the facts on which the application rests, except that during the course of a hearing in a pro-ceeding, the requests may be made by oral motion in the hearing before the agency head or the presiding officer.

#### 3 of 5 DOCUMENTS

### PENNSYLVANIA ADMINISTRATIVE CODE

Commonwealth of Pennsylvania
Pennsylvania Codes
Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.
All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

# TITLE 1. GENERAL PROVISIONS PART II. GENERAL RULES OF ADMINISTRATIVE PRACTICE AND PROCEDURE

CHAPTER 33, DOCUMENTARY FILINGS SUBCHAPTER B. SERVICE OF DOCUMENTS

Go to the Pennsylvania Administrative Code Archive Directory

1 Pa. Code § 33.31 (2016)

### § 33.31. Service by the agency

Orders, notices and other documents originating with an agency, including forms of agency action, complaints and similar process and other documents designated by the agency for this purpose, shall be served by the office of the agency by mail, except when service by another method shall be specifically required by the agency, by mailing a copy thereof to the person to be served, addressed to the person designated in the initial pleading or submittal at his principal office or place of business. If service is not accomplished by mail, it may be effected by anyone authorized by the agency in the manner provided in 231 Pa. Code Rules 400 -- 441 (relating to service of original process).

#### 4 of 5 DOCUMENTS

#### PENNSYLVANIA ADMINISTRATIVE CODE

Commonwealth of Pennsylvania
Pennsylvania Codes
Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.
All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

# TITLE 1. GENERAL PROVISIONS PART II, GENERAL RULES OF ADMINISTRATIVE PRACTICE AND PROCEDURE

# CHAPTER 33. DOCUMENTARY FILINGS SUBCHAPTER B. SERVICE OF DOCUMENTS

# Go to the Pennsylvania Administrative Code Archive Directory

1 Pa. Code § 33.34 (2016)

§ 33.34. Date of service

The date of service shall be the day when the document served is deposited in the United States mail, or is delivered in person, as the case may be.

#### 5 of 5 DOCUMENTS

# PENNSYLVANIA ADMINISTRATIVE CODE

Commonwealth of Pennsylvania
Pennsylvania Codes
Copyright (c) 2016 Pennsylvania Legislative Reference Bureau.
All rights reserved.

\*\*\* This document is current through the December 2016 supplement \*\*\*

\*\*\* Changes effective through 46 Pa.B. 6280 (October 1, 2016) \*\*\*

# TITLE 1. GENERAL PROVISIONS PART II. GENERAL RULES OF ADMINISTRATIVE PRACTICE AND PROCEDURE

CHAPTER 35. FORMAL PROCEEDINGS
SUBCHAPTER A. PLEADINGS AND OTHER PRELIMINARY MATTERS
PETITIONS

Go to the Pennsylvania Administrative Code Archive Directory

1 Pa. Code § 35.20 (2016)

§ 35.20. Appeals from actions of the staff

Actions taken by a subordinate officer under authority delegated by the agency head may be appealed to the agency head by filing a petition within 10 days after service of notice of the action.

# **ADDENDUM B**

# PENNSYLVANIA MUNICIPAL RETIREMENT SYSTEM BASE PLAN DOCUMENT

# ADDENDUM B – PMRS POLICY STATEMENTS

Pennsylvania Municipal Retirement Board	
Policy Statement 14 – 3	Utilization of
Adopted June 11, 2014	Actuarial Tables

Purpose:

The Pennsylvania Municipal Retirement Board is required to adopt all necessary actuarial assumptions essential for the calculation of retirement benefits. This policy sets forth the adopted tables and assumptions the System uses to determine actuarial equivalence for optional benefit forms of retirement and the early retirement actuarial equivalence.

Authority:

Pennsylvania Municipal Retirement Law, Act of February 1, 1947, P.L. 34, No. 15, § 104(5), 53 P.S. § 881.104(5).

Policy:

The Pennsylvania Municipal Retirement Board adopts the following actuarial assumptions to be used for optional benefit forms of retirement and early retirement actuarial equivalence being made effective on or after January 1, 2011:

- 1. Mortality Assumption: The Society of Actuaries RP 2000 table. Rates will be blended 70% male and 30% female to provide unisex factors for calculating actuarial equivalent benefits.
- 2. Interest assumption: 6.0% compounded annually for calculations effective on or after January 1, 2011 through December 31, 2012; 5.5% compounded annually for calculations effective on or after January 1, 2013.

The following actuarial assumption shall be used for optional benefit form of retirement and early retirement actuarial equivalence being made on or after December 31, 2002 through December 31, 2010:

- 1. Mortality Assumption: The 1994 Group Annuity Mortality Table without adjustment or projection will be used. Rates will be blended 70% male and 30% female to provide unisex factors for calculating actuarial equivalent benefits.
- 2. Interest assumption: 6.0% compounded annually.

Effective Date

This policy is effective immediately and shall be applied retroactively to distributions with annuity starting dates on or after December 31, 2002. It replaces Policy Statement 10-7 adopted at the November 18, 2010 meeting of the Pennsylvania Municipal Retirement Board.

Adoption Date:

Adopted at the June 11, 2014 meeting of the Pennsylvania Municipal Retirement Board.

James B. Allen, Secretary

JBA:jba 06/11/2014

Pennsylvania Municipal Retirement Board
Policy Statement
Rule Number 97-1

Allocation of Excess Interest to Municipal Account, Members' Accounts, and Retired Members' Reserve Account

# Purpose:

To formalize the procedure used by the Pennsylvania Municipal Retirement System (the System) to allocate excess interest "... as each municipality deems appropriate to the credit of the Municipal accounts, member's accounts, the member's excess investment accounts, retired members reserve accounts and total disability reserve accounts." ("Pennsylvania Municipal Retirement Law": Section 110)

# Authority:

Act 15 of 1974, "Pennsylvania Municipal Retirement Law": Section 110. Management and Investment of Fund; Interest Credits.

# Specific Cites:

Act 15 of 1974, "Pennsylvania Municipal Retirement Law": Section 104. General Powers of the Board. (12.1) and (12.2); Section 110. Management and Investment of Fund; Interest Credits.

# Policy:

The Board has been directed to credit any excess investment monies, hereafter referred to as excess interest, earned by the System to the Municipal Account, Members' Accounts, and Retired Members' Reserve Account as directed by each member municipality. It is noted that the law speaks also to the crediting of excess interest to the member's excess investment account and to the total disability reserve.

Whereas the member's excess investment account is a sub-account of each member's account maintained as part of the Members' Reserve Account, the Board has determined that it does fulfill the obligation of allocating monies to the member's excess investment account when a municipality directs that excess interest is to be allocated to the members' accounts. The Board satisfies the obligation by crediting the directed excess interest to each member's excess interest account maintained as a part of the Members' Reserve Account.

The Board further declares that the total Disability Reserve Account is a pooled account of the System. This reserve account is used to fund a portion of a disability pension when a disability benefit is not fully funded by a member's accrued benefit as of the date of disability. This account does not benefit a plan unless that plan has a member who seeks a disability pension in a given year. The Board declares that allocating excess interest to the Disability Reserve Account would not directly benefit a municipality in a subsequent year, nor would it benefit a municipality's members (active or retired) in a subsequent year. Given these facts, the Board deems it illogical to have a municipality allocate excess interest to this account and therefore directs that the municipalities

# Pennsylvania Municipal Retirement Board Policy Statement Rule Number 97-1 (Continued)

Allocation of Excess Interest to Municipal Account, Members' Accounts, and Retired Members' Reserve Account

# Policy: (Cont'd.)

not be given the option of allocating excess interest to the total Disability Reserve Account.

The Board directs that the Secretary of the System or the Secretary's designee inform the municipalities of the excess interest award value to the municipalities' plan(s) within two weeks of the Board's adoption of the excess interest award. The written announcement shall be mailed to each plan's designated contact person and shall contain the specific dollar amount awarded as excess interest. This amount shall be calculated by multiplying each plan's assets by the percentage of the adopted excess interest award. Assets refer to monies held in the Members' Reserve, the Municipal Reserve, and the Retired Members' Reserve (the assets held in the name of a retired member – the present value as determined by the System's actuary) as of December 31 of the previous year. The announcement shall show not only the value of the total excess interest awarded to the plan but also the dollar amount of the award generated from the previous year-end balances of the reserve accounts (Municipal, Members, and Retired Members).

The plan's governing body shall notify the System in writing of the manner in which the governing board wants the excess interest allocated among the three reserves. In addition, if any excess interest is allocated to either active or retired members, it must be done in a non-discriminatory, systematic manner, and the process shall be uniformly applied to all members. The Board directs that staff monitor the allocation process and contact the municipality should a questionable or unusual allocation decision be presented to the System.

If the plan's allocation decision is not returned to the System by November 30, the System shall allocate all awarded excess interest to the plan's municipal account. The System will implement the municipalities' allocation decisions effective December 31 of the year the Board declared the excess interest award. Once the allocation has been made (December 31), the decision is irrevocable.

Active members who terminate membership in the System after December 31 of the previous year shall not be entitled to any excess interest award. Any award that would have been allocated to the member's account shall be deposited instead to the municipal account of the plan.

# Pennsylvania Municipal Retirement Board Policy Statement Rule Number 97-1 (Continued)

Allocation of Excess Interest to Municipal Account, Members' Accounts, and Retired Members' Reserve Account

# Policy: (Cont'd)

Retired members who terminate membership in the System after December 31 of the previous year shall not be entitled to any excess interest award. Instead, any award that would have been allocated to the retired member's benefit shall be deposited to the municipal account of the plan.

Members who would have been allocated excess interest money as an active member but who retired on or after December 31 of the previous year shall have said allocation annuitized and added to the monthly benefit said member is getting as a retired member of the plan.

Municipalities that withdraw their plan(s) from the System on or after December 31 of the previous year shall be entitled to any excess interest declared by the Board in the current year which is based upon the plan's assets as of December 31 of the previous year. The System shall award said excess interest to the governing boar dof the withdrawn plan on December 31 of the current year. A plan must be a member of the System on December 31 to participate in any allocation of excess interest in the following year.

Any former member who has lost rights to membership in the System due to termination of employment in a member plan but has not withdrawn the accumulated member deductions by December 31 of the year in which membership rights ended shall not be eligible for allocated excess interest.

Effective Date:

This rule is to be implemented effective immediately and applicable only on a prospective basis.

Adoption Date:

Adopted at the September 18, 1997, meeting of the Pennsylvania

Municipal Retirement Board.

Pennsylvania Municipal Retirement Board	Disability Reserve	
Policy Statement 07 - 5		

# Purpose:

The Pennsylvania Municipal Retirement Law establishes a separate reserve fund, Total Disability Reserve Account (Disability Reserve), to assist plans in providing disability benefits through a pooled risk vehicle. This statement sets out the funding and utilization policy of the Disability Reserve.

Background: Municipalities may elect to provide disability retirement benefits. When a municipality has elected to provide disability benefits, a common design is to provide 30% income replacement for non-service or standard disability and 50% income replacement for an in-service disability. The law requires the Pennsylvania Municipal Retirement System (PMRS) to include in the plan's annual obligation a charged term cost for the value of these benefits.

> When a participant of a plan with disability benefits applies and is approved by the Pennsylvania Municipal Retirement Board for a disability retirement benefit, all or a portion of the funding of the benefit is transferred from the Disability Reserve to the Retired Member's Reserve Account (Retiree Reserve). When determining the amount of the transfer from the Disability Reserve a value of the disability benefit is determined using the applicable actuarial factors consistent with the member's demographics and the plan's benefit formula.

Once the value of the disability benefit is determined, PMRS determines the accrued value of the member's normal retirement benefit. The value of the member's accrued normal retirement benefit is subtracted from the value of the member's approved disability benefit and the remainder is the amount of money that is transferred from the Disability Reserve to the Retiree Reserve to fully fund the approved disability benefit.

Any monies in the Member's Account (except those monies voluntarily contributed to fund an optional benefit and the regular interest credited to them) and any monies in the Member's Excess Interest Account are subtracted from the value of the accrued normal retirement benefit to determine the amount of monies charged to the member's employer's Municipal Account to fund the benefit not funded by monies from the Disability Reserve.

### Authority:

Pennsylvania Municipal Retirement Law, Act of February 1, 1974, P.L. 34, No. 15, § 104(5), 53 P.S. § 881.104(5).

# Policy:

The Pennsylvania Municipal Retirement Board adopts the following funding procedures to ensure adequate financial resources are available to provide for disability retirement liabilities for those participating municipalities that elect to provide disability benefits.

- 1. Funding An annual cost, expressed as a percentage of a plan's covered compensation, shall be determined by the system's actuary and included in the Minimum Municipal Obligation of each plan that provides disability benefits. This cost is to be used for the funding of the Disability Reserve. The annual cost shall be equal to the projected disability cost in excess of the projected accrued benefit cost factoring in the probability of a disability occurrence for the one year term covered by the Minimum Municipal Obligation. Such amounts charged shall be deposited to the Disability Reserve.
- 2. Distributions On approval of a disability retirement by the Pennsylvania Municipal Retirement Board, the difference in the value of the disability benefit and the funds identified for transfer to the Retiree Reserve based upon the member's accrued normal retirement benefit as of the date of disability shall be determined and that amount shall be transferred from the Disability Reserve to the Retiree Reserve to fully fund the disabled retiree benefit.
- 3. Funding Limitation The assets in the Disability Reserve shall be determined as of December 31 of each year. If the value exceeds 1.5 times the highest total of all required disability transfers to the Retiree Reserve in any one of the last three years, all assets in excess of 1.5 times the highest transfer value shall be transferred from the Disability Reserve and placed in the unallocated assets for distribution in the year of determination.
- 4. Shortfall Funding To the extent that transfers from the Disability Reserve to the Retiree Reserve exceed the value of assets in the Disability Reserve and the balance in the Disability Reserve is determined to be a negative value as of December 31, assets from the unallocated earnings of PMRS shall be deposited in the Disability Reserve to bring the Disability Reserve balance to "0" at year end.

Effective Data

This policy is to be effective immediately.

Date:

Adoption Date:

Adopted at the September 20, 2007 meeting of the Pennsylvania Municipal Retirement Board.

James B. Allen, Secretary

JBA:jba 09/07/07

Pennsylvania Municipal Retirement Board Policy Statement 05-2

**Excess Interest Award Process** 

Purpose:

To document how the Pennsylvania Municipal Retirement Board determines whether and the extent to which Excess Interest is available to be distributed to the member plans.

Authority:

The Pennsylvania Municipal Retirement Law, Act 15 of 1974 (referred to as "the Act"), Sections 104 (General Powers of the Board) and 110 (Management and Investment of Fund; Interest Credits).

Background:

The Pennsylvania Municipal Retirement Board is required to determine annually the amount of investment earnings on the System assets in excess of that required for allocation to regular interest and expenses. This "excess" is defined in the law as "Excess Interest." The Board adopted a procedure to determine the excess interest in 1984. This Policy Statement is intended to memorialize the process and make it available to the public. The Board is adding one new stipulation to the process – the establishment of a minimum award threshold. This additional provision is to limit the administrative burden associated with the allocation process in years when the award would be considered "de minimis."

Policy:

The Board requires the following calculations to be made as part of the excess interest determination process.

- 1. At the close of the year, staff with the confirmation of the independent auditing firm under contract to the Board, shall make all required allocations of regular interest to the System's accounts.
- 2. Staff and consulting actuary shall then determine the expected administrative expense for the following year that is not expected to be met from the \$20 per plan member administrative charge.
- 3. Staff with the confirmation of the independent auditing firm shall determine the *Market value* of the System's investment portfolio as of year end along with the audited balances of the *System's Reserve accounts* (Member, Municipal, Retired, Disability).
- 4. These numbers (See Steps 1, 2 & 3) will be provided to the consulting actuary who will be responsible for performing the next set of calculations.
- 5. The consulting actuary will add to the *System's Reserve accounts* the projected expenses from Step 2. This total will become the *Preliminary value*.

# Policy: (Con't.)

- 6. The consulting actuary will then subtract from the *Market value* of the System's investment portfolio the *Preliminary value*. This result will be the *Available surplus*.
- 7. The consulting actuary will then determine the *New surplus* by subtracting from the *Available surplus* last year's *Final surplus*. New surplus is limited to be no greater than Available surplus
- 8. Next the consulting actuary will calculate the ratio "m" of *Available surplus* (See Step 6) to *Market value*.
- 9. Next the consulting actuary will calculate the ratio "n" of *New surplus* (See Step 7) to *Market value*.
- 10. The percentage of *New surplus* to be used for the *Excess interest* (e) is derived from the following formula:

$$e = \frac{0.10 + 8m}{1.0 + 8n}$$

- 11. The consulting actuary then determines the *Trial excess interest allocation* by multiplying (e) (See Step 10) times *New surplus* (See Step 7).
- 12. The consulting actuary will also determine *Trial surplus* by subtracting *Trial excess interest allocation* (See Step 11) from *Available surplus* (See Step 6).
- 13. The consulting actuary will determine *Trial margin* by dividing *Trial surplus* (See Step 12) from *Market value* (See Step 3).
- 14. If *Trial margin* (See Step 13) is less than ten percent (10%), then trial figures become final figures and the *Actuarial value* is set as *Market value* less *Final surplus* (See Step 12).
- 15. If *Trial margin* (See Step 13) is equal to or greater than ten percent (10%), the *Final excess interest allocation* is *Available surplus* (See Step 6) less ten percent (10%) of *Market value* (See Step 3), *Final surplus* is ten percent (10%) of *Market value*, and *Actuarial value* is ninety percent (90%) of *Market value*.

Once the *Final excess interest allocation* is determined, the *Excess interest percentage* is calculated by the consulting actuary. This is derived by first adding the Member Reserve balance, the Municipal Reserve balance and the actuarially determined Present value of the Retired Reserve balance to determine *eligible reserves*. The Excess interest percentage is the *Final excess interest allocation* divided by the *eligible reserves*.

Pennsylvania Municipal Retirement Board Policy Statement 05-2 (Con't) **Excess Interest Award Process** 

# Policy: (Con't.)

If the Excess interest percentage is less than five tenths of one percent (0.5%), there shall be NO Excess interest awarded, the Actuarial value shall be set to the Preliminary value and the Final surplus shall be set to the Available surplus. If the Excess interest percentage is five tenths of one percent (0.5%) or greater, the resulting Excess interest percentage shall be awarded and the plans notified as to their proportion of the total Final excess interest allocation.

The actual allocation of the year's *Excess interest* will be posted to the plans' accounts as of December 31 of the following year. To be eligible to receive a portion of the excess interest awarded for any year, a municipality's plan must have been in PMRS on December 31 of the year for which the allocation was made.

# Effective

Date:

This policy is to be implemented effective immediately.

<u>Adoption</u>

Adopted at the July 21, 2005 meeting of the Pennsylvania Municipal

Date:

Retirement Board.

James B. Allen, Secretary

Pennsylvania Municipal Retirement Board	
Policy Statement 08 – 4	Overpayment Policy
Adopted November 20, 2008	

Purpose:

To provide direction to the Pennsylvania Municipal Retirement staff (Staff) on the collection, requests for compromise, and write-off of delinquent debts and accounts due the Pennsylvania Municipal Retirement System (the "System").

Authority

The Pennsylvania Municipal Retirement Law, Act 15 of 1974, Section 104.

Background

An overpayment occurs when a former member, annuitant, beneficiary or survivor annuitant ("Debtor") is paid a benefit exceeding his/her entitlement. The following represent the major causes of overpayments by the System:

- 1. Delayed notification of the death of a member or annuitant.
- 2. Inaccurate information reported by the member or municipality.
- 3. Administrative errors by Staff.
- 4. Misinterpretation of benefit provisions.

Procedures:

- A. Debtors who have been overpaid shall be notified in writing of the reason and the amount of the overpayment and offered the following repayment methods:
  - 1. A lump-sum payment within 90 days of the receipt of notification.
  - Monthly installment payments in an amount agreed upon between the former member or beneficiary and the Secretary of the Pennsylvania Municipal Retirement System (Secretary). Installment periods extending beyond sixty months shall require the approval of the Pennsylvania Municipal Retirement Board (Board).
  - If applicable, an actuarial reduction spread over the life of the annuitant, survivor annuitant, or beneficiary may be applied to the monthly check. Reductions of more than 20% of the monthly check shall require the approval of the Board

The debtor shall be given 15 days to respond to the first notice, which shall be sent via certified mail return receipt requested to the last known address on file for the debtor. If there is no response to the first notice, a second and third notice, if necessary, shall be sent by regular first-class mail at 15-day intervals.

If no response is received after the third notice and the amount of the overpayment is \$1,000 or less:

1. In the case of a former member or beneficiary, a write-off request will be completed and submitted to the Office of Attorney General, Financial Enforcement Section. If the write-off request is approved, the outstanding balance shall be written off, collection efforts shall stop, and the account shall be closed, but if the former member returns to service, the debt shall be recovered from future benefits that accrue to the re-enrolled member.

Pennsylvania Municipal Retirement Board
Policy Statement 08 – 4 (Continued)
Adopted November 20, 2008

Overpayment Policy

2. In the case of an annuitant, survivor annuitant or beneficiary, an actuarial reduction shall be applied to the ongoing annuity, when available. If a beneficiary is not receiving a monthly annuity from the System or if the annuity is not large enough to accommodate the actuarial reduction and there is still a balance remaining of less than \$1,000; after 2 documented attempts to receive the funds have been made, a write-off request will be completed and submitted to the Office of Attorney General, Financial Enforcement Section. If the write-off request is approved, the outstanding balance shall be written off, collection efforts shall stop, and the account shall be closed.

If no response is received after the third notice and the amount of the overpayment is more than \$1,000, the claim shall be referred to the Office of the Attorney General, Financial Enforcement Section for attempted collection of the debt. The System shall ask the Attorney General to return the claim to the System if the debt is not collectible. If the debt is returned as uncollectible:

- 1. In the case of a former member, the debt shall be written off to balance the member's account but if the former member returns to service, the debt shall be recovered from future benefits that accrue to the re-enrolled member.
- 2. In the case of an annuitant, survivor annuitant or beneficiary, the debt shall be permanently written off.
- B. If payments are made by the System to a Debtor after the Debtor's date of death resulting in an overpayment, the legal representative for the estate of the Debtor shall be advised in writing of the reason and the amount of the overpayment and the following repayment methods:
  - 1. A lump-sum payment within 90 days of the receipt of notification.
  - 2. Monthly installment payments in an amount agreed upon between the legal representative and the Secretary. Installment periods extending beyond sixty months shall require the approval of the Board.

The legal representative shall be given 15 days to respond to the first notice, which shall be sent via certified mail return receipt requested to the last known address on file. If there is no response to the first notice, a second and third notice, if necessary, shall be sent by regular first-class mail at 15-day intervals.

If no response is received after the third notice and the amount of the overpayment is \$1,000 or less:

1. An actuarial reduction shall be applied to the annuity of the survivor annuitant or beneficiary, if available. Reductions of more than 20% of the monthly check shall require approval by the Board.

Pennsylvania Municipal Retirement Board	
Policy Statement 08 – 4 (Continued)	Overpayment Policy
Adopted November 20, 2008	

2. If a Debtor is not receiving a monthly annuity from the System or if there is no benefit payable or the annuity is not large enough to accommodate the actuarial reduction and there is still a balance remaining of less than \$1,000; a write-off request will be completed and submitted to the Office of Attorney General, Financial Enforcement Section. If the write-off request is approved, the outstanding balance shall be written off, collection efforts shall stop, and the account shall be closed.

If the amount is more than \$1,000, the claim shall be referred to the Office of the Attorney General, Financial Enforcement Section for attempted collection of the debt. The System shall ask the Attorney General to return the claim to the System if the debt is not collectible. It shall be written off on a permanent basis.

If there is no legal representative of the estate of the Debtor and the Secretary has reason to believe that the overpayment was received by a person without due authority ("Unauthorized Person"), then Staff shall make a reasonable effort to obtain the identity of the Unauthorized Person and follow the notice procedures to a legal representative stated above.

If after reasonable investigation, such Unauthorized Person cannot be identified and there is no death benefit payable from which to deduct the overpayment:

- 1. If the amount is \$1,000 or less, a write-off request will be completed and submitted to the Office of Attorney General, Financial Enforcement Section. If the write-off request is approved, the debt shall be written off, collection efforts shall stop, and the account shall be closed.
- 2. If the amount is more than \$1,000, the claim shall be referred to the Office of the Attorney General, Financial Enforcement Section for attempted collection of the debt. The System shall ask the Attorney General to return the claim to the System if the debt is not collectible. It shall be permanently written off.
- C. If a write-off request or settlement is not approved by the Office of the Attorney General, the claim shall be referred to the Office of the Attorney General, Financial Enforcement Section for attempted collection of the debt.
- D. In all cases, the System reserves the right to confer with its Office of Chief Counsel to request delegation from the Office of the Attorney General to pursue collection of the debt in accordance with the Commonwealth Attorneys Act. 71 P.S. § 732-204(c).

Pennsylvania Municipal Retirement Board	
Policy Statement 08 – 4 (Continued)	
Adopted November 20, 2008	

Overpayment Policy

# Hardship:

If the debtor wishes to appeal the collection of the overpayment on the basis of financial hardship, the following procedures shall be followed:

- 1. The Staff shall notify the debtor in writing that reasonable evidence must be provided within 30 days to support the appeal. Where health is the issue, some form of medical evidence from an informed source shall be required. For financial reasons, details of income and expenditure (bills) shall be required.
- 2. The Secretary, or his or her designee, shall review the information provided. If the Secretary determines that additional information is needed in order to make a recommendation to the Board, he or she may request a meeting with the Debtor. The Secretary may also request the Debtor to provide a notarized affidavit certifying the information provided.
- 3. If the Secretary, or his or her designee, feels that a hardship exists, he or she shall recommend to the Board forgiveness or settlement of the debt.
- 4. If the Board decides to write-off or settle the debt, the System shall complete and submit the appropriate STD forms including supporting documentation to the Office of the Attorney General, Financial Enforcement Section for approval.

**Exceptions:** 

Exceptions to this policy shall require Board action and the Board reserves its right to make such exceptions when it deems it appropriate.

**Effective** 

Date:

Date:

This policy is to be effective immediately and apply to all current Debtors,

regardless of the date on which the debt was incurred.

Adoption

Adopted at the November 20, 2008 meeting of the Pennsylvania Municipal

Retirement Board.

James B. Allen, Secretary

JBA:jba 10/29/08 Pennsylvania Municipal Retirement Board Policy Statement Rule 03-1 Permanent Employment

Purpose:

To set forth guidance on the definition of permanent employment as used in Act 15 of 1974, the Pennsylvania Municipal Retirement Law (referred to as "the Act").

Authority:

The Pennsylvania Municipal Retirement Law, Act 15 of 1974, Section 106.

Background:

The Pennsylvania Municipal Retirement Board is authorized to prepare information to be distributed to members and municipalities showing "the methods of administration" the Board uses in managing the plans enrolled in the Pennsylvania Municipal Retirement System (PMRS). The Board has found it necessary to share with existing municipal members and potential members guidance on the term "employed on a permanent basis" as used in sections 203 and 402 of the Act. The reference is to those municipal employees who must be enrolled in the PMRS administered pension plan.

Policy:

The Board confirms its understanding of the law that any individual hired with the expectation of being in an active, regularly scheduled paid position for more than one year is required to be considered employed on a permanent basis, regardless of whether the employment is part-time or full-time. The actual number of hours worked per week is not a consideration in the determination of permanent employment.

# Implementation:

The Board directs that all municipalities that have municipal plans in PMRS be provided with information advising of their responsibility to specifically agree to enroll all employees employed on a permanent basis. The municipality's agreement to do so shall be memorialized in all contracts for benefits entered into from the effective date of this policy forward. The municipality shall also be required to certify that this requirement is met via the certification process used on the System's Quarterly Report of Contributions (PMRB-21).

When there is a dispute over whether an individual is eligible to be enrolled in a PMRS administered municipal plan, PMRS shall address the question of the individual's status upon a request to review the matter from the individual or the municipality. The request shall be set forth in writing.

Pennsylvania Municipal Retirement Board Policy Statement Rule 03-1 (Con't) Permanent Employment

Implementation: (Con't)

PMRS shall review the request and confirm the particulars of that individual's employment history with the enrolled municipal plan sponsor. Circumstances to be identified include, but are not limited to, the following:

- 1. The specific offer of employment, such as the letter offering employment, the resolution of the employer formally creating the position, or any public announcement of the position's vacancy.
- The conditions set forth in the offer of employment including hours worked, obligations to report to work and expected term of employment.
- 3. The funding of the position, and whether the individual is being paid under the same federal Employer Identification Number as other individuals enrolled in the pension plan.
- 4. Any written personnel policy adopted by the municipality.
- 5. Actual employment and payment records for the individual in question,

During the review, PMRS' Secretary shall be empowered to issue an "Order to Show Cause" if deemed necessary to the determination of the status of the individual with the PMRS administered pension plan.

The Secretary shall notify the individual and the municipality in writing of the administrative decision of PMRS. The parties to the administrative decision shall have all rights of appeal granted to parties to PMRS administrative decisions, including a formal appeal to the Board.

If PMRS determines that an individual should have been enrolled in the plan, enrollment will be effective as of the date the individual became eligible for enrollment, absent extenuating circumstances. PMRS shall calculate the required member contributions that should have been made during the time period in question, if any are required by the plan.

If member contributions should have been made, the individual shall be required to pay into the System the required contributions and the regular interest that would have been credited had the individual been enrolled in the System. The individual may pay the obligation in a lump sum or over a period of up to five years through payroll deductions, so long as the individual remains an active member of the plan. If not paid in a lump sum, the amount due shall be assessed interest at the regular interest rate. The municipality's liability will also be determined and reflected as an obligation of the municipality.

Pennsylvania Municipal Retirement Board
Policy Statement
Rule 03-1 (Con't)

Permanent Employment

**Effective** 

Date:

This policy is to be implemented effective immediately.

Adoption

Adopted at the July 17, 2003 meeting of the Pennsylvania Municipal

Date:

Retirement Board.

James B. Allen, Secretary

Pennsylvania Municipal Retirement Board Policy Statement 10 – 2 Adopted March 17, 2010

# Plan Withdrawal Procedure

Purpose:

To document how the Pennsylvania Municipal Retirement Board (Board) processes a plan's request to withdraw from the Pennsylvania Municipal Retirement System (System).

# Authority:

The Pennsylvania Municipal Retirement Law, Act 15 of 1974 (referred to as "the Act"), contains three articles (Articles II, III, and IV) that allow for municipalities to enroll their pension plan in the System. Each article contains a section that also allows a municipality to withdraw from the System. (See Sections 214, 316, and 412). Each of the previously cited sections also cites Section 104 (General Powers of the Board) which authorizes the Board to adopt rules and regulations for the proper administration of the System. The Board has adopted regulations to this end. (See 16 Pa. Code, 81.10 -Withdrawal Provisions.)

# Background:

The Pennsylvania Municipal Retirement Board is required to take action on the application of a plan that seeks permission to withdraw from the System. While the Act and the adopted regulations set forth the specific requirements that must be met, the Board wishes to specify how the application will be processed and the steps that will be followed once a plan's application to withdraw has been approved.

### Policy:

<u>Requirements for Withdrawal</u> - A municipality that files an application with the Board for permission to withdraw from the System must meet all of the following requirements:

- 1. the plan has been enrolled in the System for a period of at least five years;
- 2. the municipality has met all of the plan's financial obligations to the System;
- 3. the legislative body of the municipality has passed an ordinance or, if the legislative body does not possess the authority to adopt ordinances, a resolution, signifying its intention to withdraw from the System;
- 4. the municipality has certified to the Board that an affirmative vote approving withdrawal from the System had been obtained from at least seventy-five per cent of all of the municipal employees (the term "municipal employee" for the purpose of considering a plan's withdrawal has been defined by regulation to include a plan's active members, inactive members, vested members and retired members);
- the municipality has acknowledged its responsibility to assume and provide for all future benefit payments to the existing active, inactive, vested and retired members and their beneficiaries effective upon the withdrawal; and,

Pennsylvania Municipal Retirement Board Policy Statement 10 – 2 (Con't) Adopted March 17, 2010

Plan Withdrawal Procedure

6. the application has specified a date for the plan withdrawal to become effective, provided if there are retired members in active pay status the effective date must be the first day of a month no earlier than the month after which the Board is scheduled to take action on the withdrawal application.

<u>Board Consideration of Application</u> - The Board shall take action within ninety (90) days of receipt of an application for permission to withdraw from the System. Should no regular Board meeting be scheduled to occur within ninety (90) days of receipt of an application, the Secretary shall arrange with the Board Chair to schedule a special meeting to take action on the application.

Concurrently with the scheduling of the application for action by the Board, the Secretary shall <u>provide</u> the Office of Chief Counsel with a copy of the application and any associated documents. The Office of Chief\_Counsel shall advise the Board as to whether or not the application meets the requirements of the law and Board adopted regulations. The Secretary shall also cause a review of the plan's accounts to be undertaken and a determination to be made as to whether the plan's financial obligations to the System have been met. The Secretary shall report the results of this review to the Board.

If the Board approves the municipality's application to withdraw its plan, the municipality shall be entitled to receive a refund of the assets of the plan calculated in accordance with the procedure set forth below. The Board may require the withdrawing municipality to enter into a written agreement with the Board terminating its contractual relationship with the Board and fixing the respective rights of the parties. Additionally, the withdrawing municipality may be required to obtain individual waivers or releases from affected members. If the board disapproves the application of the municipality to withdraw, the Secretary shall notify the municipality of the Board's decision and advise the municipality of the Board's reason or reasons for disapproval.

<u>Calculation of Plan Assets to be Refunded</u> – When the Board approves a municipality's application to withdraw a plan from the System, the Secretary shall cause to be determined as of the effective date of withdrawal, the total of all amounts then standing to the credit of the plan in the members' accounts, the members' excess investment accounts, the municipal account and the retired members' reserve accounts. When determining the amount then standing to the credit of the plan in the retired members' accounts, the Secretary shall utilize Policy Statement 10-3. This total shall be considered the plan's credited assets.

Pennsylvania Municipal Retirement Board Policy Statement 10 – 2 (Con't) Adopted March 17, 2010

Plan Withdrawal Procedure

The Secretary shall also determine whether or not the actuarial value of the System's assets exceed the System's market value of its investments in the most current actuarial valuation accepted by the Board. If the actuarial value of assets exceeded the market value of investments, the ratio of market value to actuarial value shall be multiplied times the plan's credited assets. The resulting product shall be considered the plan's net refund. If the actuarial value of assets is less than the market value of investments, the plan's credited assets shall be the plan's net refund.

In the event that the withdrawal of the plan is due to the dissolution of the employing municipal entity and all of the assets and liabilities of the withdrawing plan are to be transferred and distributed to a PMRS member plan or plans; then, regardless of the ratio of the System's market value to actuarial value, the withdrawing plan's credited assets shall be the plan's net refund.

Specifics of the Plan's Net Refund – Once the determination of a withdrawing plan's net refund is made, the Secretary shall cause said amount to be made payable to the plan's withdrawing municipality. The net refund may be made in a single payment or in multiple payments depending on the municipality's certification of all plan payments into the System. If the withdrawing plan has retirees in an active payment status, the Secretary shall make every attempt to refund, at a minimum, that portion of the plan's net refund attributable to the retired members' accounts prior to the date the municipality must begin to make monthly benefit payments.

All payouts are to be considered a trustee-to-trustee payment. A complete accounting of the plan's net refund, as well as the System's financial statements for the plan for the year in which the plan withdraws shall be provided to the withdrawing municipality.

Effective Date:

This policy is to be implemented effective immediately.

Adoption Date:

Adopted at the March 17, 2010 meeting of the Pennsylvania Municipal Retirement Board.

James B. Allen, Secretary

Pennsylvania Municipal Retirement Board	Portability:
Policy Statement	Individual Member
Rule Number 97-2	

Purpose:

The law provides that the Board transfers service credits when an employee leaves one Pennsylvania Municipal Retirement System (PMRS) member municipality and joins another. This benefit is called portability. The Board, in order to give policy guidance to staff, adopts the following statements on portability.

Authority:

Act 15 of 1974, "Pennsylvania Municipal Retirement Law": Section 405.

Policy:

**STATEMENT 1:** Portability is optional.

An employee who leaves the employ of one PMRS member municipality (Municipality A) and joins the employ of another PMRS member municipality (Municipality B) shall have one year from the date of separation from Municipality A's plan to exercise the portability benefit authorized in the law. Employment (but not necessarily enrollment in Municipality B's plan) must occur within one year of separation from eligibility in Municipality A's plan. Municipality B's plan enrollment effective date as a member municipality in PMRS must be on or before the date the employee is hired. An employee shall have the rights to exercise his rights to portability but said rights have to be affirmatively selected by the eligible employee by completing the appropriate PMRS form relating to the instituting of portability within one year of separation from Municipality A.

**STATEMENT 2:** The member's benefit will be determined under the contract provisions in effect in Municipality B.

In determining the benefit to be received by the employee who exercises portability, the contract and plan benefit structure of Municipality B shall be the controlling document in determining the benefit under which the member will be allowed to retire or receive a benefit. However, in no event will the member receive an accrued benefit from Municipality B which has a present value less than that earned in Municipality A as of the time of separation from Municipality A.

### **STATEMENT 3:** Service credits re combined.

For an employee electing to institute portability, the years of credited service shall be based on all years of credited service earned by the employee (credited service in both Municipality A and Municipality B).

Pennsylvania Municipal Retirement Board
Policy Statement
Rule Number 97-2 (Continued)

Portability: Individual Member

**STATEMENT 4:** The transfer amount is the present value of the accrued benefit in Municipality A's plan as of date of separation from Municipality A.

When an individual elects to implement the portability benefit, his accumulated contributions (member contributions and regular interest thereon) and any credited excess interest shall be transferred and credited to his new account in Municipality B. The difference between the accumulated contributions and the present value of the accrued benefit of the member at the time of separation from Municipality A shall be determined (or if the plan is a defined contribution plan – the employer contributions made to the member's account and credited interest thereon). This determined amount shall be transferred from Municipality A's municipal account (or if Municipality A's plan is a defined contribution plan – from the member's account) to Municipality B's municipal account (or if Municipality B's plan is a defined contribution plan – to the member's account).

It is recognized that the potential exists for the transfer to either overfund the accrued benefit under Municipality B's plan (if the member is leaving from a higher benefit plan to go to a lower benefit plan) or, conversely, the transfer may be insufficient to meet the accrued liability of the member under Municipality B's plan (if Plan B's benefit structure is better than Plan A's benefit structure). The actual consequences will depend upon the benefit structures in effect in the two municipalities. At time of benefit receipt, the employee shall have a present value of benefits at least equal to the amount that has been transferred from Municipality A's plan to Municipality B's plan.

# Effective

Date:

This rule is to be implemented effective immediately and is intended as a continuation and restatement of the Board's policy adopted on November 20, 1986.

# Adoption

Date:

Adopted at the November 20, 1997 meeting of the Pennsylvania Municipal Retirement Board.

Pennsylvania Municipal Retirement Board	Portability:
Policy Statement	Plan Spin-offs
Rule Number 98-1	"

# Purpose:

The law provides that the Board transfers service credits when an employee leaves one Pennsylvania Municipal Retirement System (PMRS) member municipality and joins another. This benefit is called portability. The Board, in rule Number 97-2, set policy to guide staff when an individual utilizes the portability benefit. The Board finds, however, that when a member municipality elects to spin off an employee or group of employes into a new, separate plan to be governed by a new municipality, there is a need to adopt additional guidelines. Rule Number 98-1 is intended to address the portability of individuals involved in a plan spin-off.

# **Authority**:

Act 15 of 1974, "Pennsylvania Municipal Retirement Law": Section 405.

# Policy:

When the employer (Employer A) of a PMRS administered plan seeks to spin off one or more employee members and transfer the employee(s) to an employer (Employer B) created by Employer A and Employer B seeks to have the Pennsylvania Municipal Retirement System administer a newly created pension plan, the System shall provide portability for the affected, transferred employee members. This situation is typically found in the creation of an authority by a municipality or in the creation of a regional government (Council of Government – COG).

The calculation of the asset value associated with an employee's portability benefit in a spin-off situation shall be determined based upon the relationship between the member's existing pension plan and the newly created plan.

STATEMENT 1: When Employer B establishes a plan that is identical or better in benefit design than the plan that existed for the employee(s) under Employer A, the assets equal to the actuarial accrued liability under Employer A's plan for the effected employees will be transferred from Employer A's plan to Employer B's plan. The employee(s) shall be entitled to the benefits in effect in the plan from which they finally separate from employment.

This calculation method shall be used so that the newly established pension plan is not created with an immediate unfunded liability. It is assumed that the spin-off in this circumstance is intended to start the new employer's pension plan with no unfunded liability for service credits earned with the previous employer.

Pennsylvania Municipal Retirement Board	Portability:
Policy Statement	Plan Spin-offs
Rule Number 98-1 (Continued)	

Because the effect of the transfer of service credits for the affected employee(s) is such that no lesser benefit will be earned in the new plan, there is no need to guarantee that the employee will be unharmed in such a transfer.

**STATEMENT 2:** When Employer B establishes a plan with benefits less than those that existed for the employee(s) prior to the spin-off, the calculation of the assets and liabilities associated with the portability benefit shall be done in accordance with the Board's Rule Number 97-2 policy statement.

The Board believes that Employer A should not be forced to fund the new plan at a level higher than Employer B's actual obligation to the employee(s). Because the effect of the transfer of service credits for the affected employee(s) is such that a lesser benefit could be earned in the new plan, there is a need to guarantee that the employee will not receive a reduction in benefits as a result of such a transfer. That guarantee is provided in the Board's policy statement Rule Number 97-2.

# Effective

Date:

This rule is to be implemented effective immediately.

# Adoption

Date:

Adopted at the January 22, 1998 meeting of the Pennsylvania Municipal Retirement Board.

Pennsylvania Municipal Retirement Board Policy Statement 10 – 3 Adopted March 17, 2010 Calculation of Withdrawing Plan's Portion of Retired Member's Reserve

Purpose:

To adopt a procedure to calculate the amount of money "... standing to the credit of the municipality... in the retired member's reserve accounts of the system (.)" when a municipality is withdrawing a plan from the System's administration.

Authority:

Act 15 of 1974, "The Pennsylvania Municipal Retirement Law" Section 104 General Powers of the Board, Paragraph (10).

Specific Cites:

Act 15 of 1974, "The Pennsylvania Municipal Retirement Law"; Section 214 Withdrawal Provisions and Section 412 Withdrawal Provisions.

Policy:

The calculation of the amount remaining to the credit of a withdrawing plan in the retired member's reserve account shall be made as of the effective date of the withdrawal utilizing the actuarial assumptions and tables in use by the System on the effective date of withdrawal.

The withdrawal payout shall be the "reverse" transfer value (actuarial present value) of each plan member in payment status as of the effective date of withdrawal provided the System's retired member's reserve account has as of the System's most recently completed actuarial valuation, assets in excess of the actuarial present value of future expected payments to retired pensioners and beneficiaries of all the plans of the System (Retiree Actuarial Value). This provision shall apply only to those benefits which had been fully funded through transfers from the members, municipal, and/or disability reserves as well as any allocation of excess interest. (The reverse transfer values for pay-as-you-go adhoc COLA benefits are not to be calculated.)

If the assets in the System's retired member's reserve are less than the System's Retiree Actuarial Value as of the most recently completed actuarial valuation, then the ratio of the assets to the Retiree Actuarial Value shall be applied to the calculated reverse transfer value and the product shall be deemed the amount then standing to the credit of the withdrawing municipality in the retired member's reserve account, and it will be the amount paid out to the withdrawing municipality.

Pennsylvania Municipal Retirement Board Policy Statement 10 – 3 (Con't.) Adopted March 17, 2010

Calculation of Withdrawing Plan's Portion of Retired Member's Reserve

In the event that the withdrawal of the plan is due to the dissolution of the employing municipal entity and all of the assets and liabilities of the withdrawing plan are to be transferred to a PMRS member plan or plans; then, regardless of the ratio of the assets in the System's retiree reserve to the System's Retiree Actuarial Value, the amount credited to the withdrawing plan in the retired members' reserve account shall be equal to the actuarial present value of each plan member in payment status as of the effective date of withdrawal.

The actuarial experience of the retired members and their beneficiaries that has developed prior to the effective date of the plan's withdrawal shall not be considered in the calculation of the Retired Member's Reserve Account payout. Only those retired members or their beneficiaries who have a benefit in pay status on the effective date of withdrawal shall be considered as a part of the municipal plan's retired member's reserve account balance.

Effective Date:

This rule is to be implemented effective immediately and applicable only on a prospective basis.

Adoption Date: Adopted at the March 17, 2010 meeting of the Pennsylvania Municipal Retirement Board.

James B. Allen, Secretary

# **EXHIBIT 2**



# PENNSYLVANIA MUNICIPAL RETIREMENT SYSTEM DEFINED BENEFIT PLAN Adoption Agreement 001

#### **PMRS Plan Effective Date:**

Plan as applicable.)

(Effective date of Plan initially administered by PMRS and restated under this document.)

2008 Plan Year; or (2) the first day of the first Plan Year of the PMRS Plan or Original

#### **Original Plan Effective Date:**

(Effective date of Plan initially administered by Municipality or third party administrator or PMRS and restated under this document.)

#### ARTICLE I: DEFINITIONS

Any capitalized terms used in this Adoption Agreement but not defined herein shall be given the meaning set forth in the Base Plan Document and Trust.

#### 1.23 COMPENSATION.

The total amount of all payments, direct or indirect, made by the Municipality to an Member for services rendered to the Municipality, for a calendar year which ends within a Plan Year, as defined in Code Section 3401(a) for purposes of tax withholding at the source (as reported to the Employee on Form W-2 for such year). Compensation shall include before-tax or salary deferral contributions made to this Plan or any other plan of the Municipality, under a Code Section 132(f)(4) qualified transportation plan or under Code Sections 125, 402(g)(3), 457 or 414(h), on behalf of a Member for such Plan Year. The term Compensation shall **exclude** the following (select all exclusions):

Overtime (as defined under the Fair Labor Standards Act)

Bonus Payments (Regular)

Bonus Payments (Special)

Unused Accrued Leave Payments

(if unused accrued leave is included for purposes of Compensation, it should be excluded for purposes of calculation of Credited Service under Section 1.24)

Unused Sick Leave Payments

(if unused sick leave is included for purposes of Compensation, it should be excluded for purposes of calculation of Credited Service under Section 1.24)

Payments under the "The Pennsylvania Workmen's Compensation Act"

Incentive Payments		
Education Incentive Payments		
Longevity Payments		
Payments in Lieu of Offered Benefits		
Allowance for Meals		
Allowance for Clothing and/or Equipment		
All other Compensation except regular salary not otherwise enumerated in the Base Plan Document or in this Adoption Agreement		
Other (Specify Short Name):		
Definition:		
CREDITED SERVICE.		
Method of Measurement (Select One)		
Hours of Service Method		
Elapsed Time Method		
Adjustments to Credited Service	Include	Exclude
Maximum years of Credited Service		
Maximum Years:		
Unused sick leave		
(converted to Credited Service under the following formula:		
(if included for purposes of Credited Service, should be excluded for purposes of Compensation under Section 1.23)		
Unused annual leave		
(converted to Credited Service under the following formula:		
(if included for purposes of Credited Service, should be excluded for purposes of Compensation under Section 1.23)		
Worker's Compensation Leave		
Service as Employee before the Municipality's PMRS Plan Enrollment Date		
Service as Employee before the Original Plan Effective Date		
Service as Employee under the non-PMRS Original Plan		
Service as Employee before the terminated Prior Plan of Municipality		
Service as Employee under the terminated Prior Plan of Municipality		
Service as Employee after the terminated Prior Plan of Municipality		
Name of terminated Prior Plan:		
Service as Employee before the merged Prior Plan of Municipality		
Service as Employee under the merged Prior Plan of Municipality		
Service as Employee after the merged Prior Plan of Municipality		
Name of merged Prior Plan:		

# Method of Measurement (Select One) Hours of Service Method Elapsed Time Method EMPLOYEE. Plan Type (Select one) **Municipal Police Officers** Municipal Firefighters Non-Uniform Municipal Employees (Select all that apply) Eligible Employees for Mandatory Membership (Select all that apply) Full-time Employees not subject to a collective bargaining agreement with a bargaining unit authorized with the Municipality Defined as regularly scheduled to perform at least ( ) Hours of Service per week Full-time Employees subject to a collective bargaining agreement with one of the specified bargaining units below authorized with the Municipality: Defined as regularly scheduled to perform at least \_\_\_\_\_\_(\_\_\_) Hours of Service per week Elected Officials (Employee employed exclusively by virtue of election and employed concurrent with a term of office) Temporary Employees (Employee expressly hired for a definite period less than ( ) months (must be less than twelve (12)) Seasonal Employees (Employee expressly hired for a specific task for a duration of less than twelve (12) months) Part-time Employees regularly scheduled for \_\_\_\_\_\_(\_\_\_\_) or more hours per week Eligible Employees for Optional Membership/Employee Election (Select all that apply) No Eligible Employees Elected Officials (Employees employed exclusively by virtue of election and employed concurrent with a term of office) Temporary Employees (Employee expressly hired for a definite period less than \_\_\_\_ (\_\_\_\_) months (must be less than twelve (12)) Seasonal Employees (Employee expressly hired for a specific task for a duration of less than twelve (12) months) Part-time Employees regularly scheduled for ( ) or more hours per week

1.34

1.35

**ELIGIBILITY SERVICE.** 

#### Ineligible Employees

All individuals not otherwise identified as Eligible Employees in this Section Individuals specified as follows:

(Category cannot be age or service-based and cannot name specific individuals or a finite group)

## The average annual annualized Compensation earned and paid during the: 1.38 FINAL AVERAGE SALARY. (Select One) The average annual annualized Compensation earned and paid during the: Average Monthly Compensation (\_\_\_\_\_) consecutive calendar months of the most recent \_\_\_\_ consecutive calendar months of Credited Service which produces the highest Final Average Salary Average Annual Compensation \_\_\_\_\_(\_\_\_) consecutive calendar years of the most recent \_\_\_\_\_(\_\_\_\_\_(consecutive calendar years of Credited Service which produces the highest Final Average Salary Final Monthly Compensation \_ (\_\_\_\_) most recent consecutive calendar months of Credited Service which determines the Final Average Salary 1.41 HOURS OF SERVICE. An Employee shall be awarded a Year of Service upon completion of \_\_\_\_\_ (\_\_\_\_\_) Hours of Service. (The number of Hours of Service cannot exceed 2,000.) . 1.61 **NORMAL RETIREMENT AGE (Applicable to In-Service Distributions)** In-Service Distributions Not Allowed In-Service Distributions Allowed in Accordance with Section 10.05 (Continuing Employment after Normal Retirement Age) In-Service Distributions Allowed in Accordance with Section 10.06 (Re-employment after Benefit Commencement Date) Normal Retirement Age for Allowable In-Service Distributions Eligible Employees (Select all that apply below) Date the Member attains age sixty-five (65) Date the Member attains age sixty-two (62) The later of the date: Member attains age sixty (60) and Member completes five (5) Years of Vesting Service The later of the date: Member attains age fifty-five (55) and

Member completes ten (10) Years of Vesting Service

	Date the Member's age plus completed Years of Vesting Service equals or exceeds eighty (80)
	Date the Member's age plus completed Years of Vesting Service equals or exceeds() (The numerical value shall not be less than eighty (80))
	Date the Member completes twenty-five (25) Years of Vesting Service (Must be combined with one of the other options)
	To use the three options below, the date the Member meets the specified conditions cannot be greater than the date the member attains age 65 nor less than the date the member attains age 55 and, in any event, may not be less than the representative typical retirement date for such Member's occupation.
	Date the Member attains age():
	The later of the date:
	Member attains age () Member completes () Years of Vesting Service
	Other:
	Police Officers, Firefighters and other qualified public safety employees as defined under Code Section 72(t)(10) have the following additional options (Select all that apply below)
	Date the Member attains age fifty (50)
	Date the Member completes twenty (20) Years of Vesting Service
	Date the Member's age plus completed Years of Vesting Service equals or exceeds seventy (70)
	Date the Member's age plus completed Years of Vesting Service equals or exceeds  (10) (The numerical value shall not be less than seventy (70))
	To use the three options below, the date the Member meets the specified conditions cannot be greater than the date the member attains age 65 nor less than the date the member attains age 50 and, in any event, may not be less than the representative typical retirement date for such Member's occupation.
	Date the Member attains age():
	The later of the date:
	Member attains age ()
	Member completes() Years of Vesting Service
	Other:
	(Normal Retirement Age designation cannot be earlier than Superannuation Retirement Pension eligibility under Section 5.01.)
1.69	PLAN.
	Plan name as adopted by the Municipality:
	Defined Benefit Plan

# 1.70 PLAN ENTRY DATE. **Employee's Employment Commencement Date** After Completion of Eligibility Criteria under Section 2.01 First day First day of the following month First day of the following calendar year quarter First day of following calendar year Other: (Option must be completed in a manner that results in Member entering Plan on the earlier of: (i) the first day of the plan year beginning after the date on which the employee has met the minimum age and service requirements; or (ii) six month after the date the requirements are met.) 1.71 **PLAN YEAR** The Plan Year shall be the: Calendar Year Twelve month period beginning \_\_\_\_\_ and ending \_ 1.97 **VESTING SERVICE.** Method of Measurement (Select One) Hours of Service Method Elapsed Time Method Adjustments to Vesting Service Include Exclude Service as Employee before the Municipality's PMRS Plan Enrollment Date Service as Employee before the Original Plan Effective Date Service as Employee under the non-PMRS Original Plan Service as Employee before the terminated Prior Plan Service as Employee under the terminated Prior Plan Service as Employee after the terminated Prior Plan Name of terminated Prior Plan: \_\_

Service as Employee before the merged Prior Plan Service as Employee under the merged Prior Plan Service as Employee after the merged Prior Plan

Name of merged Prior Plan:

# ARTICLE II: MEMBERSHIP

2.01	MEMBERSHIP ELIGIBILITY.
	Eligibility Date (Select One)
	Employee's Employment Commencement Date
	Date on which the Employee completes () months of Eligibility Service
	Date on which the Employee completes () days of Eligibility Service
	Other date (specify):
	(Service requirements cannot exceed twelve (12) months.)
	ARTICLE III: MUNICIPALITY CONTRIBUTIONS
3.03	TIME OF PAYMENT OF CONTRIBUTION.
	Contribution Remittance (Select One)
	On an annual basis (not later than December 31)
	At the same time and frequency of Member's contributions as specified in Section 4.01 or Section 4.02
4.01	MEMBER PRE-TAX PICK-UP CONTRIBUTIONS.
	Contribution Requirement (Select One)
	Not Required
	Required in an amount equal to percent (%) of Compensation
	Contribution Remittance (Select One)
	On a payroll basis
	Per Weekly Period
	Per Bi-Weekly Period
	Per Semi-Monthly Period
	Per Monthly Period
	Per Other Period:
	On a monthly basis
	On a quarterly basis
4.02	MEMBER AFTER-TAX CONTRIBUTIONS.
	Contribution Requirement (Select all that apply)
	Not Permitted
	Required in an amount equal to percent () of Compensation (Section 4.02(a))

	Contribution Remittance (Select One)
	On a payroll basis
	Per Weekly Period
	Per Bi-Weekly Period
	Per Semi-Monthly Period
	Per Monthly Period
	Per Other Period:
	On a monthly basis
	On a quarterly basis
4.03	OTHER MEMBER CONTRIBUTIONS.
	Service Purchase Contributions (Select one)
	Not Permitted
	Permitted
	<u>Prior Plan Transfer Contributions</u> (Select one)
	Not Permitted
	Permitted
	Prior Plan Termination Contributions (Select one)
	Not Permitted
	Permitted
	(Such contributions shall be subject to the limits on the purchased Permissive Service Credit (including Non-Qualified Service Credit) set forth in Article XI of the Base Plan Document.)
4.08	REPAYMENT OF MEMBER ACCOUNT.
	Minimum Repayment Amount (Select one)
	An amount not less than percent (%) of the previously refunded amount including interest
	An amount equal to the lesser of (a): the repayment for one (1) year of Credited Service or (b): 100% of the previously refunded amount including interest
	Repayment Time Period (Select one)
	At any time between the Member's Reemployment Commencement Date and the Member's subsequent Termination of Employment
	Repayment (not to exceed five (5) years) must be made within (complete only one):
	() days of the Member's Reemployment Commencement Date
	() months of the Member's Reemployment Commencement Date
	( ) years of the Member's Reemployment Commencement Date

# ARTICLE V: SUPERANNUATION RETIREMENT PENSION

#### 5.01 OFFERING OF SUPERANNUATION RETIREMENT PENSION.

Eligible Employees (Select all that apply below)
Date the Member attains age sixty-five (65)
Date the Member attains age sixty-two (62)
The later of the date:
Member attains age sixty (60) and
Member completes five (5) Years of Vesting Service
The later of the date:
Member attains age fifty-five (55) and
Member completes ten (10) Years of Vesting Service
Date the Member's age plus completed Years of Vesting Service equals or exceeds eighty (80)
Date the Member's age plus completed Years of Vesting Service equals or exceeds() (The numerical value shall not be less than eighty (80))
Date the Member completes twenty-five (25) Years of Vesting Service (Must be combined with one of the other options)
To use the three options below, the date the Member meets the specified conditions cannot be greater than the date the member attains age 65 nor less than the date the member attains age 55 and, in any event, may not be less than the representative typical retirement date for such Member's occupation.
Date the Member attains age():
The later of the date:  Member attains age()  Member completes() Years of Vesting Service
Other:
Police Officers, Firefighters and other qualified public safety employees as defined under Code Section 72(t)(10) shall have the following additional options (Select all that apply below)
Date the Member attains age fifty (50)
Date the Member completes twenty (20) Years of Vesting Service
Date the Member's age plus completed Years of Vesting Service equals or exceeds seventy (70)
Date the Member's age plus completed Years of Vesting Service equals or exceeds() (The numerical value shall not be less than seventy (70))
To use the three options below, the date the Member meets the specified conditions cannot be greater than the date the member attains age 65 nor less than the date the member attains age 50 and, in any event, may not be less than the representative typical retirement date for such Member's occupation.
Date the Member attains age():

	The later of the date:
	Member attains age ()
	Member completes () Years of Vesting Service
	Other:
5.02	AMOUNT OF SUPERANNUATION RETIREMENT PENSION.
	Shall be calculated using the following Pension Benefit Formula(s):
	For Basic Benefit (Select all that apply)
	Maximum Annual Benefit Amount
	percent (%) of Member Final Average Salary
	Single Tiered Formula (Unit Credit Plan)
	Member Final Average Salary multiplied by
	Fixed amount of dollars (\$) multiplied by the Member Years of Credited Service
	Multi-Tiered Formula (Unit Credit Plan)
	Member Final Average Salary multiplied by percent (%) multiplied by the Member Years of Credited Service during the period to
	Member Final Average Salary multiplied by percent (%) multiplied by the Member Years of Credited Service during the period to
	Member Final Average Salary multiplied by percent (%) multiplied by the Member Years of Credited Service during the period to
	For Service Increment Benefit (Select one)
	Not Permitted
	Two and one half percent (2.5%) of the Basic Benefit multiplied by completed Years of Credited Service in excess of twenty-five (25) subject to a maximum Service Increment Benefit of one-thousand two hundred dollars (\$1,200) annually
	Two and one half percent (2.5%) of the Basic Benefit multiplied by completed Years of Credited Service in excess of twenty-five (25) subject to a maximum Service Increment Benefit of one-thousand two hundred dollars (\$1,200) annually for each year of applicable Credited Service up to a maximum five (5) additional years of such Credited Service
	Fixed dollar amount of one-thousand two hundred dollars (\$1,200) annually after completion of one (1) Year of Credited Service in excess of twenty-five (25)

(Benefit accrual rate must be no less than 0.5% of Final Average Salary per year of Credited Service inclusive of Service Increment Benefit, if any)

# 5.04 INVOLUNTARY/VOLUNTARY LUMP SUM PAYMENT OF SUPERANNUATION RETIREMENT PENSION.

Involuntary Lump Sum Payment (Amounts less than \$5,000)
Required
Not Required
Voluntary Lump Sum Payment (Amounts less than \$5,000)
Permitted
Not Permitted

#### ARTICLE VI: EARLY RETIREMENT PENSION

#### 6.01 OFFERING OF EARLY RETIREMENT PENSIONS.

The Plan offers: (Select all that apply)

No Early Retirement Pension

An Unreduced Early Retirement Pension

A Reduced Early Retirement Pension

#### 6.02 ELIGIBILITY FOR EARLY RETIREMENT PENSION.

**Retirement Pension under Section 5.01.)** 

For Unreduced Early Retirement Pension (Select all that apply)
Date the Member attains () years of age
Date the Member completes () years of Vesting Service
Date the Member age plus the Member Years of Vesting Service equals or exceeds()
The later of the date:
Member attains () years of age
Member completes () years of Vesting Service
Date the Member is involuntarily terminated (provided that the Member has at least eight (8) years of Vesting Service)
For Reduced Early Retirement Pension (Select all that apply)
Date the Member attains() years of age
Date the Member completes () years of Vesting Service
Date the Member age plus the Member Years of Vesting Service equals or exceeds()
The later of the date:
Member attains () years of age
Member completes () years of Vesting Service
Date the Member is involuntarily terminated (provided that the Member has at least eight (8) years of Vesting Service)

DB201707-F Page 11 of 22 DB Adoption Agreement

(Eligibility criteria selected must be earlier than eligibility criteria for Superannuation

#### 6.03 <u>AMOUNT OF EARLY RETIREMENT PENSION.</u>

For Reduced Early Retirement Pension/Pension Benefit Reduction Factors (Select one)

Pension benefit reduction for each month between the Benefit Commencement Date and the Member Superannuation Retirement Date

Actuarial Reduction	using Actuari	al Equivalence	as	defined	in Secti	on 1.0	)4 of th	e Base	Plar
Document									
Annual rate of		percent (		_%)					
Other reductive factor	r or schedule (	must be unifor	n to	all Mer	nbers) (s	pecify	v):		

#### **ARTICLE VII: DISABILITY PENSION**

#### 7.01 OFFERING OF DISABILITY PENSION.

The Plan offers: (Select all that apply)

No Disability Pension

**Disability Pension** 

Service-Connected Disability Pension

#### 7.02 ELIGIBILITY FOR DISABILITY PENSION.

For Disability Pension: (Select all tha	nt apply)
Member's Plan Entry Date	
Date the Member attains	() years of age Date the
Member completes	() years of Vesting Service
The later of the date:	
Member attains	_() years of age
Member completes	() years of Vesting Service
For Service Connected Disability Per	nsion: (Select all that apply)
Member's Plan Entry Date	
Date the Member attains	() years of age
Date the Member completes	() years of Vesting Service
The later of the date:	
Member attains	() years of age
Member completes	( ) years of Vesting Service

# 7.03 **AMOUNT OF DISABILITY PENSION.** For Disability Pension (Select all that apply) Benefit Formula: percent ( %) of Final Average Salary percent ( %) of Superannuation Retirement Pension calculated to reflect the Member's Final Average Salary and Credited Service as of the date of Disability. Adjustments/Reductions to the Benefit Payments: No adjustments or reductions Payments under the "The Pennsylvania Workmen's Compensation Act" Payments under the "The Pennsylvania Occupational Disease Act" Payments under the Social Security Disability Insurance (SSDI) Program For Service Connected Disability Pension (Select all that apply) Benefit Formula: percent (\_\_\_\_\_%) of Final Average Salary percent ( %) of Superannuation Retirement Pension calculated to reflect the Member's Final Average Salary and Credited Service as of the date of Disability. Adjustments/Reductions to the Benefit Payments: No adjustments or reductions Payments under the "The Pennsylvania Workmen's Compensation Act" Payments under the "The Pennsylvania Occupational Disease Act" Payments under the Social Security Disability Insurance (SSDI) Program ARTICLE VIII: DEFERRED VESTED PENSION 8.04 COMPUTATION AND PAYMENT OF DEFERRED VESTED PENSION. <u>Involuntary Lump Sum Payment (Amounts less than \$5,000)</u> Permitted

DB201707-F

Not Permitted

Permitted
Not Permitted

Voluntary Lump Sum Payment (Amounts less than \$5,000)

## **VESTING SCHEDULE.** 100% Vested upon Plan Entry Date 100% Vested upon Specified Years of Vesting Service % Vested Less than \_\_\_\_\_ ( \_\_\_ ) years of Vesting Service 0% Equal to or greater than \_\_\_\_\_(\_\_\_) years of Vesting Service 100% (Not to exceed fifteen (15) years of Vesting Service) Percent Vested upon Completed Years of Vesting Service % Vested % ( \_\_\_) years ( ) years % ( ) years \_\_\_\_% \_\_\_\_(\_\_\_) years \_\_\_\_\_ (\_\_\_\_) years % \_\_\_\_\_ (\_\_\_\_) years (Not to exceed twenty (20) years of Vesting Service) ARTICLE IX: DEATH BENEFITS PRE-RETIREMENT DEATH BENEFIT. Offering of Pre-Retirement Death Benefit Not Offered (Distribution of Minimum Death Benefit only) Offered to a Member who has met the requirements: (Select all that apply) To receive a Superannuation Retirement Pension under Section 5.01 To receive an Early Retirement Pension under Section 6.02 Of minimum Vesting Service under Section 8.05 Completion of \_\_\_\_\_ (\_\_\_\_) Years of Vesting Service under Section 8.05 PAYMENT OF RETIREMENT BENEFIT - NORMAL & OPTIONAL FORMS OF DISTRIBUTION

# ARTICLE X:

#### 10.01 NORMAL FORM OF BENEFIT.

8.05

9.01

Single Life Annuity for the life of the Member

Single Life Annuity with 50% Spouse/Minor Children Survivor

Payable for the life of the Member, and one-half the monthly amount payable for the life of the Spouse or Minor Children as applicable following the death of the Member.

Single Life Annuity with 100% Spouse/Minor Children Survivor

Payable for the life of the Member, and the same monthly amount payable for the life of the Spouse or Minor Children as applicable following the death of the Member.

(All Normal Form payments shall be adjusted as appropriate to comply with Code Section 415(b))

#### 10.02 OPTIONAL FORMS OF BENEFIT.

The optional forms of benefit offered to the Member.

Single Life Annuity with Minimum Net Present Value.

Single life annuity with guaranteed total payment equal to the Actuarial Equivalence of straight life annuity determining as Benefit Commencement Date

Single Life Annuity with Ten (10) Years of Certain Payments

Payable for the life of the Member, with a minimum 120 monthly payments if the Member death occurs prior to 120 monthly payments being paid to the Member

Single Life Annuity with Twenty (20) Years of Certain Payments

Payable for the life of the Member, with a minimum 240 monthly payments if the Member death occurs prior to 240 monthly payments being paid to the Member

Joint Life Annuity with 50% Survivor Annuitant Benefit

Payable for both the life of the Member and the Survivor Annuitant, with the Survivor Annuitant receiving 50% of the monthly benefit that had been received by the Member

Joint Life Annuity with 100% Survivor Annuitant Benefit

Payable for both the life of the Member and the Survivor Annuitant, with the Survivor Annuitant receiving 100% of the monthly benefit that had been received by the Member

Lump Sum Payment of Employee Contributions and Excess Interest Investment Account with a Normal or Optional Form of Annuity Benefit

The Normal or Optional Form of Benefit will be actuarially reduced in an amount equal to the lump sum payment.

#### 10.03 COST OF LIVING ADJUSTMENT.

# Offering of Cost of Living Adjustment ("COLA") Not Offered Offered at the Discretion of the Municipality

An ad hoc, one-time adjustment elected by the Municipality and implemented with a Plan amendment or addendum specifying the COLA amount, the eligible payees and the beginning adjustment date.

adjustment dute.
Offered upon allocation of Excess Interest under Section 15.10
Proportionately by Monthly Retirement Benefit (equal percentage)
In Equal Dollar Amounts per Eligible Payee
Offered on a Continuing Basis and Applied Every () Year(s)
Fixed applied rate in the amount of percent (%)
Adjustable rate in an amount equal to the most recent calendar year annual change in the Consumer Price Index ("CPI COLA")
Maximum and Minimum Limits on CPI COLA
No Maximum and Minimum Limits
Maximum and Minimum Annual Limits Follows: (Select all that apply)
Maximum limit of percent (%)
Minimum limit of percent (%)
Maximum Cumulative Limits Based on: (Select all that apply)
percent (%) of Final Average Salary
percent (%) of Initial Monthly Retirement Benefit
percent (%) of Simple Addition of COLA Percentage Rates
Payees Eligible for COLA (Select all that apply)
All Retired Members
Retired Members having Retired on or before: (insert date)
Retired Members having Retired on or after: (insert date)
Other Eligibility for Retired Members: Having a balance in an Excess Interest Account in

#### 10.05 CONTINUATION OF EMPLOYMENT AFTER NORMAL RETIREMENT AGE.

accordance with Section 15.10

A Member continuing as an Employee (as defined in Section 1.34 of this Adoption Agreement) after Normal Retirement Age:

May elect to Retire and commence payment of Superannuation Retirement Benefit without a Severance from Credited Service or a Termination of Employment

Shall not be able to commence Superannuation Retirement Benefit without a Termination of Employment or Retirement

#### 10.06 REEMPLOYMENT OF RETIRED MEMBER.

After Reemployment Commencement Date as an Employee (as defined in Section 1.34), the Retired Member:

Shall elect to continue or cease receiving his Accrued Benefit

Shall cease receiving his Accrued Benefit

# ARTICLE XI: MISCELLANEOUS PROVISIONS AFFECTING THE CREDITING OF SERVICE

11.01	SERVICE UPON REEMPLOYMENT. (Select one)
	Not Applicable; Plan has Member Contributions
	Credited Service shall be restored regardless of Breaks in Service
	Credited Service shall be restored if the Member has incurred less than () consecutive Breaks in Service (must be equal to or greater than five (5))
11.02	SERVICE PRIOR TO ORIGINAL PLAN EFFECTIVE DATE.
	Inclusion of Credited Service Prior to the Original Plan
	Not Included
	Included subject to the following limits: (Select all that apply)
	No limit on Credited Service Prior to Original Plan
	A maximum percent (%) of such Credited Service A
	maximum () years of such Credited Service
	Excluding such Prior Credited Service as follows:
	Member Purchase of Credited Service Prior to Original Plan
	Not permitted to purchase such Credited Service
	Member is permitted to purchase such Credited Service as follows: (Select one)
	purchased. percent (%) of the Actuarial Equivalence of the Service to be
	Other method or formula as follows:

	Member Payment Time Period for Purchase of Credited Service Prior to Original Prior Payment Time Period for Purchase of Credited Service Prior to Original Prior Payment Time Period for Purchase of Credited Service Prior to Original Prior Prior Payment Time Period for Purchase of Credited Service Prior to Original Prior Pri	ginal Plan			
	(Select all that apply)				
	Member payment must be made in whole within () days of amount by the Municipality	of payment n	otification		
	Member may elect to make the required payment by payroll deduction for () years (maximum five (5) years).	a period not	to exceed		
	The payment will be adjusted to include interest at the rate estable compounded annually. Interest shall begin on the first day of the month of notification of the payment amount by the Municipality shall end on the preceding the final payment	the month following			
	Other time period as follows:				
11.03	Payments over multiple periods will be adjusted to include interest at the rate Board compounded annually. Interest shall begin on the first day of the morn month of notification of the payment amount by the Municipality shall end of month preceding the final payment  QUALIFIED MILITARY SERVICE.	th following	g the		
	Intervening Military Service				
	Return to Employment	Yes	No		
	Repayment of Mandatory Employee Contributions required to receive Credited Service				
	Death During Military Service				
	Credited Service granted to Members who die during Qualified Credited Military Service				
	Disability During Military Service				
	Credited Service granted to Members who become Disabled during Qualified Credited Military Service				
	Vesting Service granted to Members who become Disabled during Qualified Credited Military Service				

#### 11.04 TRANSFER OF SERVICE AND ASSETS FROM ANOTHER SYSTEM PLAN.

Plan will allow the transfer of Member Credited Service, Vesting Service and Contribution Account assets from or to other System Plans

Plan will not allow the transfer of Member Credited Service, Vesting Service and Contribution Account assets from or to other System Plans

### 11.05 PRIOR PLAN SERVICE CREDIT.

<u>Transfer Contributions</u>
Credited Service Purchase allowed (Subject to the limits of Section 11.02)
Credited Service Purchase not allowed
<u>Termination Contributions</u>
Credited Service Conversion allowed (Subject to the limits of Section 11.02)
Credited Service Conversion not allowed
Inclusion of Prior Plan Credited Service
Not Included
Included subject to the following limits: (Select all that apply)
Limited to Credited Service Before the Prior Plan
Limited to Credited Service During the Prior Plan
Limited to Credited Service After the Prior Plan
A maximum percent (%) of such Limited Credited Service
A maximum () years of such Limited Credited Service
Excluding such Limited Credited Service as follows:
Member Purchase of such Prior Plan Credited Service
Not permitted to purchase such Credited Service
Member is permitted to purchase such Credited Service as follows: (Select one)
percent (%) of the Actuarial Equivalence of the Service to be purchased.
Other method or formula as follows:
Member Payment Time Period for Purchase of Prior Plan Credited Service
(Select all that apply)
Member payment must be made in whole within () days of payment notification amount by the Municipality
Member may elect to make the required payment by payroll deduction for a period not to exceed () years (maximum five (5) years).
Payments over multiple periods will be adjusted to include interest at the rate established by the Board compounded annually. Interest shall begin on the first day of the month following the month of notification of the payment amount by the Municipality shall end on the last day of the month preceding the final payment.
Other time period as follows:
Payments over multiple periods will be adjusted to include interest at the rate established
by the Board compounded annually. Interest shall begin on the first day of the month following the month of notification of the payment amount by the Municipality shall end

on the last day of the month preceding the final payment

#### 11.06 TRANSFER OF LOANS.

Transferred from Prior Plan

Shall be permitted

Shall not be permitted

#### 11.07 PERMISSIVE SERVICE CREDIT.

The Plan will not allow the purchase of Permissive Service Credit The Plan will allow the purchase of Permissive Service Credit

Maximum Amount of Allowable Permissive Service Credit

No limit on allowable years of service

Years of service allowed to be purchased shall not exceed \_\_\_ ( ) years

#### Applicable Service Types (Select all that apply)

Federal government service

Commonwealth of Pennsylvania government service (including school district service)

Government service with Municipalities located within the Commonwealth of Pennsylvania

Government service with states other than the Commonwealth of Pennsylvania:

Government service with other municipalities located outside the Commonwealth of Pennsylvania:

#### 11.08 NON-QUALIFYING SERVICE CREDIT.

The Plan will not allow the purchase of Nonqualified Service Credit

The Plan will allow the purchase of Nonqualified Service Credit

Maximum Amount of Allowable Nonqualified Service Credit

Years of Nonqualified Service Credit allowed to be purchased shall not exceed \_\_\_\_\_(\_\_\_) years (maximum of five (5))

#### ARTICLE XIII: ADMINISTRATION

#### 13.04 NOTICE TO MUNICIPALITY.

Municipality:				, Pennsylvania
Employer Identifica	tion Number:			
Address 1:				
Address 2:				
City, ST Zip:		, PA		
Contact Name:				
Contact Position:				
Phone Number:	()	Fax Number:	()	
Email Address				

### ARTICLE XV: PENNSYLVANIA MUNICIPAL RETIREMENT FUND

#### 15.10 ALLOCATION OF EXCESS INTEREST.

**Municipality Allocation** 

Between three account types: Municipal Account, Active and Deferred Vested Member Accounts and Payee Accounts (Select one)

[Signatures on the following page]

The Municipality hereby agrees to the provisions of this Adoption Agreement, Base Plan Document and Trust, and in witness of its agreement, the Municipality by its duly authorized officers has executed this Adoption Agreement, on the date specified below.

IN WITNESS WHEREOF, we have hereunto set our hands and seal the day, month and year above written.

ATTEST:		
BY:	BY:	Head of Governing Authority
		Head of Governing Authority
DATE:		
ATTEST		PENNSYLVANIA MUNICIPAL RETIREMENT BOARD
BY:	BY:	
PMRS Secretary		PMRS Board Chair
DATE:		
Approved as to form and legality:		
BY:	BY:	
Chief Counsel, PMRS		Office of General Counsel
BY:		
Office of Attorney General		

This Plan is an important legal document. Failure to properly fill out this Adoption Agreement may result in disqualification of this Plan. PMRS will inform you of any amendments made to the Base Plan Document. The PMRS mailing address for U.S. Postal Service delivery is P.O. Box 1165, Harrisburg, PA 17108-1165. The PMRS street address for overnight/courier service delivery is 1721 N. Front Street, 3<sup>rd</sup> Floor, Harrisburg, PA 17102-2315.

You may rely on an opinion letter issued by the Internal Revenue Service as evidence that this Plan is qualified under Code Section 401 only to the extent provided in Revenue Procedure 2015-36.

You may not rely on the opinion letter in certain other circumstances or with respect to certain qualification requirements, which are specified in the opinion letter issued with respect to the Plan and in Revenue Procedure 2015-36. In order to have reliance in such cases, an individual application for a determination letter must be made to Employee Plans Determinations of the Internal Revenue Service.

# **EXHIBIT 3**



# PENNSYLVANIA MUNICIPAL RETIREMENT SYSTEM DEFINED BENEFIT PLAN

#### **Adoption Agreement Amendment**

The undersigned, **Salisbury Township, Pennsylvania** ("Municipality"), pursuant to Article XVI of the Base Plan Document, is amending its Adoption Agreement having the effective date and the expiration date as shown below. The Municipality makes the following elections granted under the provisions of the Base Plan Document:

#### PLAN AMENDMENT

Amendment Effective Date:

Adoption Agreement Effective Date:

Application:

Amendment Expiration Date:

January 1, 2019

General

December 31, 2019

General Description: Establishes the Required Member Contribution amount for all

Active Members for Calendar year 2019 only.

Affected Members: All Active Members.

#### AMENDED ADOPTION AGREEMENT SECTIONS

The Sections of the Adoption Agreement below are applicable to this Amendment and will be effective for the Affected Members between Amendment Effective Date and the Amendment Expiration Date. All other sections of the applicable Defined Benefit Plan Adoption Agreement 001 remain the same during the period between the Amendment Effective Date and the Amendment Expiration Date.

4.01	MEMBER PRE-TAX PICK-UP CONTRIBUTIONS.
	Contribution Requirement (Select One)
[ ]	Not Required
[X]	Required in an amount equal to one percent (1.00%) of Compensation
	Contribution Remittance (Select One)
[X]	On a payroll basis
	[] Per Weekly Period
	[X] Per Bi-Weekly Period
	[ ] Per Semi-Monthly Period
	[ ] Per Monthly Period
	[ ] Per Other Period:
[]	On a monthly basis
[]	On a quarterly basis



The Municipality hereby agrees to the provisions of this Adoption Agreement Amendment, and in witness of its agreement, the Municipality by its duly authorized officers has executed this Adoption Agreement Amendment, on the date specified below.

IN WITNESS WHEREOF, we have hereunto set our hands and seal the day, month and year above written.

ATTEST:		SALISBURY TOWNSHIP
BY	BY	Head of Governing Authority
		Head of Governing Authority
DATE:		
ATTEST		PENNSYLVANIA MUNICIPAL RETIREMENT BOARD
BY:	BY	
Secretary		Board Chair
DATE:		
Approved as to form and legality:		
BY:	BY	
Chief Counsel, PMRS		Office of General Counsel
BY:		
Office of Attorney General		

This Plan is an important legal document. Failure to properly fill out this Adoption Agreement Amendment may result in disqualification of this Plan. PMRS will inform you of any amendments made to the Base Plan Document. The address of PMRS is 1721 North Front Street, Harrisburg, PA 17102.

You may rely on an opinion letter issued by the Internal Revenue Service as evidence that this Plan is qualified under Code Section 401 only to the extent provided in Revenue Procedure 2015-36.

You may not rely on the opinion letter in certain other circumstances or with respect to certain qualification requirements, which are specified in the opinion letter issued with respect to the Plan and in Revenue Procedure 2015-36. In order to have reliance in such cases, an individual application for a determination letter must be made to Employee Plans Determinations of the Internal Revenue Service.

# **EXHIBIT 4**

# Salisbury Township Nonuniformed Employees Defined Contribution Pension Plan

#### **Article I — General Definitions**

§ 101	In General	9
§ 102	Accounting Date	9
§ 103	Administrator	9
§ 104	Alternate Payee	9
§ 105	Authorized Leave of Absence.	10
§ 106	Beneficiary	10
§ 107	Code	10
§ 108	Compensation.  (a) In General	10 10 11 11 11
§ 109	Effective Date	
§ 110	Eligible Spouse	
§ 111	Employer	
§ 112	ERISA	12
§ 113	Fiduciary	12
§ 114	Investment Manager	12
§ 115	Normal Retirement Age.	12
§ 116	Participant(a) Active Participant(b) Inactive Participant	12
§ 117	Plan or Plan and Trust	12
§ 118	Plan Year	13
§ 119	Qualified Employee.	13
§ 120	Related Employer	13
§ 121	Separation from Service.  (a) In General	14 14 14
§ 122	Sponsor.	
§ 123	Trust	15

§ 124	Trust Fund.	15
§ 125	Trustees	15
§ 126	Wages.	15
	Article II — Participation & Service	
§ 201	In General	16
§ 202	Entry Date	16
·	(a) In General	16
	(b) Rehired Employees(c) Uniformed Service	
§ 203	Required Information.	
§ 204	Year of Service	
g 204	(a) In General.	
	(b) Workers' Compensation	
	(c) Unpaid Leave.	
	(d) Service Before Age 18	17
	Article III — Accounting	
§ 301	Accounts	17
Ü	(a) Paper Accounts of Participants	
	(b) Paper Accounts of the Plan.	
	(c) Alternate Payee Accounts(d) Segregation of Assets	
	(1) Paper Accounts.	17
	(2) Segregated Accounts.	
§ 302	(3) Record KeepingIncome of the Trust Fund	
9 302	(a) Definition	
	(b) Annual Determination	
	(c) Special Determination.	
	(d) Termination of Trust(e) Valuation of Assets upon Segregation	
	(f) Return of Segregated Assets to General Pool of Investments	18
	(g) Distribution from Segregated Group of Assets(h) Other Determination.	
§ 303	Allocation of Income.	
8 000	(a) In General	
	(b) Early Employer Contributions Account	
§ 304	Valuation of Assets	19
§ 305	Beneficiaries.	19
Δrti	cle IV — Additions to Employer Contribution Accounts	
§ 401	Annual Allocation(a) In General	
	(b) Increase	
§ 402	Qualified Recipients	20
§ 403	Funding of Credits; Employer Contributions	20
§ 404	Payment of Employer Contributions.	
	(a) In General	
	(b) Uniformed Service(c) Additional Contributions for Interest on Late Contributions	
	(d) Mistake of Fact	

§ 405	Multiple Employers.	22
	(a) Employer Contributions	
	(b) Forfeitures.	22
	Article V — Employee Contributions	
0.504	• •	
§ 501	No Contributions	22
	Article VI — Maximum Additions	
§ 601	General Definitions	23
•	(a) Compensation.	23
	(1) In General	
	(2) Payment During the Limitation Year	
	(3) Payment Prior to Severance from Employment(4) Regular Pay After Severance from Employment	
	(5) Leave Cashouts	
	(6) Qualified Military Service	
	(7) Back Pay	
	(8) Maximum Amount Which May Be Treated As Compensatio	
	(A) General Rule	
	(B) Short Years(b) Employer	
	(c) Limitation Year.	
§ 602	Definitions Relating to Defined Contribution Limitations	
3 002	(a) Annual Additions.	
	(b) Defined Contribution Plan.	
	(c) Maximum Permissible Amount	
	(1) In General	
	(2) Short Year	25
§ 603	General Rule	25
§ 604	Procedure for Reducing Contributions	25
	(a) Timing of Reductions.	25
	(b) Priority vs. Medical Plans	
	(c) Priority vs. Earlier Defined Contribution Plan Allocations	26
	(d) Priority vs. Contemporaneous Allocations Under All Defined Contribution Plans	26
	(1) Nondeductible Employee Contributions	
	(2) Elective Deferrals	
	(3) Profit Sharing Employer Contributions	26
	(4) Money Purchase Employer Contributions	
	(5) Other Allocations.	
	(6) Forfeitures(e) Matching Contributions	
§ 605	Conformance to Code Section 415	27
	Article VII — Rollovers & Transfers to the Plan	
§ 701	Rollovers	27
_		
§ 702	Plan-to-Plan Transfers	27
	Article VIII — Vesting & Forfeitures	
§ 801	Vesting	27
-	(a) Employer Contribution Accounts.	
	(b) Normal Retirement Age	
	(c) Death or Disability	
	(d) Plan Termination	28

§ 802	Forfeiture of Nonvested Employer Contributions	28
§ 803	Application of Forfeitures.	28
§ 804	Restoration in the Case of Reemployment Following a Forfeitur  (a) Return of Formerly Non-Vested Participant.  (b) No Limitations.	28
	Article IX — Loans & Withdrawals	
§ 901	Loans	29
§ 902	Withdrawals.	29
	Article X — Beneficiaries	
§ 1001	Designation of Beneficiaries	29
§ 1002	Procedure	29
§ 1003	Revocation	30
§ 1004	Default Beneficiaries	30
	Article XI — Commencement of Benefits	
§ 1101	In General.	30
§ 1102	Commencement Date	30
	(a) Immediate Payment(b) Amounts Allocated After Commencement of Benefits	
§ 1103	Reemployment of Participant.	
§ 1104	Production of Information	
	Auticle VII - Forms of Deposits	
	Article XII — Form of Benefits	
§ 1201	Form of Distribution	
	(b) Deceased Participant	
§ 1202	Direct Rollovers of Distributions.	31
	(a) In General	
	(b) Definitions(1) Direct Rollover	
	(2) Distributee	
	(3) Eligible Rollover Distribution.	
	(4) Eligible Retirement Plan(A) In General	
	(B) After-Tax Employee Contributions.	
	(c) Automatic Rollovers.	
§ 1203	Compliance with Code § 401(a)(9)	33
	Article XIII — Benefits: Miscellaneous Provisions	
§ 1301	Provision of Benefits	33
§ 1302	Notice Requirements.	33
§ 1303	Spendthrift Provisions	33
	(a) General Rule.	
	(b) Qualified Domestic Relations Orders.	34

§ 1304	Qualified Domestic Relations Orders	34
	(a) Definition	
	(1) Rights Recognized(2) Required Provisions	
	(3) Prohibited Provisions	
	(4) Permitted Provision.	
	(b) Procedure	
	(1) Notification.	
	(2) Establishment of Procedure(c) Alternate Payee Accounts	
	(1) Creation	
	(2) Disposition.	35
	(A) To Alternate Payee	
	(B) Return to Participant's Accounts(d) Compliance with Qualified Domestic Relations Order	
\$ 120E	Facility of Payment.	
§ 1305		
§ 1306	Unclaimed Distribution.	
	(a) Segregation(b) Payment	
	(b) Payment(c) Escheat	
§ 1307	Survivors of Participant Who Dies While Performing Qualified	
8 1007	Military Service	36
	Article XIV — Claims Procedure	
§ 1401	Filing a Claim.	37
§ 1402	Notice of Denial	37
§ 1403	Review of Denial.	37
	(a) Petition	37
	(b) Rights	
	(c) Decision(d) Compliance with Local Agency Law	
	(a) complained that 200a / goldy _a	
	Article XV — Trust & Investments	
§ 1501	Establishment & Acceptance of Trust	38
§ 1502	Trustees	38
	(a) Qualification	38
	(b) Initial Trustees.	
	(c) Joint Trustees(d) Resignation	
	(e) Removal	
	(f) Successor & Additional Trustees	
	(g) Transfer of Assets to New Trustees	39
§ 1503	Investment of the Trust Fund	39
	(a) In General	
	(b) Location(c) Employer Securities	
C 1EO4		
§ 1504	Life Insurance Policies and Annuity Contracts	
	(b) Requirements for Permitted Contracts.	
§ 1505	Participant-Directed Investments	
§ 1506	Investment by Investment Managers	
2 1200	(a) In General	
	(b) Qualification	
	(c) Acceptance & Communication	

	(d) Security Transactions(e) Release and Indemnification of Trustee	
§ 1507	Other Powers of the Trustees.	
3 1007	(a) Purchase of Property.	
	(b) Disposition of Property	41
	(c) Exercise of Ownership Rights	
	(d) Registration of & Title to Investments(e) Borrowing	
	(f) Collection	
	(g) Retention of Cash	
	(h) Retention of Property Acquired	
	(i) Execution of Instruments	
	(j) Settlement of Claims & Debts(k) Employment of Agents & Counsel	
	(I) Incorporation	
	(m) Pooling of Assets	42
	(n) Legal Actions.	
	(o) Necessary Acts.	
§ 1508	Limiting Directions from the Administrator	
§ 1509	Distributions from the Trust Fund.	43
§ 1510	Administrative Payments.	43
	(a) Compensation of Trustees and Investment Managers	
	(b) Expenses(c) Taxes	
0 1511		
§ 1511	Accounting(a) Record Keeping	
	(b) Reports to Administrator.	
	(c) Discharge from Liability	
§ 1512	Immunity	44
3	(a) Persons to whom Responsible	
	(b) Ordinary Negligence.	
	(c) Permitted Reliance	
	(1) Action by the Administrator(2) Other Writings	
§ 1513	Purpose: Exclusive Benefit Rule	
	Standard of Care.	
§ 1514	Standard of Care	40
	Auticle VVI - Douticinent Directed Investments	
	Article XVI — Participant-Directed Investments	
§ 1601	Program Authorized	45
§ 1602	Definitions.	45
	(a) Affiliate	
	(b) Average Annual Total Return.	
	(1) In General(2) Transitional Rule	
	(c) Designated Investment Alternative	
	(d) Total Annual Operating Expenses	46
§ 1603	Investment Instructions.	
	(a) In General	
	(b) Restrictions and Procedures(1) In General	
	(2) Range of Risk and Return Characteristics.	
	(3) General Frequency Standard	
	(4) Core Investment Funds.	
	(5) Investment Alternative Available to Receive Transfers (6) Employer Securities	
	(7) Annuity Contracts	40

	(c) Compliance With Instructions.	
	(d) Unavailable Investments.	
§ 1604	Core Investment Funds	. 48
§ 1605	Notice of Limited Liability	. 49
§ 1606	Disclosure of Plan-Related Information	. 49
	(a) General Information.	
	(b) Administrative Expenses	
	(1) Initial and Annual Disclosure	
	(2) Quarterly Disclosure(c) Individual Expenses	
	(1) Initial and Annual Disclosure	
	(2) Quarterly Disclosure	50
	(d) Changes to Disclosed Information	51
§ 1607	Automatic and Periodic Disclosure of Investment-Related	
	Information	_
	(a) Identifying Information(b) Performance Data	
	(1) Return Not Fixed	
	(2) Fixed Return.	
	(c) Benchmarks	51
	(d) Fee & Expense Information.	
	(1) Return Not Fixed	
	(2) Fixed Return	
	(f) Glossary	
	(g) Annuity Options	
§ 1608	Special Rules Relating to Automatic Disclosure of Investment-	
	Related Information.	. 53
	(a) Annuity Options	
	(b) Fixed Return Investments.	53
§ 1609	Comparative Format For Furnishing Investment-Related Informat	ion.54
	(a) In General	
	(b) Additional Information.	
	(c) Use of Model Comparative Chart.	54
§ 1610	Investment-Related Information to be Provided Subsequent to Investment.	54
0.4044		-
§ 1611	Investment-Related Information to be Provided Upon Request	. 55
§ 1612	Miscellaneous Rules Relating to Disclosures	
	(a) Fees & Expenses	
0.4040	(b) Understandable Disclosures	55
§ 1613	Independent Control by Participants	
	(a) In General(b) Improper Influence	
	(c) Concealment of Material Non-Public Facts	
	(d) Incompetent Participant.	
	(e) Transactions Involving a Fiduciary	56
	(f) Investment Advice	56
§ 1614	Incidents of Ownership Appurtenant to Participant Investments	. 56
§ 1615	Segregation; Expenses	. 56
§ 1616	ERISA § 404(c) Requirements.	. 56
5		
	Article XVII — Administration	
§ 1701	In General	. 57
J -	(a) In General	
	(b) Members of the Committee	
	(c) Conduct of the Committee.	57

§ 1702	Powers & Duties	57
	(a) In General	
	(b) Delegation(c) Designation of Chief Administrative Officer	
	(d) Employment of Professionals & Others	
	(e) Records.	
	(f) Notifications	
	(g) Reports, Documents, and Communications	58
§ 1703	Direction of the Trustees	58
	(a) Direction to Request Approval	
	(b) Funding Policy and Method(c) Duty to Question Direction by Administrator	
§ 1704	Compensation & Expenses.	
_		
§ 1705	Standard of Care	58
	Article XVIII — Fiduciaries	
§ 1801	Prohibition Against Certain Persons Holding Positions under Plan.	
§ 1802	Bonding	59
§ 1803	Duty of Care	59
§ 1804	Duty of Loyalty.	60
Ū	(a) Self-Dealing	
	(b) Adverse Interests.	60
§ 1805	Prohibited Transactions	60
	(a) General Rule.	60
	(b) Prohibited Transactions.	
	(c) Disqualified Persons(d) Definitions	
§ 1806	Indemnification.	
3 1000	indennincation.	
	Article XIX — Amendment, Termination & Merger	
§ 1901	Amendment	61
	(a) In General	61
	(b) Prohibited Amendments	
	(1) Exclusive Benefit(2) Accrued Rights	
	(3) Trustees.	
§ 1902	Termination	
3 1302	(a) Right to Terminate Plan.	
	(b) Plan Accounts	
	(c) Termination & Liquidation of the Trust	
	(d) Termination of Trust Without Termination of Plan	
	(e) Termination or Spin-Off by Related Employer	
§ 1903	Merger of Plans; Transfer of Assets.	
	(a) Definition(b) Accrued Rights	
	Article XX — Miscellaneous	
§ 2001	Acquittance	63
§ 2002	Limitation of Liability.	63

§ 2003	Legal Actions	63
	(a) Necessary Parties	
	(b) Notice	
	(c) Final Judgment	63
§ 2004	Delegation of Authority by Employer	63
§ 2005	Clerical Errors.	63
§ 2006	Construction	64
§ 2007	Gender & Number	64
§ 2008	Headings	64
§ 2009	Severability.	64
§ 2010	Employment Rights.	64
§ 2011	Communications	
	(a) To the Administrator or Trustees	65
	(b) By the Administrator, Trustees, or Employer	65
§ 2012	Public Employee Pension Forfeiture Act	65
§ 2013	Type of Plan	65

### **Article I — General Definitions**

### § 101 In General.

When used in this Plan with initial capital letters, the words and phrases defined in the following sections of this Article shall have the following meaning, unless the context in which they are used clearly indicates a different meaning.

# § 102 Accounting Date.

The term "Accounting Date" shall mean the last day of each Plan Year.

#### § 103 Administrator.

The term "Administrator" shall mean the Plan Administrator described in Article XVII.

### § 104 Alternate Payee.

The term "Alternate Payee" shall mean a person entitled to receive, by virtue of a Qualified Domestic Relations Order (see § 1304), some of the benefits under this Plan of a Participant.

#### § 105 Authorized Leave of Absence.

The term "Authorized Leave of Absence" shall mean any absence authorized by the Employer under the Employer's standard personnel practices, *provided that* all persons under similar circumstances must be treated alike in the granting of such leaves. An absence due to service in the uniformed services of the United States shall be considered an Authorized Leave of Absence if the employee complies with all of the requirements of federal law in order to be entitled to reemployment and in fact does return to employment with the Employer within the period provided by law.

### § 106 Beneficiary.

The term "Beneficiary" shall mean a person who has the right to receive benefits under this Plan as a result of the death of a Participant or Alternate Payee. (See Article X for the method by which Participants may designate their beneficiaries).

#### § 107 Code.

The term "Code" shall mean the Internal Revenue Code of 1986, as amended (Title 26, U.S. Code). Reference to a section of the Code shall mean that section as it may be amended or renumbered from time to time, or any corresponding provision of any future legislation that amends, supplements or supersedes that section.

### § 108 Compensation.

(a) In General. Except as provided in subsections (b) through (e), the "Compensation" of a Qualified Employee for a given year (or other period for which a determination is being made) shall mean the Qualified Employee's total Wages from the Employer actually paid, made available, or includible in gross income for the year (or other determination period).

#### (b) Adjustments.

- (1) Certain Non-Taxable Compensation. Except as provided in paragraph (2), "Compensation" for a given year (or other determination period) shall also include amounts that would otherwise be includible in the Qualified Employee's gross income for the given year (or other determination period) but for the application of Code § 125 (relating to cafeteria plans), Code § 457 (relating to compensation deferred under an eligible deferred compensation plan for state and local governments and tax exempt organizations), Code § 414(h)(2) (relating to employee contributions to governmental plans that are picked up by the employing unit and thus are treated as employer contributions), or Code § 132(f)(4) (relating to qualified transportation fringes), or any elective deferrals (within the meaning of Code § 402(g)(3)) for the given year (or other determination period), but only with respect to contributions made to plans maintained by the Employer.
- (2) Cash Payments for Waiver of Benefits. "Compensation" shall *not* include cash payments made through a cafeteria plan due to the full or partial waiver of health or other coverage, even though such payments are includible in the Qualified Employee's Wages or gross income. The purpose of this subsection (b) is to insure that a Participant shall receive the same pension benefits under this Plan regardless of whether the Participant elects through a cafeteria plan to receive health or other coverage or waive the receipt of such coverage, and regardless of the health or other coverage options he/she elects through a cafeteria plan. "Compensation" is

the same as it would be if the Employer had maintained coverages which covered all Qualified Employees, required no employee contributions, and provided no incentives to Qualified Employees to select any particular coverage option.

- (c) Compensation During Periods of Uniformed Service. In the case of a period during which a Qualified Employee is serving in the uniformed services of the United States, the employee's "Compensation" shall be computed—
- (1) at the rate the Qualified Employee would have received but for the uniformed service; or
- (2) in the case that the determination of such rate is not reasonably certain, on the basis of the Qualified Employee's average rate of Compensation during the 12-month period immediately preceding the period of uniformed service (or, if shorter, the period of employment immediately preceding such period).

### (d) Maximum Amount Which May Be Treated As Compensation.

- (1) General Rule. The "Compensation" of a Qualified Employee for any given Plan Year shall not exceed the amount in effect for such year under Code § 401(a)(17), as adjusted for changes in the cost of living. (For any year beginning in 2019, the amount is \$280,000.00.)
- (2) Short Years. If Compensation is ever required to be determined for a period of time which contains fewer than 12 months, the amount of effect for such period under Code § 401(a)(17) shall be equal to the amount in effect under Code § 401(a)(17) for the calendar year in which the period begins, multiplied by a fraction whose numerator is equal to the number of months in the period, and whose denominator is equal to 12.
- (e) Modified Definition of Compensation for Purposes of Certain Provisions. For purposes of Article VI (relating to Maximum Additions), the term "Compensation" shall be modified as described in § 601(a).

#### § 109 Effective Date.

The "Effective Date" shall mean **June 1, 2019**, the date on which this Plan becomes effective.

# § 110 Eligible Spouse.

The term "Eligible Spouse" shall mean, with respect to any amount of benefit payments, the spouse to whom a Participant was married on the **earlier of** the date such benefit payments commenced under this Plan, **or** the date of his/her death (*except* to the extent a former spouse is to be treated as an Eligible Spouse under a Qualified Domestic Relations Order).

### § 111 Employer.

The term "Employer" shall mean the Sponsor and all Related Employers which have adopted this Plan, and their successors.

#### § 112 ERISA.

The term "ERISA" shall mean the Employee Retirement Income Security Act of 1974 (P.L. 93-406), as amended (29 U.S. Code § 1001 *et seq.*). Reference to a section of ERISA shall mean that section as it may be amended or renumbered from time to time, or any corresponding provision of any future legislation that amends, supplements, or supersedes that section.

### § 113 Fiduciary.

The term "Fiduciary" shall mean the Trustees, the Administrator, any Investment Manager, and any other person who exercises any discretionary authority or discretionary control respecting the management of the Plan; or who exercises any authority or control respecting the management or disposition of Plan assets; or who renders investment advice for a fee or other direct or indirect compensation with respect to any monies or property of the Plan or has any authority or responsibility to do so; or has any discretionary authority or discretionary responsibility in the administration of the Plan.

### § 114 Investment Manager.

The term "Investment Manager" shall mean an investment manager appointed under § 1506.

### § 115 Normal Retirement Age.

The "Normal Retirement Age" under this Plan shall mean age 60.

# § 116 Participant.

The term "Participant" shall mean an "Active Participant" or an "Inactive Participant":

- (a) Active Participant. An "Active Participant" shall mean a Qualified Employee who is currently an Active Participant in this Plan (see Article II).
- **(b) Inactive Participant.** An "Inactive Participant" shall mean any person, other than an Active Participant, who had previously been an Active Participant, and still has accounts with positive balances in the Plan.

#### § 117 Plan or Plan and Trust.

The terms "Plan" or "Plan and Trust" shall mean the **Salisbury Township Nonuniformed Employees Defined Contribution Pension Plan and Trust**, as set forth in this Plan document, as it may be amended from time to time.

#### § 118 Plan Year.

The term "Plan Year" shall mean any 12 consecutive month period beginning on January 1 and ending on the following December 31. However, the first Plan Year under this Plan shall be a short Plan Year beginning on June 1, 2019 and ending on December 31, 2019.

### § 119 Qualified Employee.

The term "Qualified Employee" shall mean, as of any given date, any person employed by the Employer for a stated salary or compensation in a position for which work is regularly scheduled for thirty-five (35) or more hours per week (or would be so scheduled except for authorized sick time, holidays, vacation time, leave, and similar paid or unpaid time off); provided such person is neither—

- (a) a person who was employed by the Employer at any time before Jure 1, 2019 in a position which qualified the person to participate in the Employer's defined benefit plan for nonuniformed employees maintained through the Pennsylvania Municipal Retirement System;
- **(b)** a nonresident alien who receives no remuneration from the Employer which constitutes income from sources within the United States (within the meaning of the Code):
- (c) a person who is included in a unit of employees covered by a negotiated collective bargaining agreement which does not provide for his/her inclusion as a Qualified Employee eligible for participation in this Plan, *provided that* retirement benefits were the subject of good faith bargaining and less than two percent of the employees of the Employer who are covered pursuant to that agreement are "professionals" as defined in Treas. Regs. § 1.410(b)-9(g);
- (d) an employee of a police department or fire department organized and operated by the Employer, if the employee provides police protection, firefighting services, or emergency medical services for any area within the jurisdiction of the Employer; nor
- (e) the Township Treasurer, the Township Tax Collector, a *per diem* employee or an elected official, other than those who are employed by the Employer in an independent employee status. In the later event, the person shall only be considered a Qualified Employee with respect to the independent employee status.

No self-employed individual or leased employee who is not a common-law employee of the Employer may be a Qualified Employee or a Participant in this Plan.

# § 120 Related Employer.

The term "Related Employer" shall mean any—

- (a) corporation which is a member of a controlled group of corporations (as defined in Code § 414(b)) which includes the Sponsor;
- **(b)** trade or business (whether or not incorporated) which is under common control (as defined in Code § 414(c)) with the Sponsor;
- (c) member of an affiliated service group (as defined in Code  $\S 414(m)$ ) which includes the Sponsor; and
- (d) other entity required to be aggregated with the Sponsor pursuant to Code § 414(o) and the regulations thereunder;

provided that for purposes of Article VI (relating to Maximum Additions), the definitions of Code §§ 414(b) and (c) shall be read as modified by Code § 415(h).

### § 121 Separation from Service.

- (a) In General. The term "Separation from Service" shall mean the end of a continuous period of employment of a given person by the Employer (or any Related Employer) and may result from retirement, death, resignation, involuntary termination, unauthorized absence, or by failure to return to active employment with the Employer (or any Related Employer) or to retire by the date on which an Authorized Leave of Absence expires. For purposes of the preceding sentence only, periods of Authorized Leaves of Absence and temporary lay-offs are considered to be periods of employment by the Employer. A person "Separates from Service" if he incurs a Separation from Service.
- **(b) Temporary Lay-Offs.** If the Employer (or any Related Employer) shall terminate a person's employment due to insufficient work for such person and shall indicate that the termination is temporary and that the Employer (or Related Employer) anticipates being able to re-employ the person within six (6) months, the termination shall be considered a "temporary lay-off" and not a "Separation from Service." In that case, if the person does not return to active employment with the Employer (or any Related Employer) immediately upon recall and within six (6) months, he shall incur a "Separation from Service" as of the **earlier** of:
  - (1) the date specified in any recall notice as the date to return to work, or
  - (2) the date six (6) months after the temporary lay-off began.
- (c) Transfers Among Related Employers. The term "Separation from Service" shall not include transfers between employers all of whom are included within the definition of "Employer" or "Related Employer," or the mere cessation of a person's status as a "Qualified Employee" if he remains in the employment of the Employer (or any Related Employer).

#### (d) Transfer of Business.

- (1) A person shall not incur a "Separation from Service" if the Employer or any Related Employer transfers the trade or business for which the person performs services to an unrelated transferee, but the person continues to work for the trade or business. Thereafter, the person shall incur a "Separation from Service" if he does so under the provisions of this § 121 as modified by substituting the transferee of the trade or business (and his related employers) for the Employer (and Related Employers).
- (2) A person shall not incur a "Separation from Service" if the corporation for which he works shall cease to be included within the definition of Employer or Related Employer (e.g., through the sale of its stock), but the person continues to work for the corporation. Thereafter, the person shall incur a "Separation from Service" if he does so under the provisions of this § 121 as modified by substituting the corporation for which he works (and its related employers) for the Employer (and Related Employers).
- (e) Uniformed Service. In the case of a person who is absent to perform service in the uniformed services of the United States and who could be entitled to reemployment with the Employer under the Uniformed Services Employment and Reemployment Rights Act of 1994, 38 U.S.C. § 4301 *et seq.*, following the completion of such service, the person shall incur a Separation from Service (and be entitled to commence the receipt of Plan benefits) as of the earliest of the following dates—
- (1) the date elected by the person (which may not be earlier than the date the person files the election with the Administrator);

- (2) the date of the person's death; or
- (3) the date the Administrator determines that the person no longer is legally entitled to reemployment with the Employer,

provided that if the person complies with all of the requirements of federal law in order to be entitled to reemployment and, does in fact return to employment with the Employer within the period provided by law, the person shall thereafter be treated as not having incurred a Separation from Service with respect to the period of uniformed service. The person shall not be required or permitted to return any benefits received from the Plan prior to his/her return to employment with the Employer.

### § 122 Sponsor.

The term "Sponsor" shall mean **Salisbury Township**, Lehigh County, Pennsylvania, a Pennsylvania first class township and municipal corporation, and its successors.

#### § 123 Trust.

The term "Trust" shall mean the trust established for this Plan in § 1501.

#### § 124 Trust Fund.

The term "Trust Fund" shall mean any and all assets held under the Plan or the Trust by the Trustees.

#### § 125 Trustees.

The term "Trustees" shall mean those individuals or corporations who, at any given time are the trustees of the Trust (see § 1502).

# § 126 Wages.

The term "Wages" shall mean wages as defined in Code § 3401(a) and all other payments of compensation to an employee by the Employer (in the course of such employers' trade or business) for which the Employer is required to furnish the employee a written statement under Code §§ 6401(d), 6051(a)(3), and 6052. See Treas. Regs. §§ 1.6041-1(a), 1.6041-2(a)(1), 1.6052-1, 1.6052-2, 31.6051-1(a)(1)(i)(C). Compensation shall be determined without regard to any rules under Code § 3401(a) that limit covered employment based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Code § 3401(a)(2)). (This amount is the amount shown on the "Wages, Tips, and Other Compensation" box on Form W-2 for federal income tax purposes.)

# **Article II — Participation & Service**

### § 201 In General.

In order to be eligible to become an Active Participant in this Plan, a person must be a Qualified Employee. The eligible person shall become an Active Participant at the time provided in § 202, and shall remain an Active Participant only so long as he/she remains a Qualified Employee. After the person ceases to be a Qualified Employee, he/she shall become an Inactive Participant until all of his/her Plan accounts are distributed, or until he/she becomes an Active Participant again.

### § 202 Entry Date.

- (a) In General. A person shall become an Active Participant as of the first day the person satisfies the eligibility requirements of § 201.
- (b) Rehired Employees. Subject to subsection (c), a person who Separates from Service and who later returns to employment with the Employer as a Qualified Employee shall become an Active Participant again upon his/her return to employment with the Employer as a Qualified Employee. A person who ceases to be a Qualified Employee but does not Separate from Service, and who becomes a Qualified Employee again shall become an Active Participant again on the date he becomes a Qualified Employee again.
- (c) Uniformed Service. A Qualified Employee who Separates from Service as a result of service in the uniformed services of the United States and who returns to employment with the Employer at a time when the Employer is legally obligated to reemploy the person under the Uniformed Services Employment and Reemployment Rights Act of 1994, 38 U.S.C. § 4301 *et seq.*, shall be treated as not having incurred that Separation from Service, shall be treated as having been a Qualified Employee during the period of such uniformed service, and shall be treated as having remained an Active Participant during the period of such uniformed service.

# § 203 Required Information.

The Administrator may require a Qualified Employee to submit relevant information to the Plan in connection with his/her entry into participation in this Plan. The Administrator shall be fully protected from any loss which may result from the Qualified Employee's failure to submit such information or from the Plan's reliance on incorrect information.

### § 204 Year of Service.

(a) In General. To determine the number of Years of Service credited to an individual, one divides the total number of days the person was employed by the Township as a Qualified Employee by three hundred sixty-five and one-quarter (365.25). For this purpose, time is generally measured for any continuous period of employment as a Qualified Employee from the first day the person performs an hour of service as a Qualified Employee, and time in non-continuous separate periods of employment as a Qualified Employee are aggregated.

- **(b)** Workers' Compensation. Notwithstanding subsection (a), days within a period for which payment is received under the Pennsylvania Workmen's Compensation Act shall not be counted.
- (c) Unpaid Leave. Notwithstanding subsection (a), any days while on an unpaid Leave of Absence shall not be counted, except for periods of uniformed service described in § 202(c) for which the person is considered to have remained a Qualified Employee or any other period required by law to be counted for retirement plan vesting purposes.
- (d) Service Before Age 18. Notwithstanding subsection (a), no days shall be counted which occur before the person's eighteenth (18th) birthday.

# **Article III — Accounting**

### § 301 Accounts.

- (a) Paper Accounts of Participants. The Administrator shall create and maintain adequate records to disclose the interest in the Trust Fund of each Participant who was a Qualified Employee at any time on or after the Effective Date and for each Beneficiary of such a Participant. Such records shall be in the form of individual accounts, created and closed as appropriate. Credits and charges shall be made to such accounts in the manner described in this Plan. Each such Participant and Beneficiary shall have an Employer Contribution Account to hold employer contributions to the Plan for the Participant, and the earnings thereon.
- **(b) Paper Accounts of the Plan.** The Administrator shall create and maintain adequate records to disclose the portions of the Trust Fund which represent amounts not currently allocated to the accounts of persons with an interest in the Plan. There shall be two such accounts created with respect to each particular Employer which maintains this Plan (unless there would be zero balances in any such account):
- (1) Forfeiture Account (to hold amounts forfeited from the accounts of Inactive Participants plus any earnings).
- (2) Early Employer Contributions Account (to hold amounts contributed to the Plan for a Plan Year before the end of that Plan Year; see § 404(a)).
- (c) Alternate Payee Accounts. Alternate Payee Accounts under this Plan and Trust shall be created at such times as provided in § 1304(c) (relating to Alternate Payee Accounts under Qualified Domestic Relations Orders).

#### (d) Segregation of Assets.

- (1) Paper Accounts. The accounts described in subsections (a) through (c) together account for all of the assets of the Trust Fund. The maintenance of these individual accounts is for accounting purposes only. A segregation of the assets of the Trust Fund to each account shall not be required.
- (2) Segregated Accounts. The Trustees may segregate the assets of the Trust Fund if they so desire. If the Trustees earmark any assets to the accounts of specific Participants, they shall first obtain the consent of the Participant or shall earmark such assets ratably among the accounts of all Participants.

(3) **Record Keeping.** Whenever assets are segregated, the Trustees shall maintain adequate records to disclose which of the accounts described in subsections (a) through (c) or portions of such accounts) are identified with which segregated group of assets.

### § 302 Income of the Trust Fund.

- (a) **Definition.** For the purposes of this Section, the "Income" of the Trust Fund or any group of assets shall mean the net gain or loss of the Trust Fund or group of assets from investments, as reflected by interest payments, dividends, realized and unrealized gains and losses on securities and other property, and administrative and other expenses paid from the Trust Fund or the group of assets. All administrative and other expenses which are fairly chargeable to one or more segregated groups of assets shall be charged to those assets alone, regardless of the fund from which those expenses were initially paid. All administrative and other expenses which are fairly chargeable to all of the assets of the Trust Fund shall be charged pro rata against all of the segregated groups of assets, regardless of the fund from which those expenses were initially paid.
- **(b) Annual Determination.** The Trustees shall determine the Income of the Trust Fund since the most recent determination as of each Accounting Date.
- (c) **Special Determination.** If a distribution of benefits is to begin from the general Trust Fund during a given month, the Trustees shall determine the Income of the Trust Fund since the most recent determination as of the end of the month immediately preceding the calendar month of the distribution.
- (d) Termination of Trust. The Trustees shall determine the Income of the Trust Fund since the most recent determination upon the Termination of the Trust and liquidation of its assets.
- (e) Valuation of Assets upon Segregation. The Trustees shall determine the Income of a group of assets at any time they choose to segregate certain assets from the larger group.
- (f) Return of Segregated Assets to General Pool of Investments. Whenever any segregated group of assets are to be merged with another group of assets, the Trustees shall determine the Income of the two groups of assets in the Trust Fund since their most recent valuation.
- (g) Distribution from Segregated Group of Assets. The Trustees shall determine the Income of a segregated group of assets whenever they are to make a distribution from such segregated group of assets.
- (h) Other Determination. The Trustees may determine the Income of the Trust Fund or any segregated portion of it since its most recent valuation at any time they think it prudent to do so (e.g., the Trustees may decide to determine the Income for certain segregated bank accounts, mutual funds, or other assets quarterly, monthly, or even daily); provided that elections to make such determinations are not made in a fashion likely to discriminate among persons with an interest in the Plan.

### § 303 Allocation of Income.

(a) In General. Except as provided in subsection (b), the Income of each segregated portion of the Trust Fund shall be allocated to the accounts described in § 301(a) through (c) as of the day such Income is determined, but only to those accounts which have a positive balance

on such day and which have been assigned to that segregated portion of the Trust Fund. If more than one of those accounts have been assigned to that segregated portion of the Trust Fund, the Income will be allocated among the accounts according to each account's portion of the following total:

- (1) the balances in the accounts as of the last asset valuation date (to the extent the accounts are assigned to that segregated portion of the Trust Fund);
- less (2) the amount of any distributions or transfers from the accounts during the period concerned (to the extent the segregated portion of the Trust Fund is used to make such distributions or transfers) multiplied by a fraction whose numerator is the number of days in the period after the date of distribution or transfer and whose denominator is the total number of days in the period;
- **plus (3)** the amount of any contributions or transfers to the accounts during the period concerned (to the extent the contributions or transfers are assigned to the segregated portion of the Trust Fund) **multiplied by** a fraction whose numerator is the number of days in the period after the date as of which the contributions or transfers were credited to the accounts and whose denominator is the total number of days in the period.
- (b) Early Employer Contributions Account. Any positive net Income which is allocated under subsection (a) to the Early Employer Contributions Account during a Plan Year shall be allocated as of the last day of the Plan Year among the Employer Contribution Accounts of the Qualified Recipients for that Plan Year by increasing the percentage in effect under § 401 (relating to Additions to Employer Contribution Accounts—Annual Allocation) as described in that section. All of such Income must be used to provide additional credits for Qualified Recipients beyond those which would have been received in the absence of such Income.

### § 304 Valuation of Assets.

In determining the value of Trust Fund assets for any purpose under this Plan, assets shall be valued on the basis of their fair market value as of the valuation date.

### § 305 Beneficiaries.

Accounts originally maintained on behalf of a deceased Participant shall be maintained on behalf of his current Beneficiaries (*see* Article X).

# Article IV — Additions to Employer Contribution Accounts

#### § 401 Annual Allocation.

(a) In General. Subject to the provisions of Article VI (relating to Maximum Additions), § 404(b) (relating to Uniformed Service), and subject to modification under subsection (b), as of the last day of each Plan Year, the Employer Contribution Account of each Qualified

Recipient (see § 402) shall be credited with an amount equal to **eight percent** (8.00%) of the Qualified Recipient's Compensation attributable to service as a Qualified Employee (or service in the uniformed services of the United States recognized under this Plan due to service as a Qualified Employee) for the portion of the Plan Year during which he/she was an Active Participant.

- **(b) Increase.** The percentage under subsection (a) shall be increased for any given Plan Year if and to the extent necessary so that the total amount of credits provided for that Plan Year under this Section and § 404(b) (relating to Payment of Employer Contributions—Uniformed Service), as modified by Article VI (relating to Maximum Additions) is not less than the sum of:
  - (1) the greater of—
- (A) the total amount which would be credited for all Qualified Recipients for that Plan Year under subsection (a) prior to the application of this subsection (b), plus the amount of credits during that Plan Year under § 404(b) (relating to Uniformed Service); or
- **(B)** the maximum amount of withdrawals able to be made for that Plan Year from the Forfeiture Account under § 403 (relating to Funding of Credits); **plus**
- (2) any Income to be allocated to the Employer Contribution Accounts of Qualified Recipients for the Plan Year under § 303(b) (relating to Early Employer Contributions Account); and
- (3) any amount to be allocated to the Employer Contribution Accounts of Qualified Recipients for the Plan Year under § 404(d)(2)(A) (relating to contributions of state aid in excess of the amount required).

### § 402 Qualified Recipients.

For purposes of this Article IV, a "Qualified Recipient" for any Plan Year shall mean a person who was an Active Participant on any day during the Plan Year.

# § 403 Funding of Credits; Employer Contributions.

The credits described in § 401 and § 404(b) shall be funded—

- (a) First, by withdrawals from the Forfeiture Account (including amounts forfeited on the last day of the Plan Year); and
- **(b)** Then, if necessary, by contributions to the Plan and Trust Fund from the Employer, which the Employer hereby covenants and agrees to make, which may be from funds of the Employer and/or grants from the Commonwealth of Pennsylvania and/or others.

# § 404 Payment of Employer Contributions.

(a) In General. Employer contributions under this Article for a given Plan Year are due to be paid to the Trustees not later than December 31 of the Plan Year. All amounts contributed before the end of the Plan Year shall be held unallocated in a separate Early Employer Contributions Account until the end of the Plan Year, when they shall be withdrawn and allocated as if they were contributed on the last day of the Plan Year.

- **(b) Uniformed Service.** Employer contributions under this Article IV for a Participant with respect to any period of service in the uniformed services of the United States shall be made at the *later* of—
  - (1) the time set forth in subsection (a); or
- (2) within a reasonable period of time after the Participant returns to employment with the Employer or any Related Employer,

provided that the Participant returns to employment at a time when the Employer or any Related Employer is legally obligated to reemploy the person under the Uniformed Services Employment and Reemployment Rights Act of 1994, 38 U.S.C. § 4301 *et seq.*, and any amendments, supplements, or successor legislation.

The amount of such contributions shall *not* be adjusted for any earnings or forfeitures which may otherwise have accrued to the benefit of the Participant during the period between the time when the contributions would have been made had the Participant not provided service in the uniformed services of the United States, and the time when the contributions were actually made.

- (c) Additional Contributions for Interest on Late Contributions. If any amount of Employer contributions for a Plan Year remains unpaid as of December 31 of that Plan Year, the amount of Employer contributions for that Plan Year shall be increased by interest on the unpaid amount as of December 31, from January 1 of the Plan Year until the date of payment at a rate equal to the interest assumption used for the required actuarial valuation report under the Municipal Pension Plan Funding Standard and Recovery Act, 53 PA. STAT. ANN. § 891.101 *et seq.*, or the discount rate applicable to treasury bills issued by the Treasury Department of the United States with a six-month maturity as of the last business day in December of the Plan Year, whichever is greater, expressed as a monthly rate and compounded monthly. Such "interest" contributions shall be treated as Income of the Plan for the period after the end of the Plan Year.
- (d) Mistake of Fact. The actual required Employer contributions for a given Plan Year cannot be known until the end of the Plan Year, since an employee's qualification for a contribution and the amount of his/her Compensation for the Plan Year cannot be determined until that time. Consequently, the estimates made by the Employer in determining the minimum municipal obligation for the Plan Year and in making contributions to the Plan during the Plan Year most likely will be either higher or lower than the actual required contributions. Therefore:
- (1) If the amount in the Early Employer Contributions Account as of the last day of the Plan Year is insufficient to cover the required allocations to Participants under § 401 (subject to modifications under Article VI, relating to Maximum Additions), the Employer must contribute the amount of the shortfall to the Plan as soon as possible, together with interest as described in subsection (c).
- (2) If there otherwise would remain a positive balance in the Early Employer Contributions Account as of the last day of the Plan Year after making the required allocations to Participants under § 401 (subject to modifications under Article VI, relating to Maximum Additions, and after any Income earned by the Early Employer Contributions Account for the Plan Year is allocated to other accounts as described in § 303(b) and § 401(b)(2)), then:
- (A) That portion of the Early Employer Contributions Account as of the last day of the Plan Year which otherwise would remain and which is attributable to grants by the Commonwealth under the General Municipal Pension System State Aid Program shall be transferred to the Employer's defined benefit plan for nonuniformed employees through the Pennsylvania Municipal Retirement System, together with the Income earned on that amount from the last day of the Plan Year until the date of the transfer, *except* that if such portion does not exceed Two

Hundred Dollars (\$200.00), then such portion shall instead be allocated as of the last day of the Plan Year among the Employer Contribution Accounts of the Qualified Recipients for that Plan Year by increasing the percentage in effect under § 401 (relating to Additions to Employer Contribution Accounts—Annual Allocation) as described in that section.

- **(B)** That portion of the Early Employer Contributions Account which is *not* attributable to grants by the Commonwealth under the General Municipal Pension System State Aid Program shall either be:
- (I) returned to the Employer (if so directed by the Administrator within one year after the date the contributions were made to the Plan); **or**
- (II) treated as a new Employer contribution for the following Plan Year, and retained in the Early Employer Contributions Account.
- (3) For purposes of this Plan, Employer contributions to the Early Employer Contributions Account shall be deemed to be attributable to grants under the General Municipal Pension System State Aid Program to the extent the governing body of the Employer allocates the grants to this Plan, and all amounts allocated to Participant accounts under § 401 and § 404(b) from the Early Employer Contributions Account shall be deemed to be attributable to grants under the General Municipal Pension System State Aid Program until the amount of such grants allocated to this Plan has been exhausted.

### § 405 Multiple Employers.

For any Plan Year in which more than one employer is included within the definition of "Employer," the following rules shall apply:

- (a) **Employer Contributions.** Employer contributions to be allocated to the account of a given Participant shall be made by his particular employer.
- **(b) Forfeitures.** For the purpose of allocating forfeitures, amounts derived from the contributions of a particular employer shall be allocated only to employees of that particular employer. Separate subaccounts shall be maintained in the Forfeiture Account for this purpose.

# **Article V** — Employee Contributions

#### § 501 No Contributions.

No employee contributions shall be required or permitted under this Plan.

# **Article VI — Maximum Additions**

### § 601 General Definitions.

When used in this Article, the words and phrases defined in this Section shall have the following meaning, unless the context in which they are used clearly indicates a different meaning:

- (a) Compensation. The term "Compensation" for a Participant for a Limitation Year shall mean:
- (1) In General. All wages within the meaning of Code § 3401(a) (for purposes of income tax withholding at the source), and all other payments of compensation to an employee by the Employer (in the course of such employers' trade or business) for which the Employer is required to furnish the employee a written statement under Code §§ 6401(d), 6051(a)(3), and 6052, plus amounts that would be included in wages but for an election under Code §§ 125(a), 132(f)(4), 402(e)(3), 402(h)(1)(B), 402(k), or 457(b). Compensation shall be determined without regard to any rules under Code § 3401(a) that limit covered employment based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Code § 3401(a)(2)).
- (2) Payment During the Limitation Year. Except as otherwise provided in this subsection (a), in order to be taken into account for a Limitation Year, amounts under paragraph (1) must be actually paid or made available to the Participant (or, if earlier, includible in the gross income of the Participant) within the Limitation Year. For this purpose, an amount is treated as paid on a date if it is actually paid on that date or it would have been paid on that date but for an election under Code §§ 125, 132(f)(4), 401(k), 403(b), 408(k), 408(p)(2)(A)(i), or 457(b).
- (3) Payment Prior to Severance from Employment. Except as otherwise provided in paragraphs (4) and (5), in order to be taken into account for a Limitation Year, amounts under paragraph (1) must be paid or treated as paid to the Participant (in accordance with the rules of paragraph (2)) prior to the Participant's severance from employment (within the meaning of Treas. Regs. § 1.415(a)-1(f)(5)) with the Employer. Thus, for example, "Compensation" generally does not include severance pay or parachute payments.
- (4) Regular Pay After Severance from Employment. Notwithstanding paragraph (3), a payment of an amount described in paragraph (1) after severance from employment will be considered "Compensation" for the Limitation Year which includes the date of severance from employment if it—
- (A) is regular compensation for services during the Participant's regular working hours, or compensation for services outside the employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments;
- **(B)** would have been paid to the Participant prior to a severance from employment if the Participant had continued in employment with the Employer; and
- (C) is paid by the *later* of 2.5 months after severance from employment with the Employer maintaining the plan or the end of the Limitation Year which includes the date of severance from employment.
- **(5) Leave Cashouts.** Notwithstanding paragraph (3), a payment of an amount described in paragraph (1) after severance from employment will be considered "Compensation" for the Limitation Year which includes the date of severance from employment if it—

- (A) is payment for unused accrued bona fide sick, vacation, or other leave, but only if the Participant would have been able to use the leave if employment had continued; and
- **(B)** is paid by the later of 2.5 months after severance from employment with the Employer maintaining the plan or the end of the Limitation Year which includes the date of severance from employment.
- (6) Qualified Military Service. For purposes of this subsection (a), a Participant who is in qualified military service (within the meaning of Code § 414(u)(5)) shall be treated as receiving Compensation from the Employer during such period of qualified military service equal to—
- (A) the Compensation the Participant would have received during such period if the Participant were not in qualified military service, determined based on the rate of pay the Participant would have received from the Employer but for absence during the period of qualified military service; or
- **(B)** if the compensation the Participant would have received during such period was not reasonably certain, the Participant's average Compensation from the Employer during the 12-month period immediately preceding the qualified military service (or, if shorter, the period of employment immediately preceding the qualified military service).
- (7) **Back Pay.** Payments awarded by an administrative agency or court or pursuant to a bona fide agreement by an employer to compensate an employee for lost wages are Compensation within the meaning of this subsection (a) for the Limitation Year to which the back pay relates, but only to the extent such payments represent wages and compensation that would otherwise be included in compensation under this subsection (a.)

### (8) Maximum Amount Which May Be Treated As Compensation.

- (A) General Rule. Notwithstanding anything to the contrary in this subsection (a), the "Compensation" of a Participant for any Limitation Year shall not exceed the amount in effect for such year under Code § 401(a)(17), as adjusted for changes in the cost of living. (For any Limitation Year beginning in 2019, the amount is \$280,000.00.)
- **(B) Short Years.** If Compensation is ever required to be determined for a short Limitation Year which contains fewer than 12 months, the amount of effect for such Limitation Year under Code § 401(a)(17) shall be equal to the amount in effect under Code § 401(a)(17) for the calendar year in which the Limitation Year begins, multiplied by a fraction whose numerator is equal to the number of months in the period, and whose denominator is equal to 12.
- **(b) Employer.** The term "Employer" shall mean the Sponsor and all Related Employers, and, to the extent required under Treas. Regs. § 1.415(f)-1, a predecessor employer within the meaning of that regulation.
- (c) Limitation Year. The term "Limitation Year" shall mean, for this Plan, those periods which are coextensive with the Plan Year after the Effective Date of this Plan. Limitation Years for other plans shall be as elected for those plans.

# § 602 Definitions Relating to Defined Contribution Limitations.

When used in this Article, the words and phrases defined in this Section shall have the following meaning, unless the context in which they are used clearly indicates a different meaning:

(a) Annual Additions. The term "Annual Additions", for a Participant in any given Limitation Year with respect to the Defined Contribution Plans maintained by the Employer, shall have the meaning given to the term under Treas. Regs. § 1.415(c)-1(b) (which generally

includes all employer contributions, employee contributions, and forfeitures credited to the Participant's accounts for the Limitation Year).

**(b) Defined Contribution Plan.** The term "Defined Contribution Plan" shall have the meaning provided in Treas. Regs. § 1.415(c)-1(a)(2) (including mandatory employee contributions to a defined benefit plan maintained by the Employer which are not treated as employer pick-up contributions under Code § 414(h)(2), and employee contributions to a separate account in a defined benefit plan maintained by the Employer to the extent that benefits are based on the separate account).

#### (c) Maximum Permissible Amount.

- (1) In General. Subject to the special rules relating to certain medical benefits and employee stock ownership plans under Treas. Regs. § 1.415(c)-1(e) and (f), the term "Maximum Permissible Amount", for any Limitation Year, shall mean the **lesser** of—
- (A) The defined contribution dollar limitation in effect for the Limitation Year under Code  $\S 415(c)(1)(A)$  as adjusted under Code  $\S 415(d)$  for changes in the cost of living (\$56,000 for 2019); or
  - **(B)** 100% of the Participant's Compensation for the Limitation Year.
- (2) Short Year. If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12 consecutive month period, for purposes of the short Limitation Year, the number in paragraph (1)(A) shall be multiplied by the following fraction:

#### number of months in the short Limitation Year (including fractional parts of a month) 12

If the Plan is terminated as of a date other than the last day of the Limitation Year, the Plan is deemed to have been amended to change its Limitation Year and the Maximum Permissible Amount shall be determined by prorating for the resulting short Limitation Year.

#### § 603 General Rule.

The amount of Annual Additions which would otherwise be allocated under this Plan on behalf of any Participant during any Limitation Year (whether of this Plan or any other plan of the Employer) shall be reduced (under the procedures of § 604) to the extent necessary and possible so that the total amount of Annual Additions which may be allocated on behalf of any Participant during that Limitation Year (whether of this Plan or any other plan of the Employer) under all of the Defined Contributions Plans maintained by the Employer shall not exceed the Maximum Permissible Amount.

# § 604 Procedure for Reducing Contributions.

- (a) **Timing of Reductions.** Reductions pursuant to this Article in the amount of contributions and allocations made on behalf of a Participant shall be made as soon as is administratively feasible, preferably before contributions and allocations are made.
- **(b) Priority vs. Medical Plans.** Annual Additions under this Plan shall be reduced in full before Annual Additions are reduced under any individual medical accounts (as defined in

Code  $\S 415(l)(2)$ ) or post-retirement medical accounts for key employees (as described in Code  $\S 419A(d)$ ) maintained by the Employer.

- (c) Priority vs. Earlier Defined Contribution Plan Allocations. Annual Additions under this Plan which are allocated as of later dates shall be reduced in full before any earlier allocations under this or any other defined contribution plan of the Employer are reduced.
- (d) Priority vs. Contemporaneous Allocations Under All Defined Contribution Plans. Annual Additions under this Plan which are allocated on the same day as other Annual Additions under this Plan or under other Defined Contribution Plans, shall be reduced according to the order of priority which follows (to the extent necessary). Where a full reduction is not necessary under any given category, the amount of Annual Additions to be reduced under this Plan shall be determined by the following product:

The total amount of Annual Additions to be reduced in that category.

The amount of Annual Additions allocated under this Plan in that category.
The total amount of Annual Additions allocated under all defined contribution plans in that category.

(1) Nondeductible Employee Contributions. First, nondeductible employee contributions under this Plan and other Defined Contribution Plans of the Employer shall be reduced.

X

- (2) Elective Deferrals. Second, elective deferrals under any other plans of the Employer shall be reduced.
- (3) Profit Sharing Employer Contributions. Third, allocations which would be attributable to nonelective employer contributions under profit sharing plans of the Employer shall be reduced. These reductions shall be used to increase the amount of additions to be allocated to Participants who have not reached their Code § 415 limit (as provided in this Article VI). After each such allocation, the provisions of this Article VI and similar Code § 415 provisions in other plans executed to that point shall be re-executed in accordance with the new allocations. Successive reductions and reallocations under this paragraph shall continue until all employer contributions are allocated to Participants consistent with the Code § 415 limitations or all eligible Participants have received the maximum amount permitted under this Article VI and similar Code § 415 provisions in other plans. Thereafter, the contributions from the Employer shall be reduced.
- (4) Money Purchase Employer Contributions. Fourth, allocations which would be attributable to nonelective employer contributions under this and other money purchase pension plans of the Employer shall be reduced.
- (5) Other Allocations. Fifth, any allocations not described in paragraphs (1) through (4) or (6) of this subsection (d), or in subsection (e), shall be reduced.
- **(6) Forfeitures.** Finally, allocations which would be attributable to withdrawals from forfeiture accounts of all Defined Contribution Plans of the Employer shall be reduced.
- **(e) Matching Contributions.** Allocations which would be attributable to employer matching contributions made with respect to employee contributions or elective deferrals under Defined Contribution Plans of the Employer that are reduced by virtue of this Article VI, shall not be made and the Employer shall not be required to make such contributions (since there are no underlying contributions to match).

#### § 605 Conformance to Code Section 415.

The limitations provided by this Article are intended to comply with Code § 415 and the regulations promulgated thereunder. To the extent there is any discrepancy between this Article and Code § 415 and related regulations, or any ambiguity in the terms of this Article, the discrepancy or ambiguity (whether this Article is more or less stringent than Code § 415 and related regulations) shall be resolved in such a way as to give full effect to the provisions of Code § 415 and regulations promulgated thereunder.

### Article VII — Rollovers & Transfers to the Plan

#### § 701 Rollovers.

No person may roll over any money or property to the Plan and Trust Fund which was received from other qualified plans, individual retirement accounts/annuities, Code § 403(a) or § 403(b) annuities, Code § 457 eligible deferred compensation plans, or any other plan, account, annuity, or arrangement (whether received directly or indirectly through an Individual Retirement Account or Annuity, and whether as a "Direct Rollover" under Code § 401(a)(31) or a rollover via the Participant).

# § 702 Plan-to-Plan Transfers.

The Trustees may not accept transfers of cash or assets to the Plan and Trust Fund from other qualified plans, individual retirement accounts/annuities, Code § 403(a) or § 403(b) annuities, Code § 457 eligible deferred compensation plans, or any other plan, account, annuity, or arrangement.

# Article VIII — Vesting & Forfeitures

# § 801 Vesting.

(a) Employer Contribution Accounts. Except as provided in subsections (b) through (d), the vested portion of the Employer Contribution Account of any Participant or Beneficiary shall be a percentage of the account balance determined in accordance with the following schedule (see § 204):

Years of Service	<u>Vested Portion</u>
less than 5	0%
5 or more	

- **(b) Normal Retirement Age.** The balance in all the Plan accounts created for any Participant who has attained the Normal Retirement Age (age 60) at a time when he is a Qualified Employee (or become a Qualified Employee after attaining the Normal Retirement Age) shall be 100% vested.
- (c) **Death or Disability.** The balance in all the Plan accounts of any Participant or Beneficiary shall be 100% vested at all times after the Separation from Service of the Participant for whom the account was created if the Separation from Service occurred due to the death of the Participant or as a result of a condition which rendered him/her unable to perform all of the essential functions of his/her employment with or without reasonable accommodation, *provided that* such condition was not caused by—.
  - (1) chronic or excessive use of intoxicants, drugs, or narcotics;
  - (2) intentionally self-inflicted injury or intentionally self-induced sickness; or
  - (4) an unlawful act or enterprise on the part of the individual.
- (d) Plan Termination. The balance in all the Plan accounts of any Participant or Beneficiary shall become 100% vested upon the termination or complete discontinuance of Employer contributions under this Plan and Trust. In the event of a partial termination of the Plan, the accounts of those Participants (and Beneficiaries) included in that part of the Plan which has terminated shall become 100% vested.

### § 802 Forfeiture of Nonvested Employer Contributions.

In the case of any Participant who is not 100% vested in all of his Plan accounts (see § 801), the nonvested portion of the Participant's Plan accounts shall be forfeited —

- (a) on the date the Participant Separates from Service, if he/she has no vested balance in any account under the Plan; and
- (b) on any date, after the Participant Separates from Service, that additional amounts are credited to his/her account under the Plan and he/she has no vested balance in any account under the Plan.

# § 803 Application of Forfeitures.

All funds forfeited under § 802 shall be allocated to the Forfeiture Account. Amounts in the Forfeiture Account under this Plan shall be applied on the last day of each Plan Year under the provisions of Article IV to provide allocations to the accounts of other Participants and reduce Employer contributions to the Plan for the given Plan Year.

# § 804 Restoration in the Case of Reemployment Following a Forfeiture.

(a) Return of Formerly Non-Vested Participant. If a Participant who suffered a forfeiture under § 802 becomes a Qualified Employee again, then the Employer shall contribute to the account(s) from which the forfeiture was made an amount equal to the amount which was

forfeited from such account (without any adjustment for imputed interest or imputed Plan Income or loss). The contributions specified in this subsection (a) shall take place as of the date the person becomes a Qualified Employee again.

**(b) No Limitations.** The Employer contributions under this § 804 shall not be subject to any limitations provided in Article IV (relating to Employer contributions) or Article VI (relating to Maximum Additions), and shall not be treated as Employer contributions for those purposes.

# Article IX — Loans & Withdrawals

#### § 901 Loans.

The Trustees shall not make any loans to Participants, Alternate Payee, or Beneficiaries from the Trust Fund.

#### § 902 Withdrawals.

A Participant or Alternate Payee may not elect to withdraw any funds from any of his accounts under this Plan, including his Employee Contribution Account, prior to the time for distribution under Article XI.

# **Article X — Beneficiaries**

# § 1001 Designation of Beneficiaries.

Each Participant and Alternate Payee (if permitted by the Qualified Domestic Relations Order) may designate any person or persons (natural or legal) as his/her Beneficiary or Beneficiaries to whom his/her Plan benefits are to be paid if he/she dies before receipt of all such benefits. Beneficiaries may be designated primarily, contingently, jointly, or successively.

# § 1002 Procedure.

Beneficiary designations shall be made on a form prescribed by the Administrator and will only be effective if filed with the Administrator during the Participant's or Alternate Payee's lifetime.

#### § 1003 Revocation.

Each effective beneficiary designation filed with the Administrator by a Participant or Alternate Payee will revoke all previously filed designations by such person. The revocation of a beneficiary designation shall not require the consent of any designated beneficiary or the Participant's spouse.

#### § 1004 Default Beneficiaries.

If a Participant (or Alternate Payee whose Beneficiary is entitled to receive Plan benefits) fails to designate a Beneficiary in the manner provided in § 1001 and § 1002, **or** if all the Beneficiaries designated by a deceased Participant or Alternate Payee die before him/her or before a complete distribution of his/her benefits, the benefits with respect to the Participant or Alternate Payee shall be paid to those of his/her survivor(s) who are highest in the following list:

- (a) his/her Eligible Spouse;
- **(b)** his/her surviving spouse;
- (c) his/her children, in equal parts;
- (d) his/her parents, in equal parts;
- (e) his/her estate.

# **Article XI — Commencement of Benefits**

# § 1101 In General.

The Plan benefits of a Participant (or his/her Beneficiary) shall commence after the Participant has Separated from Service. The Separation from Service may be for any reason, including normal retirement (age 60), death, disability, voluntary quit, or involuntary termination.

#### § 1102 Commencement Date.

Except as otherwise provided in this Article XI—

- (a) Immediate Payment. Benefits derived from the Participant's Plan accounts (as of the date of distribution) shall commence within ninety (90) days after the Participant Separates from Service.
- (b) Amounts Allocated After Commencement of Benefits. All benefits derived from amounts credited to a Participant's Plan accounts after his benefits commence under subsection (a), shall commence within ninety (90) days after they were contributed to the Plan and so credited.

### § 1103 Reemployment of Participant.

Notwithstanding anything to the contrary contained in this Article XI, no benefits shall commence to any Participant under § 1102 if the Participant is re-employed by the Employer or any Related Employer by the time the benefits would otherwise commence. In that case, such benefits shall only commence after the next event under this Article XI which permits or requires a distribution of benefits.

#### § 1104 Production of Information.

Notwithstanding anything to the contrary contained in this Article XI, no benefits shall be paid under this Plan to any recipient until an administratively reasonable period of time after the recipient shall file with or make available to the Plan Administrator such information as the Plan Administrator may require to determine that the recipient is entitled to receive such benefits under this Plan at that time, or to administer the payment of such benefits.

### Article XII — Form of Benefits

### § 1201 Form of Distribution.

- (a) Living Participant. Except as provided in § 1202 (relating to Direct Rollovers of Distributions) or § 1304 (relating to Qualified Domestic Relations Orders), if the Participant is living at the time benefits commence with respect to his Plan accounts, then the benefits which commence shall be distributed to the Participant in a lump sum cash distribution.
- **(b) Deceased Participant.** Except as provided in § 1202 (relating to Direct Rollovers of Distributions) or § 1304 (relating to Qualified Domestic Relations Orders), **if** the Participant is not living at the time benefits commence with respect to his Plan accounts, **then** the benefits which commence shall be distributed to the Participant's Beneficiary in a lump sum cash distribution.

# § 1202 Direct Rollovers of Distributions.

- (a) In General. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Distributee's election under this Article, a Distributee may elect, at the time and in the manner prescribed by the Administrator, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a Direct Rollover.
- **(b) Definitions.** When used in this Section, the words and phrases defined in this subsection shall have the following meaning:
- (1) **Direct Rollover.** A "Direct Rollover" is a payment by the Plan to the Eligible Retirement Plan specified by the Distributee.
- (2) Distributee. A "Distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former

employee's spouse or former spouse who is the Alternate Payee under a Qualified Domestic Relations Order, as defined in Code § 414(q), are "Distributee's" with regard to the interest of the spouse or former spouse. A "Distributee" also includes the employee's or former employee's nonspouse designated beneficiary, in which case the distribution can only be transferred to a traditional IRA (under Code § 408(a) or (b)) or Roth IRA (under Code § 408A) established on behalf of the nonspouse designated beneficiary for the purpose of receiving the distribution.

- (3) Eligible Rollover Distribution. An "Eligible Rollover Distribution" is any distribution of all or any portion of the balance to the credit of the Distributee, *except* that an Eligible Rollover Distribution does *not* include:
- (A) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's designated beneficiary, or for a specified period of ten (10) years or more;
- $(\mathbf{B})$  any distribution to the extent such distribution is required under Code § 401(a)(9); and
  - (C) any hardship distribution.

### (4) Eligible Retirement Plan.

- (A) In General. Except as provided in subparagraph (B), an "Eligible Retirement Plan" is—
- (I) an individual retirement account or annuity described in Code § 408(a) or (b);
- (II) a Roth individual retirement account or annuity described in Code § 408A;
- (III) a qualified trust described in Code § 401(a), including both defined benefit and defined contribution plans;
  - (IV) an annuity plan described in Code § 403(a);
  - (V) an annuity contract described in Code § 403(b); or
- (VI) an eligible deferred compensation plan described in Code § 457(b) which is maintained by an eligible governmental employer described in Code § 457(e)(1)(A), and which agrees to separately account for the amounts transferred into such plan from this Plan, that accepts the Distributee's Eligible Rollover Distribution.
- **(B) After-Tax Employee Contributions.** In the case of any portion of a distribution that consists of after-tax employee contributions which are not includible in gross income, an "Eligible Retirement Plan" is—
- (I) an individual retirement account or annuity described in Code § 408(a) or (b) or a Roth individual retirement account or annuity described in Code § 408A; or
- (II) a qualified plan under Code § 401(a), or an annuity contract described in Code § 403(b), that agrees to separate accounting for amounts so transferred (and earnings thereon), including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible,

that accepts the Distributee's Eligible Rollover Distribution.

(c) Automatic Rollovers. In the event of a mandatory distribution made from the Plan to a Participant in an amount greater than One Thousand Dollars (\$1,000.00), which is made before the Participant attains age 62 and without the Participant's consent, if the Participant does not elect to have such distribution paid directly to an Eligible Retirement Plan specified by the Participant in a Direct Rollover in accordance with this Section, and does not affirmatively elect to receive the distribution directly from the Plan at the time and in the manner prescribed by the

Administrator, then the Administrator will pay the distribution in a Direct Rollover to an individual retirement account or annuity described in Code § 408(a) or (b) designated by the Administrator. The Administrator shall notify the Participant of this possibility as required by law, and shall notify the Participant that the distribution may be transferred to another individual retirement plan. *See* IRS Notice 2005-5; 29 CFR 2550.404a-2.

### § 1203 Compliance with Code § 401(a)(9).

The provisions of this Plan are designed to satisfy the latest commencement of benefit and minimum distribution requirements of Code § 401(a)(9) and the regulations promulgated thereunder in that all benefits are to commence no later than ninety (90) days after the Participant's Separation from Service (which is earlier than the date benefits must commence from a governmental plan under Code § 401(a)(9)), and are paid in full in a lump sum as a cash distribution or a direct rollover. To the extent any provision of this Plan does not provide for the payment of the minimum distribution required through any given date under Code § 401(a)(9) and the regulations promulgated thereunder, this Plan shall be deemed amended in such a way as to comply with the minimum requirements of Code § 401(a)(9) and the regulations promulgated thereunder.

# **Article XIII — Benefits: Miscellaneous Provisions**

### § 1301 Provision of Benefits.

The Administrator shall direct the Trustees to provide benefit payments to the appropriate recipients from time to time in accordance with the provisions of this Plan. The **entire** vested portion of all accounts created with respect to a Participant shall be used to provide benefits for the Participant or his Beneficiaries or Alternate Payees under this Plan.

# § 1302 Notice Requirements.

The Administrator shall provide each person receiving benefits under this Plan with the notice required under Section 402(f) of the Code (regarding federal income tax treatment of Plan benefits and rollover rights) within 180 days prior to the date of distribution. To the extent possible, the notice shall be based on statements supplied by the U.S. Secretary of the Treasury.

# § 1303 Spendthrift Provisions.

(a) General Rule. Except as provided in subsection (b), benefits payable under this Plan shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, change, garnishment, execution, or levy of any kind, either voluntary or involuntary, including any such liability which is for alimony or other payments for support of a spouse, former spouse, or any other relative or dependent of the Participant before actually being received by the Participant, Former Participant, Beneficiary, or Alternate Payee under the terms of the Plan, except with respect to federal income tax withholding. Any attempt to anticipate,

alienate, transfer, assign, pledge, encumber, change, or otherwise dispose of any right to benefits payable under this Plan shall be void. The Trustees and the Employer shall not be liable for or subject to, in any manner, the debts, contracts, liabilities, engagements or torts of any person entitled to benefits under this Plan.

**(b) Qualified Domestic Relations Orders.** Notwithstanding the provisions of subsection (a), the Administrator may direct the Trustees to comply with a Qualified Domestic Relations Order (as described in § 1304).

### § 1304 Qualified Domestic Relations Orders.

- (a) **Definition.** A Qualified Domestic Relations Order is a judgment, decree, or order (including approval of a property settlement agreement) made pursuant to a state domestic relations law (including community property law) that relates to the provision of child support, alimony payments or marital property rights to a spouse, former spouse, child, or other dependent of a Participant (hereinafter referred to as an "Alternate Payee"), **which** was entered before January 1, 1985, **or** which—
- (1) **Rights Recognized.** Creates or recognizes a right on the part of the Alternate Payee to receive all or a portion of the benefits payable on behalf of a Participant under this Plan;

### (2) Required Provisions. Specifies—

- (A) the name and last known mailing address (if any) of the Participant and each Alternate Payee covered by the order;
- **(B)** the amount or percentage of the Participant's Plan benefits to be paid to any Alternate Payee, or the manner in which such amount or percentage is to be determined; **and**
- (C) the number of payments or the period to which the order applies and each Plan to which the order relates:
  - (3) **Prohibited Provisions.** Does not require the Plan to do any of the following:
- (A) provide any type or form of benefit or any option not otherwise provided under the Plan;
- **(B)** pay any benefit in the form of a Joint and Survivor Annuity with respect to the Alternate Payee and his or her subsequent spouse;
  - (C) pay any benefits to an Alternate Payee before the earlier of—
    - (I) the date on which the Participant is entitled to a distribution under the

#### Plan, or

#### (II) the later of—

- (i) the date the Participant attains age 50, or
- (ii) the earliest date on which the Participant could begin receiving benefits under the Plan if the Participant incurred a Separation of Service with the Employer;
  - (**D**) provide increased benefits; or
- (E) pay benefits to an Alternate Payee that are required to be paid to another Alternate Payee under a prior Qualified Domestic Relations Order; and
- (4) **Permitted Provision.** May or may not provide that an Alternate Payee who had been married to the Participant for at least one year will be treated as an Eligible Spouse with respect to the portion of the Participant's benefit in which such Alternate Payee has an interest.

#### (b) Procedure.

- (1) **Notification.** Upon receipt of any judgment, decree, or order (including approval of a property settlement agreement) relating to the provision of payment by the Plan to an Alternate Payee pursuant to a state domestic relations law, the Administrator shall promptly notify the affected Participant and any Alternate Payee of:
  - (A) the receipt of such judgment, decree, or order; and
- (B) the Administrator's procedure for determining whether or not the judgment, decree, or order is a Qualified Domestic Relations Order.
- (2) Establishment of Procedure. The Administrator shall establish a procedure to determine the status of a judgment, decree, or order as a Qualified Domestic Relations Order and to administer Plan distributions in accordance with them. Such procedure shall—
  - (A) be in writing;
- **(B)** permit an Alternate Payee to designate a representative for receipt of communications from the Administrator;
- (C) include a provision specifying the notification requirements set forth in paragraph (1);
- (D) include a provision describing the Alternate Payee Accounts provided in subsection (c); and
- (E) include such other provisions as may be required by regulations promulgated by the Secretary of the Treasury.

#### (c) Alternate Payee Accounts.

(1) Creation. During any period in which the Administrator or a court (or other tribunal) of competent jurisdiction is determining whether a judgment, decree, or order is a Qualified Domestic Relations Order, the Administrator shall create separate accounts under this Plan ("Alternate Payee accounts") and shall credit such accounts with the amounts, if any, which would have been payable to each Alternate Payee during such period (as they would have become due) if the judgment, decree, or order had already been determined to be a Qualified Domestic Relations Order. The amounts credited to the Alternate Payee accounts shall be debited from the accounts of the Participant potentially subject to the putative Qualified Domestic Relations Order. The Alternate Payee accounts need not be segregated from the general assets of the Trust Fund; they only must be accounted for separately.

#### (2) Disposition.

- (A) To Alternate Payee. If a judgment, decree, or order is determined to be a Qualified Domestic Relations Order within 18 months after the date on which the first payment would be required to be made under the judgment, decree, or order, the Administrator shall direct the Trustees to pay the amounts in Alternate Payee accounts created with respect to such judgment, decree, or order to the Alternate Payees.
- **(B) Return to Participant's Accounts.** All amounts in Alternate Payee accounts created with respect to such judgment, decree, or order shall be returned to the accounts with respect to the Participant from which they were derived upon the **earliest** of the following events:
- (I) the date 18 months after the date on which the first payment would be required to be made under the judgment, decree, or order;
- (II) the conclusive determination that such judgment, decree, or order is **not** a Qualified Domestic Relations Order; **or**
- (III) the termination, partial termination, or complete discontinuance of Employer contributions to the Plan and Trust.

Such returned amounts shall be paid at such time and in such manner as is otherwise provided in this Plan (*except* that any amounts already due for distribution shall be paid to the proper recipient immediately).

(d) Compliance with Qualified Domestic Relations Order. If a judgment, decree, or order is conclusively determined to be to be a Qualified Domestic Relations Order, the Administrator shall direct the Trustees to provide benefits under the Plan in accordance with such Qualified Domestic Relations Order.

### § 1305 Facility of Payment.

Whenever the Administrator determines that a person entitled to receive any payment of a benefit or installment is under a legal disability or is incapacitated in any way so as to be unable to manage his financial affairs, the Administrator may direct the Trustees to make payments to such person, to his legal representative, to a relative, or to a friend of such person for his benefit. Any payment of a benefit or installment in accordance with the provisions of this Section shall be a complete discharge from any liability for the making of such payment under the provisions of the Plan.

#### § 1306 Unclaimed Distribution.

- (a) **Segregation.** If, after diligent inquiry, the Administrator is unable to locate a person for the purpose of distribution of benefits under this Plan by the end of the Plan Year following the Plan Year in which the distribution was to have been made, the Administrator shall direct the Trustees to segregate the person's unclaimed Plan benefits in a separate interest bearing account under the Plan. Such separate account shall be entitled to all income it earns and shall bear all expenses it incurs.
- **(b) Payment.** If a person entitled to benefits segregated in an account under subsection (a) files a claim for benefits under this Plan and the Administrator approves such claim, the Administrator shall direct the Trustees to pay the segregated amounts over to the claimant.
- (c) Escheat. Amounts which remain unclaimed in an account under subsection (a) upon the termination and liquidation of this Plan and Trust or, if earlier, at the time when such property shall escheat under applicable state law (or be delivered to the state under applicable abandoned and unclaimed property law), shall be distributed to the state with jurisdiction over the amounts. In the event of a distribution under this subsection, the Plan and Trust shall have no further responsibility for such amounts.

# § 1307 Survivors of Participant Who Dies While Performing Qualified Military Service.

Notwithstanding any provision of this Plan to the contrary, the survivors of any Participant who dies while performing qualified military service are entitled to any additional benefits (*other than* contributions relating to the period of qualified military service, but including vesting service credit for such period and any ancillary life insurance or other survivor benefits) that would have been provided under the Plan had the Participant resumed employment on the day preceding the Participant's death and then terminated employment on account of death.

# **Article XIV — Claims Procedure**

### § 1401 Filing a Claim.

A Participant, Beneficiary, or Alternate Payee shall make a claim for benefits under this Plan by filing a written request with the Administrator on a form supplied by the Administrator.

#### § 1402 Notice of Denial.

If the Administrator denies a request for benefits under § 1401 in whole or in part, it shall notify the claimant of the same in writing within 60 days of the date the request was filed with the Administrator. Any notice of denial shall contain—

- (a) the reason for the denial;
- **(b)** specific references to the Plan provisions on which the denial is based;
- (c) a description of any additional information needed to perfect the claim and an explanation of why such information is necessary; and
- (d) an explanation of the Plan's claim procedure, including the opportunity for review under § 1403.

### § 1403 Review of Denial.

- (a) **Petition.** Within 60 days of the receipt of a notice of denial under § 1402, a claimant may petition the Administrator in writing for a review of the denial.
- **(b) Rights.** With respect to any review under this Section, the claimant shall have the right—
  - (1) to a hearing within 60 days of the receipt of the notice of denial under § 1402;
  - (2) to representation;
  - (3) to review pertinent documents;
- (4) to submit comments in writing within 60 days of the receipt of the notice of denial under § 1402; and
  - (5) to all rights afforded under subsection (d).
- (c) **Decision.** The Administrator shall issue a written decision at the conclusion of a review under this Section within 60 days following the hearing, or, if the claimant waives a hearing, within 120 days following the Administrator's receipt of a petition for such review under subsection (a). Such decision shall give specific reasons for the decision and provide specific references to the plan provisions on which it is based.
- (d) Compliance with Local Agency Law. All reviews under this § 1403 shall comply with the provisions of the Local Agency Law, 2 PA. CONS. STAT. § 551 et seq.

# **Article XV — Trust & Investments**

### § 1501 Establishment & Acceptance of Trust.

The Trustees shall receive all contributions paid to them under this Plan in cash or other property approved by the Administrator for acceptance by the Trustees. All property so received, together with income on such property, shall be held, managed, and administered in trust pursuant to the terms of this Plan agreement, and shall constitute the Trust Fund under this Plan. The Trustees shall be responsible only for such sums as shall actually be received by them as Trustees. They shall have no duty to collect any sums from the Employer or the Participants, and shall have no duties and responsibilities other than those set forth in this Plan and Trust or as imposed by applicable law. The Trustees may segregate invested assets. However, notwithstanding any other provision of this Plan, the Trustees may only earmark specific investments to the accounts of specific persons if the persons consent or if the investments are purchased ratably.

### § 1502 Trustees.

- (a) Qualification. A Trustee under this Plan may be any individual or corporation not prohibited from serving as a Trustee under § 1801.
- **(b) Initial Trustees.** The initial Trustees shall be the members of the Committee described in § 1701(b), *provided* that no person shall be a Trustee unless and until he/she signs a document accepting the Trust and agreeing to perform the duties of a Trustee under this Plan and Trust.
- (c) **Joint Trustees.** If at any time there is more than one (1) Trustee, the decision of the majority of the Trustees shall determine the actions of the Trustees. Notwithstanding the foregoing, the Administrator may allocate responsibilities among the Trustees from time to time by written notice to the Trustees. In such case, a Trustee (the "**first Trustee**") shall not be liable, either individually or as a Trustee, for any breaches of duty or losses to the Plan and Trust arising out of the acts or omissions of a co-Trustee in connection with areas of responsibility allocated to the co-Trustee to the exclusion of the first Trustee **unless**—
- (1) the first Trustee participates knowingly in, or knowingly undertakes to conceal, an act or omission of the co-Trustee, including knowing such act or omission is a breach of duty;
- (2) by the first Trustee's failure to comply with the standards set forth in § 1803 with respect to his own areas of responsibility, the first Trustee enables the co-Trustee to commit a breach of duty; or
- (3) the first Trustee has knowledge of a breach of duty by the co-Trustee and fails to make reasonable efforts under the circumstances to remedy the breach.
- **(d) Resignation.** Any Trustee may resign at any time upon 15 days notice in writing to the Board of Commissioners of the Sponsor and the Administrator.
- (e) Removal. The Board of Commissioners of the Sponsor may remove one or more of the Trustees at any time upon 15 days notice in writing to the Trustees and the Administrator. In addition, unless otherwise provided by Ordinance or Resolution of the Board of Commissioners, a Trustee who is a members of the Committee described in § 1701(b) shall automatically be removed from office upon his resignation or removal from the Committee under § 1701(b) or upon the expiration of his term of office on that Committee and the appointment of a successor.

- (f) Successor & Additional Trustees. Upon the removal, resignation, or death of a Trustee, the Board of Commissioners of the Sponsor may appoint a successor Trustee *provided* that the Board must act to insure that there will be at least one Trustee. In addition, at any time the Board of Commissioners of the Sponsor may appoint one or more additional Trustees. All Trustees appointed under this subsection shall have the same powers and duties as those conferred upon the initial Trustees under this Plan and Trust. Unless otherwise provided by Ordinance or Resolution of the Board of Commissioners and unless the Board of Commissioners has appointed a corporate Trustee, all persons appointed to the Committee under § 1701(b) shall automatically be appointed as Trustees of this Trust as well, *provided* that no person shall be a Trustee unless and until he/she signs a document accepting the Trust and agreeing to perform the duties of a Trustee under this Plan and Trust.
- (g) Transfer of Assets to New Trustees. Upon the appointment of a successor or additional Trustee under subsection (f), the former body of Trustees (or their personal representatives) shall assign, transfer, and pay over to the new or reconstituted body of Trustees the funds and properties then constituting the Trust Fund. However, the former body of Trustees are authorized to reserve such sum of money as may seem to them advisable for payment of their fees and expenses in connection with the settlement of their accounts; any balance of such reserve remaining after the payment of such fees and expenses shall be paid over to the new or reconstituted body of Trustees as soon as possible.

### § 1503 Investment of the Trust Fund.

- (a) In General. Except as provided in this Section and §§ 1505 (relating to Participant-directed investments), 1506 (relating to Investment Managers), and 1508 (relating to limiting directions from the Administrator), the Trustees shall have the power to invest and re-invest at any time all money or other property of any description held by them and constituting part of the Trust Fund. They may make such investments in any manner they deem advisable (subject to the duty of care required under § 1803 and the other fiduciary requirements of Article XVIII) which are permitted by law, and will not be limited to investments which are lawful for trustees. However, in making investments, the Trustees shall keep in mind the need for a certain degree of liquidity in order to provide benefits under this Plan.
- **(b)** Location. The Trustees may not maintain the indicia of ownership of any assets of the Plan and Trust outside the jurisdiction of the United States District Courts.
- (c) Employer Securities. The Trustees shall not invest in any securities of the Employer or any Related Employer.

# § 1504 Life Insurance Policies and Annuity Contracts.

- (a) Restrictions on Purchase. The Trustees may not invest in insurance policies on the lives of Participants, and may only invest in annuity contracts on the lives of Participants which are retirement annuity policies, retirement income endowment policies, disability income policies, a combination of such policies, or other annuity contracts permitted under Pennsylvania law and approved by the Department of the Auditor General for inclusion in municipal pension plans eligible for general municipal pension system state aid.
- **(b) Requirements for Permitted Contracts.** The Trustee may apply for permitted annuity contracts on any day of any month, and may continue to hold contracts on the lives of Inactive Participants which were purchased earlier. Each contract shall provide that the Trustees shall be the owner of such contract while it is held under this Trust, and that it may be cash

surrendered or exchanged for another policy before the annuitant attains the Normal Retirement Age (at least). All rights, options, and privileges which are available by the terms of such contracts shall be vested *exclusively* in or exercised *solely* by the Trustees. In the event of any conflict between the terms of this Plan and the terms of any insurance contract purchased hereunder, the Plan provisions shall control.

### § 1505 Participant-Directed Investments.

Participants, Beneficiaries, and/or Alternate Payees may direct the Trustees to invest all or a portion of the amounts allocated to their Plan accounts in particular investments in accordance with the provisions of Article XVI.

### § 1506 Investment by Investment Managers.

(a) In General. The Administrator may appoint one or more persons qualified under subsection (b) to be Investment Managers under this Plan, with powers to manage, acquire, and/or dispose of specified assets in the Trust. Each such appointment shall specify the powers granted to the Investment Manager, the assets involved, and the duties and responsibilities, if any, of the Trustees with respect to the assets subject to investment by the Investment Manager. The Administrator shall have the authority and responsibility for establishing operational and administrative procedures to coordinate the activities of the Trustees and any Investment Manager. The Trustees shall have no obligation to take any action with respect to those assets of the Trust subject to the direction of an Investment Manager without receipt of written directions acceptable to the Trustees from the Investment Manager, and the Trustees shall be under no duty to review such directions. The Administrator may revoke an appointment under this Section or change its terms at any time. Upon receipt of a written notice from the Administrator of the resignation or removal of an Investment Manager, the Trustees shall assume management responsibility for the assets previously appointed to the direction of the Investment Manager. The Investment Manager may not invest in any assets in which the Trustees are precluded from investing under §§ 1503 (relating to Investment of the Trust Fund) or 1504 (relating to Life Insurance Policies and Annuity Contracts).

#### (b) Qualification. An Investment Manager must be either—

- (1) registered as an investment adviser under the Investment Advisers Act of 1940;
- (2) a bank (as defined in the Investment Advisers Act of 1940); or
- (3) an insurance company qualified to perform the duties of an Investment Manager under the laws of more than one State.
- (c) Acceptance & Communication. A person appointed under subsection (a) shall not exercise the powers of an Investment Manager until he has acknowledged in writing that he is a Fiduciary under this Plan and until the appointment and acknowledgment have been transmitted to the Administrator and the Trustees.
- (d) Security Transactions. If an Investment Manager appointed pursuant to this Section elects to place security transactions directly with a broker or dealer, the Trustees shall not recognize such transactions unless and until the Trustees have received instructions or confirmations from the Investment Manager in such manner of communication customary to the Trustees. Should the Investment Manager direct the Trustees to utilize the services of any person with regard to the assets under its management or control, such instructions shall specifically set forth the actions to be taken by the Trustees as to such services. In the event that an Investment

Manager places security transactions directly or directs the utilization of services, the Investment Manager shall be solely responsible for the acts of the persons utilized. The sole duty of the Trustees as to such transactions shall be incident to the Trustees' practices as a custodian.

- (e) Release and Indemnification of Trustee. To the extent that the Trust is subject to the direction of an Investment Manager—
- (1) the Trustees shall not be responsible nor have any liability for acting pursuant to any direction of the Investment Manager or failing to act in the absence of any direction from the Investment Manager (except as may otherwise be imposed by applicable law), and shall not be required to consult with or advise the Administrator or the Employer regarding the investment quality of any investments; and
- (2) the Employer shall indemnify and hold the Trustees harmless from any and all losses or claims which arise with respect to the Trust,

#### unless the Trustees—

- (A) knowingly participate in or knowingly conceals an act or omission of the Investment Manager, knowing such act to be a breach of fiduciary duty;
- **(B)** have enabled the Investment Manager to commit a breach by failing to discharge the Trustees' duties in accordance with the fiduciary requirements of Article XVIII and applicable law; **or**
- (C) has knowledge of a breach of fiduciary responsibility by the Investment Manager and fails to make reasonable efforts under the circumstances to remedy the breach.

### § 1507 Other Powers of the Trustees.

Subject to the other provisions of this Article and the provisions of Article XVIII (relating to Fiduciaries), the Trustees shall be entitled to exercise, in their own discretion, the following powers regarding the administration of the Trust Fund:

- (a) **Purchase of Property.** To purchase, or subscribe for, any securities or other property and to retain the same in trust, regardless of whether such property is specifically authorized as a legal investment for trust funds under applicable law;
- **(b) Disposition of Property.** To sell, exchange, convey, transfer, mortgage, pledge, lease, grant options with respect to, or otherwise dispose of any securities or other property held by the Trustees, by private contract or at public auction. No person dealing with the Trustees shall be bound to see to the application of the purchase money or to inquire into the validity, expediency, or propriety or any such sale or other disposition;
- (c) Exercise of Ownership Rights. To vote any stock, bonds, or other securities; to give general or special proxies or powers of attorney, with or without powers of substitution; to exercise any conversion privileges, subscriptions rights, or other options, and to make any payment incidental thereto; to oppose, or to consent to, or otherwise participate in, corporate reorganizations or other changes affecting corporate securities, and to delegate discretionary powers, and to pay any assessments or charges in connection therewith; to manage, operate, improve, develop, repair, and preserve any real property or any oil, gas, or mineral properties, royalties, or interests; and generally to exercise any of the powers of an owner with respect to stock, bonds, securities, or other property held as part of the Trust Fund;
- (d) Registration of & Title to Investments. To cause any securities or other property held as part of the Trust Fund to be registered in the name(s) of the Trustees or in the name(s) of

one or more nominees of the Trustees, or to hold any investments in bearer form, so long as the books and records of the Trustees shall at all times show that all such investments are part of the Trust Fund;

- **(e) Borrowing.** To borrow or raise money for the purposes of the Trust in such amount and upon such terms and conditions as the Trustees shall deem advisable. For any sum so borrowed, the Trustees may issue a promissory note as Trustees and secure repayment by pledging all, or any part, of the Trust Fund. No person lending money to the Trustees shall be bound to see to the application of the money lent or to inquire into the validity, expediency, or propriety of any such borrowing;
- **(f) Collection.** To collect and receive any and all money and other property of whatsoever kind or nature due or owing or belonging to the Trust Fund and to give full discharge and acquittance therefor; and to extend the time of payment of any obligation at any time owing to the Trust Fund, as long as such extension is for a reasonable period, and continues reasonable interest;
- (g) Retention of Cash. To keep such portion of the Trust Fund in cash or cash balance as the Trustees may from time to time deem to be in the best interests of the Trust, without liability for interest thereon;
- (h) Retention of Property Acquired. To accept and retain for such time as the Trustees may deem advisable any securities or other property received or acquired as Trustees under this Plan, whether or not such securities or other property would normally be purchased as investments under this Plan:
- (i) Execution of Instruments. To make, execute, acknowledge, and deliver any and all documents or transfer and conveyance and any and all other instruments that may be necessary or appropriate to carry out the powers granted under this Plan;
- (j) Settlement of Claims & Debts. To settle, compromise, or submit to arbitration any claims, debts, or damages due or owing to or from the Trust Fund; to commence or defend suits or legal or administrative proceedings; and to represent the Trust Fund in all suits and legal and administrative proceedings;
- (k) Employment of Agents & Counsel. Subject to the prohibitions of § 1801, to employ suitable agents, counsel, consultants, specialists, and accountants, any one of whom may also be so engaged by the Employer; to pay their reasonable expenses and compensation in the event the Employer has not so paid them; and to rely exclusively upon, and be fully protected in any action taken in good faith in relying upon, any opinions or reports which shall be furnished by any such accountant, counsel, specialist, or other consultant.
- (I) Incorporation. To organize and incorporate under the laws of any state (or participate in the organization or incorporation of) a corporation for the purpose of acquiring and holding title to any property which the Trustees are authorized to acquire for the Trust Fund and to exercise with respect thereto any of the powers, rights, and duties they have with respect to other assets of the Trust Fund;
- (m) Pooling of Assets. To transfer any of the assets of this Trust to any pooled investment fund or group trust (including those which have one or more trustees who are Trustees under this Trust) which has been ruled by the Internal Revenue Service to be, and which is, a qualified trust exempt from tax under the Code and which has been established for the purpose of permitting separate qualified pension and profit sharing trusts to pool some or all of their funds for investment purposes and as to which it has been ruled by the Internal Revenue Service that the pooling of funds by the separate trusts will not adversely affect the qualified status of the separate trusts. Any such common trust fund shall constitute an integral part of this

Plan and Trust. The commingling of assets of this Trust with assets of other qualified participating trusts in such pooled funds is specifically authorized;

- (n) Legal Actions. To prosecute, defend against, or participate in any legal actions involving the Trust or any Trustee in the manner and to the extent the Trustees deem advisable. The Trustees need not participate in any litigation concerning the Trust or the Trustees' management of the Trust unless first indemnified against expense by the Employer in a form satisfactory to the Trustee, unless the litigation is occasioned by the negligence or fault of the Trustee and the Trustee is found to be negligent or at fault pursuant to such litigation; and
- (o) Necessary Acts. To do all such acts, take all such proceedings, and exercise all such rights and privileges as the Trustees may deem necessary to administer the Trust Fund and carry out the purposes of this Plan and Trust, even though not specifically mentioned in this document.

### § 1508 Limiting Directions from the Administrator.

The Trustees shall comply with any directions given by the Administrator under § 1703 (relating to limiting directions from the Administrator) as promptly as possible. The Trustees shall not be responsible for any loss which may result from compliance with the directions of the Administrator or the failure or refusal of the Administrator to approve any actions which require the Administrator's approval, and the Employer shall indemnify and hold the Trustees harmless for any such loss.

### § 1509 Distributions from the Trust Fund.

From time to time, the Trustees shall make payments out of the Trust Fund to such persons, in such manner, for such purposes, and in such amounts as may be specified in written directions by the Administrator. Such directions must be accompanied by a certificate executed by the Administrator or its designate that the payment is in accordance with this Plan. Once made, the amount of any such payment shall no longer constitute a part of the Trust Fund. The Trustees shall not be responsible in any way for the application of such payments or for the adequacy of the Trust Fund to meet and discharge any and all liabilities under the Plan.

# § 1510 Administrative Payments.

- (a) Compensation of Trustees and Investment Managers. Trustees who are also officers or employees of the Employer shall receive no compensation for their services as Trustees under this Plan. All other Trustees may be paid such reasonable compensation as shall from time to time be agreed upon in writing by the Board of Commissioners of the Sponsor and the Trustees. Investment Managers shall be paid such fees as shall from time to time be agreed upon in writing by the Investment Manager and the Administrator. Trustee and Investment Manager fees may be paid by the Employer, but unless or until so paid they shall constitute a charge upon the Trust Fund.
- **(b) Expenses.** The Trustees shall be reimbursed for any reasonable expenses, including reasonable counsel fees, incurred by them in the administration of the Trust. Such reimbursement may be made by the Employer, but unless or until so paid it shall constitute a charge upon the Trust Fund.

(c) Taxes. All taxes of any and all kinds whatsoever that may be levied or assessed (under existing or future laws) upon or with respect to the Trust Fund or the income of the Trust Fund shall be paid from the Trust Fund.

## § 1511 Accounting.

- (a) **Record Keeping.** The Trustees shall keep accurate and detailed accounts of all investments, receipts, disbursements, and other transactions under this Trust. All accounts, books, and records relating to such transactions shall be open to inspection and audit at all reasonable times by any person designated by the Administrator.
- (b) Reports to Administrator. Within 60 days following the close of each fiscal year of the Trust, within 60 days following the effective date of the termination of the Plan or Trust, and within 60 days after the removal or resignation of a Trustee, the Trustees shall file with the Administrator a written account of all investments, receipts, disbursements, and other transactions affected by them during such fiscal year (or during the period from the close of the last fiscal year to the date on which the Trustee resigned or was removed, or the effective date of the termination of the Plan or Trust). Such account shall also set forth the current value of the Trust Fund and its assets. Neither the Administrator nor any other person shall be entitled to any further accounting by the Trustees, except as provided by law.
- (c) **Discharge from Liability.** Thirty-one (31) days after an accounting has been filed under subsection (b), the Trustees shall be forever released and discharged from all liability and accountability to anyone with respect to the propriety of their actions and transactions shown in the accounting, *except* with respect to any actions or transactions as to which the Administrator has objected in a writing filed with the Trustees before such time. If such an objection is filed, the Trustees shall, unless the matter is compromised with the Administrator, file its account in any court of competent jurisdiction for audit and adjudication.

## § 1512 Immunity.

- (a) **Persons to whom Responsible.** No person other than the Employer or the Administrator may require an accounting or bring an action against the Trustees with respect to the Trust created under this Plan or their actions as Trustees.
- **(b) Ordinary Negligence.** The Trustees shall not be liable for the making, retention, or sale of any investment or reinvestment made by them as provided under this Plan, nor for any loss to, or diminution of, the Trust Fund unless caused by their own gross negligence, willful misconduct, or lack of good faith.
  - (c) **Permitted Reliance.** The Trustees shall be fully protected in relying upon—
- (1) Action by the Administrator. A certification by the Administrator or by any person designated by the Administrator (under § 1702(b)) with respect to any instruction or direction of the Administrator. The Trustees may rely upon any such designation until they have received a revocation of same;
- (2) Other Writings. Any instrument, certificate, or paper believed by them to be genuine and be signed or presented by the proper person or persons; the Trustees shall be under no duty to make any investigation or inquiry as to any statement contained in any such writing, but may accept the writing as conclusive evidence of the truth and accuracy of such statements.

#### § 1513 Purpose: Exclusive Benefit Rule.

Except as provided in § 404(d) (relating to disposition of contributions made on the basis of a mistake of fact), all assets of the Trust Fund, including investment income, shall be retained for the exclusive benefit of Participants, Alternate Payees, and Beneficiaries, and shall be used to pay benefits to such persons or to pay administrative expenses of the Plan and Trust Fund to the extent not paid by the Employer; they shall not revert to or inure to the benefit of the Employer. The Trustees shall exercise all powers and discharge all duties under this Plan and Trust solely in the general interest of Participants, Alternate Payees, and Beneficiaries.

#### § 1514 Standard of Care.

The Trustees, all agents, counsel, consultants, specialists, and accountants retained by them under § 1507(k), and all Investment Managers under § 1506 shall be subject to the fiduciary requirements detailed in Article XVIII.

# **Article XVI** — Participant-Directed Investments

## § 1601 Program Authorized.

Participants may direct the Trustees to invest all or a portion of the amounts allocated to their Plan accounts in particular investments in accordance with the provisions of this Article. A Participant may also direct the Trustees to resume responsibility for any portion of such investments. Participant directions shall expire to the extent that amounts which have been so directed are forfeited or distributed, or are invested in investments which are no longer permitted under the participant-directed investment program. Investment directions by a Participant under this Article shall relieve the Administrator and the Trustees of all fiduciary responsibilities in the management of such funds. Beneficiaries and Alternate Payees have the same rights under this Article as Participants.

## § 1602 Definitions.

For purposes of this Article—

- (a) Affiliate. The term "Affiliate" shall include:
- (1) any person directly or indirectly controlling, controlled by, or under common control with the person. For purposes of this paragraph (1), the term "control" means, with respect to a person other than an individual, the power to exercise a controlling influence over the management or policies of such person;
- (2) any officer, director, partner, employee, an employee of an affiliated employer, relative (as defined in ERISA § 3(15)), brother, sister, or spouse of a brother or sister, of the person; and
- (3) any corporation or partnership of which the person is an officer, director, or partner.

#### (b) Average Annual Total Return.

- (1) In General. The term "Average Annual Total Return" shall mean the average annual compounded rate of return that would equate an initial investment in a designated investment alternative to the ending redeemable value of that investment calculated with the before tax methods of computation prescribed in Securities and Exchange Commission Form N-1A, N-3, or N-4, as appropriate, except that such method of computation may exclude any frontend, deferred, or other sales loads that are waived for the Participants of the covered individual account plan. Nothing in this Article requires disclosure of returns for periods before the inception of a Designated Investment Alternative.
- (2) Transitional Rule. For Plan Years beginning before October 1, 2021, if the Administrator reasonably and in good faith determines that it does not have the information on expenses attributable to the Plan that is necessary to calculate, in accordance with paragraph (1), the 5-year and 10-year Average Annual Total Returns for a Designated Investment Alternative that is not registered under the Investment Company Act of 1940, the Administrator may use a reasonable estimate of such expenses or the most recently reported Total Annual Operating Expenses of the Designated Investment Alternative as a substitute for such expenses, and shall inform Participants of the basis on which the returns were determined.
- (c) Designated Investment Alternative. The term "Designated Investment Alternative" shall mean a *specific* investment identified by a Plan Fiduciary as an available investment alternative under the participant-directed investment program. An investment alternative permitted under the participant-directed investment program which is *not* specifically identified by a Plan Fiduciary, such as an investment alternative covered by a general rule that allows Participants to invest in any asset administratively feasible for the Plan to hold and not otherwise prohibited under the program, is *not* a "Designated Investment Alternative", and the information production and other requirements applicable to Designated Investment Alternatives shall not apply to such a non-specified investment alternative.
- (d) Total Annual Operating Expenses. The term "Total Annual Operating Expenses" shall mean:
- (1) In the case of a Designated Investment Alternative that is registered under the Investment Company Act of 1940, the annual operating expenses and other asset-based charges before waivers and reimbursements (e.g., investment management fees, distribution fees, service fees, administrative expenses, separate account expenses, mortality and expense risk fees) that reduce the Designated Investment Alternative's rate of return, expressed as a percentage, calculated in accordance with the required Securities and Exchange Commission form, e.g., Form N-1A (open-end management investment companies) or Form N-3 or N-4 (separate accounts offering variable annuity contracts); or
- (2) In the case of a Designated Investment Alternative that is not registered under the Investment Company Act of 1940, the sum of the fees and expenses described in the following subparagraphs before waivers and reimbursements, for the Designated Investment Alternative's most recently completed fiscal year, expressed as a percentage of the Designated Investment Alternative's average net asset value for that year:
- (A) Management fees as described in the Securities and Exchange Commission Form N-1A that reduce the Designated Investment Alternative's rate of return;
- **(B)** Distribution and/or servicing fees as described in the Securities and Exchange Commission Form N-1A that reduce the Designated Investment Alternative's rate of return; and
- (C) Any other fees or expenses not included in subparagraphs (A) or (B) that reduce the Designated Investment Alternative's rate of return (e.g., externally negotiated fees, custodial expenses, legal expenses, accounting expenses, transfer agent expenses, recordkeeping

fees, administrative fees, separate account expenses, mortality and expense risk fees), excluding brokerage costs described in Item 21 of Securities and Exchange Commission Form N-1A.

#### § 1603 Investment Instructions.

(a) In General. The Trustees shall identify a specific fiduciary or agent to receive investment instructions, and all investment instructions must be made through such fiduciary or agent. All Participant investment instructions shall be made on such written forms (or in accordance with such online procedures) as may be prescribed by the Trustees or the identified fiduciary or agent. Instructions may relate to amounts allocated to the Participant's accounts to date and/or to amounts as they are so allocated in the future. Each Participant who is qualified to participate in the participant-directed investment program shall have an opportunity to obtain written confirmation of such instructions.

#### (b) Restrictions and Procedures.

- (1) In General. The Trustees may promulgate nondiscriminatory rules restricting Participant directions to such times, investments, amounts, and features as may be necessary or desirable to avoid undue administrative expenses or complexity in the overall operation of the Participant-directed investment program, *provided* that such rules comply with requirements of this subsection (b) and of § 1604. For example, without limiting the foregoing, the Trustees may restrict investment to identified specific investment alternatives.
- (2) Range of Risk and Return Characteristics. The Trustees shall permit a sufficient number and variety of investment alternatives to provide Participants with a reasonable opportunity to materially affect the potential return on amounts in their accounts and the degree of risk to which such amounts are subject.
- (3) General Frequency Standard. The rules applicable to any given investment alternative made available under the participant-directed investment program must permit Participants to give investment instructions with a frequency which is appropriate in light of the market volatility to which the investment alternative may reasonably be expected to be subject.
- (4) Core Investment Funds. The rules applicable to each of the investment alternatives included in the group of "core funds" described in § 1604 must permit a Participant to give instructions no less frequently than once within any three month period.
- (5) Investment Alternative Available to Receive Transfers. Under the rules established by the Trustees, either—
- (A) at least *one* of the "core funds" described in § 1604 must permit a Participant to transfer *into* that fund as frequently as Participants are permitted to give investment instructions with respect to any investment alternative included in the participant-directed investment program which permits Participants to give investment instructions more frequently than once within any three month period; or
- (B) with respect to each investment alternative which permits Participants to give investment instructions more frequently than once within any three month period, Participants are permitted to direct their investments from such investment alternative to a Liquid Investment as frequently as they are permitted to give investment instructions with respect to such investment alternative, and, with respect to the Liquid Investment, Participants are permitted to direct investments from the Liquid Investment to at least one of the "core funds" described in § 1504 as frequently as they are permitted to give investment instructions with respect to that core fund. For purposes of this subparagraph (B), a "Liquid Investment" is an income producing, low risk, liquid fund, subfund, or account.

- (6) Employer Securities. The participant-directed investment program shall not permit Participant to directly or indirectly acquire any securities of the Employer or any Related Employer or Affiliate.
- (7) Annuity Contracts. If the Trustees permit investment in annuity contracts under the participant-directed investment program, an Active Participant may only direct investment in annuity contracts which are on his/her own life.
- (c) Compliance With Instructions. The fiduciary or agent receiving the instructions shall be obligated to comply with such instructions, unless an instruction, if implemented—
- (1) would not be permitted under this Article or would otherwise not be in accordance with the provisions of this Plan;
- (2) would not comply with the procedures, limitations, or restrictions established by the Trustees for the participant-directed investment program;
- (3) would cause a Fiduciary to maintain the indicia of ownership of any assets of the Plan outside the jurisdiction of the district courts of the United States other than as permitted under the standards of ERISA § 404(b) and 29 C.F.R. § 2550.404b-1;
  - (4) would jeopardize the Plan's tax qualified status under the Code;
  - (5) could result in a loss in excess of a Participant's account balance;
  - (6) would result in a prohibited transaction described in Plan § 1805;
- (7) would result in the acquisition of a "collectible", as that term is defined in Code § 408(m) and the regulations thereunder;
  - (8) would generate income that would be taxable to the Plan;
  - (9) would result in a direct or indirect—
- (A) sale, exchange, or lease of property between the Employer or any Affiliate of the Employer and the Plan except for the acquisition or disposition of any interest in a fund, subfund, or portfolio managed by the Employer or any Affiliate of the Employer;
  - **(B)** loan to the Employer or any Affiliate of the Employer;
- (C) acquisition or sale of any employer real property (as defined in ERISA  $\S$  407(d)(2)); or
  - (D) acquisition or sale of any employer security; or
  - (10) would violate any applicable law, statute, regulation, rule, order, or decree.
- (d) Unavailable Investments. In the event the Trustees find that an investment meeting the requirements of this Trust cannot be procured for a Participant under this Article, or a given investment is or becomes unavailable, the Trustees shall report the same to the Participant as soon as practicable and request further instructions.

#### § 1604 Core Investment Funds.

Throughout the time that a Participant-directed investment program is in effect under this Section, the permitted investment alternatives under the program shall include a group of at least three (3) investment funds—

(1) each of whose underlying assets are diversified so as to minimize the risk of large losses;

- (2) each of which has materially different risk and return characteristics;
- (3) which in the aggregate enable the Participant by choosing among them to achieve a portfolio with aggregate risk and return characteristics at any point within the range normally appropriate for the Participant;
- (4) each of which, when combined with investments in the other funds in the group, tends to minimize through diversification the overall risk of a Participant's portfolio; and
  - (5) each of which is either—
- (A) An investment company described in § 3(a) of the Investment Company Act of 1940, or a series investment company described in § 18(f) of the Investment Company Act of 1940, or any of the segregated portfolios of such company;
- **(B)** A common or collective trust fund or a pooled investment fund maintained by a bank or similar institution, a deposit in a bank or similar institution, or a fixed rate investment contract of a bank or similar institution;
- (C) A pooled separate account or a fixed rate investment contract of an insurance company qualified to do business in a state; or
- **(D)** Any entity whose assets include plan assets by reason of a plan's investment in the entity (such as a "group trust" as defined in Rev. Rul. 81-100).

These investment funds may be Designated Investment Alternatives, or they may merely be permitted without specific identification under a general rule allowing investment in broad categories of assets or in all assets which are administratively feasible for the Plan to hold.

## § 1605 Notice of Limited Liability.

The Administrator (or person designated by the Administrator to act on its behalf) shall provide each Participant who is qualified to participate in the participant-directed investment program with an explanation that the Plan is intended to constitute the kind of plan described in ERISA § 404(c) and 29 C.F.R. § 2550.404c-1, and that the Fiduciaries of the Plan may be relieved of liability for any losses which are the direct and necessary result of investment instructions given by such Participant.

#### § 1606 Disclosure of Plan-Related Information.

The Administrator (or person designated by the Administrator to act on its behalf) shall provide to each Participant the following Plan-related information, based on the latest information available to the Plan:

- (a) General Information. On or before the date on which a Participant can first direct his investments, and at least annually thereafter:
- (1) an explanation of the circumstances under which Participants may give investment instructions (including the persons to whom instructions may be given, the times when instructions may be given, and the manner in which instructions may be given);
- (2) an explanation of any specified limitations on such instructions under the terms of the Plan, including any restrictions on transfers to or from a Designated Investment Alternative (such as absolute restrictions, minimum investment periods, penalties, or valuation adjustments);

- (3) a description of or reference to plan provisions or provisions of the participant-directed investment program relating to the exercise of voting, tender, and similar rights appurtenant to a Participant's investment in an investment alternative, as well as any restrictions on such rights;
- (4) an identification of any Designated Investment Alternatives offered under the Plan;
  - (5) an identification of any designated Investment Managers; and
- (6) a description of any "brokerage windows," "self-directed brokerage accounts," or similar Plan arrangements that enable Participants to select investments beyond those designated by the Plan. If the participant-directed investment program does not limit the investment alternatives to Designated Investment Alternatives, a general statement of the types of non-identified investments that are permitted and the types which are prohibited, as provided under the rules established for the participant-directed investment program, shall be a sufficient description of such investments, provided that Participants are encouraged to obtain and review materials relating to any such non-identified investments prior to actually making an investment.

#### (b) Administrative Expenses.

(1) Initial and Annual Disclosure. On or before the date on which a Participant can first direct his investments, and at least annually thereafter, an explanation of any fees and expenses for general plan administrative services (e.g., legal, accounting, recordkeeping), which may be charged against the individual accounts of Participants and are not reflected in the Total Annual Operating Expenses of any Designated Investment Alternative, as well as the basis on which such charges will be allocated (e.g., pro rata, per capita) to, or affect the balance of, each individual account.

#### (2) Quarterly Disclosure. At least quarterly, a statement that includes—

- (A) the dollar amount of the fees and expenses described in paragraph (1) that are actually charged (whether by liquidating shares or deducting dollars) during the preceding quarter to the Participant's account for individual services;
- **(B)** a description of the services to which the charges relate (*e.g.*, plan administration, including recordkeeping, legal, accounting services); **and**
- (C) if applicable, an explanation that, in addition to the fees and expenses disclosed pursuant this paragraph (2), some of the Plan's administrative expenses for the preceding quarter were paid from the Total Annual Operating Expenses of one or more of the Plan's Designated Investment Alternatives (e.g., through revenue sharing arrangements, Rule 12b-1 fees, sub-transfer agent fees).

#### (c) Individual Expenses.

(1) Initial and Annual Disclosure. On or before the date on which a Participant can first direct his investments, and at least annually thereafter, an explanation of any fees and expenses that may be charged against the individual account of a Participant on an individual, rather than on a plan-wide, basis (e.g., fees attendant to processing Plan loans or Qualified Domestic Relations Orders, fees for investment advice, fees for brokerage windows, commissions, front- or back-end loads or sales charges, redemption fees, transfer fees, and similar expenses, and optional rider charges in annuity contracts) and which are not reflected in the Total Annual Operating Expenses of any Designated Investment Alternative.

#### (2) Quarterly Disclosure. At least quarterly, a statement that includes—

(A) the dollar amount of the fees and expenses described in paragraph (1) that are actually charged (whether by liquidating shares or deducting dollars) during the preceding quarter to the Participant's account for individual services; and

- (B) a description of the services to which the charges relate (e.g., loan processing fee).
- (d) Changes to Disclosed Information. If there is a change to the information described in subsections (a), (b)(1), or (c)(1), each Participant must be furnished a description of such change at least thirty (30) days, but not more than ninety (90) days, in advance of the effective date of such change, unless the inability to provide such advance notice is due to events that were unforeseeable or circumstances beyond the control of the Administrator, in which case notice of such change must be furnished as soon as reasonably practicable.

# § 1607 Automatic and Periodic Disclosure of Investment-Related Information.

Except as provided in § 1608, the Administrator (or person designated by the Administrator to act on its behalf) shall provide to each Participant the following investment-related information on or before the date on which a Participant can first direct his investments, and at least annually thereafter, based on the latest information available to the Plan and in the format described in § 1609:

#### (a) Identifying Information. Such information shall include:

- (1) The name of each Designated Investment Alternative; and
- (2) The type or category of the investment (e.g., money market fund, balanced fund (stocks and bonds), large-cap stock fund, employer stock fund, employer securities).

#### (b) Performance Data.

- (1) **Return Not Fixed.** For Designated Investment Alternatives with respect to which the return is not fixed, the Average Annual Total Return of the investment for 1-, 5-, and 10-calendar year periods (or for the life of the alternative, if shorter) ending on the date of the most recently completed calendar year; as well as a statement indicating that an investment's past performance is not necessarily an indication of how the investment will perform in the future.
- (2) Fixed Return. For Designated Investment Alternatives with respect to which the return is fixed or stated for the term of the investment, both the fixed or stated annual rate of return and the term of the investment. If, with respect to such a Designated Investment Alternative, the issuer reserves the right to adjust the fixed or stated rate of return prospectively during the term of the contract or agreement, the current rate of return, the minimum rate guaranteed under the contract, if any, and a statement advising Participants that the issuer may adjust the rate of return prospectively and how to obtain (e.g., telephone or Web site) the most recent rate of return required under this Section.
- (c) Benchmarks. For Designated Investment Alternatives with respect to which the return is not fixed, the name and returns of an appropriate broad-based securities market index over the 1-, 5-, and 10-calendar year periods (or for the life of the alternative, if shorter) comparable to the performance data periods provided under subsection (b)(1), and which is not administered by an Affiliate of the investment issuer, its investment adviser, or a principal underwriter, unless the index is widely recognized and used.

#### (d) Fee & Expense Information.

(1) **Return Not Fixed.** For Designated Investment Alternatives with respect to which the return is not fixed:

- (A) The amount and a description of each shareholder-type fee (fees charged directly against a Participant's investment, such as commissions, sales loads, sales charges, deferred sales charges, redemption fees, surrender charges, exchange fees, account fees, and purchase fees, which are not included in the Total Annual Operating Expenses of any Designated Investment Alternative) and a description of any restriction or limitation that may be applicable to a purchase, transfer, or withdrawal of the investment in whole or in part (such as round trip, equity wash, or other restrictions);
- **(B)** the Total Annual Operating Expenses of the investment expressed as a percentage (*i.e.*, expense ratio), calculated in accordance with § 1602(d);
- (C) the Total Annual Operating Expenses of the investment for a one-year period expressed as a dollar amount for a \$1,000 investment (assuming no returns and based on the percentage described in subparagraph (B);
- (**D**) a statement indicating that fees and expenses are only one of several factors that Participants should consider when making investment decisions; **and**
- (E) a statement that the cumulative effect of fees and expenses can substantially reduce the growth of a Participant's retirement account and that Participants can visit the Employee Benefit Security Administration's Web site for an example demonstrating the long-term effect of fees and expenses.
- (2) Fixed Return. For Designated Investment Alternatives with respect to which the return is fixed for the term of the investment, the amount and a description of any shareholder-type fees and a description of any restriction or limitation that may be applicable to a purchase, transfer, or withdrawal of the investment in whole or in part.
- **(e) Internet Web Site Address.** An Internet Web site address that is sufficiently specific to provide Participants access to the following information regarding the Designated Investment Alternative:
  - (1) The name of the Designated Investment Alternative's issuer;
- (2) The Designated Investment Alternative's objectives or goals in a manner consistent with Securities and Exchange Commission Form N-1A or N-3, as appropriate;
- (3) The Designated Investment Alternative's principal strategies (including a general description of the types of assets held by the investment) and principal risks in a manner consistent with Securities and Exchange Commission Form N-1A or N-3, as appropriate;
- (4) The Designated Investment Alternative's portfolio turnover rate in a manner consistent with Securities and Exchange Commission Form N-1A or N-3, as appropriate;
- (5) The Designated Investment Alternative's performance data described in subsection (b), updated on at least a quarterly basis, or more frequently if required by other applicable law; and
- (6) The Designated Investment Alternative's fee and expense information described in subsection (d).
- **(f) Glossary.** A general glossary of terms to assist Participants in understanding the Designated Investment Alternatives, or an Internet Web site address that is sufficiently specific to provide access to such a glossary along with a general explanation of the purpose of the address.
- (g) Annuity Options. If a Designated Investment Alternative is *part of* a contract, fund, or product that permits Participants to allocate contributions towards the *future* purchase of a stream of retirement income payments guaranteed by an insurance company, the information set forth in § 1608(a)(1) through (7) with respect to the annuity option, to the extent such informa-

tion is not otherwise included in investment-related fees and expenses described in subsection (d).

# § 1608 Special Rules Relating to Automatic Disclosure of Investment-Related Information.

- (a) Annuity Options. In the case of a Designated Investment Alternative that is a contract, fund, or product that permits Participants to allocate contributions towards the current purchase of a stream of retirement income payments guaranteed by an insurance company, the Administrator (or person designated by the Administrator to act on its behalf) shall provide the following information with respect to each such option *in lieu of* the information required by § 1607(a) through (e):
  - (1) The name of the contract, fund, or product;
- (2) The option's objectives or goals (e.g., to provide a stream of fixed retirement income payments for life);
- (3) The benefits and factors that determine the price (e.g., age, interest rates, form of distribution) of the guaranteed income payments;
- (4) Any limitations on the ability of a Participant to withdraw or transfer amounts allocated to the option (e.g., lock-ups) and any fees or charges applicable to such withdrawals or transfers;
- (5) Any fees that will reduce the value of amounts allocated by Participants to the option, such as surrender charges, market value adjustments, and administrative fees;
- (6) A statement that guarantees of an insurance company are subject to its long-term financial strength and claims-paying ability; and
- (7) An Internet Web site address that is sufficiently specific to provide Participants access to the following information:
  - (A) The name of the option's issuer and of the contract, fund, or product;
  - (B) Description of the option's objectives or goals;
- (C) Description of the option's distribution alternatives/guaranteed income payments (e.g., payments for life, payments for a specified term, joint and survivor payments, optional rider payments), including any limitations on the right of a Participant to receive such payments;
- (**D**) Description of costs and/or factors taken into account in determining the price of benefits under an option's distribution alternatives/guaranteed income payments (e.g., age, interest rates, other annuitization assumptions);
- **(E)** Description of any limitations on the right of a Participant to withdraw or transfer amounts allocated to the option and any fees or charges applicable to a withdrawal or transfer; **and**
- (F) Description of any fees that will reduce the value of amounts allocated by Participants to the option (e.g., surrender charges, market value adjustments, and administrative fees).
- **(b) Fixed Return Investments.** In the case of a Designated Investment Alternative with respect to which the return is fixed for the term of the investment, the Administrator (or person designated by the Administrator to act on its behalf) shall, *in lieu of* complying with the requirement of § 1607(e), provide an Internet Web site address that is sufficiently specific to provide

Participants access to the following information regarding the Designated Investment Alternative:

- (1) The name of the Designated Investment Alternative's issuer;
- (2) The Designated Investment Alternative's objectives or goals (e.g., to provide stability of principal and guarantee a minimum rate of return);
- (3) The Designated Investment Alternative's performance data described in § 1607(b)(2), updated on at least a quarterly basis, or more frequently if required by other applicable law; and
- (4) The Designated Investment Alternative's fee and expense information described in § 1607(d)(2).

# § 1609 Comparative Format For Furnishing Investment-Related Information.

- (a) In General. The Administrator (or person designated by the Administrator to act on its behalf) shall provide the information required under § 1607 or § 1608 in a chart or similar format that is designed to facilitate a comparison of such information for each Designated Investment Alternative available under the Plan, that prominently displays the date, and that includes:
- (1) A statement indicating the name, address, and telephone number of the Administrator (or person(s) designed by the Administrator to act on its behalf) to contact for the provision of the information required by § 1611 (relating to Investment-Related Information to be Provided Upon Request);
- (2) A statement that additional investment-related information (including more current performance information) is available at the listed Internet Web site addresses (see § 1607(e), § 1608(a)(7), and § 1608(b)); and
- (3) A statement explaining how to request and obtain, free of charge, paper copies of the information required to be made available on a Web site pursuant to § 1607(e), § 1608(a)(7), and § 1608(b).
- **(b)** Additional Information. Nothing in this Article shall preclude the Administrator (or person designated by the Administrator) from including additional information that the Administrator (or designee) determines appropriate for such comparisons, provided such information is not inaccurate or misleading.
- (c) Use of Model Comparative Chart. The Administrator (or person designated by the Administrator) shall be deemed to have satisfied the requirements of this Section if it uses and accurately completes the model comparative chart in the Appendix to 29 C.F.R. § 2550.404a-5.

# § 1610 Investment-Related Information to be Provided Subsequent to Investment.

The Administrator (or person designated by the Administrator to act on its behalf) shall furnish to each investing Participant, subsequent to an investment in a Designated Investment Alternative, any materials provided to the Plan relating to the exercise of voting, tender, and similar rights appurtenant to the investment, to the extent such rights are passed through to such Participant under the terms of the Plan.

#### § 1611 Investment-Related Information to be Provided Upon Request.

The Administrator (or person designated by the Administrator to act on its behalf) shall furnish to each investing Participant, either at the times specified in § 1607 or upon request, the following information related to Designated Investment Alternatives—

- (a) Copies of prospectuses (or, alternatively, any short-form or summary prospectus, the form of which has been approved by the Securities and Exchange Commission) for the disclosure of information to investors by entities registered under either the Securities Act of 1933 or the Investment Company Act of 1940, or similar documents relating to Designated Investment Alternatives that are provided by entities that are not registered under either of these Acts:
- **(b)** Copies of any financial statements or reports, such as statements of additional information and shareholder reports, and of any other similar materials relating to the Plan's Designated Investment Alternatives, to the extent such information is provided to the Plan;
- (c) A statement of the value of a share or unit of each Designated Investment Alternative as well as the date of the valuation; and
- (d) A list of the assets comprising the portfolio of each Designated Investment Alternative which constitute "plan assets" within the meaning of 29 C.F.R. § 2510.3-101, and the value of each such asset (or the proportion of the investment alternative which it comprises).

### § 1612 Miscellaneous Rules Relating to Disclosures.

- (a) Fees & Expenses. Except as otherwise explicitly required under this Article, in any disclosure of information by the Administrator to Participants under this Article, fees and expenses may be expressed in terms of a monetary amount, formula, percentage of assets, or per capita charge.
- **(b) Understandable Disclosures.** The information required to be prepared by the Administrator for disclosure under this Article shall be written in a manner calculated to be understood by the average Plan Participant.

# § 1613 Independent Control by Participants.

- (a) In General. The Trustees shall not interfere with the exercise of independent control by Participants regarding transactions related to the participant-directed investment program (including, without limitation, the acquisition or disposition of investments, and the exercise of any voting, tender, and similar rights appurtenant to a Participant's ownership interest in an investment alternative).
- **(b) Improper Influence.** No Plan Fiduciary or Plan sponsor shall subject any Participant to improper influence with regard to any transaction related to the participant-directed investment program.
- (c) Concealment of Material Non-Public Facts. No Plan Fiduciary shall conceal any material non-public facts regarding an investment from a Participant, *unless* the disclosure of such information by the Plan Fiduciary to the Participant would violate any provision of federal law or any provision of state law which is not preempted by ERISA.

- (d) Incompetent Participant. No Plan Fiduciary shall accept the instructions of any person whom the Plan Fiduciary knows is legally incompetent.
- (e) Transactions Involving a Fiduciary. In the case of any transaction permitted under the participant-directed investment program and this Article which involves the sale, exchange, or leasing of property between the Plan and a Plan Fiduciary or an Affiliate of a Plan Fiduciary, or a loan to a Plan Fiduciary or an Affiliate of a Plan Fiduciary, the Participant shall not be required to pay more than, or shall not receive less than, "adequate consideration" (as defined in ERISA § 3(18) and the regulations thereunder) in connection with the transaction. (In general terms, "adequate consideration" for a security traded on a registered national securities exchange is the prevailing price on such market; for a security not so traded for which there is a generally recognized market, the current offering price for the security; and for an asset other than a security for which there is a generally recognized market, the fair market value of the asset as determined in good faith by the Plan Fiduciary.)
- **(f) Investment Advice.** Neither the Employer nor any Plan Fiduciary should provide any investment advice to a Participant related to the participant-directed investment program.

## § 1614 Incidents of Ownership Appurtenant to Participant Investments.

The Trustees may provide each Participant who has a Plan account invested in whole or in part in any investment directed by the Participant under this Article with a reasonable opportunity to give instruction with respect to voting, tender, or similar rights appurtenant to the Participant's ownership interest in that investment. To the extent that the Trustees do not pass through such rights to Participants, they shall not be protected from fiduciary liability with respect to the exercise or non-exercise of those rights.

# § 1615 Segregation; Expenses.

The portion of any Participant's Plan accounts which are invested according to Participant instructions under this Article shall be segregated from the rest of the Trust Fund. Such portion shall not share in any gains or losses of the general Trust Fund, but shall instead reap the benefits and bear the expenses of the segregated investments. Where more than one Participant is investing in the same investment, it is not necessary that the assets allocable to each Participant be physically segregated from those of other Participants, so long as they are segregated from other investments and the Trustees maintain adequate records to disclose the portions of the investment associated with each Participant. The Plan and Trust may charge a Participant's account for the reasonable expenses of carrying out investment instructions, *provided* that procedures are established to periodically inform the Participant of the actual expenses incurred with respect to his/her individual account.

# § 1616 ERISA § 404(c) Requirements.

Although this Plan is not subject to ERISA Title I, Part 4 fiduciary requirements, the Participant-directed investment program established under this Article is generally intended to satisfy the requirements established by the U.S. Department of Labor for "ERISA § 404(c) plans", and this Plan shall be deemed to incorporate by reference all requirements for ERISA § 404(c) plans not otherwise stated, except to the extent the Plan's language clearly evidences an intent not to follow a particular requirement. (*Cf.* 29 C.F.R. § 2550.404c-1 and § 2550.404a-5.) Nonetheless, any failure by a Plan Fiduciary to satisfy any of the provisions of this Article that

are designed to comply with requirements for ERISA § 404(c) plans, shall not subject the Plan, any Fiduciary, or the Employer to any liability to any Participant, Alternate Payee, or Beneficiary *except* to the extent such liability would exist under Pennsylvania law for a municipal government retirement plan even if the ERISA § 404(c) plan requirements had not been included in this Article, and for those purposes it shall be deemed that such provisions are not included in this Plan.

## **Article XVII — Administration**

## § 1701 In General.

- (a) In General. The Plan Administrator and named fiduciary of this Plan shall be the Committee described in this Section.
  - **(b) Members of the Committee.** The Committee shall consist of:
    - (1) the Salisbury Township Manager;
    - (2) the Chief Administrative Officer of this Plan (see § 1702(c);
- (3) one member of the Salisbury Township Board of Commissioners, who shall be appointed by resolution of the Board of Commissioners, and shall serve at the pleasure of the Board of Commissioners;
- (4) one Qualified Employee who is not included within any collective bargaining unit, who shall be appointed by resolution of the Board of Commissioners, and shall serve at the pleasure of the Board of Commissioners; and
- (5) one Qualified Employee who is included within the collective bargaining unit covering certain full-time public works employees, who shall be appointed by the collective bargaining representative of that unit, and serve at its pleasure.

Any person described in paragraphs (3) through (5) may resign upon written notice to the Board of Commissioners and the Committee.

(c) Conduct of the Committee. The Committee shall meet no less than annually. The Committee may adopt procedures, by-laws, and regulations as it deems necessary for the conduct of its affairs. A majority of the members of the Committee shall constitute a quorum, and any action of the Committee shall require the affirmative vote of a majority of those present at a meeting or, if action without a meeting is permitted by the Committee's procedures, by-laws, and regulations, the affirmative vote in writing of a majority of the members of the Committee.

## § 1702 Powers & Duties.

(a) In General. The Administrator shall administer the Plan in accordance with its terms and applicable law, shall direct the Trustees to make payments in accordance with the Plan from the Trust under § 1509, and shall have all powers necessary to carry out the provisions of the Plan. The Administrator shall have absolute and exclusive discretion to decide all issues arising in the administration, interpretation, and application of the Plan, including eligibility for benefits. The Administrator may from time to time set forth rules of interpretation and administration,

subject to modification as appropriate in the light of experience. No such rule will be ineffective by reason of the fact that such rule may amend the purely administrative provisions of the Plan or conform to any changes in the Plan or applicable law relating to qualified retirement plans. Decisions and rules established by the Administrator shall be conclusive and binding on all persons. The Administrator shall act without discrimination among persons similarly situated at any given time, although it may change its policies from time to time.

- **(b) Delegation.** Subject to the prohibitions of § 1801, the Administrator may delegate to any person or group of persons its authority to perform any act under this Plan and Trust, including those matters involving the exercise of discretion, *provided* that such delegation shall be subject to revocation at any time at the Administrator's discretion.
- (c) Designation of Chief Administrative Officer. The Township Finance Director shall be the Chief Administrative Officer of the Plan for purposes of the Municipal Pension Plan Funding Standard and Recovery Act, 53 PA. STAT. ANN. § 895.101 *et seq.*, subject to the control of the Administrator.
- (d) Employment of Professionals & Others. Subject to the prohibitions of § 1801, the Administrator may appoint such accountants, counsel, specialists, consultants, and other persons as it may deem necessary or desirable in connection with the administration of this Plan, including persons who may also be engaged by the Employer or who may be Trustees. The Administrator shall be entitled to rely exclusively upon, and shall be fully protected in any action taken in good faith by it in relying upon, any opinions or reports which shall be furnished to it by any such accountant, counsel, specialist, or other consultant.
- **(e) Records.** The Administrator shall keep a record of all its proceedings and acts, and shall keep all such books of account, records, and other data as may be necessary for the proper administration of the Plan.
- **(f) Notifications.** The Administrator shall notify the Trustees of all its actions, and, when required by law, it shall also notify any other interested persons of its actions.
- (g) Reports, Documents, and Communications. The Administrator (or Chief Administrative Officer, as applicable) shall prepare and file all reports and documents required to be filed with a governmental agency, shall prepare and provide or make available all reports and documents required to be provided or made available to persons with an interest under the Plan, and shall communicate with employees and other persons with respect to all matters relating to the Plan and Trust, including rights and benefits under this Plan.

#### § 1703 Direction of the Trustees.

- (a) **Direction to Request Approval.** The Administrator may at any time direct the Trustees in writing to obtain the written approval of the Administrator before exercising certain of the powers granted the Trustees under this Plan and Trust. Any such direction may be of a continuing nature or otherwise, and may be revoked in writing by the Administrator at any time.
- **(b) Funding Policy and Method.** The Administrator shall, from time to time, establish a funding policy and investment objectives and guidelines for the Trust consistent with the purposes of the Plan and state and federal law, and shall direct the Trustees to comply with such policy, objectives, and guidelines. The Administrator shall periodically review the operation of the Trust and all financial reports, investment reviews, and other reports prepared for the Plan or Trust.
- (c) **Duty to Question Direction by Administrator.** Neither the Trustees nor any other person shall be under any duty to question a direction by the Administrator under this Section.

#### § 1704 Compensation & Expenses.

Members of the Committee shall serve without compensation for their services. All expenses incident to the administration of the Plan by the Administrator, including but not limited to fees of accountants, counsel, consultants, and other specialists, and other costs of administering the Plan, may be paid by the Employer, but until and unless they are paid by the Employer they shall constitute a charge upon the Trust Fund.

#### § 1705 Standard of Care.

The Administrator and all accountants, counsel, specialists, consultants, and others retained by it under § 1702(d) shall be subject to the fiduciary requirements detailed in Article XVIII.

### Article XVIII — Fiduciaries

# § 1801 Prohibition Against Certain Persons Holding Positions under this Plan.

No person may serve under this Plan and Trust as a Fiduciary **if** he has been convicted of any of the crimes enumerated in ERISA § 411 **until** after the expiration of 13 years from the later of conviction or release from imprisonment (or such earlier period as allowed under ERISA § 411).

# § 1802 Bonding.

Every Fiduciary and every other person who handles funds or other property under this Plan (except properly capitalized corporate fiduciaries organized, doing business, and authorized to exercise trust powers under the laws of the Commonwealth of Pennsylvania or the United States, and their directors, officers, and employees) should be bonded in the same manner as if this Plan and Trust were subject to ERISA § 412 (which generally requires a bond not less than 10% of the amount of funds handled, though not less than \$1000 nor more than \$500,000) and the Plan may pay the costs and expenses of such bonds.

# § 1803 Duty of Care.

To the extent of their powers, the Fiduciaries shall discharge their duties with respect to the Plan and Trust-

(a) with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character;

- **(b)** by diversifying the investments of the Plan and Trust so as to minimize the risk of large losses **unless** under the circumstances it is clearly prudent not to do so; **and**
- (c) in accordance with the documents and instruments governing the Plan and Trust to the extent they are consistent with the applicable provisions of ERISA, the Code, and other laws.

## § 1804 Duty of Loyalty.

- (a) **Self-Dealing.** Fiduciaries shall not deal with the income or assets of the Plan in their own interests or for their own accounts, nor shall they receive any consideration for their own personal accounts from any party dealing with the Plan in connection with a transaction involving the income or assets of the Plan.
- **(b)** Adverse Interests. A Fiduciary shall not act in any transaction involving the Plan on behalf of a party (or represent a party) whose interests are adverse to the interests of the Plan or the interests of the Participants, Alternate Payees, or Beneficiaries, whether in his individual capacity or any other.

#### § 1805 Prohibited Transactions.

- (a) General Rule. Fiduciaries shall not engage on behalf of the Plan or Trust, either directly or indirectly, in any Prohibited Transaction or in any other transaction which is prohibited under Code § 503.
- **(b) Prohibited Transactions.** For purposes of this Section, the term "Prohibited Transaction" means any transaction in which the Plan or Trust:
- (1) lends any part of its income or corpus, without receipt of adequate security and a reasonable rate of interest, to a Disqualified Person;
- (2) pays any compensation, in excess of a reasonable allowance for salaries or other compensation for personal services actually rendered, to a Disqualified Person;
- (3) makes any part of its services available on a preferential basis to a Disqualified Person;
- (4) makes any substantial purchase of securities or any other property, for more than adequate consideration in money or money's worth, from a Disqualified Person;
- (5) sells any substantial part of its securities or other property, for less than an adequate consideration in money or money's worth, to a Disqualified Person; or
- (6) engages in any other transaction which results in a substantial diversion of its income or corpus to a Disqualified Person.
- (c) **Disqualified Persons.** For purposes of this Section, the term "Disqualified Person" means—
  - (1) the Employer;
  - (2) a person who has made a substantial contribution to the Plan or Trust;
- (3) a member of the family (as defined in Code  $\S 267(c)(4)$ ) of an individual described in paragraph (2); and

- (4) a corporation controlled by a person described in paragraphs (1) or (2) through the ownership, directly or indirectly, of 50% or more of the total combined voting power of all classes of stock entitled to vote or 50% or more of the total value of shares of all classes of stock of the corporation.
- (d) **Definitions.** The terms used in this Section shall be interpreted in the same manner as corresponding terms utilized in Code § 503.

#### § 1806 Indemnification.

The Employer hereby agrees to indemnify any officer, director, elected official, or employee of the Employer for any expenses, penalties, damages, or other pecuniary losses which such person may suffer as a result of his responsibilities, obligations, or duties in connection with the Plan or fiduciary activities actually performed in connection with the Plan, **but only** to the extent that—

- (a) fiduciary liability insurance is not available to cover the payment of such item; and
- (b) the person is not being relieved of his fiduciary responsibilities and liabilities to the Plan for breaches of fiduciary obligations.

# **Article XIX — Amendment, Termination & Merger**

## § 1901 Amendment.

- (a) In General. Subject to the provisions of subsection (b) and applicable collective bargaining obligations, the Board of Commissioners of the Sponsor shall have the right at any time, and from time to time, to amend in whole or in part, any or all of the provisions of this Plan and Trust, by Ordinance.
- **(b) Prohibited Amendments.** No amendment under this Section shall be effective to the extent that it shall—
- (1) Exclusive Benefit. Authorize or permit any part of the Trust Fund to revert to or become the property of the Employer or any Related Employer, or to be used or diverted to purposes other than the exclusive benefit of the Participants, Beneficiaries, and Alternate Payees, except as permitted under ERISA and the Code for qualified retirement plans which are government plans;
- (2) Accrued Rights. Cause any reduction in the accrued benefit of any Participant, Alternate Payee, or Beneficiary except as permitted under applicable law;
- (3) **Trustees.** Affect the rights, duties, or responsibilities of the Trustees without the written consent of the Trustees.

#### § 1902 Termination.

- (a) **Right to Terminate Plan.** The Sponsor shall have the right to discontinue its contributions to the Plan and Trust and terminate its participation under this Plan at any time by Ordinance, subject to any applicable collective bargaining responsibilities.
- **(b) Plan Accounts.** Following a termination of the Plan or complete discontinuance of Employer contributions to the Plan, the amounts in the Forfeiture Account and Early Employer Contributions Account shall be allocated on the day of termination or discontinuance as if they were end-of-Plan-Year Employer contributions under Article IV.
- (c) Termination & Liquidation of the Trust. Following a termination of the Plan, the Sponsor shall terminate the Trust. In that event, the Trustees shall distribute the accounts of all Plan Participants, Alternate Payees, and Beneficiaries to such persons as quickly as possible under the provisions of Articles XI, XII, and XIII as if the Plan and Trust termination were an event described in § 1101.
- (d) Termination of Trust Without Termination of Plan. The Sponsor may also elect to terminate the Trust *without* terminating the Plan so long as the Sponsor directs the Trustees to transfer the assets of the Trust Fund to another funding medium for the Plan consistent with applicable law concerning qualified retirement plans and plans of Pennsylvania first class townships, and § 1513 (relating to Exclusive Benefit Rule).
- (e) Termination or Spin-Off by Related Employer. Any Related Employer (or former Related Employer) which has adopted this Plan and Trust may at any time elect to terminate its participation in this Plan and Trust by written notice to the Administrator and the Trustees, to the extent permitted by state and federal law. In such event, the Trustee shall segregate assets attributable to employer and employee contributions (and liabilities allocable to investments thereof) made by or with respect to employees of the Related Employer from the Trust Fund and distribute such assets in accordance with the written directions of the (former) Related Employer (consistent with applicable law concerning qualified retirement plans and § 18-1613 (relating to Exclusive Benefit Rule)). Any election by a Related Employer under this subsection (d) shall be deemed an amendment and separation of the Related Employer's plan and trust from the provisions of this Plan and Trust Agreement, and not a termination of the Related Employer's plan, unless the Related Employer specifically terminates its plan.

## § 1903 Merger of Plans; Transfer of Assets.

- (a) **Definition.** For purposes of this Section, the term "merger" shall mean any merger or consolidation of this Plan and/or the Trust Fund with any other plan, or any transfer of the assets or liabilities of the Plan and/or the Trust Fund to any other plan.
- **(b) Accrued Rights.** The terms of any merger must specify that if this Plan or its successor were to terminate immediately after the merger, each Participant shall receive a benefit which is not less than he/she would have received in the event this Plan terminated immediately before such merger.

## **Article XX — Miscellaneous**

## § 2001 Acquittance.

This Plan and Trust is purely voluntary on the part of the Employer. Except as provided in this Plan and Trust document, neither the establishment of the Trust, any modification thereof, the creation of any fund or account, nor the payment of any benefits shall be construed as giving to any Participant or any other person any legal or equitable right against the Employer, any officer or Employee of the Employer, the Trustees, or the Administrator. Neither the Trustees, the Administrator, nor the Employer in any way guarantees the Trust Fund from loss or depreciation, nor do they guarantee any payment to any person. The liability of such persons to make any payments hereunder is limited to the available assets of the Trust Fund.

## § 2002 Limitation of Liability.

Each Employee who becomes a Participant under this Plan expressly agrees and understands that neither the Employer, the members of the Administrator, the Trustees in their individual capacity, nor any of their officers and agents shall be subject in any way to any suit or litigation, or to any personal liability for any reason whatsoever in connection with this Plan and Trust or its operation, *except* for their willful neglect or fraud.

## § 2003 Legal Actions.

In any action or proceeding involving the Trust Fund, its administration, or any of its constituent property—

- (a) Necessary Parties. The only necessary parties shall be the Employer, the Administrator, and the Trustees;
- **(b) Notice.** No Employees or former Employees of the Employer, Alternate Payees, Beneficiaries, or any other person having or claiming to have an interest in the Trust Fund or under the Plan shall be entitled to any notice or process; **and**
- (c) **Final Judgment.** Any final judgment which is either not appealed or appealable shall be binding and conclusive on all parties, the Administrator, and all persons having or claiming to have any interest in the Trust Fund or under the Plan.

# § 2004 Delegation of Authority by Employer.

Whenever any Employer is permitted or required to do or perform any act, matter, or thing under this Plan, it shall be done or performed by any officer duly authorized to perform same by the Employer.

#### § 2005 Clerical Errors.

If the Administrator discovers that a person who should have received any contribution or allocation under this Plan for any Plan Year did not, or did not receive as large a contribution or

allocation as he should have, the Employer shall make a contribution for such person in the amount erroneously omitted, plus the amount of Income which would have been earned by such contribution had it been timely made.

Conversely, if the Administrator discovers that a person received a contribution or allocation under this Plan for any Plan Year to which he was not entitled, the amount of such erroneous contribution or allocation shall be deducted from the person's account and transferred to the Early Employer Contributions Account (where it shall be treated as being a contribution derived from state aid).

#### § 2006 Construction.

This Plan and Trust Agreement shall be construed and administered according to the laws of the United States of America and the Commonwealth of Pennsylvania. Further, this Plan and Trust Agreement shall be construed and administered so as to conform to the applicable requirements for qualification under Code §§ 401(a) and 501(a) and shall be deemed amended automatically to conform to such legal requirements as in effect from time to time to the extent necessary.

#### § 2007 Gender & Number.

Whenever any words are used in this Plan and Trust in the masculine gender, they shall be construed as though they were also used in the feminine gender in all appropriate cases. Whenever any words are used in either the singular or plural form, they shall be construed as though they were also used in the other form in all appropriate cases.

## § 2008 Headings.

Article, section, subsection, paragraph, subparagraph, clause, subclause, and other headings are included in this document for convenience only and shall in no manner be construed as a part of this Plan and Trust Agreement.

# § 2009 Severability.

Any provision of this Plan which is prohibited or unenforceable shall be ineffective to the extent of such prohibition or unenforceability without invalidating or rendering unenforceable the remaining provisions of this Plan. To the extent permitted by applicable law, the Employer and Trustees hereby waive any provision of law which renders any provision of this Plan prohibited or unenforceable in any respect.

# § 2010 Employment Rights.

Nothing contained in this Plan and Trust shall be construed or interpreted as giving any employee of the Employer the right to be retained in the service of any Employer or shall affect or impair any terms of employment with any Employer, the right of any Employer to control its employees, and the right of any Employer to terminate the service of any employee at any time.

#### § 2011 Communications.

- (a) To the Administrator or Trustees. All elections, designations, requests, notices, instructions, or other communications made to the Administrator or the Trustees shall be in such form as may be prescribed by the Administrator or the Trustees and shall be mailed by first-class mail or delivered to such location as shall be specified by the Administrator or the Trustees, or transmitted to them in some other manner approved by the Administrator or Trustee (e.g., fax, e-mail, action on an internet web page, etc.). The communication shall be deemed to have been given and delivered only upon actual receipt thereof at such location.
- **(b)** By the Administrator, Trustees, or Employer. All notices, statements, reports, or other communications from the Administrator, the Trustees, or the Employer to any person with an interest under this Plan shall be deemed to have been duly given when delivered to, or when mailed by first-class mail, postage prepaid and addressed to such person at his address last appearing on the records of the Administrator, the Trustees, or the Employer.

## § 2012 Public Employee Pension Forfeiture Act.

Notwithstanding anything to the contrary contained in this Ordinance, all benefits with respect to a Participant under this Plan are subject to partial or complete forfeiture in accordance with the provisions of the Public Employee Pension Forfeiture Act, 43 PA. STAT. ANN. § 1311 et seq. Any amounts so forfeited which are subject to restitution to the Employer or another person or entity under the Act shall be so paid in accordance with the restitution order. Any amounts so forfeited which are not subject to restitution to the Employer or other person or entity under that Act shall, at such time as the Administrator reasonably determines that all proceedings related to the forfeiture and/or restitution under the Act have commenced and have concluded, be allocated among the accounts of persons who were Active Participants in this Plan as of the date of the forfeiture in proportion to their Compensation for the last Plan Year which ended before the date of the forfeiture.

# § 2013 Type of Plan.

This Plan is a money purchase, defined contribution pension plan.

### **SALISBURY TOWNSHIP**

Lehigh County, Pennsylvania
Board of Commissioners Workshop Meeting
May 23, 2019
7:30 p.m.

(approximate time)

# **Topics of discussion:**

• Presentation by and discussion with Finance Director Paul Ziegenfus regarding a draft Fire Fund Capital Plan and a draft 2020 Budget timeline